

## **Faculty Handbook**

Produced by the Office of the Senior Vice President and Provost
Approved by the Virginia Tech Board of Visitors
September 9, 2013

This document is subject to change. Please refer to the <u>provost's website</u> for the most recent *Faculty Handbook* information.

<u>University policies</u> are available online, as are many important procedures maintained by <u>purchasing</u>, <u>human resources</u>, and the <u>controller's office</u>. These websites will be updated as policies and procedures change. Please refer to them for issues not addressed in the *Faculty Handbook*.

Virginia Tech does not discriminate against employees, students, or applicants for admission or employment on the basis of race, gender, disability, age, veteran status, national origin, religion, sexual orientation, or political affiliation. Anyone having questions concerning discrimination should contact the <u>Office for Equity and Access</u>.

# 2013-2014 Faculty Handbook Table of Contents

	CHAPTER ONE
1.0	Mission and Governance of the University
1.1	Mission of the University
1.2	Governance of the University
1.2.1	University Policies, Administrative Policies, and Presidential Policy Memoranda
1.2.2	Governance by Shared Responsibility
1.2.3	Governance Structure
1.2.4	University Council
1.2.5	Commission on Administrative and Professional Faculty Affairs
1.2.6	Commission on Equal Opportunity and Diversity
1.2.7	Commission on Faculty Affairs
1.2.8	Commission on Graduate Studies and Policies
1.2.9	Commission on Outreach and International Affairs
1.2.10	Commission on Research
1.2.11	Commission on Staff Policies and Affairs
1.2.12	Commission on Student Affairs
1.2.13	Commission on Undergraduate Studies and Policies
1.2.14	Commission on University Support
1.2.15	University Advisory Council on Strategic Budgeting and Planning
1.2.16	University Committees
1.3	Board of Visitors
1.4	Other Governance Organizations
1.4.1	College Faculty Associations
1.4.2	Staff Senate
1.4.3	Student Government Association
1.4.4	Graduate Student Assembly
1.4.5	Faculty Senate
1.4.5.1	Committee on Reconciliation

1.4.5.2	Faculty Review Committee
1.4.5.3	Committee on Faculty Ethics
1.5	Central Administrative Organization
1.5.1	President
1.5.2	Senior Vice President and Provost
1.5.3	Vice President for Administration
1.5.4	Vice President for Alumni Relations
1.5.5	Vice President for Development and University Relations
1.5.6	Vice President for Diversity and Inclusion
1.5.7	Vice President for Finance and Chief Financial Officer
1.5.8	Vice President for Information Technology and Chief Information Officer
1.5.9	University Treasurer
1.6	University Academic Administration
1.6.1	College Deans
1.6.2	Departmental Administration
1.6.3	Dean of University Libraries
1.6.4	Director of Virginia Cooperative Extension
1.6.5	Vice President and Dean for Graduate Education
1.6.6	Vice President and Dean for Undergraduate Education
1.6.6	Vice President for the National Capital Region
1.6.7	Vice President for Outreach and International Affairs
1.6.8	Vice President for Research
1.6.9	Vice President for Student Affairs
	CHAPTER TWO
2.0	Employment Policies and Procedures for All Faculty
2.1	Types of Positions
2.2	Faculty Definition
2.3	Faculty Categories
2.3.1	The Collegiate Faculty
2.3.2	University Libraries Faculty with Continued Appointment or on the

	Continued Appointment-Track
2.3.3	Virginia Cooperative Extension Faculty with Continued Appointment or
2.3.4	on the Continued Appointment-Track Administrative and Professional Faculty
2.3.5	Special Research Faculty
2.3.5.1	Affiliated Research Faculty
2.3.6	Non-Tenure-Track Instructional Faculty
2.4	Faculty Compensation Plan
2.5	General Procedures for Faculty and Administrative Searches and Appointments
2.5.1	Faculty Search Procedures
2.5.2	Appointment of Department Heads or Chairs
2.5.3	Appointment of Deans
2.5.4	Appointment of Associate and Assistant Deans
2.5.5	Appointment of Senior Vice President and Provost and Vice Presidents
2.5.6	Appointment of the President
2.5.7	Criminal Background and Driving Record Checks
2.5.8	Terms of Faculty Offer and Final Approval of Appointment
2.5.9	Faculty Credentialing Guidelines
2.6	Terms of Appointment
2.6.1	Academic Year Appointments
2.6.2	Calendar Year Appointments
2.6.3	Research Extended Appointments
2.6.4	Restricted Appointments
2.6.5	Summer and Winter Appointments
2.6.6	University-Sponsored Applications for Permanent Residency
2.6.7	Dual Career Assistance Program
2.7	Professional Responsibilities and Conduct
2.7.1	Statement of Principles of Ethical Behavior
2.7.1.1	Allegations of Unprofessional or Unethical Conduct
2.7.2	Virginia Tech Principles of Community
2.7.3	University Safety and Security Policy

2.7.4	Campus and Workplace Violence Prevention
2.7.5	Health and Safety
2.7.6	Statement of Business Conduct Standards
2.7.7	Policy on Anti-Discrimination and Harassment Prevention
2.7.8	Policies Related to Electronic Communications
2.7.8.1	Privacy of Electronic Communications
2.7.8.2	Guidelines for Acceptable Use of Information Systems
2.7.9	Policy on Misconduct in Research
2.8	Advanced Study at Virginia Tech
2.9	Leaves
2.9.1	Leave Report
2.9.2	Educational Leave
2.9.3	Military Leave
2.9.4	Administrative Leave
2.9.5	Annual Leave and Holidays
2.9.6	Sick Leave
2.9.7	Family Leave
2.9.7.1	Family Medical Leave Act (FMLA)
2.9.7.2	Additional Faculty Leave Benefits for Faculty on Regular, Salaried Appointments
2.9.7.3	Genetic Information Nondiscrimination Act of 2008
2.9.8	Leave Without Pay
2.9.9	Disaster Relief Leave
2.10	Change of Duty Station or Special Leave
2.11	Retirement, Resignation, and Non-Reappointment
2.11.1	Retirement
2.11.1.1	Retirement Transition Program for Faculty with Tenure or Continued Appointment
2.11.2	Resignation
2.11.3	Non-Reappointment
2.11.3.1	Non-Reappointment for Faculty on Tenure-Track or Continued Appointment-Track

2.11.3.2	Non-Reappointment for Non-Tenure-Track Instructional Faculty on Regular Appointments
2.11.3.3	Non-Reappointment for Special Research Faculty on Regular Appointments
2.11.3.4	Non-Reappointment for Administrative and Professional Faculty on Regular Appointments
2.11.4	Unclaimed Personal Property
2.12	Reduction in Force
2.12.1	Reduction in Force Under Conditions of Financial Exigency
2.12.2	Reduction in Force for Academic Program Restructuring or Discontinuance
2.13	Severance Benefits
2.13.1	Alternative Severance Option
2.14	Consulting and Outside Employment
2.14.1	Consulting Activities
2.14.2	Technical Assistance Program
2.14.3	Outside Employment and External Activities Other Than Consulting
2.15	Conflicts of Interest and Commitment
2.15.1	Faculty Commitment to the University
2.15.2	Statement of Principles
2.15.3	Procedures to Monitor and Approve Involvement in External Activities
2.15.4	Compliance with the Virginia Conflict of Interests Act
2.15.5	Compliance with Federal Conflict of Interests Guidelines
2.15.6	Definitions
2.15.7	Disclosure of External Activities and Potential Conflict of Interest and/or Commitment
2.15.8	Appeal Process
2.15.9	Compliance
2.16	Overload Payments for Credit or Non-Credit Instruction
2.16.1	Participation in and Compensation for Non-Credit Continuing and Professional Education Activities
2.16.2	Participation in and Compensation for Credit Continuing and Professional Education Activities
2.17	Faculty Involvement with the Virginia Tech Carilion School of Medicine

2.17.1	Faculty Buyout Agreements with Virginia Tech Carilion School of Medicine
2.17.2	Faculty Overload Payment Agreements with Virginia Tech Carilion School of Medicine
2.18	Use of University Facilities
2.18.1	Required Use of Continuing and Professional Education Program Services and Facilities
2.19	Use of University Letterhead
2.20	Political Activities
2.21	Indemnity
2.22	Geographical Transfer Policy
	CHAPTER THREE
3.0	Employment Policies and Procedures for Tenured and Tenure-Track Faculty
3.1	Standard Faculty Ranks
3.1.1	Assistant Professor
3.1.2	Associate Professor
3.1.3	Professor
3.2	Honored Faculty Appointments
3.2.1	Endowed Chairs, Professorships, and Fellowships
3.2.1.1	Eminent Scholar Program
3.2.2	Alumni Distinguished Professor
3.2.3	University Distinguished Professor
3.2.4	Emeritus or Emerita Designation
3.3	Procedures for Faculty Appointments with Tenure
3.3.1	Part-Time Tenure-Track and Tenured Appointments
3.3.1.1	Part-Time Term Tenure-Track and Tenured Appointments
3.3.1.2	Permanent Part-Time Tenure-Track or Tenured Appointments
3.4	Promotion and Tenure
3.4.1	Tenure Eligibility
3.4.2	Probationary Period
3.4.2.1	Extending the Tenure Clock
3.4.3	Guidelines for the Calculation of Prior Service

3.4.4	Evaluation Procedures for Promotion and Tenure
3.4.4.1	Departmental Evaluation
3.4.4.2	College Evaluation
3.4.4.3	University Evaluation
3.4.4.4	Promotion and Tenure Guidelines
3.4.5	Appeals of Decisions on Reappointment, Tenure, or Promotion
3.4.5.1	Probationary Reappointment
3.4.5.2	Tenure Decision
3.4.5.3	Promotion Decision
3.5	Annual Evaluation, Post-Tenure Review, and Periodic Review of College and Departmental Administrators
3.5.1	Annual Evaluation and Salary Adjustments
3.5.2	Unsatisfactory Performance
3.5.3	Departmental Minimal Standards
3.5.4	Post-Tenure Review
3.5.5	Periodic Review of Deans, Department Heads, Senior Administrators, and Vice Presidents
3.6	Imposition of a Severe Sanction or Dismissal for Cause
3.6.1	Adequate Cause
3.6.2	Imposition of a Severe Sanction
3.6.3	Dismissal for Cause
3.7	Faculty Grievance Policy and Procedures
3.7.1	Faculty Reconciliation and Mediation Services
3.7.2	The Formal Grievance Procedure
3.7.3	Timeliness of Grievance and Procedural Compliance
3.7.4	Valid Issues for Grievance
3.7.5	Particular Concerns and Definitions
3.7.6	Overview of the Formal Grievance Process for Tenured and Tenure- Track Faculty
3.8	Study-Research Leave
3.9	Research Assignment
3.10	Modified Duties

	CHAPTER FOUR
4.0	Employment Policies and Procedures for Faculty with Continued Appointment or on the Continued Appointment-Track
4.1	University Libraries Faculty with Continued Appointment or on the Continued Appointment-Track
4.2	Virginia Cooperative Extension Faculty with Continued Appointment or on the Continued Appointment-Track
4.3	Library and Extension Faculty Ranks
4.3.1	Extra-Collegiate Instructor
4.3.2	Assistant Professor
4.3.3	Associate Professor
4.3.4	Professor
4.3.5	Emeritus or Emerita Designation
4.4	Procedures for Appointments with Continued Appointment
4.4.1	Part-Time Continued Appointment and Continued Appointment-Track Appointments
4.4.1.1	Part-Time Term Continued Appointment and Continued Appointment- Track Appointments
4.4.1.2	Permanent Part-Time Continued Appointment and Continued Appointment-Track Appointments
4.5	Continued Appointment and Promotion
4.5.1	Continued Appointment Eligibility
4.5.2	Probationary Period
4.5.2.1	Extending the Continued Appointment Clock
4.5.3	Guidelines for the Calculation of Prior Service
4.5.4	Evaluation Procedures for Promotion and Continued Appointment
4.5.4.1	Division-Level Evaluation
4.5.4.2	Composition of Library and Extension Division-Level Committees
4.5.4.3	Procedures and Recommendations of Library and Extension Division- Level Committees
4.5.4.4	Review and Recommendations by the Dean of University Libraries and Director of Virginia Cooperative Extension
4.5.4.5	University Evaluation
4.5.5	Appeals of Decisions on Reappointment, Continued Appointment, or Promotion
4.5.5.1	Probationary Reappointment

4.5.5.2	Continued Appointment Decision
4.5.5.3	Promotion Decision
4.6	Annual Evaluation and Post-Continued Appointment Review
4.6.1	Annual Evaluation and Salary Adjustments
4.6.2	Unsatisfactory Performance
4.6.3	Divisional Minimal Standards
4.6.4	Post-Continued Appointment Review
4.7	Imposition of a Severe Sanction or Dismissal for Cause
4.7.1	Adequate Cause
4.7.2	Imposition of a Severe Sanction
4.7.3	Dismissal for Cause
4.8	Faculty Grievance Policy and Procedures
4.8.1	Faculty Reconciliation and Mediation Services
4.8.2	The Formal Grievance Procedure
4.8.3	Timeliness of Grievance and Procedural Compliance
4.8.4	Valid Issues for Grievance
4.8.5	Particular Concerns and Definitions
4.8.6	Overview of the Formal Grievance Process for Faculty with Continued Appointment or on the Continued Appointment-Track
4.9	Study-Research Leave
4.10	Research Assignment
4.11	Modified Duties
4.12	Consulting Activities for Virginia Cooperative Extension Faculty
	CHAPTER FIVE
5.0	Employment Policies and Procedures for Non-Tenure-Track Instructional Faculty
5.1	Non-Tenure-Track Instructional Faculty Ranks
5.1.2	Visiting Professor
5.1.3	Adjunct Professor
5.1.4	Professor of Practice Ranks
5.1.5	Clinical Faculty Ranks

5.1.6	Instructor Faculty Ranks
5.2	Policies Related to Non-Tenure-Track Instructional Appointments
5.2.1	Initial Appointment
5.2.2	Reappointment
5.2.3	Annual Evaluations and Merit Adjustments
5.2.4	Promotion Guidelines for Non-Tenure-Track Faculty Appointments
5.2.5	Appeal Procedures
5.3	Termination Procedures for Non-Tenure-Track Faculty on Regular Appointments
5.3.1	Dismissal for Cause
5.3.2	Termination of Appointment During the Contract Period
5.4	Participation in Governance
5.5	Participation on Graduate Committees
5.6	Eligibility to Serve as a Principal Investigator
5.7	Faculty Grievance Policy and Procedures
5.7.1	Faculty Reconciliation and Mediation Services
5.7.2	The Formal Grievance Procedure
5.7.3	Timeliness of Grievance and Procedural Compliance
5.7.4	Valid Issues for Grievance
5.7.5	Particular Concerns and Definitions
5.7.6	Overview of the Formal Grievance Process for Non-Tenure-Track Instructional Faculty
	CHAPTER SIX
6.0	Employment Policies and Procedures for Special Research Faculty
6.1	Office of Research Human Resources
6.2	Special Research Faculty Appointments
6.3	Research Associate Ranks
6.3.1	Research Associate
6.3.2	Senior Research Associate
6.3.3	Research Scientist
6.3.4	Senior Research Scientist

6.3.5	Postdoctoral Associate
6.4	Project Associate Ranks
6.4.1	Project Associate
6.4.2	Senior Project Associate
6.4.3	Project Director
6.5	Research Professor Ranks
6.5.1	Research Assistant Professor
6.5.2	Research Associate Professor
6.5.3	Research Professor
6.6	Matrix of Research Ranks
6.7	Affiliated Research Faculty
6.8	Searches for Special Research Faculty
6.9	Terms of Faculty Offer and Documentation of Credentials
6.9.1	Restricted Appointments
6.9.2	Multi-Year Restricted Appointments
6.9.3	Regular Appointments
6.9.4	Calendar Year versus Academic Year Appointments
6.10	Job Descriptions
6.11	Annual Evaluations
6.12	Merit and Special Adjustments
6.13	Reappointment
6.14	Termination Procedures for Special Research Faculty
6.14.1	Dismissal for Cause
6.14.2	Termination of Appointment for Faculty on Restricted Contracts
6.15	Effort Certification Compliance Issues for Special Research Faculty Members
6.16	Instructional Responsibilities for Special Research Faculty Members
6.17	Faculty Grievance Policy and Procedures
6.17.1	Faculty Reconciliation and Mediation Services
6.17.2	The Formal Grievance Procedure
6.17.3	Timeliness of Grievance and Procedural Compliance

6.17.4	Valid Issues for Grievance
6.17.5	Particular Concerns and Definitions
6.17.6	Overview of the Formal Grievance Process for Special Research Faculty
	CHAPTER SEVEN
7.0	Employment Policies and Procedures for Administrative and Professional Faculty
7.1	Faculty Rank and Title
7.1.1	Standard Faculty Rank
7.2	Policies Related to Administrative and Professional Faculty Appointments
7.2.1	Protection of Academic Freedom
7.2.2	Initial Appointment and Reappointment
7.2.3	Degree Verification
7.2.4	Academic Year Appointments for Administrative and Professional Faculty
7.3	Annual Evaluations
7.3.1	Periodic Evaluation of Deans, Vice Presidents, and Directors of Major Organizational Units
7.3.1.1	Reviews for Senior Academic Administrators
7.3.1.2	Reviews for the Provost, Vice Presidents, and Senior Staff Reporting to the President, and Other Non-Academic Administrators
7.4	Salary Adjustments
7.5	Teaching Credit Classes and Overload Compensation for Administrative and Professional Faculty
7.6	Non-Reappointment, Reassignment, Removal, and Imposition of Sanctions Other Than Dismissal
7.6.1	Non-Reappointment of Administrative and Professional Faculty on Regular Appointments
7.6.2	Non-Reappointment of Administrative and Professional Faculty on Restricted Appointments
7.6.3	Reassignment
7.6.4	Removal for Just Cause
7.6.5	Imposition of Sanctions Other Than Dismissal
7.6.6	Abolition of Position
7.7	Grievance Policy and Procedures for Administrative and Professional Faculty

7.7.1	Faculty Reconciliation and Mediation Services
7.7.2	The Formal Grievance Procedure
7.7.3	Timeliness of Grievance and Procedural Compliance
7.7.4	Valid Issues for Grievance
7.7.5	Particular Concerns and Definitions
7.7.6	Overview of the Formal Grievance Process for Administrative and Professional Faculty
7.8	Leave
7.9	Consulting Activities for Virginia Cooperative Extension Faculty
	CHAPTER EIGHT
8.0	Employment Policies and Procedures for Graduate Assistants
8.1	Graduate Student Appointments
8.2	Required Teaching Credentials for Graduate Teaching Assistants
8.3	Additional Employment by Graduate Students with a Full-Time Assistantship Contract
	CHAPTER NINE
9.0	Instruction-Related Policies
9.1	Assignment of Academic Responsibilities
9.1.1	Summer and Winter Sessions
9.1.2	Independent Study and Undergraduate Research
9.1.3	Graduate Program Standards and Policies
9.2	Scheduling of Classes
9.3	Registration Procedures
9.3.1	Drop-Add Period
9.3.2	Force-Add Requests
9.3.3	Class Rolls
9.4	Textbooks and Other Instructional Materials
9.4.1	Faculty-Authored Course Materials
9.5	Grading Systems
9.6	Course Grading
9.6.1	Syllabus and Performance Expectation

9.6.2	Class Attendance
9.6.2.1	Religious Holidays
9.6.3	Final Examinations
9.6.4	Undergraduate Student Appeals
9.6.5	Graduate Student Appeals
9.6.6	Grade Adjustments for Undergraduate Honor Suspects
9.6.7	Change of Grade
9.6.8	Final Grade Reports
9.7	Instruction-Related Responsibilities
9.7.1	Office Hours
9.7.2	Tutoring
9.7.3	Students with Disabilities
9.8	The Undergraduate Honor System
9.8.1	Undergraduate Offenses and Sanctions
9.8.2	Faculty Participation
9.8.3	The Honor Pledge
9.9	Honor System for Graduate Students and Students in the College of Veterinary Medicine
9.10	Classroom Conduct
9.11	Teaching Evaluation
9.11.1	Student Evaluation of Courses and Instructors
9.11.2	Other Evaluation of Courses and Instructors
9.12	Student Record Policy
9.12.1	Academic Records
9.13	Undergraduate Student Advising
9.14	Identifying and Referring the Distressed Student
9.15	Faculty Awards for Teaching and Advising
9.15.1	The Wine Award for Excellence in Teaching
9.15.2	The University Sporn Award for Excellence in Teaching Introductory Subjects
9.15.3	Certificates of Teaching Excellence

9.15.4	Alumni Teaching Excellence Awards
9.15.5	Diggs Teaching Scholar Awards
9.15.6	Academy of Teaching Excellence
9.15.7	Alumni Award for Excellence in Undergraduate Academic Advising
9.15.8	Alumni Award for Excellence in Graduate Academic Advising
9.15.9	Provost's Award for Excellence in Undergraduate Academic Advising
9.15.10	Academy of Advising Excellence
9.16	Faculty Awards for Research, Extension, Outreach, and Service
9.16.1	Alumni Award for Research Excellence
9.16.2	Alumni Award for Extension Excellence
9.16.3	Alumni Award for Outreach Excellence
9.16.4	Alumni Award for Excellence in International Education
9.16.5	Alumni Award for Excellence in International Outreach
9.16.6	Alumni Award for Excellence in International Research
9.16.7	Academy of Outreach Excellence
9.16.8	Academy of Faculty Service
9.17	Awards for Creating Good Teaching and Learning Environments
9.17.1	Exemplary Department Award
9.17.2	Diggs Rountable
9.17.3	XCaliber Award
	CHAPTER TEN
10.0	Research, Creative, and Scholarly Activities
10.1	Procedures for Research and Scholarship
10.1.1	Departmental Research
10.1.2	Core Research
10.1.3	Sponsored Research
10.1.4	Preparation of Proposals for Sponsored Projects
10.1.5	Laboratory Services and Facilities
10.1.6	Ownership and Control of Research Results
10.1.7	Classified Research

10.1.8	Securing Theses and Dissertations
10.1.9	Publication of Research
10.1.10	International Travel Supplemental Grants Program
10.2	Policy on Misconduct in Research
10.2.1	Definition
10.2.2	Activities Covered
10.2.3	Procedures for Reporting, Investigating, and Resolving Misconduct in Research
10.3	Removal of a Principal, Co-Principal, or Lead Investigator
10.3.1	Appeal of Removal as a Principal, Co-Principal, or Lead Investigator
10.4	Effort Certification and Salary Charges to Sponsored Grants and Contracts
10.4.1	Effort Certification
10.4.2	Compliance Issues Related to Summer Research Appointments for Nine-Month Faculty Members
10.4.3	Compliance Issues for Special Research Faculty Members
10.5	Policy on Intellectual Property
10.5.1	Authority and Policy of the Intellectual Properties Committee
10.5.2	Policy Guidelines
	CHAPTER ELEVEN
11.0	Faculty Benefits Program
11.1	Required Benefits
11.1.1	Group Life Insurance
11.1.2	Long-Term Disability Insurance
11.1.3	Faculty Retirement
11.1.3.1	Virginia Retirement System
11.1.3.1.1	The Hybrid Retirement Plan
11.1.3.2	Optional Retirement Plan
11.1.4	Short-Term Disability Income Protection
11.2	Optional Programs
11.2.1	Long-Term Care
11.2.2	Accidental Death and Dismemberment Insurance

11.2.3	Health Insurance
11.2.4	Medical Flexible Spending Reimbursement Account
11.2.5	Dependent Care Flexible Spending Reimbursement Account
11.2.6	Employee Assistance Program
11.2.7	Tax-Deferred Investments/Deferred Compensation/Cash Match
11.2.8	Credit Unions
11.2.9	Charitable Deductions
11.2.10	Optional Life Insurance
11.3	Special Programs
11.3.1	Unemployment Insurance
11.3.2	Severance Benefits
11.3.3	Workers' Compensation
11.3.3.1	Reporting Work-Related Injuries
	CHAPTER TWELVE
12.0	Additional Information
12.1	Emergency Information

#### CHAPTER ONE

## 1.0 Mission and Governance of the University

## 1.1 Mission of the University

Virginia Polytechnic Institute and State University (Virginia Tech) is a public land-grant university serving the Commonwealth of Virginia, the nation, and the world community. The discovery and dissemination of new knowledge are central to its mission. Through its focus on teaching and learning, research and discovery, and outreach and engagement, the university creates, conveys, and applies knowledge to extend personal growth and opportunity, advance social and community development, foster economic competitiveness, and improve the quality of life.

## **1.2** Governance of the University

This section of the Faculty Handbook describes the structure of the university governance system. The board of visitors is the primary governing body of the university. The board appoints the president of the university, who serves as the chief executive. The president may delegate authority to the senior vice president and provost and vice presidents.

# 1.2.1 University Policies, Administrative Policies, and Presidential Policy Memoranda

University policies are available at <a href="www.policies.vt.edu">www.policies.vt.edu</a>. University policies are generally applicable to more than one office or department of the university. University Council and the university commissions constitute the main resident bodies for university policy formation. The university commissions formulate and recommend policies to the University Council, which in turn makes recommendations to the university president. Final authority rests with the university president and the board of visitors.

Administrative policies are issued by the vice presidents who are responsible for the accuracy and timeliness of policies and procedures relating to their areas. This responsibility includes proper notification of changes and updates to these policies and procedures, and conducting a review of policies annually.

Presidential policy memoranda provide information regarding policies and procedures that apply to specific situations, groups, or individuals. Presidential policy memoranda are issued by the university president and are available at www.policies.vt.edu/policymemos.

## 1.2.2 Governance by Shared Responsibility

There is a wide recognition of the complexity of university governance and general acknowledgment of the need for faculty, staff, and student participation in the conduct of university affairs. The University Council, university commissions, university advisory council, and university standing committees provide an organizational structure through which faculty, staff, student, and administrative responsibilities are shared and fulfilled.

The University Council and university commissions constitute the main resident bodies for policy formulation. Their memberships include representatives from administration, Faculty Senate, Staff Senate, college faculty associations, administrative and professional faculty, Graduate Student Assembly, Student Government Association, and representatives from other interest groups where appropriate. The university commissions formulate and recommend policies to the University Council, which in turn makes recommendations to the president of the university. Final authority rests with the president of the university and the board of visitors.

The constitutions and bylaws of the University Council, Faculty Senate, Staff Senate, and Graduate Student Assembly are found on their respective Virginia Tech websites. The constitution and bylaws of the Student Government Association are available from their office. Persons concerned with the governance of the university are encouraged to consult the constitutions and bylaws of these organizations. The constitution and bylaws of University Council contain the specific membership lists for University Council, the commissions, and the committees. Also available are lists of the specific persons who hold membership on the various bodies. These lists are updated as the personnel change. The membership lists are maintained by the president's office and are available on the governance website.

Virginia Polytechnic Institute and State University

#### 1.2.3 Governance Structure

Governance Structure Board of Visitors (www.bov.vt.edu) 13 members appointed by Governor, The Pres. Of the Bd. Of Agriculture & Consumer Services, Presidents of Faculty and Staff Senates, undergrad and grad reps (nonvoting members) President Advisory Council on Strategic Budgeting and Planning University Council 76 Members, 5 non-voting (www.governance.vt.edu) Commission on Faculty Commission on Administrative Affairs and Professional Faculty Affairs Commission on Staff Policies Commission on Student Affairs and Affairs Commission on Equal Opportunity Commission on Outreach and International Affairs and Diversity Commission on Undergraduate Commission on Research Studies and Policies Commission on Graduate Commission on University Studies and Policies Support

## 1.2.4 University Council

The functions of the University Council are to advise the president on matters of university governance; to accept functions and authority delegated to it by the president; and to review and make recommendations on matters of concern to the faculty, staff, students, and administration. Any council member, in accordance with its constitution and bylaws, may place such matters on the University Council agenda. The University Council refers appropriate matters to the commissions, advisory councils, Faculty Senate, Staff Senate, groups, or individuals for consideration and recommendation.

## 1.2.5 Commission on Administrative and Professional Faculty Affairs

The charge of the Commission on Administrative and Professional Faculty Affairs is to study, formulate, and recommend to University Council policies and procedures affecting the employment and working conditions of administrative and professional faculty. Areas for consideration include: morale of administrative and professional faculty; procedures for appointing, evaluating, disciplining, recognizing, and promoting administrative and professional faculty; benefits, educational and personal leave, and extra-university professional activity; and matters of equity and diversity that affect the university's professional environment. The commission is also responsible for reviewing grievances advanced to the executive level and to advise the senior vice president and provost or associate vice president for human resources prior to action.

## 1.2.6 Commission on Equal Opportunity and Diversity

The charge of the Commission on Equal Opportunity and Diversity is to study, formulate, and recommend to University Council policies and procedures as they relate to the university's responsibilities for equal opportunity, affirmative action, accessibility, and compliance; diversity planning and evaluation; diversity training and education; assessment of institutional climate; and similar matters of equity and diversity that affect the university. In collaboration with other university commissions and/or units, the commission addresses issues of diversity and equity as they relate to recruitment, retention, and advancement of faculty, staff, and students, particularly those from historically underrepresented groups and from international populations; student life; academic policies and support; curriculum; research, scholarship, and creative activity; and outreach.

## 1.2.7 Commission on Faculty Affairs

The charge of the Commission on Faculty Affairs is to study, formulate, and recommend to the University Council policies and procedures affecting employment and professional standing of members of the collegiate faculties and library or extension faculty on continued appointment. Areas for consideration include: commencement, academic freedom and faculty morale; procedures for appointing, evaluating, disciplining, recognizing, and promoting faculty—including department heads and chairs; tenure, benefits, academic and personal leave, extrauniversity professional activity; and matters of equity and diversity that affect the university's professional environment.

#### 1.2.8 Commission on Graduate Studies and Policies

The charge of the Commission on Graduate Studies and Policies is to study, formulate, and recommend to University Council policies and procedures concerning on- and off-campus graduate academic matters. Areas for consideration include: admissions, academic progress, degree requirements, commencement, registration, scheduling, curricula, courses, advising, instruction, teaching, research, financial assistance (including assistantships, scholarships, fellowships, and tuition), library resources, and other matters affecting the graduate student academic environment.

#### 1.2.9 Commission on Outreach and International Affairs

The charge of the Commission on Outreach and International Affairs is to study, formulate, and recommend to University Council policies and procedures concerning the engagement of the university in service, outreach, and international affairs. Areas for consideration include: Virginia Cooperative Extension; continuing and professional education; economic development—including community resource and leadership development; liaison with affiliated corporations and institutes; international programs; and other matters affecting service, outreach, and international affairs. The Commission on Outreach and International Affairs liaisons with the Commission on Undergraduate Studies and Policies on issues regarding education abroad and international exchange programs.

#### 1.2.10 Commission on Research

The charge of the Commission on Research is to study, formulate, and recommend to University Council policies and procedures concerning research. Areas for consideration include: sponsored programs, core programs, and interdisciplinary research; intellectual properties; animal care and human subjects; indirect costs and overhead; research facilities, centers, and institutes; library resources; liaisons with affiliated corporations and institutes; and other matters that affect research.

#### 1.2.11 Commission on Staff Policies and Affairs

The charge of the Commission on Staff Policies and Affairs is to study, formulate, and recommend to University Council policies and procedures affecting the working conditions of classified and university staff and to promote staff participation in the university community. Areas for consideration include: staff morale, evaluation, communication with supervisors, recognition, and career development; staff relations with administrators, faculty, and students; benefits and personal leave; extra-university professional activity; matters of equity and diversity that affect the university's professional environment; and other matters affecting the staff environment.

#### 1.2.12 Commission on Student Affairs

The charge of the Commission on Student Affairs is to study, formulate, and recommend to University Council policies and procedures affecting graduate and undergraduate student life and

morale. Areas for consideration include: student relations with peers, staff, faculty, administrators, alumni, and the community; student organizations, social life, and recreation; employment, placement, and counseling; residential life, health, safety, and quality of student-related services; and other matters affecting student life.

## 1.2.13 Commission on Undergraduate Studies and Policies

The charge of the Commission on Undergraduate Studies and Policies is to study, formulate, and recommend to University Council policies and procedures concerning on- and off-campus undergraduate academic matters. Areas for consideration include: library resources, admissions, academic progress, commencement, degree requirements, honor systems, study environment, inhonors programs, curricula, courses, advising, instruction, student honors and awards, financial aid, scheduling, registration, and other matters affecting the undergraduate student academic environment.

## 1.2.14 Commission on University Support

The charge of the Commission on University Support is to study, formulate, and recommend to University Council policies and procedures related to supporting the learning, discovery, and engagement missions of the university. Areas for consideration include: accounting, budget administration, purchasing, computing, communications, publications, physical facilities, parking, transportation, corporations, auxiliaries, development, institutional research, and liaisons with the Virginia Tech Foundation and the Alumni Association, and other related matters.

## 1.2.15 University Advisory Council on Strategic Budgeting and Planning

The university Advisory Council on Strategic Budgeting and Planning serves a primary advisory role for the university budgeting and planning process. It reports jointly to the president and the University Council. Areas for consideration include: monitoring the university planning and budgeting processes; participating in and advising on the development of biennial budgets, formulation of the university plan, development of university capital and facilities plans, and reconciliation of the plan and available resources; consulting on other budget and planning matters; addressing matters of policy relative to budget and planning appropriate for governance consideration and, in such instances, making recommendations to the University Council.

## 1.2.16 University Committees

University standing committees are those committees constituted on a continuing basis by the president on recommendation of the University Council to deal with matters of a university-wide interest. Each standing committee reports directly to one commission. The university standing committees are: Academic Support, Athletics, Campus Development Commencement, Computing and Communications Resources, Employee Benefits, Energy and Sustainability, Faculty Honorifics, Graduate Curriculum, Honor System Review Board, Intellectual Property, Library, Transportation and Parking, and University Curriculum Committee for Liberal

Education, and the Undergraduate Curriculum Committee. The president also constitutes ad hoc and special committees as needed.

In addition, university operational committees are constituted on a continuing basis and appointed by the senior vice president and provost, the vice president for administration, or one of the other vice presidents. These committees deal with matters of university-wide interest that fall primarily within the responsibilities of the appointing officers. Unless made the subject of specific legislation to the contrary, each university operational committee is constituted, charged, and staffed as the appointing administrative officer deems appropriate. The secretary of the University Council maintains a record of university operational committee chairs and members. Approved minutes of meetings of university operational committees are maintained by each committee and by the secretary of the University Council so that they are accessible, as required or requested, to the University Council.

#### 1.3 Board of Visitors

By statute of the Commonwealth of Virginia, the governing body of the university is the board of visitors, which exists as a corporation under the control of the Virginia General Assembly. The board is comprised of 13 members appointed by the governor—subject to confirmation by the senate—and the president of the Virginia Board of Agriculture and Consumer Services, who is an ex-officio member. In addition, the board appoints two non-voting student representatives (one undergraduate; one graduate) who serve for a one-year term and attend open sessions. Board members appointed by the governor serve for overlapping four-year terms. The current membership of the board of visitors and its by-laws are available on the Virginia Tech website at www.bov.vt.edu. A rector and vice-rector are elected annually by the membership, and the board appoints a secretary annually. The president of the Faculty Senate and the president of the Staff Senate sit with the board at all meetings, except those held in closed session, and participate in discussion without authority to vote. By law, the board meets at least once a year, but typically meets quarterly to consider policy matters and to review the progress of the university.

The board of visitors is responsible for institutional policies except those under the direct jurisdiction of the Commonwealth of Virginia. The board's responsibilities are specified by state statute. Responsibilities include, but are not restricted to, the appointment of the university president; appointment, promotion, and granting of tenure; removal of members of the faculty; the prescription of faculty responsibilities; the setting of faculty salaries; the determination of student tuition, fees, other charges; and the government and discipline of students.

The Faculty Handbook serves as a compilation of these and other policies related to the faculty of the university.

## 1.4 Other Governance Organizations

## 1.4.1 College Faculty Associations

The faculties are formally organized as faculty associations in the colleges of Agriculture and Life Sciences, Architecture and Urban Studies, Engineering, Liberal Arts and Human Sciences,

Natural Resources and Environment, Science, and Veterinary Medicine, as well as in the University Libraries and Virginia Cooperative Extension. These associations have constitutions that designate the purposes of the association, membership, officers, election procedures, standing committees and their duties, and other organizational and procedural matters. The Pamplin College of Business vests similar rights and responsibilities on its faculty members through a less formal structure.

#### 1.4.2 Staff Senate

The purpose of the Staff Senate is to create an effective staff organization that can enter into partnership for shared responsibility and cooperative action between the staff, faculty, administration, and students in order to promote the general welfare of the university.

The functions of the Staff Senate are:

- To serve as the representative body for staff employees of Virginia Tech;
- To act in an advisory capacity to the university administration and governance system;
- To appoint or recommend staff representatives to University Council, commissions, advisory councils, and committees;
- To facilitate the exchange of information between staff and the administration;
- To foster a spirit of unity and cooperation;
- To provide referral for individual concerns and problems to appropriate organizations or personnel, and
- To accept and share responsibility with the administration, faculty, and students in all efforts to attain the stated goals of the university.

#### 1.4.3 Student Government Association

The Student Government Association (SGA) is the official representative body for undergraduate students. Popular elections for officers and senators of SGA are held annually each spring. Senators are elected within academic colleges to represent proportionate enrollment in each respective college. The student senate and house make up the legislative branch of the SGA. The representatives to the house are elected or appointed by student organizations across campus.

The functions of the Student Government Association are:

- To express opinion on university affairs as the legislative branch of the SGA deems appropriate and necessary;
- To establish an effective means for advising and responding to the University council, university commissions, the university administration, the Faculty Senate, the Board of Visitors, and the public on university affairs;
- To participate in the formulation of academic and education policies which concern more than one college, division, or center at the university;
- To consider policies, programs, and other matters as the administration, college faculties, student organizations, and individuals of the university may propose;

- To afford avenues and procedures whereby communications within the university may flow freely, fully, and systematically;
- To help create, maintain, and protect a university environment conducive to the richest growth of scholarship, learning, teaching, research, and respect or dignity and rights, and
- To accept and share responsibility with the administration and faculty in all efforts to improve the stature and value of the university.

## 1.4.4 Graduate Student Assembly

The Graduate Student Assembly (GSA) is the governing and representative body of approximately 4,500 on-campus and 2,400 off-campus graduate students. The graduate students of each department offering graduate work elect two members to the assembly. Every year the GSA governing board and delegate body work to improve campus life, scholarly development, and graduate community by creating and implementing goals based on the issues and concerns of graduate students at Virginia Tech.

The functions of the Graduate Student Assembly are:

- To represent the interests of graduate students in all university activities and to facilitate the exchange of information between university governance structure and graduate students;
- To solicit, codify, and promote graduate student opinions and concerns and to develop and recommend policies concerning graduate students to the university governance system;
- To work with the administration of the graduate school and other appropriate commissions and committees to improve the quality of graduate educational programs, graduate research activity, and graduate teaching programs, and
- To work within the university governance structure to improve graduate student life and social programs.

## **1.4.5** Faculty Senate

The purpose of the Faculty Senate is to create an effective faculty organization that can enter into partnership for shared responsibility and cooperative action between the faculty, staff, administration, and students in order to promote the general welfare of the university.

The membership of the Faculty Senate consists of elected faculty members. A faculty member is eligible to be elected to the Faculty Senate and to vote in the election of faculty senators if the faculty member holds: (1) the rank of professor, associate professor, assistant professor, or instructor [this includes faculty in the clinical professor series, professor of practice series, and all ranks of instructor]; (2) a full-time and continuing appointment to the university; (3) an appointment or tenure in an academic department, or in the library, or as an extra-collegiate faculty member.

Although otherwise qualified, those faculty members in administrative positions at the college level, the university level, or the library, and those faculty members studying for a degree at the university are ineligible.

Faculty Senate members, though uninstructed representatives of their constituents, have the responsibility to seek the opinions of their electorate. Having done so, they make decisions and vote on matters brought before the Faculty Senate according to their own reasoned judgment.

The functions of the Faculty Senate are:

- To establish within the laws applicable to the university an effective means for advising and responding to the university commissions, the University Council, the administration, and the board of visitors on university affairs;
- To express opinion on university affairs as the Faculty Senate deems appropriate and necessary;
- To participate in the formulation of academic and educational policies that concern more than one college, division, or center;
- To consider policies, programs, and other matters that the administration, college faculties, student organizations, and individuals of the faculty may propose;
- To afford avenues and procedures whereby communications within the university may flow freely, fully, and systematically;
- To help create, maintain, and protect a university environment conducive to the richest growth of scholarship, learning, teaching, research, service, and respect for human dignity and rights;
- To accept and share responsibility with the administration, staff, and students in all efforts to improve the stature and usefulness of the university, and
- To advise, respond to, and participate in the Faculty Senate of Virginia, thereby providing university faculty with a voice in matters of broader concern to faculty across the commonwealth.

The following standing committees aid the Faculty Senate in performing its functions: Committee on Reconciliation, Faculty Review Committee, and Committee on Faculty Ethics.

The Faculty Senate cabinet is comprised of the officers and representatives of each of the colleges and the library. It is concerned primarily with the general business of the Faculty Senate. The president of the Faculty Senate may create additional work groups and ad hoc committees, as necessary to address specific issues or concerns of the faculty.

#### 1.4.5.1 Committee on Reconciliation

The Committee on Reconciliation is composed of eight tenured faculty members eligible for membership in the Faculty Senate. Faculty appointed to administrative positions with responsibility for recommending promotions, salary adjustments, and distribution of teaching, research, and extension assignments are ineligible for membership. The functions of this committee are to offer advice and counsel to faculty members who seek it, particularly in relation to disputes with immediate supervisors or university administrators.

The committee has a designated role within the grievance process to assist in resolving disputes that are eligible for consideration as a grievance if so requested by the faculty member. See the relevant grievance procedure for guidance on valid and ineligible issues for grievance that define the committee's purview in relation to the grievance process. Faculty members who wish to

engage the assistance of the committee in the context of a potential grievance submit their request concurrently to the Office of the Senior Vice President and Provost and the chair of the Committee on Reconciliation within 30 calendar days of the time when the faculty member knew or should have known of the event or action that is the basis for the potential grievance. The senior vice president and provost automatically grants a 60-day postponement of grievance timelines for the Committee on Reconciliation to attempt to resolve the complaint between the parties.

The Faculty Senate Committee on Reconciliation may also be consulted by faculty members concerning serious disagreements with immediate supervisors or other university administrators concerning issues that are not eligible for consideration within the grievance process. In such instances, the committee contacts the relevant administrator to determine if there is an interest and willingness to explore informal resolution of the dispute. The Office of the Senior Vice President and Provost need not be notified to initiate discussions with the committee in such instances.

Cooperation and candor from all members of the university community are prerequisite to the successful functioning of the Committee on Reconciliation in investigating serious and delicate cases. Confidential information concerning personnel and academic issues may be shared with the committee. The committee keeps no written records and treats all matters with utmost sensitivity.

In conducting its work, the Committee on Reconciliation takes special measures to assure that participating members have no conflict of interest in the matter. In contrast to the Faculty Review Committee (see section 1.4.5.2), which has responsibility for formally investigating a grievance through hearings, calling witnesses, and collecting and assessing evidence prior to rendering its judgment, the Committee on Reconciliation operates more informally as a facilitator. It meets with the respective parties to determine if there is common ground for resolution of the matter, facilitating a solution that is agreeable to the principal parties and consistent with university policy and practice.

Generally the faculty member initiates the request for assistance of the committee; however, an administrator may seek the committee's involvement in resolving a dispute with a faculty member. Both parties to the dispute must agree to be participants in the reconciliation process. If no resolution is reached, the matter reverts to formal review as part of the grievance process if the matter is eligible for such consideration.

## **1.4.5.2** Faculty Review Committee

The Faculty Review Committee includes a minimum of two faculty members from each college and two each from the library and the extra-collegiate extension faculty. Additional members from colleges are appointed in direct proportion to the colleges' representation in the Faculty Senate. The president of the Faculty Senate, in consultation with the cabinet, appoints the chair. The chair is either a member of the senate or a member of the Faculty Review Committee (FRC). Committee members, who are appointed by the Faculty Senate president in consultation with the

senate cabinet, must, at the time of their initial appointment, have served on their college or equivalent promotion and tenure committee or in the Faculty

Senate. Members cannot serve on a university or college promotion and tenure committee and on the Faculty Review Committee simultaneously. In the event that no eligible faculty members from a particular college or the library or the extra-collegiate extension are identified as willing and able to fill a vacant position on the FRC, the Faculty Senate president in consultation with the senate cabinet appoints the appropriate number of tenured faculty members to complete the membership of the Faculty Review Committee.

The functions of this committee are: to provide faculty review of faculty grievances and to consider appeals in the promotion and tenure or continued appointment process when the provost does not concur with a positive recommendation from the University Committee on Promotion and Tenure (section 3.4.5, "Appeals of Decisions on Reappointment, Tenure, or Promotion") or the University Committee on Promotion and Continued Appointment (section 4.5.5, "Appeals of Decisions on Reappointment, Continued Appointment, or Promotion").

The Faculty Review Committee reports annually to the Faculty Senate cabinet on the number of cases handled, their disposition, and on the effectiveness of the review process.

Separate grievance procedures exist for administrative and professional faculty and for special research faculty. See the relevant sections of this handbook for a description of those appeal processes.

## 1.4.5.3 Committee on Faculty Ethics

Purpose: The Committee on Faculty Ethics receives and considers charges of violations of faculty ethics that abrogate the principles of ethical behavior set forward in section 2.7, "Professional Responsibilities and Conduct." The Committee on Faculty Ethics (CFE) also acts to promote knowledge of and adherence to the principles of ethical behavior.

The CFE is not used to seek remedies such as monetary damages. The CFE does not act when legal, mediation, or other proceedings are initiated or are ongoing by complainants that render the CFE's pursuit of an investigation redundant or disruptive. The CFE does not act on matters that are referred or should be referred for investigation and action to an administrative officer, supervisor, or another appropriately charged committee in accordance with university policies and procedures. Complaints concerning personnel actions taken by a supervisor are handled by the applicable grievance procedure. If ethical issues arise from, or remain unresolved following such proceedings, the CFE is available to receive or consider charges of violations of the principles in section 2.7, "Professional Responsibilities and Conduct."

Composition: The Committee on Faculty Ethics is composed of one faculty member from each college, one from the library faculty and one from the extension faculty with continued appointment. The president of the Faculty Senate with the advice of the senate cabinet appoints committee members. All committee members hold tenure or continued appointment. The senate president designates one committee member to serve as chair. Voting members of the CFE serve

a two-year appointment and are limited to serving two consecutive terms. A quorum of the CFE consists of two-thirds of the appointed members.

While the Committee on Faculty Ethics is not a legal body, conflicts of interest may affect members' judgment on a given case. All members of the CFE are required to disclose any potential conflicts of interest with regard to cases that come before the CFE. Decisions concerning recusal are made by the individual member in consultation with the committee as a whole. In the event of a conflict of interest concerning the chair of the Committee on Faculty Ethics, the chair steps down from that position during the course of any discussion or investigation of that case. In such a situation, the president of the Faculty Senate appoints an interim chair from within the current CFE membership. Decisions with regard to the chair's recusal from the committee itself are made in consultation with the committee and the Faculty Senate president.

Procedures: Anyone may bring incidents to the CFE by writing to the chair of the CFE. The CFE will decide if a violation may have occurred and may conduct investigations, as it deems appropriate. The CFE will inform those involved of its findings and the processes and rationale by which the findings were formulated as described in the CFE operating procedures. When it is determined that a breach of faculty ethics has occurred, the CFE's findings along with its recommendations, will be reported to appropriate administrative personnel including, when appropriate, the Faculty Senate president. Strict confidentiality will be maintained. The CFE operating procedures are available on the Faculty Senate website.

The CFE chair will provide a summary of its activities to the Faculty Senate annually.

#### 1.5 Central Administrative Organization

#### 1.5.1 President

The president, the chief executive officer of the university, is selected by the board of visitors to execute approved policies and to administer the university, and holds office without term, subject to the pleasure of the board. The president is charged with the overall organization and supervision of the university and all of the state agencies, services, and regulatory activities, which have been placed under the president's supervision by the board of visitors. The president serves as the authorized officer through whom communication takes place between the board and the faculty, the board and the students, and the board and the other officers of administration or instruction employed by the university.

University officers who report to the president include the senior vice president and provost, vice president for administration, vice president for alumni relations, vice president for development and university relations, vice president for finance and chief financial officer, vice president for diversity and inclusion, vice president for information technology and chief information officer, university legal counsel, the chief of staff to the president, the executive director of governmental relations, and the director of intercollegiate athletics.

The directors and executive officers of the individual agencies, services, and regulatory activities report directly to the vice president under whom the president has placed each organization.

#### 1.5.2 Senior Vice President and Provost

The senior vice president and provost, who serves as the university's chief executive officer in the president's absence, assists the president in the administration, coordination, and development of the learning, discovery, and engagement programs. Reporting to this officer are the college deans, dean of University Libraries, executive directors of the Center for the Arts and the Virginia Tech Carilion Research Institute, director of the Institute for Creativity, Arts, and Technology, vice president and dean for graduate education, vice president for outreach and international affairs, vice president for research, vice president and executive director of the national capital region, vice president for student affairs.

In addition, the senior vice president and provost is responsible for maintaining the university's relations with the academic program function of the State Council of Higher Education for Virginia (SCHEV).

Also reporting the senior vice president and provost are the vice provost for undergraduate academic affairs, the vice provost for faculty affairs, the vice provost for enrollment and degree management, and the vice provost for resource management and institutional effectiveness.

Also reporting to the senior vice president and provost are the vice provost for undergraduate academic affairs, the vice provost for faculty affairs, the vice provost for enrollment and degree management, and the vice provost for resource management and institutional effectiveness.

#### 1.5.3 Vice President for Administration

Reporting to the president, the vice president for administration provides essential operating functions in support of the university's mission. The vice president assures a safe and supportive campus environment through proper management of the university's physical and human resources (excluding teaching and research faculty); planning, design, construction, maintenance, and operation of the physical plant and utilities; sustainability initiatives and programs, parking and transportation, mail services, printing services, campus planning; real estate services; space management; compliance with governmental, environmental, health, and safety regulations; university police (including the Virginia Tech Rescue Squad); emergency management; and human resource functions of the university, including organizational and professional development and equity and access including compliance with the Americans with Disabilities Act (ADA).

Reporting directly to the vice president are the associate vice presidents for facilities services and human resources, the assistant vice presidents for business services and university planning, the directors of emergency management and environmental health and safety, and the Virginia Tech police chief.

In addition, the vice president represents Virginia Tech with peer institutions, the local community, federal and state agencies, and corporate representatives to assist the university in meeting long- range objectives and emerging demands. The vice president also represents the university by participating as a member of related community/organization committees and boards.

#### 1.5.4 Vice President for Alumni Relations

The vice president for alumni relations oversees the mission and goals of alumni programs. This officer serves as secretary-treasurer of the Virginia Tech Alumni Association; coordinates committees and activities of the alumni board of directors; guides strategic planning and budgeting of alumni programs and services; and shapes the communication program that promotes alumni programs and alumni support of the university.

The Alumni Association has numerous programs and activities designed to promote Virginia Tech to alumni and friends of the university, enhance financial support, and encourage alumni involvement. Among the most prominent programs of the Alumni Association are its network of alumni chapters, class reunions and academic college homecomings, special programs and seminars, legislative advocacy, career services support, professorship and scholarship programs, and a merchandising/ group tours program. The Alumni Relations staff maintains records on the university's alumni and receives university internal requests for approved mailings to alumni.

Reporting to this vice president is the associate vice president for alumni relations.

## 1.5.5 Vice President for Development and University Relations

The vice president for development and university relations is responsible for the development and maintenance of effective relationships with the university's various constituencies, serving as the chief fundraiser and responsible for overseeing the university's overall communication strategy. This officer secures resources that enhance the academic quality of the institution; promotes public understanding and support for Virginia Tech; enhances the perception and knowledge of various publics of the university's programs and accomplishments; and creates a positive public impression of the university and its faculty, staff, students, and programs.

Reporting directly to the vice president are the senior associate vice president for development and principal gifts, and the associate vice presidents for development, advancement services, and university relations. The vice president works closely with the Virginia Tech Foundation, the Alumni Association, and the Virginia Tech Athletic Fund to focus on overall advancement of the university.

This officer also serves as one of the two executive vice presidents of the Virginia Tech Foundation.

## 1.5.6 Vice President for Diversity and Inclusion

The vice president for diversity and inclusion provides leadership and direction for the continuous assessment and improvement of university-wide efforts to create and sustain a more diverse and inclusive community of learners. This senior officer reports directly to the president, and advises the president and university leadership on policies, programs, practices, and resources needed to achieve excellence, diversity, and effectiveness in the learning, discovery, and engagement activities of the university.

The vice president for diversity and inclusion works collaboratively with other university units to build a capacity for recognizing and valuing difference. The vice president encourages and supports positive interactions between and among diverse populations, and maximizes the benefits of a multicultural perspective. In collaboration with individuals and organizational units across the campus, the vice president assists and advocates for those persons who have special needs and/or significant cultural differences, and who may have concerns about their experiences at Virginia Tech.

The vice president's team is comprised of the directors for business affairs, special projects, diversity education/training, and multicultural student engagement/outreach; the manager of diversity initiatives; and the coordinators for outreach/education and multicultural alumni outreach/programs.

#### 1.5.7 Vice President for Finance and Chief Financial Officer

The vice president for finance and chief financial officer supports the university leadership by identifying, obtaining, and allocating the resources needed to achieve Virginia Tech's mission and the goals and objectives of the university's strategic plan. This officer develops innovative solutions to challenging problems by using sound financial management practices and extensive financial planning. The vice president develops and maintains active relationships with key staff in the Commonwealth of Virginia's executive branch and general assembly to ensure a positive and effective working environment for the university. The vice president oversees the university accounting processes (including financial relationships with the commonwealth and regulatory relationships with the state and federal governments); supports the university's internal operations with innovative procurement operations; oversees an identification/debit/security card system known as Hokie Passport; and enables the development and enhancement of effective business practices throughout the university, including financial/business information systems that provide electronic paperless services to faculty, staff, and students.

Reporting directly to this vice president are assistant vice president for budget and financial planning, university bursar, assistant vice president of capital assets and financial management, university controller, coordinator of university budget and financial systems, and the directors of Hokie Passport, internal audit, purchasing, surplus property, and risk management.

## 1.5.8 Vice President for Information Technology and Chief Information Officer

The vice president for information technology and chief information officer reports directly to the university president and manages Virginia Tech's information technologies.

Under the vice president's direction, the information technology organization provides, secures, and maintains information systems that allow the university to accomplish its missions of learning, discovery, and engagement. The vice president fosters outreach by developing partnerships with communities and promoting the capabilities of advanced networking and communications. As a national leader in building technological resources, the information technology organization supports and preserves the work of Virginia Tech's faculty, staff, and students.

Reporting to this vice president are the deputy chief information officer and director of administration and planning; the associate vice presidents for learning technologies, enterprise systems, and research computing; the executive directors of converged technologies for security, safety, and resilience, network infrastructure and services, and strategic initiative; the director of secure enterprise technology initiatives; the information technology security officer; and the chief technology architect.

## 1.5.9 University Treasurer

The university treasurer manages the financial assets and debt of the university.

University public funds are managed in accordance with state law and investment policy in the name of the university. All financial assets supporting the programs of Virginia Tech are managed in accordance with university goals and objectives. Debt is issued in the name of the university as authorized by the university's board of visitors and is managed to maintain the university's debt rating of AA by Standard & Poor's and Aa2 by Moody's.

## 1.6 University Academic Administration

## 1.6.1 College Deans

The deans of the colleges of Agriculture and Life Sciences, Architecture and Urban Studies, Pamplin College of Business, Engineering, Liberal Arts and Human Sciences, Natural Resources and Environment, Science, and the Virginia-Maryland Regional College of Veterinary Medicine are responsible for the academic activities of their respective colleges.

These responsibilities include the allocation and administration of resources, appointment and evaluation of faculty and support staff, and curriculum development. The department heads or chairs in each college report directly to their respective deans for all matters related to the programs of the colleges.

The college deans are appointed by the senior vice president and provost, and may be reappointed indefinitely. Periodic evaluations of their effectiveness in this capacity occur every five years.

## 1.6.2 Departmental Administration

The colleges are subdivided into academic departments and/or schools. Departments are under the supervision of department heads, chairs, or school directors, who report to the dean of the college.

Department heads or chairs are responsible for the growth and vigor of the academic programs, for the recruitment of new faculty, and for the administration of the curriculum and the budget of their departments. In certain cases, some of these responsibilities may be delegated to program chairs of specific disciplines.

Department heads or chairs serve for fixed-length terms, specified by the dean. The dean, in consultation with departmental faculty, takes into account the periodic review required prior to reappointment and decides the length of term and procedures for renewal. The president or the senior vice president and provost authorizes the appointment.

As an integral part of shared governance of the university, faculty committees are formed in departments to make recommendations and otherwise assist the head or chair in curricular modification, in the selection of new faculty, and in the determination and application of policies.

## 1.6.3 Dean of University Libraries

The dean of University Libraries directs the University Libraries in providing the Virginia Tech community with information collections and services necessary to support the learning, discovery, and engagement programs of the university. The dean allocates and administers resources, and appoints and evaluates faculty and staff in support of University Libraries goals. The dean reports to the senior vice president and provost.

Reporting directly to this dean are the associate dean for learning and outreach, associate dean for research and informatics, and directors of assessment and access services, collection and technical services, facilities, information technology services, research/instructional services, and special collections.

The dean of the libraries is appointed by the senior vice president and provost, and may be reappointed indefinitely. A periodic evaluation of his or her effectiveness in this capacity occurs every five years.

## 1.6.4 Director of Virginia Cooperative Extension

The director of Virginia Cooperative Extension (VCE) reports to the dean of the College of Agriculture and Life Sciences and is responsible for the administration of VCE programs in cooperation with the U.S. Department of Agriculture and state and local governments, as well as

the Cooperative Extension <u>Service</u> Program at Virginia State University. The director is responsible for the VCE programs in agriculture and natural resources, family and consumer sciences, 4-H youth development, and community viability. Since three colleges provide these services, the director administers these programs under the guidance of a committee chaired by the senior vice president and provost. Committee members include the vice president for finance and chief financial officer, vice president for outreach and international affairs, vice president for research, director of VCE, director of the Agricultural Experiment Station, and deans of the colleges of Agriculture and Life Sciences, Natural Resources and Environment, and Veterinary Medicine, .

#### 1.6.5 Vice President and Dean for Graduate Education

The vice president and dean for graduate education is responsible for the development, administration, and evaluation of graduate education throughout the university. The graduate education program currently has over 4,500 graduate students enrolled on campus and an additional 2,400 graduate students enrolled at several extended campus centers in Virginia. The vice president and dean for graduate education reports to the senior vice president and provost.

The vice president and dean for graduate education works in partnership with faculty, program chairs, department heads or chairs, deans, graduate students (particularly through the Graduate Student Association), and the Commission on Graduate Studies and Policies (CGSP) to further graduate education at Virginia Tech. The vice president and dean develops academic policies and procedures; assists in preparing new graduate degree options/programs; implements graduate enrollment management strategies; oversees the graduate admissions process and the progress of graduate students; accepts and approves theses/dissertations; certifies successful completion of graduation requirements; manages tuition remissions and scholarship/fellowship programs; implements the health insurance subsidy program; coordinates university-wide graduate recruitment efforts; coordinates university-wide graduate student orientation and graduate teaching assistant training; assures federally-mandated compliance; and administers the graduate honor system.

Reporting directly to the vice president and dean are the associate deans, and the directors of graduate admissions and academic progress, graduate student recruiting, graduate student services, and the graduate student ombudsperson.

## 1.6.6 Vice President for the National Capital Region

The vice president for the national capital region assists the president and the senior vice president and provost in developing and implementing strategic directions for the region. In the National Capital Region (NCR), this vice president coordinates services and operations across multiple locations; enhances the visibility and presence of Virginia Tech and its programs; develops new opportunities for teaching and learning, discovery, and engagement; assists in the development of viable business models for programs; and leverages assets of the region to faculty on the main campus in Blacksburg.

Reporting to this vice president are the deputy director of the NCR regional administration; directors of the center for technology, security, and policy, the global issues initiative, the research development team, and information services; the associate director of research program development; and the manager of public relations and marketing.

#### 1.6.7 Vice President for Outreach and International Affairs

The vice president for outreach and international affairs leads the university's efforts to be fully engaged in society, here and around the world. The vice president is responsible for facilitating the exchange of knowledge that benefits both the university and society and for helping faculty disseminate research results that assist communities, business, non-profit and non-governmental organizations, and governments to solve real world societal and natural resource problems. This senior officer works with communities to address local and global economic development needs; manages U.S. government-funded projects that improve food security and natural resources management around the world; arranges opportunities for students to serve communities and organizations while learning; tracks Virginia Tech's international agreements with institutions abroad; provides visa assistance for the university's international visitors; oversees Virginia Tech centers abroad; explores new ways to leverage university resources for an effective international agenda; and spearheads implementation of the university's international strategic plan. The vice president tailors regional and online graduate and lifelong learning educational opportunities through continuing and professional education programming conducted at The Inn at Virginia Tech, Skelton Conference Center, the Hotel Roanoke and Conference Center, and at commonwealth campus centers throughout Virginia.

Reporting to this vice president are the associate vice president for international affairs; directors of outreach program development, the center for organizational and technological advancement, economic development, outreach finance and administration, Reynolds Homestead, extended campus sites in southwest Virginia, Roanoke, Richmond, and Hampton Roads, outreach programs in Southside Virginia and the Institute for Advanced Learning and Research (IALR); chair of VT-STEM (science, technology, engineering, and math) K-12 outreach; international fellow of the Center for European Studies and Architecture (CESA); and the outreach fellow for fine arts.

#### 1.6.8 Vice President for Research

Working closely with the college deans, the vice president for research provides administrative leadership to support, enhance, and advance the overall discovery mission of the university. This officer reports directly to the senior vice president and provost.

The vice president directs both operational and interdisciplinary research units within the Office of the Vice President for Research. Operational units include the Offices of Sponsored Programs, Research Compliance, Research Integrity, Laboratory Animal Resources, Finance and Controls, Research Administrative Programs, Information Technology, Research Education, Human Resources for research faculty, and Research Communications. The vice president is responsible for fostering quality research and scholarship; providing access to funding information; assisting faculty in securing and managing external funding; coordinating large-scale research projects;

enhancing the response to national research priorities and opportunities; encouraging and directing inquiry into new and emerging fields; promoting solutions and advances of basic knowledge in diverse disciplines through interdisciplinary cooperation; ensuring compliance with regulatory policies and ethical research conduct policies; supporting and promoting the research, scholarship, and creative activity of Virginia Tech's faculty, staff, and students; marketing faculty talent and university capabilities; and fostering partnerships with outside agencies, academic institutions, community organizations, and industry.

The vice president serves as chairman of the board of Virginia Tech Intellectual Properties, Inc. and works with the Virginia Center for Innovative Technology and other agencies to market and commercialize the research results of faculty, staff, and students.

Reporting to this vice president are the associate vice presidents for research, research computing, research compliance, research programs; president of Virginia Tech Intellectual Properties (VTIP); assistant vice presidents for sponsored programs administration and finance and controls; university veterinarian and director of laboratory animal resources, research integrity officer; and directors of the Fralin Life Science Institute, Institute for Society, Culture, and Environment, Virginia Bioinformatics Institute, Virginia Tech Transportation Institute, and university research centers. Reporting indirectly to this vice president are the director of the Institute for Critical Technology and Applied Science (ICTAS) and the executive director of the Virginia Tech Carilion Research Institute, and the director of the Institute for Creativity, Arts and Technology.

#### 1.6.9 Vice President for Student Affairs

The vice president for student affairs reports directly to the senior vice president and provost, and provides leadership to the university's services and programs in the Division of Student Affairs. The vice president is responsible for promoting students' academic success, personal growth, and life skills development; guiding the success of students from orientation through graduation; creating environments, programs, and services that support the university's teaching and learning focus and its mission to expand personal growth and opportunity, advance social and community development, foster economic competitiveness, and improve the quality of life; creating environments that are welcoming to all; promoting supportive and inclusive communities that contribute to a just and caring university; and offering high quality environments, programs, and services with cost-effective administration.

Reporting to this vice president are the directors of career services, campus alcohol abuse prevention center, Cook Counseling Center, Cranwell International Center, dining services, fraternity and sorority life, housing and residence life, multicultural programs and services, office of student conduct, recreational sports, Schiffert Health Center, services for students with disabilities, student centers and activities; the commandant of the Virginia Tech Corps of Cadets; and the dean of students.

#### CHAPTER TWO

#### 2.0 Employment Policies and Procedures for All Faculty

#### 2.1 Types of Positions

In compliance with regulations and statutes of the Commonwealth of Virginia, university employees fall into four distinct categories: (1) classified staff covered by the state personnel system, (2) university staff covered by the university human resources system, (3) faculty, and (4) non-salaried (wage) employees.

The university employs approximately 13,000 people on campus, throughout the state, and around the world. These employees are in a variety of position types that include teaching and research faculty (T&R), research faculty, administrative and professional faculty (A/P), adjunct wage faculty, staff (classified and university), wage, student wage, emergency hires, and others. The majority of positions are either faculty or staff.

Faculty employment is under the purview of the board of visitors, which is responsible for establishing faculty employment policies. The faculty category includes several distinct subsets of positions.

## **2.2** Faculty Definition

All faculty members are designated as members of the general faculty of the university. The general faculty is composed of those faculty members outside the staff personnel system who are appointed to carry out the learning, discovery, and engagement programs of the university; carry out general university administration; or provide academic support to those programs.

For the purpose of designation within the university and for the application of faculty policies, including especially those related to promotion and tenure or continued appointment, the general faculty is divided into five categories: collegiate faculty, library faculty on the continued appointment-track, extension faculty on the continued appointment-track, administrative and professional faculty, and research faculty.

Faculty members may be assigned to one of the following standard faculty ranks: extracollegiate instructor (in library or extension), assistant professor, associate professor, or professor.

Or, they may be assigned to one of the following non-tenure instructional ranks: lecturer, visiting assistant professor, visiting associate professor, visiting professor, adjunct associate professor, adjunct professor, assistant professor of practice, associate professor of practice, professor of practice, clinical instructor, clinical assistant professor, clinical associate professor, clinical professor, instructor, advanced instructor, or senior instructor.

Or, faculty members may be assigned to one of the following research faculty ranks: research associate, senior research associate, research scientist, senior research scientist, postdoctoral

associate, project associate, senior project associate, project director, research assistant professor, research associate professor, or research professor.

Specification of faculty rank in the library faculty, extension faculty, or the administrative and professional faculty does not imply a particular rank in any collegiate department. Library and extension faculty may be invited to hold concurrent adjunct status in a collegiate department in order to formally recognize their contributions to the undergraduate or graduate program. Policies regarding the assignment of a standard faculty rank in a collegiate department for an administrative or professional faculty member are described in section 7.1, "Faculty Rank and Title."

Typically, the Commission on Faculty Affairs provides oversight of employment policies for the collegiate, library, and extension faculty on continued appointment-track, while the Commission on Administrative and Professional Faculty Affairs provides oversight for administrative and professional faculty, and the Commission on Research provides oversight for research faculty.

#### **2.3** Faculty Categories

### 2.3.1 The Collegiate Faculty

The collegiate faculties are composed of teaching and research (T&R) faculty members with a standard rank or non-tenure-track instructional rank who are appointed to full- or part-time positions in academic departments or schools in the eight colleges. (Subsequent references to departments or schools in this handbook are subsumed under the term "department.")

T&R faculty are teaching faculty members with substantial responsibilities for undergraduate and graduate instruction, research, and outreach. Research and extension specialist positions are also T&R faculty, although they may have a relatively small instructional component. The T&R positions may be tenure-track with the ranks of assistant, associate, or full professor. Typically such positions would be regular and require a terminal degree in the field. They may also be non-tenure-track, such as the instructor ranks, clinical faculty ranks, or professor of practice ranks. These positions may be either restricted or regular.

Members of the collegiate faculty who relinquish full-time responsibilities in a collegiate department may choose to continue to have their professional development evaluated by that department and college. The same is true for someone who accepts a position in the library faculty, Virginia Cooperative Extension faculty, or in the administrative and professional faculty. The evaluation for promotion in rank or the granting of tenure is according to the criteria and procedures of promotion and tenure for collegiate faculty. Tenure already granted need not be relinquished. Annual evaluation for merit salary adjustment is based on the responsibilities of the current position.

## 2.3.2 University Libraries Faculty with Continued Appointment or on the Continued Appointment-Track

University Libraries faculty may or may not hold appointments as a collegiate faculty member. They perform a unique and indispensable function in the educational process and share many of the professional concerns of their collegiate colleagues. The university recognizes the need to protect the academic freedom of librarians in their responsibility to ensure the availability of information and ideas, no matter how controversial, so that teachers may freely teach and students may freely learn.

Employment policies and procedures for library faculty with continued appointment or on the continued appointment-track are described in chapter four of this handbook.

## 2.3.3 Virginia Cooperative Extension Faculty with Continued Appointment or on the Continued Appointment-Track

Extension faculty may or may not hold appointment as a collegiate faculty member. They are, nonetheless, subject to high professional standards in the dissemination of knowledge through the Virginia Cooperative Extension programs and in the planning and delivery of educational programs and programs of assistance to industries and local governmental agencies. In these functions, extension faculty share many of the professional concerns of their collegiate colleagues, including the need for the protection of academic freedom in these responsibilities.

Employment policies and procedures for extension faculty with continued appointment or on the continued appointment-track are described in chapter four of this handbook.

### 2.3.4 Administrative and Professional Faculty

Administrative and professional (A/P) faculty may or may not hold appointment as a collegiate faculty member. Policies regarding the assignment of a standard faculty rank in a collegiate department for an administrative or professional faculty member are described in section 7.1, "Faculty Rank and Title."

Administrative faculty, also referred to as senior administrators, typically serve in executive level leadership roles such as vice president, dean, assistant or associate vice president or dean, or director of a major unit. They perform work directly related to management of the institution, college, or an administrative department.

Professional faculty, also referred to as managers and professionals, may direct or provide support for academic, administrative, extension, outreach, athletic, or other programs. They may also provide vital university functions such as information technology, budget or finance, human resources, public relations, development, and architectural or engineering functions.

Administrative and professional faculty employment policies and procedures are described in chapter seven of this handbook.

#### 2.3.5 Research Faculty

Faculty designated to promote and expedite university research activities and who have responsibilities primarily in the research area are considered research faculty. This is a large and growing category of positions at Virginia Tech. Research faculty are typically employed on sponsored grants and contracts for a restricted period to carry out research or outreach projects as part of the university's research and outreach missions.

Research faculty ranks are: research associate, senior research associate, research scientist, senior research scientist, postdoctoral associate, project associate, senior project associate, project director, research assistant professor, research associate professor, and research professor.

Research faculty employment policies and procedures are described in chapter six of this handbook.

#### 2.3.5.1 Affiliated Research Faculty

Occasionally, individuals outside of university employment are identified to team with university faculty to enhance research opportunities through departments. To support these associations, the university has developed the affiliated research faculty program. Affiliated research appointments may be made for individuals connected to specific academic departments, or may be made through research institutes or centers. Affiliated research appointments may also be established to facilitate research partnerships. This program may address occasions where a university faculty member has a spouse or partner who also has professional academic credentials, but who has not found appropriate employment opportunities. The vice president for research reviews nominations from departments, colleges, or the provost, and approves appropriate applications. (See section 6.7, "Affiliated Research Faculty," for more information.)

#### 2.3.6 Non-Tenure-Track Instructional Faculty

The following titles are reserved for non-tenure-track faculty in instructional positions: lecturer; visiting assistant professor, visiting associate professor, and visiting professor; adjunct assistant professor, adjunct associate professor, and adjunct professor; assistant professor of practice, associate professor of practice, and professor of practice; clinical instructor, clinical assistant professor, clinical associate professor, and clinical professor; instructor, advanced instructor, and senior instructor.

Non-tenure-track instructional faculty employment policies and procedures are described in chapter five of this handbook. Information regarding appropriate credentials for instructional faculty is found in section 2.5.9, "Faculty Credentialing Guidelines," and on the provost's website.

#### 2.4 Faculty Compensation Plan

The board of visitors annually approves a faculty compensation plan, which is prepared using the parameters provided by the commonwealth's secretary of education in the Consolidated Salary

Authorization for Faculty Positions in Institutions of Higher Education. In accordance with the Consolidated Salary Authorization, the faculty compensation plan provides information about (1) the promotion and tenure process, (2) the annual evaluation and salary adjustment process for teaching and research (T&R) faculty, administrative and professional (A/P) faculty, and special research faculty, (3) salary adjustments within the evaluation period, and (4) the pay structure.

## 2.5 General Procedures for Faculty and Administrative Searches and Appointments

It is the policy of Virginia Tech to provide equal opportunity for all qualified individuals while rejecting all forms of prejudice and discrimination, including those based on race, gender, disability, age, veteran status, national origin, religion, sexual orientation, or political affiliation. Anyone having questions concerning discrimination should contact the Office for Equity and Access.

Virginia Tech is committed to ensuring that all qualified individuals with disabilities have the opportunity to take part in educational and employment programs and services on an equal basis. The aim is to provide this opportunity in an integrated setting that fosters independence and meets the guidelines of the Americans with Disabilities Act (ADA) and the Rehabilitation Act of 1973.

Reasonable accommodations are made on an individual and flexible basis. Appropriate services include: (1) support, counseling, and information, (2) academic assistance services, (3) referral services, or (4) environmental modifications. However, it is the responsibility of individuals with disabilities to make their needs known and to provide documentation of a disability. It is the responsibility of the supervisor to comply with accommodation requests made by appropriate university offices. Appeals of requests are made through procedures stated in university policy 4075, "Accommodation of Employees with Disabilities."

Virginia Tech is committed to increasing the number of women and underrepresented faculty and administrators. This commitment is stated and elaborated in the Affirmative Action Program, Executive Order 11246, and other documents filed with federal and state officials. The guidelines below are designed to ensure that searches are conducted affirmatively, resulting in greater faculty diversity.

#### **2.5.1** Faculty Search Procedures

Faculty search resources are available on the human resources website. Please refer to that website for detailed information on the search process. The procedures found there apply to all types of full-time, regular faculty positions. Search waivers may be approved under specified circumstances. Search procedures for research faculty are available on the research website.

In general, the procedures call for the establishment of a representative search committee once approval for the position is granted; development of a tailored, aggressive search strategy that usually includes national advertising in appropriate journals in the discipline; personal contacts with colleagues; follow up with female and underrepresented colleagues and doctoral students

listed in relevant directories; and other targeted efforts to identify a strong and diversified pool of candidates. Prior to selecting candidates for interview, the search committee head reviews the diversity and strength of the candidate pool with the dean, who makes a judgment as to whether additional recruitment efforts should be made. The committee reviews applications once a representative pool is established or recruitment strategies are exhausted. A limited number of candidates are usually invited for on-campus interviews. Additional guidance on the conduct of the search and interview process for department head or chair, dean, vice president, or president is provided in subsequent sections of this handbook.

Prior to tendering an offer, the department head or chair reviews the search and interview process with the dean.

Review and recommendation by the applicable departmental or school committee dealing with promotion and tenure or continued appointment shall be sought before a decision is made to extend to a candidate a firm offer that includes the granting of tenure or continued appointment, or the award of a rank higher than assistant professor.

### 2.5.2 Appointment of Department Heads or Chairs

When a vacancy occurs, the college dean requests the department to nominate members of its faculty for service on a search committee. The dean appoints the committee from among those nominated and may appoint additional members who shall constitute a minority of the committee.

The committee elects its chairperson and meets with the college dean to determine appropriate conditions of the position, such as rank and available resources. The dean should share with the search committee a realistic assessment of the college and university commitment to the department and its programs.

The position is nationally advertised as described in section 2.5.1, "Faculty Search Procedures," unless the dean and the committee agree that the position should be considered a promotional opportunity restricted to candidates from faculty of the department without national advertisement. Such a decision should be reached only in a department having a quality of program and a maturity of development to afford several well-qualified candidates from within its own ranks. The decision may be influenced by the lack of a vacant faculty position in the department.

After professional credentials of candidates are reviewed, references and colleagues of the best-qualified candidates are consulted. A limited number of candidates (ordinarily three) are invited, on approval of the college dean, to visit the campus. The search committee, the college dean, and university officials, as available and appropriate, interview the candidates. They also meet with selected students and faculty members in the department. The committee should take care to provide internal candidates with fair opportunities to make their qualifications equally well known.

The search committee seeks advice from those who meet with the candidates and makes its recommendations on the preferred candidate(s) to the college dean. The head or chair is recommended for appointment by the dean to the provost, but only after extensive consultation with the department faculty. It should, in effect, be a joint process. The appointment is subject to board of visitors' approval.

#### 2.5.3 Appointment of Deans

When a vacancy occurs, the provost requests nominations for membership on a search committee from the appropriate faculty members or faculty association. When a vacancy occurs in an academic deanship that has university responsibility cutting across college lines, the search committee shall include faculty representatives from all appropriate colleges. The provost appoints this search committee from the list of nominees and may appoint additional members who shall constitute a minority of the committee. The provost or designee serves as chairperson of the search committee. Ordinarily a national search is conducted.

After professional credentials of candidates are reviewed, references and colleagues of the best-qualified candidates are consulted. A limited number of candidates are invited to visit the campus. The search committee, representative department heads/chairs, academic deans, the vice presidents, and the president interview the candidates. Candidates also meet with selected students and faculty members of the college. The committee should take care to provide internal candidates with fair opportunities to make their qualifications equally well known.

The provost seeks advice from those who meet with the candidates and seeks agreement with the search committee on the candidate(s) to be recommended. The provost's recommendation is made to the president, who then authorizes the extension of an offer. The appointment is subject to board of visitors' approval.

### 2.5.4 Appointment of Associate and Assistant Deans

When a vacancy occurs in the position of associate dean, assistant dean, or assistant to the dean, and the position does not involve responsibility for assignment of faculty activities or recommendations on salaries and promotions, it is filled on recommendation by the dean to the provost and the president. Department heads or chairs and representative faculty should be consulted; a formal search committee is used if the appointment is not limited to an internal promotional opportunity. If the position involves responsibility for assignment of faculty activities or recommendations on salaries and promotions, the search and selection procedures are similar to those used for deans, but the dean serves as chairperson of the search committee.

## 2.5.5 Appointment of Senior Vice President and Provost and Vice Presidents

When a vacancy occurs, the president determines the procedures that will be used for identifying qualified candidates, including the decision to engage a search firm and/or to appoint a university search or screening committee. Where the position involves considerable interaction with the academic faculty, significant engagement of faculty members in the search and/or interview

process is desirable and expected. The board of visitors retains authority to approve appointments of all vice presidents.

## **2.5.6 Appointment of the President**

The board of visitors establishes the procedures for the selection of a president when the vacancy is announced.

#### 2.5.7 Conviction and Driving Record Investigation

In order to protect university interests, institutional resources, and the welfare of its students, employees, and the public, the university conducts criminal background and/or driving record checks on candidates, according to the provisions in university policy 4060, "Conviction and Driving Record Investigation."

The university recognizes that its interests in investigating employees' backgrounds must be balanced with the need to protect the privacy of employees and prospective employees. University policy and state and federal laws recognize the individual's right to privacy and prohibit university employees from seeking, using, or disclosing information except within the scope of their assigned duties. Any information related to personal history, including conviction records, is maintained in confidence. Such information is shared only with the hiring official, university legal counsel, and the Virginia Tech Police, as necessary.

A preliminary offer may be made to the selected candidate, contingent on the results of the check. However, at no time should the selected candidate be allowed to begin work before completion of the process.

Human resources coordinates the conviction and driving record check process.

#### 2.5.8 Terms of Faculty Offer and Final Approval of Appointment

All new appointments and reappointments are documented in the terms of faculty offer prepared by the department head or chair, approved according to procedures established by the dean or senior manager, signed by the candidate, and forwarded to human resources. The terms of offer (and the reappointment contract) for research faculty require prior review and approval by the Office of the Vice President for Research. See the human resources website for the terms of faculty offer templates for each type of faculty appointment. The terms of faculty offer documents the tenure or continued appointment status (tenured, tenure-track, non-tenure track, continued appointment, or continued appointment-track), appointment status (regular or restricted, effective date and, if restricted, an end date), the appointment period (academic or calendar year) and length of the appointment, assigned faculty rank, and other conditions relevant to the employment offer. If the appointment is tenure-track or continued appointment-track, reference to prior service credit should be addressed if relevant (see sections 3.4.3 and 4.5.3, "Guidelines for Calculation of Prior Service").

All letters of appointment shall make reference to further terms and conditions of employment contained in the Faculty Handbook.

The terms of faculty offer for a restricted appointment must clearly define the length of the appointment. In cases where there is no expected opportunity for continuation, the document also serves as a notice of termination. Continuation of a restricted appointment, even during the specified appointment period, is subject to the availability of funds, the need for services, and satisfactory performance; this information is included in the terms of offer template. Related letters of offer or reappointment should not contain promises that the hiring unit may not be able to keep; the university looks to the department to make good on defaults. The department head/chair's approval is required before an offer is extended.

The board of visitors holds the authority to approve all faculty appointments. The authority has been delegated to university officials for certain types of new appointments, generally including non-tenure positions and restricted appointments. Final approval by the board of visitors is required for new appointments of instructional and research faculty members on the tenure-track or continued appointment-track, including those appointed with tenure or continued appointment; faculty-ranked athletic personnel; senior administrators (such as deans and vice presidents) and their direct reports; and administrative and professional faculty members reporting directly to the president and their direct reports.

#### **2.5.9** Faculty Credentialing Guidelines

In accordance with federal law, on the first day of their employment, new employees must provide documentation of U.S. citizenship or lawful authorization to work in the United States.

For regional accreditation purposes, Virginia Tech must justify and document the qualifications of all instructors of record as outlined by the Southern Association of Colleges and Schools Commission on Colleges (SACS) faculty credentials guidelines. All faculty members, including part-time and adjunct faculty, must hold credentials appropriate to the level and subject matter they are teaching. Employment is contingent upon presentation of satisfactory documentation of credentials prior to employment.

Documenting the credentials of teaching faculty is the responsibility of the department where the course originates or is listed. This may be different than the employing department in some cases.

A completed transcript cover sheet signed by the department head/chair and an original transcript from the institution awarding the highest degree are required of all new teaching faculty members upon employment. An original transcript has an official raised seal from the institutional source; copies are not acceptable.

SACS guidelines recommend the following credentials for instructors of record by course level:

- Baccalaureate/undergraduate courses, a doctorate, terminal, or master's degree in the teaching discipline or a master's degree with a concentration in the teaching discipline (minimum of 18 graduate semester hours in the teaching discipline) is required;
- Graduate/post-baccalaureate courses, an earned doctorate/terminal degree in the teaching discipline or a related discipline is required.

The university's processes for collecting and storing the documentation of a faculty member's credentials are as follows:

- Salaried faculty members: a transcript cover sheet and the original transcript must be submitted to <u>Human Resources</u> (mail code 0318). After these are scanned into the university's document repository system, the originals will be returned to the faculty member. It is suggested that the faculty member's home department retain copies of these documents prior to submission to Human Resources.
- Adjunct and wage faculty appointments: the transcript cover sheet and original transcript must be submitted to the Office of Institutional Research and Effectiveness (mail code 0433). After these are scanned into the university's document repository system, the originals will be returned to the faculty member. It is suggested that the department retain copies of these documents prior to submission to Institutional Research.

If the instructor of record is not a faculty member in the course-origination department, that department head or chair is responsible for verifying that the faculty member's teaching credentials are appropriate for the course content and level. In such cases, a copy of the original transcript for the course-origination departmental file is acceptable if the faculty member or graduate assistant has an official transcript documented elsewhere at Virginia Tech.

Faculty credentialing guidelines and the transcript cover sheet are available on the provost's website.

Faculty credentialing guidelines for Administrative and Professional faculty are found in chapter seven of this handbook.

#### 2.6 Terms of Appointment

#### **2.6.1** Academic Year Appointments

New faculty appointments and renewals of term appointments are made in writing by the department head, chair, or dean.

Most faculty appointments in the academic units of the university are for the nine-month academic year; they are called academic year appointments. While the contractual payroll dates for the academic year are August 10 through May 9, faculty are expected to be available two weeks prior to the first day of classes and two weeks following commencement. No annual leave is awarded within the academic year, but the discretion of the department head or chair is recognized in assigning duties during periods when the university is not in session. Faculty members are expected to inform their department heads or chairs of their whereabouts during such periods.

Although the annual salary assigned for an academic year appointment covers only the academic year, the salary is paid in 24 semi-monthly installments over the calendar year, with payment occurring on the first and sixteenth day of each month. (If that day of the month falls on a Saturday, the payment is made on the preceding Friday; if Sunday, the payment is made on the

following Monday.) Payment is made directly to the faculty member's bank or financial establishment.

Faculty members whose appointments are for only part of the academic year receive a pro rata portion of the annual salary. Details of the calendar of such payments are available from human resources.

Faculty members on academic year appointments whose employment with the university ceases at the end of the academic year—or any academic term—may request (with proper notice) that all remaining installments of their earned salary be paid on the next available payroll after human resources is notified and employment has ceased.

#### 2.6.2 Calendar Year Appointments

Some faculty members have assigned responsibilities that extend throughout the calendar year, largely independent of the academic calendar. Such faculty members are on calendar year appointments with work assignments covering the full 12 months except for periods of annual leave. The kinds of positions that may call for calendar year appointments include department heads or chairs, administrative and professional faculty, and research faculty.

Faculty who assume calendar year appointments while serving in department head or other administrative roles retain the calendar year appointment only for the duration of the assignment. Upon returning to an instructional faculty position in a department characterized by academic year appointments, the faculty member resumes an academic year appointment with a corresponding adjustment in salary. (Instructional faculty on calendar year appointments prior to assuming the administrative assignment normally resume their prior calendar year appointment and salary upon completion of the assignment.)

Conversions of appointment from academic year to calendar year or reverse (or to any other appointment period acceptable under university policy) are done in accordance with standard formulas approved by the provost or vice president for administration. Any exception requires approval by the provost or vice president for administration, depending upon the reporting structure.

#### 2.6.3 Research Extended Appointments

Under certain conditions, faculty members on academic year appointments have the opportunity to extend their base nine-month appointment to a 10-, 11-, or 12-month appointment reflecting the faculty member's sponsored research responsibilities.

Academic year faculty members who have approved research extended appointments may earn and accrue annual leave proportional to their appointments. Faculty members with one, two, or three months of sponsored funding are strongly urged to convert their nine-month appointment to a 10-, 11-, or 12-month research extended appointment, which entitles them to earn and use annual leave and have the summer pay included as creditable compensation for retirement

purposes in accordance with university policies. Unused annual leave will not be compensated at the time of reconversion or separation.

Faculty members requesting a research extended appointment complete the request form available on the provost's website. Documentation of the additional months of funding must be provided. Requests for research extended appointments require approval of the department head or chair, dean (or appropriate administrator), and provost or vice president for administration (or their designees).

Research extended appointments are renewed annually with verification of sponsored funding by the department head or chair to support the continuation. The continuation request form is also available on the provost's website.

Further information regarding employment policies and practices for research faculty is available in chapter six of this handbook.

#### 2.6.4 Restricted Appointments

Appointments to the general faculty may be made for which there is no presumption or consideration of renewal. Such appointments are called "restricted" and should be so designated, with a specified term, in the terms of faculty offer. They are in contrast to renewable term appointments (often called "probationary," "tenure-track," or "continued appointment-track" appointments), tenured appointments, continued appointments, and year by year appointments of administrative and professional faculty, all of which are categorized as "regular" appointments. Restricted appointments are commonly made in the cases of research faculty employed to work on projects with external funding, visiting professorships, and other temporarily available faculty positions. See section 2.5.8, "Terms of Faculty Offer and Final Approval of Appointment."

When a person on a restricted appointment is to be continued, a formal reappointment is required and should be issued prior to the end of the existing contract. The reappointment contract again defines the conditions of the appointment. Any changes should be made explicit. If a salary increase is approved, it should be a part of the reappointment contract. The reappointment contract requires the prior approval of the department head or chair, dean, and the Office for the Vice President for Research.

Faculty members on restricted appointments earn sick leave at the rate of 5 hours per pay period under the policy that was standard for all faculty members before September 1, 1981. Sick leave does not extend beyond the date of termination of appointment. Faculty members on calendar year restricted appointments earn annual leave at the same rate as faculty on regular appointments, but earned annual leave must be taken during the term of appointment; accrued annual leave will not be paid on termination of appointment. Restricted faculty who are eligible to earn annual and sick leave may carry over their unused balances to the next leave year; however, they are not paid out upon separation.

## 2.6.5 Summer and Winter Appointments

Faculty on academic year appointments may be invited by the department head or chair to teach one or more courses in summer session for special compensation. Maximum compensation is set at 11.25 percent of the faculty member's annual salary for each scheduled three-credit semester course taught, subject to a salary limit that is determined each year.

No summer appointments are made without the consent of the faculty member involved.

Faculty members on academic year appointments may also receive special compensation for engaging in approved sponsored research, extension activities, or non-credit instructional activity conducted by continuing and professional education. The total of special compensation earned through all university programs in the summer by any faculty member on academic year appointment shall not exceed 33 ½ percent of the annual salary for the preceding academic year.

For purposes of sponsored grant and contract activity and for limitations on compensation, May 10 to August 9 designates the summer work period. Faculty members who receive summer salary from sponsored projects must certify to the effort expended on those projects during the summer period. Work on a sponsored project during the academic year for which compensation is then provided during the summer is specifically prohibited by federal regulations. Summer pay for sponsored projects is only justified by appropriate effort expended on the project during the summer period.

Only academic year faculty members who have approved research extended appointments earn and accrue annual leave proportional to their appointments. Faculty members with one, two, or three months of sponsored funding are strongly urged to convert their nine-month appointment to a 10-, 11-, or 12-month research extended appointment, which entitles them to earn and use annual leave and have the summer pay included as creditable compensation for retirement purposes in accordance with university policies. Alternatively, the faculty member can charge less than one, two, or three months of full-time salary to the sponsored project (or other sources as appropriate) and take uncompensated leave for the remainder of the summer in order to have vacation.

Winter Session Appointments: Faculty on academic or calendar year appointments may be invited by the department head or chair to teach in Winter Session. The faculty member will receive overload payment for teaching a Winter Session credit course as it is not considered part of the normal expectation for the instructional year. Compensation for teaching in the special session is negotiated by the faculty member and the department. Maximum compensation is set at 3.75% of the faculty member's annual salary for each one-credit semester course taught. An additional incentive grant may be negotiated up to a maximum of one month salary. The overload payment including any incentive grant is considered in the total allowable additional aggregate compensation of no more than 33 1/3% of annual salary.

Appropriately credentialed administrative and professional (A/P) faculty may also teach during this special session with approval of their department head. Guidelines set forth in **Policy 4071**, *Policy for Staff Employed to Teach For-Credit Courses* and **Policy 4072**, *Teaching Credit* 

Classes and Overload Compensation for Administrative and Professional Faculty Members apply.

#### 2.6.6 University-Sponsored Applications for Permanent Residency

Virginia Tech welcomes the contributions of scholars from all over the world in carrying out its learning, discovery, and engagement missions. Employer-sponsored applications for permanent residency assure the ongoing involvement of the foreign scholar in the life of the university and the work for which the employee was hired. To receive Virginia Tech sponsorship, all of the following conditions must be met:

- 1. The position must have the potential to be ongoing with successive renewals over a period of several years. For positions funded from sponsored grants or contracts, the supporting unit must demonstrate a record of sustained external funding.
- 2. The individual's appointment must be full-time and salaried, and in compliance with federal regulations, such as prevailing wage rate. The appointment may be restricted or regular, either academic or calendar year, as long as it is salaried, full-time, and there is an expectation of successive renewals over a period of several years. Wage employment does not meet this test.
- 3. The position is significant and meets institutional needs as documented by the department and validated by the approval of the relevant senior manager. Significance may be signaled, in part, by rank and title, as well as documented in the job description and supported by the credentials of the individual. These include:
  - Instructional faculty—ranks of instructor and assistant professor or above, including clinical faculty, but excluding adjunct, wage, or visiting faculty members;
  - Research faculty—all ranks except postdoctoral associates, whose appointments are limited, by definition, to four years;
  - Administrative/professional faculty with significant expertise critical to the university, and
  - Staff members with significant expertise critical to the university.
- 4. The department verifies that they wish to retain the employee in the position indefinitely subject to availability of funding, need for services, and satisfactory performance.

#### 2.6.7 Dual Career Program

Prospective candidates for faculty positions at Virginia Tech may have spouses or partners who are also seeking employment. The ability of a spouse/partner to find suitable employment is a crucial element in the recruiting process, and may be a determining factor in the couple's decision.

The spouse/partner of a faculty candidate or administrator who is being recruited to Virginia Tech is eligible for dual career services. The spouse/partner of a current faculty member who has been recently hired or is negotiating a retention package is also eligible for dual career services.

The dual career program offers job search assistance for up to one year; advice regarding résumé, curriculum vitae, and cover letter; assistance with interview preparation; and networking assistance. These services do not mean entitlement to employment or a guarantee of job

placement. Guidelines that describe procedures for Virginia Tech's hiring of dual career partners are available on the Human Resources and provost's websites.

### 2.7 Professional Responsibilities and Conduct

## 2.7.1 Statement of Principles of Ethical Behavior

The faculty of Virginia Tech believe that academic freedom is essential to attain our missions as scholars and teachers. We also recognize and accept the responsibilities attendant to academic freedom as fundamental to a scholarly community. We believe we must exercise our rights with due regard to the rights of others and we must meet our obligations fully as faculty members. We hold ourselves accountable to ensure that the faculty of Virginia Tech is recognized for its commitment and leadership to pursue knowledge, to promote the free expression of ideas, to teach our students, and to serve the citizens of Virginia.

Scholarship: Guided by a deep conviction of the worth and dignity of the advancement of knowledge, we recognize our primary responsibility to our disciplines is to seek and to state the truth. To this end, we devote our energies to developing and improving our scholarly competence. We accept the obligation to exercise critical self-discipline and judgment in using, extending, and transmitting knowledge. We practice intellectual honesty and do not compromise our freedom of inquiry.

Students: We encourage the free pursuit of learning in our students and exemplify the best scholarly and ethical standards of our disciplines. We value and promote differences among students and respect students as individuals and serve as their intellectual guides and counselors. We make every reasonable effort to foster honest academic conduct and to assure that our evaluations of students reflect each student's true merit. We respect the confidential nature of the relationship between professors and students. We avoid any exploitation, harassment, or discriminatory treatment of students and acknowledge significant academic or scholarly assistance from students. We do not engage in any romantic or sexual relationships with students whom we are in a position to evaluate by virtue of our teaching, research, or administrative responsibilities.

Instruction: We strive to be fair, compassionate, and effective teachers. We prepare classes adequately, present materials fairly, and make ourselves available to students for consultation and advice. We avoid bias and we respect diverse points of view.

Colleagues: We accept our obligations that derive from common membership in the faculty of Virginia Tech. We relate to colleagues and other university personnel in a responsible, professional, and civil manner, avoiding behaviors and actions that purposefully, consistently, and unnecessarily tend to disrupt, impede, harass, or abuse them in the performance of their assigned tasks and professional duties. We do not discriminate against colleagues, nor do we engage in romantic or sexual relationships with employees whom we are in a position to supervise or evaluate. We respect and defend free inquiry by all. In the exchange of criticisms and ideas, we show respect for the opinions of others, acknowledge significant academic or scholarly assistance, and strive to be open-minded and fair in our professional judgments. We

accept our share of faculty responsibilities for the governance of Virginia Tech and take due care in the discharge of those responsibilities.

University: We seek above all to be effective in our assigned responsibilities. We give paramount importance to these responsibilities in determining the amount and character of work done outside of Virginia Tech. Although we observe the Faculty Handbook, we maintain our right to criticize and seek revision of university policy.

Community: As members of the larger community, we have the same rights and obligations as other citizens. We measure the importance of these rights and obligations in light of our responsibilities to our disciplines, to our professions, to our students, and to Virginia Tech. When we speak or act as private persons, we avoid creating the impression of speaking or acting for Virginia Tech. As citizens engaged in a profession that depends upon freedom for its welfare and integrity, we have a particular obligation to promote conditions of free inquiry and of further public understanding of academic freedom.

#### 2.7.1.1 Allegations of Unprofessional or Unethical Conduct

The Committee on Faculty Ethics receives, investigates, and considers allegations of unprofessional or unethical conduct for all types of faculty members, except administrative and professional faculty members. See section 1.4.5.3, "Committee on Faculty Ethics," for further information on the committee charge, membership, and procedures. If the committee finds a serious breach of ethical conduct that leads to a recommendation for a severe sanction or dismissal for cause, the procedures in sections 3.6 and 4.7, "Imposition of a Severe Sanction or Dismissal for Cause," are followed in implementing such sanctions.

When the allegation is against an administrative or professional (A/P) faculty member without tenure or continued appointment, a special panel of five administrative or professional faculty members is selected to review the charges and hear the case, if appropriate. The chair of the Commission on Administrative and Professional Faculty Affairs (CAPFA) chooses panel members from among the A/P faculty at large. The CAPFA chair may invite an experienced member of the CFE to serve as a non-voting member of the panel. All potential members must disclose possible conflicts of interest concerning their participation in the case.

#### 2.7.2 Virginia Tech Principles of Community

Virginia Tech is a public land-grant university, committed to teaching and learning, research, and outreach to the Commonwealth of Virginia, the nation, and the world community. Learning from the experiences that shape Virginia Tech as an institution, we acknowledge those aspects of our legacy that reflected bias and exclusion. Therefore, we adopt and practice the following principles as fundamental to our ongoing efforts to increase access and inclusion and to create a community that nurtures learning and growth for all of its members:

We affirm the inherent dignity and value of every person and strive to maintain a climate for work and learning based on mutual respect and understanding.

We affirm the right of each person to express thoughts and opinions freely. We encourage open expression within a climate of civility, sensitivity, and mutual respect.

We affirm the value of human diversity because it enriches our lives and the university. We acknowledge and respect our differences while affirming our common humanity.

We reject all forms of prejudice and discrimination, including those based on age, color, disability, gender, national origin, political affiliation, race, religion, sexual orientation, and veteran status. We take individual and collective responsibility for helping to eliminate bias and discrimination and for increasing our own understanding of these issues through education, training, and interaction with others.

We pledge our collective commitment to these principles in the spirit of the Virginia Tech motto of *Ut Prosim* (That I May Serve).

#### 2.7.3 University Safety and Security Policy

In the interest of promoting a safe and secure working, learning, and living environment for employees, students, and visitors, the university developed policy 5615, "University Safety and Security." As part of a larger and institution-wide commitment to a safe and secure campus, the president established the Safety and Security Policy Committee and charged it with general oversight and leadership for the university's safety, emergency management, and security efforts. The policy further affirms the structure of operational committees that report to the Safety and Security Policy Committee.

Additionally, the policy recognizes the responsibility of department heads and individuals in supervisory roles to ensure that personnel within their departments are aware of safety and security policies and reporting procedures. Individuals are responsible for being aware of and complying with university policies, procedures, and applicable law.

University facilities must be used in a safe and appropriate manner so as not to endanger the university community or the general public. All faculty, staff, students, and other members of the Virginia Tech community share responsibility for the safety and security of the institution and must conduct university activities and operations in compliance with applicable federal and state regulations and university policies. Employees and students should take any threat or violent act seriously and report acts of violence or threats to the appropriate authorities. Any individual who believes there is an immediate danger to the health or safety of any member of the university community should contact the Virginia Tech Police immediately.

## 2.7.4 Campus and Workplace Violence Prevention

Virginia Tech does not tolerate acts of violence committed by or against employees, students, contractual workers, temporary employment agency workers, volunteers, visitors, or other third parties on university owned, controlled, or leased properties, or while conducting university business at another location, including representing the university at conferences or off-site meetings, or riding in university owned or leased vehicles. Violence includes, but is not limited

to, intimidation, threats, physical attack, domestic and dating violence, stalking, or property damage.

Per the university's "Weapons on Campus Regulation", the university's employees, students, and volunteers, or any visitor or other third party attending a sporting, entertainment, or educational event, or visiting an academic or administrative office building, dining facility, or residence hall, are further prohibited from carrying, maintaining, or storing a firearm or weapon on any university facility, even if the owner has a valid permit, when it is not required by the individual's job or in accordance with the relevant university student life policies.

The board of visitors established a Campus and Workplace Violence Prevention Committee, charged with developing and implementing violence prevention and education procedures, programs, and guidance. The board further endorsed the president's appointment of a Threat Assessment Team, charged with using its judgment to assess, intervene, and follow policies for individuals whose behaviors may present a threat to the safety of the campus community as appropriate; working with enforcement and mental health agencies to expedite assessment and intervention; and developing comprehensive fact-based assessments of students, employees, or other individuals who may present a threat to the university. The Threat Assessment Team is empowered to take timely and appropriate action consistent with the judgment of the team, university policy, and applicable law.

## 2.7.5 Health and Safety

University policy 1005, "Health and Safety Policy," is intended to help prevent accidents, illnesses and injuries; increase safety awareness; meet requirements of environmental, occupational health, and safety laws and regulations; reduce institutional liability; and establish safety responsibilities for members of the university community and visitors to university-owned or occupied property. All members of the university community are expected to be thoroughly familiar with their safety responsibilities, strive to follow safety practices at all times, act proactively to prevent accidents and injuries, communicate hazards to supervisors, and be prepared for emergencies that may occur in the workplace. Faculty, staff, students, and other members of the Virginia Tech community must conduct university operations in compliance with applicable federal, state, and local regulations, and Environmental Health and Safety requirements. The policy outlines the responsibilities of department heads, faculty, principal investigators, supervisors, staff, students, and contractors.

#### 2.7.6 Statement of Business Conduct Standards

Each employee makes an important contribution to the overall success of Virginia Tech by performing job responsibilities in accordance with university policies and procedures. The university's business standards provide a strong foundation of business practices to support the core missions of learning, discovery, and engagement. These standards are described in the statement of business conduct standards available on the Capital Assets and Financial Management website.

All employees are expected to ensure that business activities are conducted properly and in compliance with various federal and state laws. Many important procedures are located on the websites of the controller's office, purchasing, human resources, and university policies. The websites are updated as policies change.

### 2.7.7 Policy on Anti-Discrimination and Harassment Prevention (Policy 1025)

Virginia Tech does not discriminate against employees, students, or applicants for admission or employment on the basis of race, color, gender, disability, age, veteran status, national origin, religion, sexual orientation, political affiliation, genetic information. Such behavior is inconsistent with the university's commitments to excellence and to a community in which mutual respect is a core value as articulated in the Virginia Tech Principles of Community (see section 2.7.2) and outlined in university policy 1025. The prohibition against discrimination and harassment applies to all levels and areas of university operations and programs, to students, administrators, faculty, staff, volunteers, vendors, and contractors.

Virginia Tech is also committed to the free and vigorous discussion of ideas and issues. This policy does not allow curtailment or censorship of constitutionally protected expression, nor does it attempt to address behaviors that do not constitute discrimination or harassment. The appropriate supervisor or administrator is responsible for addressing offensive behavior that does not violate the discrimination/harassment policy.

Discrimination/harassment includes the following behaviors:

- Conduct that conditions any element of a person's employment, enrollment as a student, receipt of student financial aid, or participation in university activities on that person's age, color, disability, gender (including pregnancy), national origin, political affiliation, race, religion, sexual orientation, or veteran status, genetic information unless otherwise permitted or required by applicable law;
- Conduct of any type (oral, written, graphic, electronic, or physical) based upon a person's age, color, disability, gender (including pregnancy), national origin, political affiliation, race, religion, sexual orientation, or veteran status, genetic information and which unreasonably interferes with the person's work or academic performance or participation in university activities, or creates a working or learning environment that a reasonable person would find hostile, threatening, or intimidating;
- Conduct consisting of unwelcome sexual advances, requests for sexual favors, or other verbal
  or physical conduct of a sexual nature when submission to such conduct is made, explicitly
  or implicitly, a term or condition of an individual's employment or education; or submission
  to or rejection of such conduct is used as a basis for employment or educational decisions
  affecting an individual.

It is also a violation of this policy to retaliate against anyone for filing a complaint of discrimination/harassment or otherwise participating in the investigation of such a complaint.

It should be understood by all members of the university community that consensual amorous or sexual relationships (hereinafter referred to as consensual relationships) that occur in the context of educational or employment supervision and evaluation present serious ethical concerns.

Consensual relationships between faculty and students enrolled in their classes or students for whom they have professional responsibility as advisor or supervisor violate the policy on professional ethics and responsibilities and may be a violation of this discrimination/harassment policy. Similarly, consensual relationships between supervisors and employees they directly supervise violate university policy. Faculty members or others performing instructional or academic advising duties and supervisors involved in consensual relationships must remove themselves from any activity or evaluation that may reward or penalize the affected student or employee.

Consensual relationships between faculty and students are particularly susceptible to exploitation. The respect and trust accorded a professor by a student, as well as the power exercised by the professor in giving praise or blame, grades, recommendations for further study and future employment, make voluntary consent by the student suspect, given the fundamentally asymmetric nature of the relationship.

Faculty and supervisors should be aware that engaging in consensual relationships with students or employees they supervise could make them liable for formal action. Even when both parties have consented to the development of such a relationship, it is the faculty member or supervisor who, by virtue of his or her special responsibility, may be held accountable for unprofessional behavior. Complaints alleging discrimination/harassment, as defined above, may be filed by either party to the consensual relationship or by an aggrieved party outside the relationship.

University administrators, supervisors, faculty members, and others performing instructional or academic advising duties have an added responsibility to create and maintain a work and learning environment free of discrimination/harassment.

If an administrator, supervisor, or faculty member becomes aware of an incident that might reasonably be construed as constituting discrimination/harassment, he or she must take immediate steps to address the matter. In such cases, the administrator, supervisor, or faculty member should promptly contact the Office of Equity and Access in human resources in order to coordinate any further action that may be necessary.

Administrators, supervisors and faculty members should act whenever they learn—directly or indirectly—about discrimination/harassment. This obligation exists even if the complainant requests that no action be taken. It is not the responsibility of the complainant to correct the situation.

Administrators, supervisors, and faculty members have the legal responsibility to protect a complainant from continued discrimination, harassment, or retaliation. They must also protect persons accused of discrimination/harassment from potential damage by false allegations. Administrators and supervisors will be held accountable for dealing with and taking necessary steps to prevent discrimination/harassment.

Administrators and supervisors are responsible for informing their employees and students of this policy.

For additional information, including procedures to follow when concerns or questions about discrimination/harassment arise, contact the Office of Equity and Access in human resources.

#### 2.7.8 Policies Related to Electronic Communications

### 2.7.8.1 Privacy of Electronic Communications

Human resource policy 1.75 of the Commonwealth of Virginia states, "no user should have any expectation of privacy in any message, file, image, or data created, sent, retrieved, or received by use of the commonwealth's equipment and/or access." A policy defining the balance between the university's business needs and respect for employees' freedom of inquiry has been established to guide actions of managers in certain situations and to clarify expectations for all employees about when and how the university may access employees' communications.

Virginia Tech requires all employees to obey applicable policies and laws in the use of university computing and communications technologies.

The university does not routinely monitor or access the content of electronic communications, computer files, or voice mail of its employees, whether stored on university equipment or in transit on the university network. Content of employees' electronic communications or files are not accessed during the execution of systems support, network performance, and related security functions.

However, monitoring or access may be necessary under certain circumstances. Legal or administrative circumstances where monitoring and/or access may occur without further authorization are communications or files required to be released by law, by orders of a court, or requested in accordance with the Virginia Freedom of Information Act; approved internal audit reviews; resolution of technical problems; emergency situations involving an imminent threat of irreparable harm to persons or property; and resources assigned to a group or publicly available to any user.

If routine monitoring or the examination of employee electronic communications or files are an essential part of the work environment, the department must develop and maintain a clearly written operating policy that is regularly disseminated to the affected employees. Affected employees must be given an opportunity to comment during the development or major revision of such a policy. Prior written approval of such departmental policies is required from the relevant dean or senior manager.

Authorization for non-law enforcement university personnel to monitor or access electronic communications or files of employees is not granted casually. Such authorization requires justification based on reasonable business needs or reasonably substantiated allegations of violation of law or policy on the part of the employee. In carrying out retrieval of files or information, due respect should be accorded to confidential or personal information and legally protected files.

Requests for authorization to monitor or review electronic communications or files because of allegations of violations of policy or law by faculty or staff members may originate with supervisors. They may also originate with an investigatory authority such as the Office of Equity and Access investigating a claim of sexual harassment. Requests must be made in writing and include the rationale for the request, a description of the information or files to be accessed or retrieved, and the proposed handling and disposition of the files. In such cases, the relevant dean, senior manager (including vice presidents), or higher-level authority may grant authorization if needed.

The senior manager who is asked to consider authorization for monitoring or reviewing the electronic communications or files of an employee must use his or her best professional judgment in determining if there exists reasonable grounds to grant such authorization, with consideration given to the surrounding circumstances and environment. The senior manager maintains confidentiality in such a situation. He or she is encouraged to consult with university legal counsel or human resources in determining whether to authorize monitoring or review, and in determining if the affected employee or anyone else should be notified that the monitoring or review is taking place.

Where there is a reasonable need for access to business or educational documents and the employee is unavailable, authorization to access that employee's electronic communications should be provided by the department head/chair or director, or next higher authority. Whenever possible, the employee should be informed and asked to help obtain the needed business materials. If that help is not reasonably available, then other steps should be considered to respect the confidential or personal nature of any other materials present. The employee is promptly notified of the access and the nature of the documents or communications reviewed or obtained.

#### 2.7.8.2 Guidelines for Acceptable Use of Information Systems

Access to computer systems and networks owned or operated by Virginia Tech imposes certain responsibilities and obligations and is granted subject to university policies, and local, state, and federal laws. Acceptable use is always ethical, reflects academic honesty, and shows restraint in the consumption of shared resources. It demonstrates respect for intellectual property, ownership of data, system security mechanisms, and individuals' rights to privacy and to freedom from intimidation, harassment, and unwarranted annoyance.

In making acceptable use of resources you must use resources only for authorized purposes; protect your user ID and system from unauthorized use (you are responsible for all activities on your user ID or that originate from your system); access only files and data that are your own, that are publicly available, or to which you have been given authorized access; use only legal versions of copyrighted software in compliance with vendor license requirements; and be considerate in your use of shared resources. Refrain from monopolizing systems, overloading networks with excessive data, or wasting computer time, connect time, disk space, printer paper, manuals, or other resources.

In making acceptable use of resources you must not use another person's system, user ID, password, files, or data without permission; use computer programs to decode passwords or access control information; attempt to circumvent or subvert system or network security measures; engage in any activity that might be harmful to systems or to any information stored thereon, such as creating or propagating viruses, disrupting services, or damaging files; use university systems for commercial or partisan political purposes, such as using electronic mail to circulate advertising for products or for political candidates; make or use illegal copies of copyrighted software, store such copies on university systems, or transmit them over university networks; use mail or messaging services to harass, intimidate, or otherwise annoy another person, for example, by broadcasting unsolicited messages or sending unwanted mail; waste computing resources, for example, by intentionally placing a program in an endless loop, printing excessive amounts of paper, or sending chain letters; use the university's systems or networks for personal gain; for example, by selling access to your user ID or to university systems or networks, or by performing work for profit with university resources in a manner not authorized by the university; or engage in any other activity that does not comply with the general principles presented above.

The university considers any violation of acceptable use principles or guidelines to be a serious offense and reserves the right to copy and examine any files or information resident on university systems allegedly related to unacceptable use. Violators are subject to disciplinary action as prescribed in the honor codes and the student and employee handbooks. Offenders also may be prosecuted under laws including (but not limited to) the Communications Act of 1934 (amended), the Privacy Protection Act of 1974, the Computer Fraud and Abuse Act of 1986, the Computer Virus Eradication Act of 1989, Interstate Transportation of Stolen Property, the Virginia Computer Crimes Act, and the Electronic Communications Privacy Act. Access to the text of these laws is available through the University Libraries reference department.

#### 2.7.9 Policy on Misconduct in Research

The university endorses high ethical standards in conducting research to ensure public trust in the integrity of research results. The university recognizes that deception in research erodes the credibility of an institution and the confidence of those who might benefit from the research. The university takes all reasonable and practical steps to foster a research environment that promotes the responsible conduct of research and research training (and activities related to that research or research training), discourages research misconduct, and deals promptly with allegations or evidence of possible research misconduct. Additional information regarding the policy on misconduct in research and procedures for dealing with research misconduct is found in chapter 10 of this handbook.

## 2.8 Advanced Study at Virginia Tech

Virginia Tech encourages and supports the continuing and advanced education of its faculty and staff. Educational leave to pursue a degree elsewhere is one option available to faculty. In addition, faculty may enroll for credit courses or degree programs at Virginia Tech. The program is administered under the provisions of the general appropriations act and operates under certain constraints imposed by the state policy on educational aid to state employees.

The following provisions apply to full-time salaried faculty (including administrative and professional faculty and research faculty) who wish to take courses at Virginia Tech. Part-time salaried faculty are eligible for a partial tuition benefit. Only courses or degree programs approved in advance by the faculty member's department head/chair or supervisor are eligible for tuition waiver or reimbursement. Enrollment should not impede the normal work schedule of the department. Time spent attending class during normal work hours must be made up under a plan approved by the head or supervisor, unless the course is a work-related course required by the university.

Faculty who take courses must meet all admissions requirements, registration, and payment deadlines, just as any other student. Application for admission must be made and approval granted by the graduate school prior to the waiver of tuition for classes. If approved by the department head, faculty may register for credit or audit a total of 12 credit hours per calendar year with no more than six credit hours taken in any enrollment period—fall, spring, summer I, or summer II. (The year begins with fall term and ends with summer II.) Additional hours may be taken outside the normal work schedule with the employee paying all applicable fees in excess of those allowable for tuition waiver or reimbursement.

The university adopted policy memorandum 19, "Clarification on Faculty Earning Degrees at the University," that instructional faculty members of the rank of assistant professor or above are not eligible to become candidates for a degree or to earn an additional degree at this institution. The policy is designed to avoid the awkwardness of faculty members evaluating their colleagues in the fulfillment of degree requirements. This policy may be waived on a case-by-case basis through appeal to the Commission on Faculty Affairs (CFA). CFA may find and recommend to the Office of the Senior Vice President and Provost that in a specific case the purpose of the policy is not contradicted. This policy does not apply to degree-seeking administrative and professional faculty, or non-instructional research faculty.

#### 2.9 Leaves

Several kinds of approved leave, with or without salary, are available to faculty members as described in this section. Unapproved absence from assigned duties, which is not covered by an approved or earned leave, is subject to subsequent payroll adjustment to compensate for the absence.

When collegiate faculty members are absent during the academic year to attend meetings or consult about research funding, etc., and when these absences take fewer than five days, the department head or chair is the principal approving officer and is responsible for ensuring the adequate coverage of the duties of the absent colleague. An absence of up to two weeks is managed entirely within the college and requires the approval of the department head or chair and the dean. But, absences beyond two weeks must be forwarded through the department head or chair and dean to the provost for review and approval. A determination is made about the necessity of a leave of absence without pay or a change of duty station with pay for university approved activities away from the home location. (See section 2.10, "Change of Duty Station and Special Leave," for further information.)

#### 2.9.1 Leave Report

Salaried faculty and staff use the university leave and work report to record all types of leave. A summary of leave policies and detailed procedures to complete the report is available on the human resources website.

Calendar year faculty are required to submit leave reports in a timely manner during any leave period that leave is used and are to submit leave periods 6 and 12 for financial reporting purposes. Department heads may require leave reports to be submitted monthly by all faculty.

Regular calendar-year faculty members are eligible to donate annual leave hours to the leave sharing program. Under state policy, only full-time and part-time salaried staff employees may be recipients of leave sharing.

#### 2.9.2 Educational Leave

Leaves of absence on partial salary (not to exceed one-half salary) may be granted to faculty members for educational purposes. Such leaves are granted for formal educational advancement ordinarily leading to an advanced degree from another institution, which is of demonstrable benefit to the university and to the faculty member.

The leave is granted only if satisfactory arrangements can be made for effective continuation of the relevant program. Only that fraction of a position not supported by the partial salary of the leave is available for the appointment of a replacement faculty member during the period of the leave. Educational leaves ordinarily are granted for periods of one year or less. If the program of study is completed, or if the faculty member ceases active participation in that program, before the ending date of the approved leave, the faculty member immediately returns to full employment or resigns employment. Application should be made to the provost or vice president for administration, depending upon the reporting structure, on recommendation of the department head or chair and dean (or appropriate administrator).

On approval of educational leave with partial pay, the faculty member must sign a memorandum of agreement, which obligates the faculty member to return to full employment in the university for a period twice the time of the approved leave or to repay the university the salary received plus interest. If a leave recipient returns to the university, but resigns before completing that obligation, the salary repayment is prorated.

Policies governing advanced study at Virginia Tech without leave are covered in section 2.8, "Advanced Study at Virginia Tech."

#### 2.9.3 Military Leave

Military leave is available to all faculty members, including those on restricted, wage, or adjunct appointments. Faculty members are eligible for leave with pay for up to three weeks in a federal fiscal year (October 1 through September 30) for military duty, including training, if they are

members of any reserve component of the Armed Forces or the National Guard. Three weeks of paid military leave is the maximum allowable for one tour of duty, even when that tour encompasses more than one federal fiscal year. Employees may use accrued annual leave to continue their pay while on military leave. Employees are granted unconditional leave without pay for the duty indicated in their military orders that is not covered by military leave with pay.

To qualify for military leave, faculty members must furnish their department head or chair and human resources with copies of their orders or other documentation.

Employees are reinstated to their previous positions or to positions comparable to their previous positions provided that certain conditions are met. Requests for reinstatement must be made to human resources and should state that the individual is seeking reinstatement to employment upon return from military service. If the military leave was for a period of 31 to 180 days, the employee must apply for reinstatement within 14 days of discharge. If the military leave was for a period of 181 days or more, the employee must apply for reinstatement within 90 days of discharge. Contact human resources for a full description of military leave benefits and conditions, and guidance on all requests for military leave and/or reinstatement.

#### 2.9.4 Administrative Leave

If a faculty member is called for jury duty, subpoenaed, or summonsed to appear in court, this absence may be charged to administrative leave, except when he or she is a defendant in a criminal or civil case. This leave should be requested before it is taken. Any time spent in court as a defendant in a criminal or civil case must be charged to annual leave, overtime leave, compensatory leave, or leave without pay.

Faculty members receive full pay for administrative leave, provided a copy of the subpoena or other supporting document accompanies the leave report.

Administrative leave with pay is not granted for more time than actually required for the purpose for which it is taken. Any additional administrative leave time taken on the same day must be charged to leave without pay or appropriate leave balances and reported on the monthly leave report.

Faculty members are granted administrative leave to attend work-related hearings as a witness under subpoena or regarding a personal claim.

Administrative leave may be used when called to serve on councils, commissions, boards, or committees of the commonwealth. If a faculty member is serving as an official representative of the university, then administrative leave is not used. The service time is treated as part of the faculty member's regular work hours.

#### 2.9.5 Annual Leave and Holidays

Instructional and research faculty members on academic year appointments do not earn or accrue annual leave. Regular administrative and professional (A/P) faculty members on academic year

appointment earn annual leave only during the period of their appointment at the same rate as regular A/P faculty members on calendar year appointment. Faculty members on calendar year appointments earn two days (16 hours) of annual leave credit per month in accordance with leave regulations; after 20 years of continuous employment by the commonwealth, 18 hours of annual leave are earned per month. Faculty members on research extended appointments earn annual leave proportional to their appointments. However, unused annual leave will not be compensated at the time of reconversion or separation for faculty on research extended appointments or restricted appointments.

All faculty members who earn annual leave are expected to record the appropriate leave type on the monthly leave report if they do not work during the academic breaks. Use of annual leave requires the advance approval of the department head/chair or supervisor.

Faculty members on calendar year appointments, who are on regular appointment, may carry forward accrued annual leave to a maximum of 36 days (288 hours) at the beginning of each calendar year or may be paid up to the maximum on termination of employment. After 20 years of service, the maximum accrued leave carried forward or paid upon separation becomes 42 days (336 hours).

Annual leave must be credited before it is used. Holidays observed by university faculty members are New Year's Day, Martin Luther King Jr. Day, Memorial Day, Independence Day, Thanksgiving Day and the day following, Christmas Day, and other holidays that may be designated by the governor. If faculty members are required to work on these days due to extenuating circumstances as determined by the supervisor, or if they are designated as emergency personnel, they may take compensatory days off later, but compensatory days do not accrue as earned annual leave.

The university closes between December 25 and January 1 each year. Twelve-month faculty (and those on research extended appointments earning leave), research faculty, and administrative and professional faculty must use annual or other appropriate leave balances to cover the days not worked that have not been designated as official holidays or additional days designated as holidays by the governor.

Faculty members with accrued annual leave who temporarily change their status (for example, going on leave without pay or changing to a part-time appointment for a short period of time) should contact human resources to discuss their options and indicate their preference for either payout of their leave balance or retention of their leave balance until they resume full-time status. Calendar year faculty on study-research assignment earn annual leave at a rate that is half their normal annual leave earnings.

Regular calendar-year faculty members are eligible to donate annual leave hours to the leave sharing program. Under state policy, only full-time and part-time salaried staff employees may be recipients of leave sharing.

#### 2.9.6 Sick Leave

Faculty members in regular salaried positions are ordinarily eligible for limited sick leave at full pay under a policy approved by the board of visitors in July 1981. Under this policy, eligible faculty members have immediate protection of 26 weeks of sick leave. On return from sick leave, reaccrual to a maximum of 26 weeks takes place at the rate of one week of sick leave for each two weeks of full-time work. Isolated minor illnesses extending over a maximum of 10 days are handled at the department level with the cooperation of faculty colleagues for the covering of necessary duties. Sick leave must be recorded for absences exceeding 10 days in duration. Faculty members are strongly encouraged to record use of sick leave, even when in increments of less than 10 days. Ability to document the onset of illness can provide critical financial protection for faculty members ultimately needing long-term disability. Provision is made for prorated sick leave when partial resumption of duties can be undertaken. However, reaccrual does not begin until full-time service resumes. Faculty members whose appointments began on or after September 1, 1981, are subject to this policy.

Faculty members whose appointments began before September 1, 1981, had the option of selecting the above policy or continuing coverage under the previously existing sick leave policy. Under that policy, sick leave is accrued at the rate of 15 days (120 hours) per calendar year with no maximum accrued limitation. Sick leave credit is not given for service of less than one-half month; leave cannot be granted before it is earned. All faculty members on restricted appointment have coverage under this previously existing sick leave policy.

Faculty members in full-time restricted salary positions receive 10 hours per month of sick leave. Faculty on part-time restricted appointments earn sick leave based on their percentage of employment.

In 1999, state employees in regular or restricted positions, who were participants under the Virginia Retirement System (VRS), could choose to enroll in the Virginia Sickness and Disability Program (VSDP). VSDP provides employees with a minimum of 64 hours of sick leave and 32 hours of family/personal leave annually. These hours are reinstated each year, but do not carry over. In addition, VSDP provides salary continuation during periods of short-term disability up to six months and long-term disability to age 65 or later depending on age at the time of disability.

The sick leave plans do not provide for compensation for unused sick leave upon termination.

Under all policies, sick leave may be used for illness, accidents, and pregnancy-related conditions. Faculty under the first two plans may use sick leave for family emergencies. (For more information on leave for family emergencies and pregnancy-related conditions, see sections below on the Family Medical Leave Act and family leave.) The university may require certification of the medical necessity of the period of absence from work due to illness, accident, or pregnancy-related condition. Certification, when required, is requested from the attending physician and/or a physician designated by the university.

Faculty on academic year appointments are not covered during the period of a summer appointment under any sick leave policy.

Faculty positions are not released for replacement purposes because of sick leave; authorization by the dean is necessary when replacement is needed.

Full details of all sick leave policies and procedures, including eligibility, are available from human resources.

#### 2.9.7 Family Leave

The university recognizes that faculty members have family responsibilities that may, from time to time, make extraordinary claims on their time, making it difficult to carry out fully their responsibilities. Department heads or chairs, deans, and other supervisory personnel are urged to be as flexible as possible within existing university policy in responding to the need for leave or temporary adjustment of duties for family-related reasons.

Paid sick leave is available for pregnancy-related medical conditions, childbirth, and recovery. The length of time charged to sick leave varies and is based on the recommendation of a physician. Sick leave may also be used for family emergencies. Leave for this purpose is restricted to a total of 10 days (80 hours) during a calendar year for absence necessitated by the serious illness or death of a family member or any individual residing in the same household. Use of such leave must be recorded through the regular leave reporting system so that total usage during the period can be monitored. Restricted faculty may use 80 hours of their earned sick leave or annual leave for these purposes.

#### 2.9.7.1 Family Medical Leave Act (FMLA)

Federal law (Family Medical Leave Act) guarantees employees a minimum period of 12 workweeks of leave during a year for family care if they have been employed at Virginia Tech for at least 12 months and if they have worked at least 1,250 hours during the previous 12 months. The time frame for calculating a year is measured forward from the date the employee is approved for FMLA. Accrued sick and annual leave may be used, as appropriate; the use of paid leave should be concurrent with the approved FMLA period (i.e., run concurrently). The remainder of the 12-workweek leave period is leave without pay. The faculty member may request a longer period of leave without pay as suggested in the following section, which requires approval of the department head or chair, dean or vice president, and senior administrator. Adjunct faculty are eligible for unpaid leave under FMLA if they meet the same minimum employment requirements.

Eligible faculty are granted unpaid family or medical leave for one or more of the following: birth of a child; placement of a child with the faculty member for adoption or foster care; the care of an immediate family member (child, spouse, parent) who has a serious health condition; or a serious health condition that makes the faculty member unable to perform the function of his or her position. Prior to leave approval, the department head or chair may require documentation of

the health condition necessitating care of a family member by the employee's own health condition.

The period of up to 12 workweeks of family leave for purposes of birth or adoption may be taken intermittently or on a reduced schedule if the faculty member and department head or chair agree and it does not create an undue hardship on the program or project. In the case of medical leave because of a faculty illness or illness of a family member, leave may be taken intermittently or on a reduced schedule when medically necessary. The department head or chair may require documentation of such medical necessity. The department head or chair may reassign responsibilities or transfer the faculty member to another position with the same salary in order to accommodate the intermittent or reduced leave schedule.

On return from leave, the faculty member returns to the same position or an equivalent position with the same benefits and salary at the time leave was taken. There is no accrual of additional sick or annual leave during the leave without pay period taken, or service toward the minimum six-year requirement for consideration for research-study leave. However, if the requested amount of leave extends beyond the term of appointment of a restricted faculty member or wage/adjunct employee and reappointment is not anticipated, the department head or chair is not required to maintain the position of the faculty member on leave beyond the original termination date. The request for family or medical leave shall not constitute sufficient reason for non-reappointment, termination, or other retaliatory action.

Eligibility for family leave for the purpose of birth or adoption expires at the end of the 12-month period beginning on the date of birth or placement. The faculty member gives the department head or chair at least 30 days notice regardless of reason, whenever practicable. If leave is requested for care of a family member, documentation of the serious health condition necessitating care by the faculty member may be required.

Benefits are continued for full-time employees in accordance with state personnel policies and the federal Family and Medical Leave Act of 1993. For detailed information on which benefits will be continued for what period, faculty members should contact human resources.

The procedures for requesting FMLA are available at human resources. Unlike some other leave types, the employee and medical professionals must complete specific forms.

## 2.9.7.2 Additional Faculty Leave Benefits for Faculty on Regular, Salaried Appointments

Policy memorandum 189, "Modification of Family Sick Leave Policy," provides family leave benefits greater than the minimum guarantees set forth in the FMLA for faculty on regular, salaried (non-restricted) appointments. In addition to any paid sick leave used for pregnancy, childbirth, and recovery, up to 10 days of paid family leave may be used by either parent at the birth of a child or at the time of adoption. (If both parents are eligible Virginia Tech faculty members, then both may request and receive 10 days of leave.) This leave is also available to faculty who are under the Virginia Sickness and Disability Program (VSDP). Use of such leave must be recorded through the regular leave reporting system so that total usage during the period

can be monitored. Calendar year faculty may use annual leave to extend the paid leave period for childbirth, adoption, or family care.

Faculty members on regular, salaried appointments wishing or requiring an extended period of time for child or family care may be granted leave without pay for up to one year (academic year or calendar year depending on type of appointment) thereby guaranteeing their job during the period of leave. A second year of leave without pay may be requested and approved in unusual cases. (See section 2.9.8, "Leave Without Pay," for terms and conditions.) Sick leave and accrued annual leave if appropriate and applicable may be used prior to leave without pay.

It is in the university's interest to help employees combine new parenthood (or other temporary extraordinary family obligations) and employment when possible and preferred by the faculty member over a full leave from the university. Regular faculty members who find that they cannot carry on their normal university duties in the usual manner and fulfill their family obligations at the same time may request consideration for part-time employment at proportional pay. Assigned responsibilities for a part-time load vary depending on the needs of both the individual and department. Expectations for learning, discovery, engagement, and committee assignments should be discussed in advance with the head or supervisor. The department head or chair may require the faculty member to be assigned different responsibilities or transferred to another position at the same salary in order to accommodate the intermittent leave or reduced schedule.

Probationary faculty moving to part-time status for the purposes of child or family care receive an extension of the mandatory tenure or continued appointment review date, consistent with sections 3.4.2.1, "Extending the Tenure Clock," and 4.5.2.1, "Extending the Continued Appointment Clock."

Department heads/chairs and supervisors should be sensitive and responsive where possible in establishing work hours, course and committee scheduling, and other aspects of employment for faculty members who are new parents or who are experiencing temporary extraordinary family obligations.

A one-year probationary period extension is automatically granted to either parent (or both, if both parents are tenure-track or continued appointment-track faculty members) in recognition of the demands of caring for a newborn child or a child under five newly placed for adoption or foster care. The request should be made within a year of the child's arrival in the family.

An extension of the probationary period may also be approved on a discretionary basis for other extenuating non-professional circumstances that have had a significant impact on the faculty member's productivity, such as a serious personal illness or major illness of a member of the immediate family. (See sections 3.4.2.1, "Extending the Tenure Clock," and 4.5.2.1, "Extending the Continued Appointment Clock.")

#### 2.9.8 Leave Without Pay

Faculty members wishing to take leave from their duties without salary may request such leave from the provost or vice president for administration, depending upon the reporting structure, on the recommendation of the department head or chair and the dean (or appropriate administrator). The period of requested leave cannot exceed two years. The request must be made with sufficient notice to allow time to secure a qualified replacement. The request should include the reason for the leave, whether for personal reasons or because of opportunity for further professional development.

When approval of a leave is granted, a date is specified by which any request for extension of the leave or notification of intent not to return to the university at the conclusion of the leave must be received. The granting of the leave or of any extension is dependent on the interests of the university and those of the faculty member. Consult with human resources to determine what benefits may be purchased.

#### 2.9.9 Disaster Relief Leave

Department heads or chairs may grant release time to faculty when they are formally called to provide disaster relief services because of their specialized skill or training. Release time for faculty is not recorded in the leave system. For audit purposes, record of time off should be noted in the faculty member's departmental file, along with the written request.

#### 2.10 Change of Duty Station or Special Leave

Absence from campus or the home work location for a period of more than two weeks while carrying out university-approved activities is called change of duty station or special leave. A change of duty station may be approved in instances such as grant responsibilities, opportunity of a prestigious fellowship in residence at another institution, or similar activities of benefit to the individual faculty member and the university. When such absences involve salary payment by university general funds, either in full or in part, approval of the provost or vice president for administration, depending upon the reporting structure, on recommendation of the department head or chair and dean (or appropriate administrator) is required. Such authorization is not granted for longer than one semester. The host institution, agency, or sponsored project is expected to make a significant contribution toward the cost of the faculty member's salary and/or benefits. The provost or the vice president for administration determines whether a change of duty station involving institutional salary support or leave without pay is appropriate to the circumstances.

#### 2.11 Retirement, Resignation, and Non-Reappointment

#### 2.11.1 Retirement

State law prohibits mandatory retirement on the basis of age alone. There is no mandatory retirement age for university faculty and staff.

## 2.11.1.1 Retirement Transition Program for Faculty with Tenure or Continued Appointment

Tenured faculty members (or those with a continued appointment in the library or Virginia Cooperative Extension) who are at least 60 years of age and have at least 10 years full-time service at Virginia Tech are eligible for the faculty retirement transition program. The program has two major benefits: employer-paid medical benefits to age 65 and, where mutually agreeable, part-time employment following retirement.

#### 2.11.2 Resignation

Faculty members wishing to resign should give notice as far in advance as possible. Faculty members with instructional responsibilities are expected to provide notice of at least one full academic semester. The minimum acceptable notice for tenured, tenure-track, or non-tenure-track instructional faculty members without an instructional assignment is three months.

#### 2.11.3 Non-Reappointment

In the cases of faculty members on temporary or restricted appointments for which there is no indicated opportunity for reappointment, the letter of appointment also serves as notice of the termination of employment. The appointment is discontinued unless notified otherwise.

Research faculty members are ordinarily on restricted appointments for a fixed period because of limitations of external funding. Reappointments may be possible if such funding is renewed, but should not be assumed.

The decision not to reappoint a faculty member on a regular appointment may stem from many factors beyond unsatisfactory service, such as modification of programmatic emphasis, enrollment trends, a change in the nature of the position, or simply the intention to seek an appointee with superior qualifications or stronger potential for professional development. Non-reappointment does not require establishment or documentation of just cause.

# 2.11.3.1 Non-Reappointment for Faculty on Tenure-Track or Continued Appointment-Track

Faculty members on probationary term appointments should make no presumption of reappointment, including reappointment with tenure. Non-reappointment may be determined by the department head or chair in consultation with the dean and with the advice of a departmental personnel committee or faculty development committee.

Faculty members on probationary term appointments that will not be renewed are given notice of non-reappointment in writing within the following time limits:

- First year of employment (one-year term appointment)—February 9 of academic year or three months before end of employment year;
- Second year of employment—November 9 of the academic year or six months before end of employment year;

• Subsequent years—12 months before end of employment year (May 9 for academic year appointments).

## 2.11.3.2 Non-Reappointment for Non-Tenure-Track Instructional Faculty on Regular Appointments

Notice of non-reappointment for non-tenure-track faculty members on regular appointments is:

- At least three months before the end of the current contract for those who have been in regular appointments for less than two years;
- At least one semester before the end of the current contract for those on academic year appointments (or six months for those on a calendar year appointment) for those who have been in regular appointments for two years up to five years;
- At least one year before the end of the current contract for those on regular appointments for five years or more (May 9 for academic year appointments).

#### 2.11.3.3 Non-Reappointment for Research Faculty on Regular Appointments

Notice of non-reappointment for research faculty on regular appointments is:

- At least three months for those who have been in regular appointments for less than two years;
- At least six months for those who have been in regular appointments for two years or more;
- For those research faculty appointed to regular positions before March 2001, the notice of reappointment is 12 months.

## 2.11.3.4 Non-Reappointment for Administrative and Professional Faculty on Regular Appointments

Notice of non-reappointment for administrative and professional faculty on regular appointments is:

- At least three months before the expiration of an initial one-year appointment (for example, if the effective date of an initial one-year appointment was July 1, then written notice of non-reappointment must be made by March 31 for termination effective June 30);
- At least six months for administrative and professional faculty who have been employed by the university for more than one year, but less than two years;
- At least 12 months for administrative and professional faculty members who have been at the university two years or more.

## 2.11.4 Unclaimed Personal Property

All personal property—tangible, intangible, electronic, or other personal property—is removed by close of business on the faculty member's final day of employment at Virginia Tech. Unless prior approval is granted, the university is not responsible for keeping or maintaining personal property left by the faculty member. The university accepts no liability for lost, damaged, or destroyed personal property.

A departing faculty member may request permission to store his or her personal property beyond the last day of employment. The following stipulations apply: (1) the request to store personal property must be submitted prior to the last day of employment, (2) such a request must be submitted to the department head or authorized supervisor, and (3) the department head or authorized supervisor has absolute discretion in approving or denying the request.

#### 2.12 Reduction in Force

Termination refers to the involuntary cessation of employment of a tenured or continued appointment faculty member or of a faculty member on a fixed-term appointment before the end of the term. Termination takes place only as dismissal for adequate cause or in the case of a reduction in force.

Furlough refers to the involuntary interruption of employment of a tenured or continued appointment faculty member or of a faculty member on a fixed-term appointment before the end of the term. This differs from termination in that it conveys an intention of the university to reappoint affected faculty members within some reasonable period if circumstances permit.

A reduction in force is the termination or interruption of employment of a member of the general faculty under conditions of financial exigency or program reduction. Reduction in personnel by attrition, freezes on new hiring, across-the-board reductions of salaries and/or teaching schedules, and the offering of incentives for early retirement, whether at the program level or institution-wide, are not considered reductions in force. Rather, they are lesser remedies that may be implemented before any reduction in force.

Denial of tenure to an untenured faculty member or non-renewal of appointment of an untenured faculty member on probationary appointment, or non-renewal of appointment of an untenured member of the administrative and professional faculty, where normal procedures have been affected in each instance, is not considered a termination within the meaning of this policy.

For the purpose of the procedures outlined below, seniority refers to the number of years served at this university by a member of the general faculty in tenured, tenure-track, or functionally equivalent positions. Service need not be continuous to contribute to an individual's seniority. Years of service include those during which a faculty member is employed at least half time. Years during which a faculty member is employed less than half time will not count toward years of service for purposes of this section.

# 2.12.1 Reduction in Force Under Conditions of Financial Exigency

Reductions in force (RIF) may occur when financial conditions disallow the normal operation of programs. While the university has a right to initiate reductions in force, including those affecting tenured faculty, it is the policy of the university (to the extent consistent with the degree of financial exigency) to ensure that the rights of tenure or continued appointment are preserved; to ensure that the integrity of the university and its programs is preserved; to protect the contractual expectations of untenured faculty; to provide that the burden of corrective action is shared by the various categories of personnel of the university, including all members of the

general faculty; and to ensure that any reductions that do occur follow an orderly and predictable process.

A financial exigency is an imminent financial crisis that threatens the survival of the university and that cannot be alleviated by ordinary budgeting practices. Reductions in force in response to conditions of financial exigency are determined and implemented as follows:

**Declaration of a state of financial exigency:** Should the president determine that so extraordinary a circumstance has arisen or is anticipated that it might be necessary to terminate or interrupt the appointments of faculty members, the president may declare a state of exigency. Upon such declaration, the president forms an ad hoc committee to review the budgetary situation and the president's plan for addressing it.

Committee review: The ad hoc committee is comprised of no fewer than nine members, a majority of whom are faculty members nominated by the Faculty Senate. This includes at least one representative from each college. Where a RIF may affect the extra-collegiate faculty, at least one representative from that faculty should also be selected to serve on the committee. Any person who resigns from or otherwise discontinues his or her service on the committee is replaced by a new member chosen in the same manner as was the individual being replaced, and such replacement members are so selected that each college and, where appropriate, the extracollegiate faculty retain at least one representative. Within the constraints of time and circumstance, the committee reviews the proposal submitted by the president and any alternative remedies that are available, and recommends to the president a plan of action that may incorporate reductions in force of the administrative and support staff as well as the general faculty. The committee is charged with protecting both academic freedom and, insofar as circumstances permit, the presumption of continuous employment that tenure or continued appointment bestows, and considers as well the curricular needs and goals of the university and the effects of any anticipated actions on the future financial well-being of the institution.

**Determination of policy:** After receiving the recommendations of the ad hoc committee, the president determines the response of the university to the declared state of exigency. If the president's decision is substantially at variance with the recommendations of the committee with specific regard to the implementation of RIFs, the committee may, by majority vote, appeal the president's actions to the board of visitors. In all other matters, and in cases where the president's decision to carry out a reduction in force accords with the recommendations of the ad hoc committee, no such appeal is available. The ad hoc committee consults with the president and receives periodic reports until the state of exigency ends and the committee determines that the obligations of the university to furloughed or terminated faculty are met.

Implementation: Reductions in force are implemented either within specified programs or across the institution. Whenever a RIF is undertaken, it is guided by the following considerations:

- Insofar as circumstances permit, all temporary or part-time faculty members and those not holding tenured or tenure-track appointments or their functional equivalent are retained through the then-existing term of appointment.
- Insofar as circumstances permit, untenured faculty holding tenure-track appointments and library and other faculty holding probationary appointments are retained through the then-

existing term of appointment. No tenure-track or functionally equivalent appointment is terminated or interrupted unless and until all appropriate temporary appointments are terminated. Where reductions in force of these personnel are required, they are implemented in ascending order of rank and of seniority within rank. Whenever possible, the university provides notice of furlough or termination equivalent to that for non-reappointment schedule as set forth in section 2.11.3, "Non-Reappointment."

• Except in the most extraordinary circumstances, all tenured faculty and those on continued appointment retain their positions. Where reductions in force of tenured or continued appointment personnel are required, they are implemented in ascending order of rank and of seniority within rank. Whenever possible, the university provides at least one year's notice of furlough or termination.

**Notification:** The university provides written notification to all faculty affected by a RIF including: (a) a statement of the basis for its action, (b) a description of the manner in which the decision in question was reached, (c) a disclosure of the information and data on which the decision makers relied, (d) information regarding reappointment rights and process, and (e) information regarding procedures available for appealing the decision.

**Appeals:** The decision to furlough or terminate a member of the general faculty because of a reduction in force may be appealed in two ways.

- The affected individual may appeal through the grievance procedure specified in the relevant section of the Faculty Handbook.
- After consulting with the appropriate dean and an elected committee of faculty members from the affected program, the principal administrative officer of a program may appeal individual RIF decisions to the provost on programmatic grounds. Reductions in force of no more than one-quarter of the affected faculty in any program may be appealed in this manner.

**Replacement and Reappointment:** The university recognizes its obligation to reappoint personnel furloughed or terminated through a RIF insofar as circumstances permit within a reasonable period following such action. Accordingly, temporary personnel cannot replace a probationary term faculty member who has been furloughed or terminated through a reduction in force for a period of three years following that action. Similarly, temporary or probationary term personnel cannot replace a tenured or continued appointment faculty member who has been furloughed or terminated through a reduction in force for a period of five years following that action.

Rather, affected members of the general faculty are granted first refusal of re-established positions for which they are qualified, with positions offered in descending order of rank and seniority within rank whenever the number of qualified personnel exceeds the number of available positions. The university attempts to identify funds to enable the university to extend to affected faculty during these periods of three and five years, respectively, all health insurance benefits for which they would otherwise have qualified. On reaching age 70, or on declining at least one offer of employment in a position equivalent in tenure status, salary, and teaching load (as adjusted to reflect post-RIF changes in his or her department) to that which was terminated, each faculty member affected by a reduction in force forfeits all protections afforded by this paragraph.

For purposes of providing insurance benefits and implementing these reappointment procedures, the provost keeps the curriculum vitae and current address of each terminated or furloughed faculty member. Terminated or furloughed faculty have an obligation to maintain the accuracy and timeliness of these records; the failure to do so results in forfeiture of the protections afforded by this paragraph.

#### 2.12.2 Reduction in Force for Academic Program Restructuring or Discontinuance

Ordinarily, change to academic programs within the university is planned so that the appointments of faculty members are not compromised. Such changes are considered part of the ongoing evolution of academic programs and are subject to the usual procedures established by the colleges, relevant commissions, and the State Council of Higher Education for Virginia.

However, when extraordinary circumstances require more rapid change, it may be necessary to restructure or discontinue programs or departments in a way that leads to involuntary terminations or other alterations of appointments of faculty members with tenure or continued appointment. In such circumstances, the policy in this section applies.

It should be understood that any decision to restructure or discontinue academic programs in a way that alters faculty appointments is a university-wide responsibility and should be made to support the educational mission of the university as a whole. In all such circumstances, early and meaningful faculty participation is essential and fundamental to the process outlined in this policy.

The restructuring or discontinuing of one or more academic programs with the potential to invoke this policy may be initiated by the provost or president, by the college deans, by the collegiate faculties, or by an appropriately charged commission. If the provost determines that such restructuring or discontinuing of academic programs should be considered, a Steering Committee for Academic Restructuring, hereinafter referred to as the steering committee, is appointed as described below. The purpose of the steering committee is to evaluate and coordinate the proposed restructuring effort, and to ensure that the procedures in this section are followed.

The steering committee is composed of nine members determined jointly by the provost and the president of the Faculty Senate: two faculty members selected from the membership of the Commission on Faculty Affairs; two faculty members selected from the membership of the Commission on Undergraduate Studies and Policies; two faculty members selected from the membership of the Commission on Graduate Studies and Policies; one faculty member selected from the University Advisory Council on Strategic Budgeting and Planning; one member selected from nominations by the Faculty Senate; and the provost, or an administrative designee.

The steering committee elects a chair by vote of all members of the committee. The steering committee composition is intended to ensure that the expertise and perspectives of the relevant commissions are incorporated in the deliberations.

The provost initiates discussion of a proposed program restructuring or discontinuance with the steering committee, describing the need for the change, the proposed type and scope of restructuring effort, the educational rationale for the change, and an explanation of how it is consistent with the long-term goals of the university. If after these preliminary discussions and upon considering the advice of the steering committee, the provost decides to proceed, the provost prepares a more detailed proposal including identification of programs to be restructured or discontinued (or how they will be identified); timelines for development of specific plans by the affected programs and for the restructuring effort as a whole; and the estimated impact on the affected faculty, staff, and students, and on the university as a whole. If a budget reduction is involved, then reduction targets for any affected unit(s) must be included in the draft proposal.

The steering committee reviews the draft proposal and makes recommendations to the provost either to proceed with the proposal as written or with modifications, or to return it as insufficiently justified. The steering committee shares its recommendations with the university community.

The provost considers the steering committee's recommendations and makes every effort to develop a plan acceptable to the steering committee. If the provost decides to proceed, he or she directs the relevant dean(s) to prepare specific plans for the affected programs, based on guidelines in the following section. These plans identify which specific programs are to be reduced or eliminated; how the faculty, staff, and students will be affected; and how the rights, interests, and privileges of the faculty and staff members will be protected. If a budget reduction is involved, the specific plan must describe how the reduction targets will be met.

Under specific circumstances approved in advance by the provost and president, an option that may be available to the deans for meeting reduction targets is the Alternative Severance Option. The Alternative Severance Option (ASO) is described in section 2.13.1.

The deans submit specific plans to the provost, who reconvenes the steering committee to oversee the review and comment process. All specific plans are made available to the university community for comment for a period of not less than three weeks. The relevant commissions (including the commissions on Staff Policies and Affairs and Administrative and Professional Faculty Affairs if such employees are affected) are also asked to review and comment on the plans. The steering committee receives all comments and makes recommendations to the provost; these recommendations are also shared with the university community at large. The president and board of visitors have final authority to approve and implement all plans. Notification to affected faculty does not proceed until final approval is given.

**Guidelines for Development of College Plans:** The relevant deans should develop specific plans by involving the faculty at all levels of decision-making. Staff members should be involved as appropriate.

College-level planning for programmatic reductions follows the guidance and intent of the plan reviewed by the steering committee and approved by the provost. For the purpose of developing the specific plans, an academic program should meet one or more of the following criteria: (a) has "program" as part of its title, (b) grants a degree or a credential, (c) has a sequence of courses

with a common prefix, or (d) has been identified as an academic program in official university documents. A program is generally smaller than a department and must be larger than the activities of a single faculty member.

If restructuring requires the termination of faculty members, then the following guidelines must be followed:

- When programs are identified for restructuring or discontinuance, all faculty assigned to the program, both tenured and untenured, are potentially subject to reassignment or termination.
- Within programs identified for restructuring or discontinuance, tenured faculty must not ordinarily be terminated before untenured faculty. Termination decisions within the tenured faculty as a group or within the untenured faculty as a group should be based on rank and merit.
- Faculty members on restricted or temporary appointments should be terminated before faculty members on regular appointments.
- The number of involuntary terminations of tenured faculty members should be minimized by providing incentives for resignation, retirement, or reassignment.

**Minimum Responsibilities to Individual Faculty Members:** The University recognizes its responsibility to faculty members if this policy is implemented. All plans to restructure academic programs guarantee the following to individual faculty members:

- Notice of termination: Faculty members with tenure or continued appointment whose positions are eliminated as part of restructuring are given notice of not less than three years. Administrative and professional faculty members shall be given at least 90 days notice. (See section 7.6.2, "Non-Reappointment of Administrative and Professional Faculty on Restricted Appointments.") All other faculty members shall complete their current contracts or be given a one-year notice, whichever is less. In particular, notice of termination longer than the minimum specified above may be given to particular faculty members whose expertise is essential to closing out an academic program in which students are enrolled.
- Written notification: After final approval has been given for specific plans, written
  notification is provided to all faculty members whose appointments will be terminated or
  altered. The notification shall include a statement of the basis for its action, a description of
  the manner in which the decision was reached, a disclosure of the information and data on
  which the decision was based, and information regarding procedures available for appealing
  the decision.
- Transition assistance: Every effort is made to place affected faculty members with tenure or
  continued appointment in available openings in the university or to reassign them to
  continuing programs. Transition assistance may include training to qualify for placement in a
  related field if desired and appropriate. Where placement in another position is not possible,
  the university provides appropriate and reasonable career transition assistance such as
  clerical support, communications, office space, and outplacement services.
- Reappointment: In all cases of termination of appointment because of program reduction or discontinuance, the position of a faculty member with tenure or continued appointment cannot be filled by a replacement within a period of three years following separation unless the released faculty member was first offered reinstatement and a reasonable time in which to accept or decline.

**Appeals:** A faculty member whose appointment is terminated or altered due to program reduction or discontinuance may file a grievance as outlined in the relevant section of the Faculty Handbook. Grounds for appeal may be substantial failure to follow the procedures and standards set forth in this section. Because faculty members, through the steering committee, are involved in the review and development of recommendations guiding the restructuring or discontinuance, the determination of which programs or departments are affected cannot be a basis for appeal.

#### 2.13 Severance Benefits

The university provides severance benefits for eligible faculty who are involuntarily separated due to budget reduction, agency reorganizations, or workforce downsizings for reasons unrelated to performance or conduct. Faculty hired on restricted appointments funded from sponsored contracts or grants, or term appointments with a specified ending date, regardless of funding source, are not eligible to receive severance benefits. Non-reappointments and voluntary resignations for any reason shall are not be deemed "involuntary separation" for purposes of the severance policy.

# 2.13.1 Alternative Severance Option

Under specific circumstances approved in advance by the provost and president, an Alternative Severance Option (ASO) may be available to eligible faculty. Severance of faculty members with tenure or continued appointment must be voluntary; no tenured faculty member can be required to participate. Tenure-track and continued appointment-track faculty members are not eligible, nor are restricted employees.

The premise for any severance payment rests on the rationale of business necessity to reduce personnel expenses. When such a situation occurs, deans and senior managers will be asked to define the business operations, academic programs, departments, or units where personnel reductions will occur. An approved business plan for each participating college or vice presidential area will describe the specific units and eligibility criteria for participation in the ASO or layoff substitution process. These plans will necessarily differ. Some college and senior management do not offer the ASO as a means to reach their budget reduction targets. Not all employees who are eligible will be selected to participate if more apply than are needed to address the reductions or if an individual employee serves a critical function. Eligible employees in units with approved business plans are notified if the option is available to them.

## 2.14 Consulting and Outside Employment

## 2.14.1 Consulting Activities

The university recognizes that consulting work for external entities enhances the professional development of faculty members and provides channels for communication and outreach not otherwise available. For complete and updated information and forms consult the Conflict of Interest (COI) webpage maintained by the Office of the Vice President for Research.

This policy differentiates between external consulting and professional service activities as follows:

External consulting is professional activity related to an individual's area of expertise, where that individual generally receives compensation from a third party and is not acting as an agent of the university. Consulting may take many forms, but the guiding principle is that, in consulting, a person agrees to use his or her professional capabilities to further the agenda of a third party in return for an immediate or prospective gain. Even in cases without compensation, advance approval is required to document the proposed external activities and to ensure they do not constitute a conflict of commitment, or a conflict of interest where gifts of equipment or donations to the faculty member's laboratory may substitute for direct compensation. Provisions of the consulting policy also apply to external activities where the faculty member has a direct relationship to the external entity, such as personal or family ownership of the company. Consulting does not involve becoming an employee of the external entity.

Professional service includes service on national commissions, on boards of governmental agencies, on granting agency peer review panels, on visiting committees or advisory groups to other universities, on professional associations, and on analogous bodies. Professional service activities may involve a token honorarium and/or expense reimbursement. These activities are considered part of the faculty member's institutional responsibilities for participation in the larger scholarly academic community. Participation in external professional service activities may require supervisor approval depending on departmental practice and expectations of the position. Annual leave is not required.

Consulting arrangements may be entered into by faculty members during periods of university employment provided that:

- such advice is not part of their normal responsibility to the university and is not normally provided through Virginia Cooperative Extension, outreach programs, or other component of the university;
- the work undertaken contributes to their professional development;
- the work can be accomplished without interference with their assigned duties and does not ordinarily involve more than one day per week and does not exceed five days in any five-week period;
- university resources and facilities are not involved (except as described below in the use of university facilities policy); and
- written approval in advance is obtained from the faculty member's department head or chair, and dean or senior manager.

Faculty members whose appointments are funded in whole or in part by sponsored projects may participate in consulting when consistent with their responsibilities and in compliance with federal contract compliance and state regulations. University time available for consulting is in proportion to base salary funding from non-sponsored sources. With supervisor approval, additional consulting days may be charged to annual leave.

Consulting work should involve advisory services based on a faculty member's store of knowledge and experience in contrast to programs of research, development, or testing, which may interfere with the performance of the faculty member's duties or conflict with university interests.

In any faculty consulting arrangement, the name of the university must not be used in connection with any product or service developed as a result of such consulting nor in any connection arising out of the arrangement.

Paid consulting by faculty members is not permitted for work done for a group within the university. For example, if a faculty member advises or assists the principal investigator on a grant, there shall be no pay for the services. Such consulting is considered part of the normal duties of faculty members. Faculty members may be paid for participation in non-credit instruction offered through appropriate university units, such as Continuing and Professional Education or University Organizational and Professional Development, in accordance with overload payment policies in section 2.17 of the Faculty Handbook.

When a faculty member testifies as an expert witness, the following conditions apply: a disclaimer is given in court indicating that the faculty member is speaking as a professional and not as a representative of the university; when a faculty member is under subpoena, the university civil leave policies apply; and a faculty member may not testify in civil suits involving the Commonwealth of Virginia, except under subpoena.

Virginia Cooperative Extension employees should also be aware of specific policies covering faculty having federal appointments. (These appear in sections 4.12 and 7.9, "Consulting Activities for Virginia Cooperative Extension Faculty.")

Faculty members intending to do consulting work should also read the policy on use of university facilities. Except under the provisions specified in that policy, faculty members are not allowed to use university resources in conjunction with consulting or otherwise for private gain. This includes the parallel use of university facilities associated with consulting activities; i.e., when a faculty member is engaged in authorized consulting activities, the consulting employer may not enter into an agreement to use university resources for any purpose related to the consulting activity. Instead, when significant resources of the university are required, the employer may request that an agreement, grant, or contract be drawn with the university that provides the necessary services, including human resources. The faculty members then carry out duties attendant on the agreement as part of their assigned university duties. Because library facilities are made available to the public, their use in consulting is not regarded as being in contravention of this policy.

Because of the university's land-grant mission, it may be in the best interest of the university to impose some additional restrictions on the consulting activity of the faculty of one or more of the colleges. Therefore, an academic dean, after consulting with his or her faculty, may recommend to the provost that the faculty of that college need to satisfy additional requirements for consulting approval. The provost, after consulting with the Commission on Faculty Affairs, and with the approval of the president and the board of visitors, may require that the faculty of the affected college satisfy such additional requirements.

Oversight of faculty consulting is a responsibility of the department heads or chairs and other relevant administrative officers of the university so that a reasonable and appropriate level of external activities is maintained and normal duties are not neglected.

A consulting request must be approved by the department head or chair and dean. Notice of approval is accomplished by returning a copy of the signed approved consulting request form to the faculty member. (See related reporting forms available on the Conflict of Interest website maintained by the Office of the Vice President for Research.) Further information on the conflict of interest policy appears below and on that website.

Approval of consulting or other external activities for faculty members holding nine-month appointments is not necessary during the summer unless there is concern about conflict of interest, or the university employs the faculty member during the consulting period. When the university employs the faculty member in the summer months, university and college consulting policies apply.

Setting the consulting fee is the prerogative of the faculty member. The actual or estimated consulting income is reported on the request form to allow reviewers a full assessment of potential financial conflicts of interest. Income received for consulting work is not considered when faculty members are evaluated for annual merit salary increases.

# 2.14.2 Technical Assistance Program

Consulting agreements may be negotiated by the individual faculty member and the sponsoring organization, not involving university participation in any way, or they may be negotiated as part of a technical assistance agreement through the university. The Technical Assistance Program was created as part of the university's outreach mission to respond to requests from business and industry for the application of knowledge to a specific process-related or technical situation.

Proposals for technical assistance are small scale (generally less than \$25,000), short-term, require a rapid response, and do not involve the generation of new knowledge or the development of intellectual property. (Projects involving the generation of knowledge and/or faculty buyouts must be handled as sponsored projects.) Contracts for technical assistance are negotiated and administered by the Division of Continuing and Professional Education.

Technical assistance contracts typically identify the faculty member who will provide the needed expertise, the amount of time to be devoted to the project, the scope and estimated cost of the work, timelines for the consulting or project, and any required deliverables.

Payment to the faculty member for such consulting is negotiable and provided through university payroll. Faculty earnings for technical assistance agreements must be within the overall limitation of 33 ½ percent of annual income during the academic year for nine-month faculty members; summer earnings from all university sources are also capped at an additional 33 ½ percent for academic year faculty members. Faculty members on calendar year appointments may earn 33 ½ percent of annual income during the fiscal year. The earnings limitation is for payments from all university sources, including approved non-credit continuing education

activities. Similarly, total time involved in technical assistance, other approved consulting, and non-credit teaching must be within the constraints of this policy.

For further information on technical assistance agreements, please contact the Division of Continuing and Professional Education. Completion and approval by the department head or chair and dean of a technical assistance agreement substitutes for approval of a Request to Engage in External Activity Form 13010A usually required for approval of consulting.

## 2.14.3 Outside Employment and External Activities Other Than Consulting

Outside employment, not meeting the definition or intent of the consulting policy, requires prior approval of the supervisor and relevant university officials. Approval is contingent on assurance that the primary commitment to Virginia Tech will be fulfilled and that the proposed employment does not constitute a conflict of interest. Release time from university work is not normally available for paid activities that are primarily personal in nature, do not enhance the faculty member's professional skills, or that are not a potential benefit to the university. The faculty member must use pre-approved leave (or leave without pay) in cases where outside personal work creates a potential conflict with university responsibilities.

#### 2.15 Conflict of Commitment

A conflict of commitment arises when the external activities of a faculty member are so demanding of time, attention, or focus that they interfere with the individual's responsibilities to the university.

Faculty members have traditionally been allowed wide latitude in defining their professional agendas and their degree of involvement in external activities when those activities advance the mission or prestige of the university. The university encourages active participation by faculty members in external activities that are integral to and/or enhance their professional skills and standing or that constitute substantive outreach and public service activities.

Such activities are usually expected of faculty members to promote academic development, and to enrich their contributions to the institution, their profession, the state, and the national and world societies. Additionally, Virginia Tech encourages entrepreneurial activities by faculty, recognizing that such activities are critical to promoting economic development and meeting society's needs, provided that participation in those activities is in compliance with federal and state laws and policies, the Virginia Tech conflict of interest policy, and these guidelines.

Faculty members should make the fulfillment of their responsibilities to the university the focal point of their professional effort. They are expected to arrange their external activities so that they do not impede or compromise their university duties and responsibilities. Responsibility for ensuring commitment to the university and for reporting activities that might be perceived as compromising that commitment rests with each faculty member in consultation with his or her unit administrator (typically the department head or chair, or school or center director) and dean.

The university recognizes that the balance of external activities varies among individuals, from discipline to discipline, and from one type of proposed activity to another. That balance is affected by unit goals and changing needs for teaching, research, creative/artistic activities, extension, service, and outreach. Primary duties and responsibilities may vary from year to year for individual faculty members. Undergraduate and graduate enrollment demands, faculty-staffing levels, and changes in the nature and scope of outreach, teaching, and research within the unit may affect the primary duties and responsibilities of individual faculty. The primary judgment as to whether a faculty member is meeting his or her professional responsibilities to the unit rests with the department head and dean (or relevant senior manager).

If a faculty member is committed to engaging in an external activity that compromises his or her ability to meet university responsibilities, a leave of absence or a reduction in percent employment may be appropriate or necessary. Approval of a leave request or change in appointment depends on the needs of the college and unit and protection of university interests.

If a unit administrator, or dean, observes that a faculty member appears not to be fulfilling his or her primary responsibilities to the university, the administrator shall immediately address these concerns with the faculty member to ensure that these responsibilities are adequately met. Failure to meet primary departmental and university obligations is handled through established university procedures appropriate to the situation (for example, formal reprimand, non-reappointment, post-tenure review, or dismissal for cause). See sections of this handbook for relevant appeal processes for such actions.

Nothing in this policy statement shall be interpreted as interfering with the academic freedom of faculty members, nor with their primary responsibility to direct their own research.

## 2.16 Conflicts of Interest (Policy 13010)

Virginia Tech recognizes that external activities congruent with the professional expertise of faculty and consistent with the mission of the affiliated department can enhance professional development of the faculty member and enrich the academic experiences of students. Activities such as licensing of technology, consulting, or business start-ups can be critical to promoting economic development and meeting society's needs. Given these potential benefits, Virginia Tech has encouraged innovation and entrepreneurial activity in support of the broad missions of the institution. However, the primary professional allegiance of the faculty must be, both in fact and in perception, to the university, and the primary commitment of the faculty member's time and intellectual energies must be devoted to the education, research/scholarship, and outreach programs of Virginia Tech. While external activities are encouraged, the integrity of the institution and of the research conducted by individual faculty members and students depends on a high degree of transparency and appropriate oversight of relationships with outside entities. Elimination or careful management of disclosed potential conflicts of interest provides assurance to research sponsors, subjects, and the broader public that possible personal gain has not influenced the design, conduct, or outcomes of the research, nor inappropriately biased decision making in other university activities. This policy provides the basic framework for assessing potential conflicts of interest or commitment and outlines related procedures for the management and monitoring of external activities in a manner that will both promote and safeguard the interests and reputation of Virginia Tech, its faculty and students, and their research.

All faculty members must be committed to conducting themselves in accordance with the highest standards of integrity and ethics, described in Professional Responsibilities and Conduct in the Faculty Handbook as well as in other university policies. These standards also include identification of the potential for *conflicts of interest and commitment*, and the assurance that participation in external activities does not improperly affect the faculty member's teaching and research, relationships with students or colleagues, or otherwise jeopardize the reputation of the university. While this policy refers to "faculty," its provisions apply to all employees, including all types of faculty, staff, or employed students, whether full or part-time. Issues of conflict of interest and commitment are also addressed in the Statement of Business Conduct Standards; all employees must acknowledge receipt of the statement and agree to abide by the standards.

An *individual conflict of interest* occurs when a faculty member is in a position to advance one's own interests or that of one's family or others, to the detriment of the university.

A conflict of interest may occur when an individual's private interests influence his or her professional obligations such that an independent observer might reasonably question whether the individual's actions or decisions are determined by considerations of personal gain, financial or otherwise. A conflict of interest is assessed based on the factual context of the situation; it is not based on any implied judgment about the character or intent of the individual.

State and federal law define differently a *significant financial interest* and the conditions under which a conflict of interest may be deemed acceptable. The *Virginia State and Local Government Conflict of Interests Act* defines a prohibited conflict of interest when an employee, or a company in which an employee or member of the employee's immediate family, has a personal interest contract directly or through a subcontract with a state institution of higher education. In this instance, a *personal interest* is defined as owning three percent or more of the equity in the company or receiving payments from the entity of \$10,000 or more annually from consulting fees, salary, royalties, honoraria, rent, or other income.

In the case of grants and contracts, federal agency guidelines on conflict of interest disclosure apply in addition to state requirements. As a result, investigators are required to disclose personal financial interests beyond that required by state law in order to comply with sponsoring agency regulations. For publicly traded entities, a significant financial interest exists if the value of any remuneration received from the entity in the twelve months preceding the disclosure and the value of any equity interest in the entity as of the date of disclosure, when aggregated, exceeds \$5,000. For non-publicly traded entities, the threshold is \$5,000 for remuneration or when the investigator or investigator's spouse or immediate family members hold *any* equity interest (e.g. stock, stock option, or other ownership interest). A detailed definition of significant financial interest and a list of income and investments excluded from disclosure requirements are included in section 2.16.12 Definitions and on the COI website.

The required disclosures are those that are reasonably related to the investigator's institutional responsibilities. Institutional responsibilities are defined broadly and include but are not limited to teaching, advising, research and scholarly activities, outreach, administrative and institutional committee service, and service to professional associations or on panels such as peer, institutional, or accreditation review boards.

Individuals in administrative positions have substantial influence by virtue of their roles in professional appointments, promotions, tenure decisions, allocations of space, determinations of salary, execution of business contracts, etc., and they must take particular care to avoid relationships that have the potential to advantage the individual but adversely affect the university's interests or inject inappropriate considerations into administrative decisions. Administrators must be vigilant in exercising their authority so that their decisions are not in fact, nor do they appear in the perception of others, to be colored by personal financial interests.

# 2.16.1 Potential Conflicts Involving Research and Development Contracts or Commercialization of Intellectual Property

To ensure compliance with federal regulations and to provide consistent institutional policies and practices in relation to all research sponsors, investigators engaged in sponsored research must disclose any financial interest that meets or exceeds the federal definition of significant financial interest as detailed in the Definitions section of this policy. The investigator must present a complete picture of his or her financial interests, not just those interests related to a specific funded project. Examples of potential conflicts of interest in research and development (R&D) that must be reported by investigators are available on the COI website. Investigator obligations for disclosure extend to collaborators, consultants, and subcontractors on a sponsored project, who must certify disclosure and compliance with conflict of interest policies promulgated by their home institution or, in the absence of relevant institutional policies, must disclose in accordance with Virginia Tech policies and procedures.

Contracts between employee-owned businesses and the university that involve research and development activities and/or commercialization of intellectual property are permitted by state law provided that disclosure and approval requirements of this policy are followed. In the case of a sponsored project funded by an employee-owned business, the role of the faculty-owner in relation to the university research project must be carefully examined. The faculty-owner may only serve as principal investigator with an independent co-principal investigator responsible for financial oversight of the project, never as sole PI. Work done on behalf of a faculty owner's company in a university facility must be done in accordance with the provisions of Policy 5000, University Facilities Usage and Event Approval. Because of the high potential for appearance of an irresolvable conflict of interest in the case of an SBIR/STTR (Small Business Innovation Research or Small Business Technology Transfer) project, an individual may not serve as both the small business PI and the university subcontract PI (or co-PI), nor may the employee contribute to both the small business and university components of a single project as an employee, investigator, or consultant.

Organizational conflicts of interest may occur where the involvement of a faculty member in a consulting or advisory services contract with a federal agency may bar other faculty members from participating in a subsequent contract or grant from that agency. The limitation is based on a perceived institutional advantage derived from that prior involvement. The Office of the Vice President for Research, in consultation with the department head, dean, or unit leader as appropriate, is responsible for determining whether the risk of exclusion from bidding on a subsequent contract should result in denial of the faculty member's involvement in the initial consulting or advisory services contract.

Virginia Tech is committed to the free and open exchange of ideas and the timely dissemination of research results to the broader scholarly community and public. Providing preferential access to an outside entity to research results, materials, or products generated from university teaching, research, or administrative activities for personal financial gain is prohibited. University resources, including facilities, personnel, equipment, email, or confidential information, may not be used for faculty consulting or other non-university purposes. University letterhead may only be used for university business and not for outside consulting or other personal purposes. Employees must separate clearly and cleanly the personal financial aspects of their entrepreneurial activities from their university activities. Even though permitted, entrepreneurial activities should not be confused with assigned duties of the university.

# 2.16.2 Potential Conflicts Involving Spouses and Immediate Family Members

As a matter of state law, employees must avoid being in a position of authority over a spouse or a member of the immediate family who is also employed by the university where the spouse or family member earns \$10,000 or more during a fiscal year. An employee and his or her spouse or another member of the immediate family may both be employed by the university so long as the employee does not exercise any control over the employment conditions and activities (such as initial appointment, retention, promotion, tenure, salary, travel approval, leave of absence, or grievance review) of the spouse and is not in a position to influence those activities. Proposed exceptions and alternate reporting relationships are reviewed and approved by the senior vice president and provost (or relevant vice president for a non-academic appointment) prior to submission to the board of visitors for approval.

Faculty members have a responsibility to avoid potential conflicts of interest between their professional responsibilities in teaching, research, or administration and their personal interests in relationship to students who are also family members. The concern is assuring an absence of bias in evaluation of coursework and maintaining both the appearance and reality of fairness of treatment among all students in a class or an academic program. When there is no reasonable alternate course or instructor for a required course taught only by the conflicted faculty member, then the relevant academic associate dean and department head work with the faculty member to identify an alternate evaluator for the student-family member's work. It is the prerogative of the conflicted faculty member to agree to the selection of the faculty colleague for this alternate evaluator assignment. Graduate School policies prohibit faculty members from serving as advisor or committee member for family members (spouse or dependent immediate family member), or for individuals with whom they have a close personal relationship such as partner or extended family member, or close professional relationship such as business associate or supervisor.

Consensual relationships between faculty members (including GTAs) and students enrolled in their classes, or students for whom they have professional responsibility as advisor or supervisor, violate the policy on Professional Ethics and Responsibilities and may be a violation of policy 1025, Anti-Discrimination and Harassment Prevention Policy.

## 2.16.3 Potential Conflicts Involving Students and Post-Doctoral or Other Trainees.

The impact of a perceived or actual conflict of interest or commitment of faculty members on their students (including post-doctoral fellows and other trainees) is of special concern to the university. In particular, the university is committed to maintaining the content and quality of the educational experience for students whose research is sponsored by a for-profit business and whose faculty advisors have a financial interest or a management role in that business. The concern is even greater if the dissertation work could potentially affect the value of a company in which the faculty member has an ownership or managerial interest. It is most important that students make acceptable progress towards their degree goals. Work on a sponsored project funded by a faculty company, or working directly for a faculty-owned enterprise (or consulting), must not inhibit the student's academic advancement.

This policy recognizes that the involvement of students in faculty-owned or managed businesses or consulting may yield substantial benefit to the student, provided that certain conditions are met. Faculty members must disclose the proposed involvement of students in their external activities or in company-sponsored research. Faculty members who intend to serve as PI or co-PI on sponsored projects and have graduate students or trainees conduct a portion of the research under their supervision must develop a conflict of interest management plan to address protections for these early career scholars. Students, trainees, and lab personnel have a right to be fully informed about the source of their funding, the nature of the faculty member's personal interest or involvement, and, if relevant, any agreements concerning data collection, copyright, or patent protection arising from the research. Documentation of a discussion held with the student by the graduate program director or department head and the relevant associate dean is required to assure that the student has been informed of university and academic policies, the potential risks of the proposed involvement, and the channels available through the department or Graduate School to address perceived problems.

A faculty member with a significant financial interest in a company may not serve as sole advisor of a dissertation or thesis committee for a student who is funded through a university sponsored project supported by the advisor's company, or for a student who is employed directly by the faculty member's company. The faculty member with the conflict may serve as co-chair or committee member. If another co-chair is appointed, that faculty member must be of equal or greater rank, must not be involved with the sponsoring company, and must not be subject to undue influence by the faculty-owner.

If the student is engaged in longer term or more time-consuming participation in the advisor's consulting practice, or employed only during the summer months by the advisor-owned company, a co-advisor appointment may also be warranted, but is not automatically required.

The Graduate School shall receive a copy of any approved management plan involving a graduate student performing work for an advisor's company or consulting practice, and the signed student agreement acknowledging his or her understanding and voluntary participation. Every effort must be made by departmental and college officials to meet with the student prior to the start of the company-funded assistantship, or as quickly thereafter as possible, to review relevant policies and the terms of the student agreement. Copies of the signed student agreement

should be distributed to all members of the student's graduate committee by the department head or graduate program director to encourage their active engagement in protecting the student and assuring the integrity of the research.

Careful monitoring of the approved management strategies and annual reports of academic progress for the student is the responsibility of the department head and graduate program director, with oversight by the relevant academic dean(s).

## 2.16.4 Potential Conflicts Involving Other University Employees

The involvement of other faculty or staff members in employee-owned businesses or faculty consulting also requires caution and careful consideration. Any such employment must be outside the scope, and schedule, of their regular university duties and must be approved by the department head and other administrators as required by this policy and policy 4070, Additional/Outside Employment for Classified or University Staff. The concerns may be a potential conflict of interest if the faculty member supervises and evaluates the staff member (or a less senior faculty member), creating an opportunity for bias (favorable or unfavorable) toward the individual based on their involvement in the faculty member's (or supervisor's) personal company or consulting. There is also a potential conflict of commitment, where the employee's effectiveness in his or her primary role at VT may be compromised by the outside employment. For university or classified staff, the department head will determine if the additional employment may adversely affect the employee's performance in the university salaried position and therefore should not be approved. If a request is granted and job performance subsequently deteriorates, the department head may rescind permission for the outside/additional employment. The involvement of other faculty members in an employee-owned business or consulting is documented and approved as part of the disclosure by the employee-owner and by the faculty member's own request for involvement in external activities.

## 2.16.5 Procurement-Related Conflicts and Gifts to Employees

The purchase of goods or services by the university from an entity owned by an employee of the university, or by a member of the employee's immediate family, is a prohibited conflict of interest under state law. The conflict exists even if the purchase is being made by a department/unit other than the employee's home department. A business owned by an employee's spouse or immediate family member is thus barred from doing business with Virginia Tech to prevent any perception of inappropriate favoritism. Any proposed exception must be handled in accordance with the Virginia Public Procurement Act.

The university seeks to maintain the highest level of public trust in its procurement processes, which includes avoiding perceptions of conflicts of interest that might be generated by contracting with firms where employees were previously employed or with whom they have a close personal relationship when the employee is in a position to influence a purchasing or contracting decision. Decision makers with potential conflicts should recuse themselves from participation in the evaluation and selection process.

The university's Business Conduct Standards apply to all employees and they prohibit the acceptance of gifts and gratuities, or promises of future employment from contractors or individuals seeking to do business with the university. Meals and travel offered by vendors are included in this prohibition. Nominal gifts (such as pens, calendars, note pads, other give aways, or reception food provided by vendors at a public event such as a conference) may be accepted. More information and examples of potential conflicts related to procurement and gifts can be found on the Purchasing website: <a href="http://www.purch.vt.edu/Department/Ethics.html">http://www.purch.vt.edu/Department/Ethics.html</a>.

### 2.16.6 Disclosure Requirement

State law and federal law require that faculty members fully disclose personal financial interests that present a potential conflict of interest with one's university responsibilities. Involvement with external activities, such as consulting or outside employment, or financial relationships with external entities related to one's professional responsibilities must be approved in advance. All employees (or their immediate family members) who have a private company (or personal ownership interest meeting or exceeding state limitations) that wishes to do business with Virginia Tech must disclose this financial interest annually. All investigators (or their immediate family members) who own or have ownership interest in a private business that would reasonably appear to be related to their institutional responsibilities must disclose that financial interest and have it evaluated for potential conflicts of interest or commitment, whether or not that company intends to do business with Virginia Tech. A renewal application must be filed annually for those approved activities that extend beyond one year. When a conflict management plan is required by the circumstances, full and immediate compliance is essential to assure the integrity of university-sponsored research, teaching, outreach, and administrative activities. University officials, not the faculty member, ultimately determine whether a reported financial interest constitutes a conflict and whether an award can be accepted or must be declined given the possible unresolved issues.

Virginia Tech will follow regulations contained in the Code of Federal Regulations and state law to respond to a written request for information concerning disclosed financial conflicts of interest related to research projects funded by the Public Health Service.

# 2.16.7 Training Requirement

The complexity and importance of managing potential conflicts of interest warrant the full attention of members of the university community. Initial and periodic training concerning applicable university policies and state and federal law and the investigator's responsibilities for disclosure is required for all faculty engaged in sponsored research. Federal regulations require all Public Health Service investigators to complete training prior to the award of a grant or contract. Initial training for other investigators must be completed in conformance with procedures issued by the Vice President for Research.

Retraining related to conflict of interest is required every four years and/or whenever there is significant change in university policy related to investigator responsibilities. Any investigator found to be in non-compliance with disclosure requirements will also be required to complete immediate training.

#### 2.16.8 Related Policies

This policy must be read in conjunction with related university policies and procedures, such as intellectual property, use of facilities and university resources, procurement, and Faculty Handbook sections on consulting, outside employment, and conflict of interest and conflict of commitment. Policy guidance is provided on the Conflict of Interest (COI) website maintained by the Office of the Vice President for Research and the Standards of Conduct and Ethics website maintained by the Purchasing Department.

#### 2.16.9 Conflict of Interest Procedures

#### 2.16.9.1 Disclosure of External Activities

University policy requires reporting of a wide variety of external activities, whether they generate additional income for the faculty member, or an immediate family member, or not. Certain traditional scholarly-related activities do not require reporting, such as serving on a peer-review panel, guest lecturing at another institution, serving as a speaker or panelist at a professional society meeting or conference, or editing a scholarly journal, even when the faculty member receives an honorarium for such service. A list of common activities that do not need to be reported is available on the COI website.

Advance approval is required for consulting or outside employment. Use form 13010 for disclosure and approval of such activities on an annual basis each fall, or as needed for new activities. Include sufficient information for department and college officials to review and evaluate the proposed activities for conflict of interest or commitment. A single form may be submitted for on-going consulting activity with a single entity during the fiscal year. Continuation of that same consulting activity beyond the current fiscal year must be reported and approved annually. Separate disclosures are required for each consulting client so that potential conflicts of interest or commitment can be properly evaluated. A single disclosure denoting ownership of a consulting company with multiple unidentified clients does not allow an adequate assessment and is considered insufficient and unacceptable. Proposed external activities must be reported sufficiently in advance to allow a meaningful evaluation and approval process, including development of a management plan in the case of a conflict of interest or other legal issues posed by the proposed activity.

External activities with no anticipated impact on related sponsored research, or employment or funding of students or trainees, and within stated limitations of university policies on consulting and/or outside employment may be approved by the department head and relevant dean or senior administrator. Exceptions to the time limitations on consulting and outside or additional VT employment require approval by the provost, or the relevant vice president. The original approved form is kept in the employee's departmental personnel file. Copies are sent to the employee and the designated conflict of interest officer, and retained by the relevant dean or vice president's office.

Staff members may request approval for outside employment and/or additional employment at Virginia Tech in accordance with policy 4070 using the form *Request for Additional/Outside Employment for Classified or University Staff*.

#### 2.16.9.2 Disclosure of Potential Conflicts of Interest

Form 13010 is also used to disclose annually external activities, or relationships with external entities, that may involve a potential conflict of interest. Examples of activities that generally require disclosure are available on the COI website, such as employee (or family) ownership of a private business seeking to sponsor research at Virginia Tech, or utilizing students, trainees, or other university employees in the faculty member's private business or consulting. A newly acquired significant financial interest, such as might occur through marriage, inheritance, or establishment of a new company, must be disclosed within 30 days. Disclosures must be filed and approved prior to submission of a related proposal for sponsored research.

For faculty members with PHS/NIH funding, reimbursed or sponsored travel paid for by an external entity must also be disclosed. Disclosure includes at a minimum the purpose of the trip, identity of the sponsor/organizer, the destination, and duration. Disclosure is not required for travel reimbursed by the university from any source of university funds (e.g. departmental, foundation, sponsored project), or travel paid for (or reimbursed) by a federal, state, or local government agency, an institution of higher education, affiliated research institute, or academic teaching hospital. Special rules concerning reporting of travel apply to those employees who must complete the annual Commonwealth Statement of Economic Interest forms.

It is essential that the employee describe in detail the totality of the relationship with the external entity, including but not limited to the extent of personal or family financial interest, any involvement of students or other employees, and any project-specific considerations. Disclosures will be reviewed initially by the department head or the designated administrator for employees in non-academic units. Guidance on how to assess reported activities or relationships for potential conflicts of interest is available on the COI website. A management plan must be drafted if the faculty member expects related future research involvement with the entity.

Once approved by the department or unit head, the form is routed for review and approval by the relevant dean or vice president's office. Review and approval by the Office of the Vice President for Research is required for disclosures involving business ownership interests of investigators (or their family members), significant financial interests related to sponsored research, employment or funding of students/trainees/staff, and any proposed management plan.

When the Office of Sponsored Programs receives a proposal involving an employee-owned business either as a research sponsor or subcontractor, relevant disclosures (completed form 13010, management plan, Virginia Statement of Economic Interest) are collected and reviewed by the designated COI officer for sufficiency. A summary of the proposed award and nature of the financial interest is prepared for further review by university legal counsel, the vice president for research, the provost, and president. Contracts approved by the president are subsequently reported to the board of the visitors.

A fully approved disclosure and management plan, if relevant, must be in place prior to execution of a related sponsored award. An investigator's significant financial interest related to a sponsored project and key elements of the management plan are reported by the designated conflict of interest officer to the federal agency sponsor in accordance with agency requirements.

Should a significant financial interest be identified subsequent to project initiation, the COI officer will promptly review the financial interest and its relationship to the project and implement an interim management plan to mitigate the potential conflict, including suspending further project expenditures pending a more complete review and determination of appropriate action.

## 2.16.9.3 Virginia Statement of Economic Interest

The Commonwealth of Virginia requires separate, additional reporting of economic interests by members of the boards of visitors, senior administrators, select others with financial decision making authority, and employee business owners whose companies seek to contract with the university for research and development or commercialization of intellectual property. The *Statement of Economic Interest* is completed annually and submitted on or before January 15, and as needed in the case of a new research contract. The form must be notarized and filed with the associate vice president for human resources, who will submit it to the commonwealth authorities. For disclosures related to a pending research contract, a copy of the employee's completed statement is sent to the designated COI officer in the Office of the Vice President for Research. A current, completed statement must be filed prior to execution of an award for a related research contract.

## 2.16.9.4 Management of Potential Conflicts

A management plan is used to document the proposed strategies for eliminating, mitigating, or managing a potential conflict of interest. Approved strategies must reflect the commitment of the faculty member to university principles and values of transparency; strict adherence to the highest standards of professional ethics and conduct of research; appropriate and timely dissemination of research results; and protection of students, employees, and human subjects involved in the research. The management plan can call for implementation of several strategies designed to protect these values and the integrity of the research, research sponsors, the employee, and the university. Examples of frequently adopted strategies and sample management plans are available on the COI website. Management plans are typically project specific, reflecting the conditions of the proposed research. Management plans must fully describe the situation giving rise to the potential conflict, the proposed strategy to eliminate, mitigate, or manage the potential conflict, and the process by which the activities and the management plan will be monitored. Appropriate monitoring should include an oversight function to evaluate the effectiveness of and ensure compliance with the strategies specified in the management plan. In accordance with Public Health Service (PHS) regulations, disclosure of an investigator's significant financial interest must be made directly to the human subjects involved in the PHS-funded research project.

A faculty-owner may not serve as sole principal investigator on a university project that is funded wholly, or partially, by the external organization with which the faculty-owner is associated. The co-principal investigator must be a faculty member who is not in a subordinate position to the faculty-owner, nor associated with the faculty-owner's company, or in any way susceptible to potentially inappropriate influence by the faculty-owner. The faculty-owner may not have financial responsibility for the contract. Oversight arrangements must be carefully

crafted for those cases where the faculty-owner remains involved in the sponsored research. Faculty consulting with an external entity which then sponsors research at the university involving that faculty member also presents a potentially serious conflict of interest which must be eliminated or carefully mitigated.

All management plans are reviewed and approved by the Office of the Vice President for Research to assure consistency and compliance with applicable sponsor regulations and university policy.

Awards will be contingent upon acceptance and implementation of the plan by the faculty member. While the faculty member bears primary responsibility for carrying out the plan, department heads and deans are expected to monitor compliance and assure that protections are provided for employees and students. The faculty member must submit a report annually and at the end of a related sponsored project to the department head and the Office of the Vice President for Research to demonstrate compliance with all conditions identified in the management plan. The report must include certifications of compliance from other parties identified in the management plan, such as a research oversight committee. Such reports may also be required for submission to specific federal agency sponsors.

# 2.16.9.5 Participation of and Payment to Students

The participation of students in projects involving faculty-owners should be given particularly careful consideration. Work for faculty-owned companies or in faculty consulting provides valuable experience for undergraduate and graduate students. Nevertheless, such opportunities come with some risk that the student may be diverted from his or her educational goals or the perception that students are being used primarily for the benefit of those companies. For example, a faculty member who pressures a student to complete work related to the faculty member's company could easily affect the student's completion of graduate studies in a timely and appropriate way, thereby putting the faculty member's interest in obtaining proprietary results ahead of the student's academic or scholarly research activities. The concern is similar for the involvement of students in faculty consulting or other external activities. The risks and benefits of such involvement must be carefully weighed by departmental administrators responsible for evaluating the disclosure and approving the request, particularly where the involvement may be longer term and/or more time consuming. See section 2.3 for further discussion of issues related to student involvement in faculty members' external activities or outside businesses.

Where approved, students may be paid for involvement in faculty-owner activities in either of two ways:

1. Remuneration may be in the form of an assistantship and tuition, or wages funded by a sponsored project contracted to the university from the business or organization with which the faculty-owner is associated. The assistantship or wages are remuneration for work only within the agreed scope of that funded project and for no other tasks undertaken for the benefit of the external organization. This is no different from any other sponsored project that involves graduate research assistants or wage-earning

- students. In these circumstances, assistantships are constrained to payments within the scales published by the university.
- 2. Alternatively, the company or agency might engage students directly as employees. This is the situation experienced by most off-campus and part-time graduate students. However, it is envisaged that in the case of faculty-owned businesses, students will spend time in university facilities when not engaged in direct work for the company at the company site or in an off-campus location. [Students may not do work on behalf of the company in university facilities.] Remuneration is not limited to university scales when students are employed directly by the company and may include the cost of tuition.

Students who will be employed by either of the two methods of payment and have their research supervised by the faculty-owner must sign an agreement acknowledging that they have been informed by their graduate program director or department head and associate dean about the source of their funding, the potential concerns associated with conflict of interest, and their channels for redress if needed.

Any work done on behalf of the faculty-owner's company in university facilities must be done in accordance with sponsored program guidelines and/or policy 5000, <u>University Facilities Usage</u> and Event Approval.

# **2.16.10** Compliance

Virginia Tech expects its employees to comply fully and promptly with the policy. The Office of the Vice President for Research is responsible for overseeing the implementation of the policy and associated procedures. The designated conflict of interest officer will work cooperatively with investigators and others to resolve any minor issues of noncompliance.

Breaches of the disclosure process, including (a) failures to comply with such process, whether by virtue of an employee's refusal to respond or by his or her responding with incomplete or knowingly inaccurate information, (b) failures to remedy conflicts, and (c) failures to comply with a prescribed management plan, will be forwarded to the appropriate university unit for investigation and appropriate action. Concerns about potential bias in the research will be reviewed in accordance with procedures established by the Office of the Vice President for Research and posted on the COI website.

Instances of deliberate breach of policy - including failure to file a required disclosure form, knowingly filing an incomplete, erroneous, or misleading disclosure form, or failure to comply with prescribed monitoring procedures - will be adjudicated in accordance with applicable disciplinary policies and procedures of the university as described in the Faculty and Staff Handbooks. University procedures for imposition of a severe sanction or dismissal for cause described in section 3.6 of the Faculty Handbook provide appeal mechanisms for faculty members charged with serious breaches of university policies. Staff employees may appeal through the staff grievance process.

#### 2.16.11 Record Retention

All original signed disclosure forms and management plans related to conflicts of interest and commitment will be retained in the employee's departmental personnel file in accordance with university record retention guidelines. Copies of disclosure forms and management plans related to research will be maintained in the Office of the Vice President for Research in accordance with state and federal requirements.

# 2.16.12 Definitions

- 1. "Business" means any corporation, partnership, sole proprietorship, firm, franchise, association, trust or foundation, or any other individual or entity carrying on a business or profession, whether or not for profit. The definition excludes the University or any other entity controlled by, controlling, or under common control with the University.
- 2. "Disclosure" is the full recording or specification of the employee's relationship with an external organization or involvement in external activities.
- 3. "Faculty member" is anyone who has faculty status at Virginia Tech, regardless of rank and including research faculty and administrative and professional faculty. The complete categories of faculty are defined in Section 2.1 of the Faculty Handbook. This policy also applies to staff employees and employed students where relevant.
- 4. "Family Member" means an employee's spouse or immediate family member who is a dependent of the employee (or of whom the employee is a dependent). A potential conflict of interest may arise when the party holding the "Financial Interest" as defined in (8) below is related to the employee in ways other than spouse and dependent children. Financial interests held by this party should be disclosed by the employee to the best of his or her knowledge.
- 5. "Institutional responsibilities" are defined broadly and include but are not limited to teaching, advising, research and scholarly activities, outreach, administrative and institutional committee service, and service to professional associations or on panels such as peer, institutional, or accreditation review boards.
- 6. 6. "Investigator" means the project director or principal investigator and any other person, regardless of title or position, who is responsible for the design, conduct, or reporting of research. Collaborators, consultants, or subcontractors are considered investigators for the purposes of this policy, as is anyone identified as key personnel in the contract proposal and contract. Investigator disclosures must include the financial interests of their spouse and dependents.
- 7. "Organizational Conflict of Interest" exists when multiple university relationships with sponsors create an actual or perceived conflict of interest. Virginia Tech is treated as a single contractual entity in the federal contracting context. Separate sponsored projects and their principal investigators are considered part of the overall Virginia Tech contracting entity. Hence, the involvement of one faculty member in a consulting or advisory services contract with the agency may preclude a subsequent related contract from that agency based on potential concerns that there may have been access to source selection criteria or confidential information not in the public domain in advance of an RFP or procurement, or that the organization might be evaluating its own work products and hence open to charges of biased judgment, or that the preparation of specifications or

- statements of work might have favored the institution. Organizational conflict of interest applies to more than one project and may affect all current or even future sponsored research with a particular federal sponsor.
- 8. "Significant financial interest" is defined differently by state law and federal regulation. The Virginia State and Local Government Conflict of Interests Act uses the term "Personal Interest" and the defined thresholds for disclosure apply to all employees not engaged in sponsored research. The definition for "significant financial interest" below applies to all investigators involved in sponsored research, regardless of the source of funds for that research.
  - a. "Personal Interest" as defined by the Virginia State and Local Governments Conflict of Interests Act: A financial benefit or liability accruing to an officer or employee or to a member of his immediate family. Such interest shall exist by reason of (i) ownership in a business if the ownership interest exceeds three percent of the total equity of the business; (ii) annual income that exceeds, or may reasonably be anticipated to exceed, \$10,000 from ownership in real or personal property or a business; (iii) salary, other compensation, fringe benefits, or benefits from the use of property, or any combination thereof, paid or provided by a business or governmental agency that exceeds, or may reasonably be anticipated to exceed, \$10,000 annually; (iv) ownership of real or personal property if the interest exceeds \$10,000 in value and excluding ownership in a business, income, or salary, other compensation, fringe benefits or benefits from the use of property; (v) personal liability incurred or assumed on behalf of a business if the liability exceeds three percent of the asset value of the business; or (vi) an option for ownership of a business or real or personal property if the ownership interest will consist of (i) or (iv) above.
  - b. "Significant Financial Interest": A financial interest consisting of one or more of the following interests of the investigator (and those of the investigator's spouse and dependent children) that reasonably appears to be related to the investigator's institutional responsibilities:
    - 1) With regard to any publicly traded entity, a significant financial interest exists if the value of any remuneration received from the entity in the twelve months preceding the disclosure and the value of the equity interest in the entity as of the date of disclosure, when aggregated, exceeds \$5,000. For purposes of this definition, remuneration includes salary and any payment for services not otherwise identified as salary (e.g., consulting fees, honoraria, paid authorship); equity interest includes any stock, stock option, or other ownership interest, as determined through reference to public prices or other reasonable measures of fair market value.
    - 2) With regard to any non-publicly traded entity, a significant financial interest exists if the value of any remuneration received from the entity in the twelve months preceding the disclosure, when aggregated, exceeds \$5,000, or when the investigator (or the investigator's spouse or dependent children) holds any equity interest (e.g., stock, stock option, or other ownership interest); or

3) Intellectual property rights and interest (e.g. patents, copyrights), upon receipt of income related to such rights and interest;

PHS investigators must also disclose the occurrence of any reimbursed or sponsored travel (i.e., that which is paid on behalf of the investigator and not reimbursed to the investigator so that the exact monetary value may not be readily available), related to their institutional responsibilities. However, this disclosure requirement does not apply to travel that is reimbursed or sponsored by a federal, state, or local government agency, an institution of higher education, an academic teaching hospital, a medical center, or a research institute affiliated with an institution of higher education.

The term significant financial interest does not include the following types of financial interests:

- salary, royalties, or remuneration paid by Virginia Tech to the investigator, including intellectual property rights assigned to the institution and agreements to share in royalties related to such rights;
- income from investment vehicles, such as mutual funds and retirement accounts, as long as the investigator does not directly control the investment decisions made in these vehicles;
- income from seminars, lectures, or teaching engagements sponsored by a federal, state, or local government agency, and institutions of higher education; or
- income from service on advisory committees or review panels for a federal, state, or local government agency, an institution of higher education, an academic teaching hospital, a medical center, or a research institute that is affiliated with an institution of higher education.

# 2.17 Faculty Involvement with the Virginia Tech Carilion School of Medicine

Virginia Tech and Carilion Clinic partnered to establish the Virginia Tech Carilion School of Medicine as a separate, private, non-profit entity. Medical school faculty are drawn, in part, from the two founding partners.

Virginia Tech Carilion School of Medicine initiates, defines, and contracts for professional services requested from a Virginia Tech faculty member. The contract may be for a buyout of the faculty member's time through a sponsored project, or the faculty member may be paid directly through overload (wage) compensation. The payment mechanism reflects the level of time commitment, the ability of the department to release the faculty member from current assignments, and the needs of both Virginia Tech Carilion School of Medicine and the faculty member's department at Virginia Tech.

As part of its commitment to partnership, Virginia Tech provides faculty mentorship of medical student research projects without additional compensation or buyout.

Payments made to Virginia Tech faculty members are made through an approved Virginia Tech payroll mechanism. Virginia Tech faculty members may not hold a private consulting contract with Virginia Tech Carilion School of Medicine since this would violate the Virginia Conflict of Interests Act.

# 2.17.1 Faculty Buyout Agreements with Virginia Tech Carilion School of Medicine

A buyout of the faculty member's time is appropriate when the professional services requested are of longer duration and/or exceed 20 percent of the faculty member's time (more than one day per week, for example). A buyout may also be used in the context of shorter duration commitments if determined to be in the best interest of Virginia Tech Carilion School of Medicine, the Virginia Tech department, and the faculty member.

Buyouts work as any other sponsored project buyout, releasing salary savings to the department and/or college to hire behind as needed, and requiring approval by the department head or chair and dean.

# 2.17.2 Faculty Overload Payment Agreements with Virginia Tech Carilion School of Medicine

Overload or wage payments that are made directly to the faculty member are appropriate for short duration and/or occasional professional services rendered to Virginia Tech Carilion School of Medicine (usually up to 20 percent time or one day per week). The rate of payment is established by Virginia Tech Carilion School of Medicine as a general rate of compensation or in individual negotiation with the faculty member.

Faculty members may earn up to 33 ½ percent of their current salary through all overload wage payments, including Virginia Tech Carilion School of Medicine, continuing education, or other authorized special wage payments during the period of their Virginia Tech contract. Faculty on 10-, 11-, or 12-month research extended appointments may also earn up to this limit as overload compensation during their contract period.

Summer pay from all Virginia Tech sources (e.g., summer school, funded research paid as wages, Virginia Tech Carilion School of Medicine, etc.) for nine-month faculty members may not exceed 33 ½ percent of the prior academic year salary.

Contracts for professional service to Virginia Tech Carilion School of Medicine paid as overload compensation may not exceed the current time limitations defined in the consulting policy, which is one day per week or five days in a five-week period. Time limitations also include the accumulation of other types of authorized special or external activity, including continuing education and consulting. University policies on conflict of commitment (section 2.15.1, "Faculty Commitment to the University") set the expectation that a faculty member owes his or her primary professional responsibility to the university.

Overload agreements and payments require approval of the department head or chair and dean.

In lieu of salary compensation, a faculty member may choose to receive an equivalent contribution to an operating allocation in support of professional activities.

# 2.18 Use of University Facilities

The facilities of the university are intended for the use of its faculty, staff, students, and invited guests participating in university-approved programs and activities, sponsored by or under the direction of the university or one of its related agencies or approved organizations, or by other organizations outside the university. Refer to university policy 5000, "University Facilities Usage and Event Approval," for further guidance regarding approved uses of university facilities.

University policy 6362, "Policy on Continuing and Professional Education," requires that university or auxiliary enterprise facilities and services be used for university-sponsored continuing and professional education programs held in Blacksburg. Alternate arrangements may be made in the case of lack of availability of appropriate space or mutual agreement between the sponsoring university entity and Continuing and Professional Education.

University facilities are to be used in a manner consistent with their intended purpose. Priority of use is given to those activities related to the academic, residential, cultural, and recreational programs of the university. The facilities must be used in a safe, professional manner so as not to endanger the university community or the general public. The university may restrict access to land and buildings to protect individuals, property, and equipment.

In general, the associate vice president for facilities services is responsible for implementing policies and procedures about university facilities, including academic buildings. The vice president for administration delegates general responsibility for use of academic facilities to the deans of the colleges, with the understanding that the registrar schedules classrooms in accordance with procedures outlined in university policy 5000, "University Facilities Usage and Event Approval." The appropriate dean's permission is obtained for nonscheduled classroom functions, posting of announcements and bulletins on non-departmental bulletin boards, and for similar non-academic usage. In some instances, the dean may refer recommendations to the associate vice president and chief facilities officer.

Rooms in academic buildings may be reserved for special functions such as lectures, social meetings, and recreational activities through approval of the college deans. Requests for use of rooms in The Inn at Virginia Tech and Skelton Conference Center are sent to The Inn's space reservationist. Requests for use of the residence halls are sent to the associate vice president for student affairs and director of housing and residence life. Requests for the use of Virginia Tech athletic department facilities are sent to: (1) the athletic department facilities and scheduling manager, and (2) the student centers and activities office. Refer to university policy 5000, "University Facilities Usage and Event Approval," for specific information regarding use of other university buildings and grounds.

Faculty and staff are not allowed to use university resources for private gain. However, under the following conditions, the compensated use of specialized facilities or equipment is allowed in support of approved consulting activities:

- The facility or equipment must have a charge rate, established by the Controller's Office, which reflects all direct and indirect costs associated with the use of the facility or equipment and applies to use by parties outside the university. The charge rate is applied to the actual use.
- A Request to Engage in External Activity Form 13010A must be filed, specifying the facility or equipment to be used and estimating, in time or charges, the extent of the proposed use.
- The head or chair of the department responsible for the facility or equipment verifies, on the Request to Engage in External Activity Form 13010A, that the proposed use does not interfere with, or have priority over, anticipated university use of the facility or equipment.
- In approving the Request to Engage in External Activity Form 13010A, the faculty member's department head/chair, dean, and the provost determine that the consultation is of substantial professional merit and presents no conflict of interest in the use of the facilities or equipment. Particular care is given to the relationship of the consultation with current or potential grants or contracts and to the possibility of unfair competition with local firms and businesses.
- If a faculty member uses the equipment of a specialized service center, the faculty member is charged the "commercial" or "consulting" rate, as determined by the Controller's Office. The faculty member is billed based on actual use. The deposit is credited to the appropriate service center account.
- If the faculty member uses facilities such as those of the Virginia Tech athletic department, Squires Student Center, or the G. Burke Johnston Student Center, the faculty member is charged at a rate established by the Controller's Office for such use. The deposit is made to the appropriate department's account.
- For facilities other than specialized service centers, computing services, or other facilities for which a charge rate has been determined, the use of the facilities must be authorized and reimbursed at a rate determined by the joint collaboration of the faculty member's department head/chair and the Controller's Office.
- The use of library facilities in connection with consulting is exempt from the above regulations, since those facilities are available to the public.

# 2.18.1 Required Use of Continuing and Professional Education Program Services and Facilities

University policy 6362, "Policy on Continuing and Professional Education," requires that university or auxiliary enterprise facilities and services be used for university-sponsored continuing and professional education programs held in Blacksburg. Alternate arrangements may be made in the case of lack of availability of appropriate space or mutual agreement between the sponsoring university entity and Continuing and Professional Education. For further information, contact the Division of Continuing and Professional Education.

# 2.19 Use of University Letterhead

As the primary identifier of the university, letterhead should be used for appropriate university business. As such, university letterhead should not be used for personal business or where personal gain results. Endorsements of political personages, businesses, or products should be avoided. Discretion is advised if correspondence on university letterhead could be construed as a university endorsement.

#### 2.20 Political Activities

Candidacy for political office, service on county and state commissions, and active participation in political campaigns are recognized as individual freedoms of each faculty member. The only restriction placed upon such activities is that they not interfere with the faculty member's academic responsibilities. Faculty members must take care to assure that their positions in the university are kept separate from their political activities; it must be clear that they act as citizens in such activities, not as representatives of the university.

The university encourages interest in civic affairs. However, neither political nor community activities are considered in the annual merit evaluation of a faculty member. If income is obtained for such activities, approval must be obtained under consulting policies.

# 2.21 Indemnity

All university employees, while acting within the course and scope of their employment, are covered by the commonwealth's insurance plan and will be defended by the Office of the Attorney General in actions brought against them. Questions concerning any specific situation should be addressed to the Office of the University Legal Counsel.

# 2.22 Geographical Transfer Policy

Reassignment of a faculty member at the initiative of the university to a primary workstation located more than 50 miles from the current workstation is considered a geographical transfer. A department head or chair may request the geographical transfer of a faculty member to implement a programmatic mission of the university. The affected faculty member shall be involved in planning for the transfer prior to the submission of a formal request for transfer. The request for geographical transfer shall be transmitted in writing to a second-level administrator for approval with accompanying documentation justifying the need for the transfer of the selected individual. The justification shall describe the university program and the position to which the faculty member is being transferred. This description shall list the unique skills and knowledge required to fulfill the program's mission. The alternatives for meeting the requirements shall be outlined, along with the reasons for selecting the alternative of geographical transfer of the particular faculty member. A faculty member must be notified in writing at least six months in advance of the geographical transfer. The transferred faculty member shall be reimbursed for all allowable expenses as defined by the university and state policy. A cost of living adjustment will be added to the faculty member's base salary during the period they are employed in a high-cost area as defined for employees in the Virginia Compensation Plan issued by the state human resources director.

#### **CHAPTER THREE**

# 3.0 Employment Policies and Procedures for Tenured and Tenure-Track Faculty

## 3.1 Standard Faculty Ranks

#### 3.1.1 Assistant Professor

Appointment to the rank of assistant professor carries with it professional responsibilities in learning, discovery, and engagement. An assistant professor may be assigned responsibility for teaching graduate courses and for supervising master's theses and doctoral dissertations, as well as serving on graduate student committees. The terminal degree appropriate to the field is expected for appointment to this rank. (Further information regarding appropriate credentials for teaching faculty is found in section 2.5.9, "Faculty Credentialing Guidelines," and on the provost's website.)

#### 3.1.2 Associate Professor

In addition to the requirements for assistant professor, a person appointed as associate professor must have demonstrated substantial professional achievements by evidence of an appropriate combination of outstanding teaching, creative scholarship, and recognized performance in extension, outreach, library, or related academic and professional service.

#### 3.1.3 Professor

In addition to the requirements for associate professor, appointment to the rank of professor is contingent upon national recognition as an outstanding scholar and educator.

# **3.2** Honored Faculty Appointments

# 3.2.1 Endowed Chairs, Professorships, and Fellowships

Each college has formal procedures for the nomination and appointment to endowed chairs, professorships, and fellowships, which include review by a college honorifics committee or promotion and tenure committee. The university Faculty Honorifics Committee reviews extracollegiate nominations to endowed chairs, professorships, and fellowships. After review by the appropriate college committee, recommendations are then made by the college dean for approval by the provost and the board of visitors. Such an appointment may continue through the active career of the professor at the university, unless it is relinquished in favor of some other honored or administrative appointment or unless the appointment has specific term limitations that may be renewable.

Endowed chairs, professorships, and fellowships are established by a donor who provides an endowment to support the salary and/or operating funds of the professor. Funding levels determine whether the endowed position is a chair, professorship, or fellowship. For further

information regarding the establishment of an endowment, see the Virginia Tech Foundation website.

# 3.2.1.1 Eminent Scholar Program

The state General Assembly appropriates funds each year for the purpose of attracting eminent scholars to Virginia's colleges and universities. Institutions of higher education can request funds from this appropriation item if gifts from private donors have been specifically designated for purposes of the Eminent Scholars Program. The earnings from these endowments are intended to match state funds appropriated for this purpose. However, state appropriations may not be sufficient in some years to fully match the available private support.

## 3.2.2 Alumni Distinguished Professor

General Conditions and Definitions: The alumni distinguished professorship (ADP) recognizes extraordinary academic citizenship and distinguished service within the Virginia Tech community. In recognition of the importance of alumni to the university, the alumni distinguished professorship is a pre-eminent faculty appointment, reserved by the board of visitors for recognition of faculty who, over time, have made outstanding contributions to the instructional program of the university and, in so doing, have touched the lives of generations of Virginia Tech alumni.

The board of visitors confers upon an individual an appointment as alumni distinguished professor for a period of ten years; the appointment may be renewed. Currently, ten alumni distinguished professors may be appointed. There is no quota by college or department.

Eligibility and Criteria for Selection: Since the hallmark of the alumni distinguished professorship is distinguished contribution—over time—to Virginia Tech, newly arrived faculty are not eligible for nomination. And while there is no specified minimum number of years of service required for eligibility, the selection committee places strong emphasis on the magnitude and character of the candidate's impact on academic programs at Virginia Tech. Nominees should also have established outstanding personal records of accomplishment in creative scholarship.

Nomination and Selection Procedures: Each academic year the provost determines if there is to be one or more vacancies in the alumni distinguished professor group and, if appropriate, issues a call to the academic deans for nominations. The deans, in turn, invite nominations from academic departments. Screening procedures at departmental and collegiate levels involve appropriate personnel or executive committees in place. Nominations are accompanied by a full dossier of relevant materials including current *curriculum vitae*, letters of nomination from both the departmental and collegiate screening committees, letters of support, and other evidence attesting to the quality of the contributions of the nominee(s).

In developing recommendations the provost relies primarily on the advice and counsel of the university Commission on Faculty Affairs. The commission, in turn, invites the evaluation of nominees by the incumbent distinguished professors. The provost's recommendation(s) are sent

through the president to the executive committee of the board of directors of the Virginia Tech Alumni Association for its review and recommendation. The president makes the final recommendation to the board of visitors for its approval.

Perquisites and Responsibilities: Each alumni distinguished professor is provided a base salary supplement from the endowment established by the Alumni Association and matched by available funds, if available, from the eminent scholar program. The ADPs enjoy a salary supplement and operating allocation from the university comparable to that provided for other endowed professorships.

Each alumni distinguished professor is expected to continue in service to the department, the college, and the university at the same high level evident at the time of appointment. But in particular, within whatever latitude the department head or chair and college dean can accommodate the alumni distinguished professor is encouraged to teach, when invited, in other departments of the university or in collegiate or university courses (e.g., honors colloquia). He or she may also elect, in a given term, to divert energies from the usual classroom responsibilities to other valued activities, such as substantive curriculum revision or textbook preparation.

Alumni distinguished professors are also called upon from time to time, individually and also as a group, to render special service or to offer particular advice to the university at large.

Given the high level of performance expected of this select group of faculty, university and college administrators are cognizant of the particular needs of each individual alumni distinguished professor for appropriate support personnel and sufficient space, within acknowledged fiscal and physical constraints.

## 3.2.3 University Distinguished Professor

General Conditions and Definitions: The university distinguished professorship (UDP) is a preeminent faculty rank bestowed by the university board of visitors upon members of the university faculty whose scholarly attainments have attracted national and/or international recognition. The professorship is bestowed upon no more than 1 percent of the total full-time faculty at any time.

Nomination and Selection Procedures: Each academic year the provost determines if there will be one or more vacancies in the rank of university distinguished professors and, if appropriate, issues a call to the academic deans for nominations. The deans, in turn, invite nominations from academic departments.

Screening procedures at departmental and collegiate levels involve personnel or executive committees in place. Nominations are accompanied by a full dossier of relevant materials including current *curriculum vitae*, letters of nomination from both the department and collegiate screening committees, and letters of support and other evidence attesting to the scholarly reputation of the nominee(s).

In developing recommendations, the provost relies primarily on the advice and counsel of the university Commission on Faculty Affairs. The commission, in turn, invites the evaluation of nominees by the incumbent distinguished professors. The president makes the final judgment at the university level and, if that judgment so determines, takes the recommendation(s) to the board of visitors for approval.

Perquisites and Responsibilities: The rank of university distinguished professor is conferred by the university and is considered a university appointment (as distinct from a collegiate or departmental appointment). While the successful nominee is nominated by departmental and collegiate colleagues, and while he or she continues to serve in the discipline and department of origin, nonetheless the perquisites and responsibilities of each university distinguished professor are fixed by the university.

The president of the university annually adjusts the salary of university distinguished professors after consultation with the provost and dean of the relevant college.

The sole responsibility of the university distinguished professors is to continue their professional development at the same high level evident at the time of appointment. They are free to define the exact nature of their work after consultation with the dean of the college and the head or chair of the department. They are expected to engage fully with their colleagues into the governance of their departments.

At the same time they may freely decide to teach, when invited, in other departments of the university or in collegiate or university courses (e.g., honors colloquia). They may also elect, in a given term, to devote all of their energies to research, scholarship, or extension activities. In shaping their plans of work, the university distinguished professors take full cognizance of departmental and collegiate needs and expectations. But their principal responsibility is to serve the university well by giving their talents to the development and sharing of their competencies where, in their judgments, they are most effectively employed.

It is the university's responsibility to provide such support as seems necessary to sustain the high level of performance expected of university distinguished professors.

Tenure: Incumbents carry the rank of university distinguished professor until resignation or retirement from the university, subject to the normal standard of continuous high performance. The rank is conferred only by the university board of visitors and is altered by that body alone, on the recommendation of the president of the university.

## 3.2.4 Emeritus or Emerita Designation

The title of emeritus or emerita is conferred on retired full professors, associate professors, administrative officers, extra-collegiate faculty with continued appointment, and senior extension agents who have given exemplary service to the university, and who are specifically recommended to the board of visitors by the president and approved. Their names are listed on the university's website. University policy 4405, "Emeritus Faculty," provides further guidance

for department heads or chairs and retiring faculty members concerning emeriti status and continued involvement with the life of the university.

## 3.3 Procedures for Faculty Appointments with Tenure

An offer of faculty appointment with tenure may be made with the review and approval of the department head or chair, the department promotion and tenure committee, the dean, the provost, and the president. Ultimately, final approval rests with the board of visitors.

The dean forwards the following information to the provost and president for their consideration and decision: the candidate's application package, including cover letter, *curriculum vitae*, and letters of reference; documentation of the department promotion and tenure committee's approval of rank and tenure, and concurrence of the department head or chair and dean with as much supporting evidence as deemed appropriate; and a brief overview of the search itself—how many candidates applied, were interviewed, and what is the compelling case for the candidate.

In general, faculty recruited from a comparable university should be recommended for a position at Virginia Tech at a similar level with tenure. If the recommended appointment involves a promotion or the initial awarding of tenure, the case must be strongly justified. If an individual is coming from a university with a less extensive research mission, or a university of lesser stature, the case must also be strongly justified.

## 3.3.1 Part-Time Tenure-Track and Tenured Appointments

While tenure-track and tenured appointments are normally full time, Virginia Tech recognizes the importance of allowing flexibility in the percent employment so that faculty members can better manage the balance between their professional work and family or personal obligations over a defined period of time, or perhaps permanently. This policy is intended to encourage departments to accommodate reasonable requests for part-time appointments; however, part-time appointments are not an entitlement, and requests may be turned down when the faculty member and the department cannot agree upon a workable plan.

Departments continue to advertise for full-time tenure-track or tenured positions and must have funding for a full-time hire. Advertisements include information about university policies for flexible appointments. If desired, the faculty member requests and negotiates a part-time appointment at or after the point of hire if acceptable to the department.

Tenure-track faculty members may request a part-time appointment only for reasons of balancing work and family for the arrival or care of a child, the care of a family member, or for personal circumstances related to the health of the faculty member. In addition, they may request a *term* part-time appointment only, allowing the issue to be revisited on a defined cycle. While such term appointments can be renewed throughout the probationary period, a *permanent* part-time appointment may not be granted until tenure is awarded.

Tenured faculty members may request either term or permanent part-time appointments for reasons stated above, or to balance work at Virginia Tech with professional practice or

significant community or public service, for example, a professor who wishes to serve as a consultant in addition to an appointment at Virginia Tech; a professor who wishes to engage in entrepreneurial activity outside of university responsibilities; or a professor who runs for office for a limited term and wishes to reduce the workload at Virginia Tech for that period. Other reasonable justifications may be considered if approved by the department head or chair, dean, and provost.

# 3.3.1.1 Part-Time *Term* Tenure-Track and Tenured Appointments

Part-time tenure-track and tenured appointments are either *term* or *permanent*. *Term* part-time appointments are in increments from one semester up to two years. During the duration of a part-time term appointment, terms of the appointment are only changed via the agreement of all parties. A term agreement must specify the date on which the faculty member is expected to return to full-time status. Renewal of a term appointment should be negotiated no less than three months before the end of the current term so that the department can plan accordingly. For term part-time appointments, departments are able to use the salary savings to replace the work of the faculty member on part-time appointment.

Only the faculty member may initiate a request for conversion from full-time to part-time appointment. The reasons for the request for a change in the percentage of appointment should be clearly stated. The department chair should make a careful assessment of the needs of the department, and works with the faculty member requesting the part-time appointment to facilitate the request whenever possible. The period for which this part-time appointment is granted shall be clearly stated (renewable terms from one semester up to two years, or permanent).

The written agreement should include a careful and thorough statement of work expectations for the part-time appointment. Generally, faculty members continue to contribute to all areas of responsibility, but with reduced expectations for accomplishment proportional to the fractional appointment. Service responsibilities for faculty members on part-time appointments are generally proportional to their appointments. Faculty members on part-time appointments are not excused from regular departmental, college, or university service because of the part-time appointments.

The written agreement for either an initial appointment or a conversion of a full-time appointment to part-time status and any subsequent renewal requires the approval of the faculty member, department head or chair, dean, and provost.

An initial *term* part-time appointment, either tenured or tenure-track, may be approved to accommodate a dual career hire if funding is not immediately available to support a full-time position, or if the faculty member seeks a part-time appointment consistent with the intent of this policy. The expectation is that the subsequent reappointment, if recommended, is for a full-time position, unless the faculty member requests a renewal of the term part-time appointment in accordance with these guidelines. A part-time appointment created for a dual career hire is approved through the usual approval processes for dual career hires. (See section 2.6.7, "Dual Career Assistance Program.")

## 3.3.1.2 *Permanent* Part-Time Tenure-Track or Tenured Appointments

For *permanent* part-time tenure-track or tenured appointments with no end date, a return to a full-time appointment is not guaranteed. If tenured, the faculty member remains entitled to the tenured appointment on the part-time basis only. However, an increase in the percent appointment up to full-time may be renegotiated between the faculty member and department head or chair if mutually agreeable and funds are available. The department and the college determine the best way to cover the costs of the academic work in the case of conversion to a permanent part-time appointment.

Faculty members on part-time appointments, whether term or permanent, retain all rights and responsibilities attendant to their appointment as a tenure-track or tenured faculty member.

Part-time appointments are made for any fraction 50 percent or greater of a full appointment; faculty members receive proportional salary. Faculty members considering such appointments are strongly encouraged to meet with representatives in the <u>benefits office</u> to gain a clear understanding of the consequences of the change to their benefits. Office and laboratory space may be adjusted for longer term or permanent part-time appointments.

### 3.4 Promotion and Tenure

The university has a tradition of upholding academic freedom. It endorses the "1940 Statement of Principles on Academic Freedom and Tenure" of the American Association of University Professors and the Association of American Colleges (*AAUP Bulletin*, September 1970, pp. 323-326).

### 3.4.1 Tenure Eligibility

Tenure is an institution developed for the protection of the academic freedom of the teaching faculty in institutions of higher education. Eligibility for tenure consideration is limited to faculty members holding regular faculty appointments of 50 percent to 100 percent in academic departments. Tenure is not granted to faculty members with temporary appointments or to administrative and professional faculty. Individuals holding tenure in academic departments who are appointed to administrative positions, however, continue to hold tenure in those departments.

Full-time administrators who also hold appointments in academic departments and engage in teaching and research may be recommended for tenure in such departments.

### 3.4.2 Probationary Period

The term "probationary period" is applied to the succession of term appointments that an individual undertakes on a full- or part-time regular faculty appointment, and during which continued evaluation for reappointment and for an eventual tenured appointment takes place. The beginning of the probationary period for faculty members on term appointments is taken as July 1 or August 10 of the calendar year in which their initial full-time appointment begins, depending on whether they are on a calendar year or academic year appointment, regardless of

the month in which their services are initiated. (The probationary period for new faculty appointed for spring semester begins the following fall even though the spring contract period officially begins December 25.)

Under normal circumstances, departmental promotion and tenure committees review pre-tenure faculty members twice during the probationary period, usually their second and fourth, or third and fifth, years of service. The timing of the reviews depends upon the nature of the faculty member's discipline and must be clearly indicated in written departmental policies. The terms of offer identifies the initial appointment period. Pre-tenure reviews may be delayed if there is an approved extension as described below. Changes or variations in the standard review cycle must be documented in writing.

The initial review for a part-time faculty member should be no later than the third year of service (regardless of percent of employment) to give early feedback on their progress. At least two reviews should be conducted for part-time faculty members during their probationary period; more are recommended. The anticipated schedule for such reviews for reappointment and for the mandatory review for tenure should be documented in writing as part of the agreement for the part-time appointment. Changes should be agreed upon by the faculty member and department.

These reviews are substantive and thorough. At a minimum, departmental promotion and tenure committees must review the faculty member's relevant annual activity reports, peer evaluations of teaching, and authored materials. It is strongly suggested that promotion and tenure committees and pre-tenure faculty use the promotion and tenure dossier format (available on the provost's website) in organizing and presenting information for review.

The pre-tenure reviews should analyze the faculty member's progress toward promotion and tenure and offer guidance regarding future activities and plans. All reviews must be in writing, with the faculty member acknowledging receipt by signing and returning a copy for his or her departmental file. In addition, the promotion and tenure committee and the department head or chair meet with the faculty member to discuss the review and recommendations. Individual faculty members are also encouraged to seek guidance and mentoring from senior colleagues and the department head or chair. Pre-tenure faculty members bear responsibility for understanding departmental expectations for promotion and tenure and for meeting those expectations.

The initial appointment for assistant professors, and for associate professors and professors employed without tenure, is ordinarily for a period of not less than two years. Multiple-year reappointment may be subsequently recommended.

The maximum total period for full-time probationary appointments is six years, unless an approved extension is granted. Decision about tenure, if not made earlier, is made in the sixth year of the probationary appointment. If the tenure decision made in the sixth year is negative, a one-year terminal appointment is offered.

Pre-tenure faculty members may request a term part-time appointment as described in section 3.3.1, "Part-Time Tenure-Track and Tenured Appointments," for reasons of balancing work and family or personal health issues. In such cases, the probationary period is extended

proportionately. For example, two years of service at 50 percent count as one year of full-time service. The term appointment may be renewed. (A permanent part-time appointment may be requested and granted following award of tenure.)

In determining the mandatory tenure review year for those with partial appointments, general equivalency to full-time appointments is expected, so that approximately five years of full-time equivalent service is expected prior to the mandatory tenure review year if no tenure clock extensions are granted; six years if one year of extension is granted, and seven years if two extensions are granted. (In summing partial years of service, a total resulting in a fraction equal to or less than .5 is rounded down, and a fraction greater than .5 is rounded up.) However, review for tenure must occur no later than the tenth year of service, resulting in somewhat less full-time equivalent service (4.5 years) for a faculty member with 50 percent appointment throughout all nine probationary years prior to review. If denied tenure following a mandatory review, a one-year terminal appointment is offered.

Faculty members on part-time appointments may request a tenure clock extension in accordance with section 3.4.2.1, "Extending the Tenure Clock." (Extensions are granted in one-year increments, not prorated by the part-time appointment percentage.) However, the extension is not approved if it results in a mandatory review date beyond the tenth year.

Up to three years of appropriate service at other accredited American four-year colleges and universities may be credited toward the six-year probationary period, as specified in section 3.4.3, "Guidelines for the Calculation of Prior Service."

A faculty member on probationary appointment who wishes to request a leave of absence consults with his or her department head or chair about the effect of the leave on the probationary period, taking into account the professional development that the leave promises. The request for leave should address this matter and the provost's approval of the leave request specifies whether the leave is to be included in the probationary period.

### 3.4.2.1 Extending the Tenure Clock

A one-year probationary period extension is *automatically* granted to either parent (or both, if both parents are tenure-track faculty members) in recognition of the demands of caring for a newborn child or a child under five newly placed for adoption or foster care. The request should be made within a year of the child's arrival in the family.

An extension of the probationary period may also be approved on a discretionary basis for other extenuating non-professional circumstances that have a significant impact on the faculty member's productivity, such as a serious personal illness or major illness of a member of the immediate family. In rare cases, extraordinary professional circumstances not of the faculty member's own making may be acceptable justification for a probationary period extension, for example exceptional delays in providing critical equipment, laboratory renovations, or other elements of the committed start-up package essential to establishing a viable research program.

Faculty members who benefit from this policy are expected to fulfill their normal responsibilities during the probationary period extension unless they are also granted a period of modified duties or unless other arrangements are made. (See section 3.10, "Modified Duties.")

Probationary period extensions are granted in one-year increments. A cumulative total of two years is normally the maximum probationary period extension for any combination of reasons. Requests should be made within a year of the qualifying event or extenuating circumstance. The provost may approve exceptions to these limitations.

Requests for a probationary period extension are submitted in writing to the department head or chair. (A form is available on the <u>provost's website</u>.) Approval is automatic for new parents. Documentation of medical reasons (other than childbirth or adoption) is required prior to approval; documentation of other extenuating circumstances may also be required. Approvals by the department head or chair, dean, and provost are required for probationary period extensions. The faculty member may appeal denial of the request to the next higher level.

It is very important that all individuals and committees participating in tenure reviews understand that any individual who receives a probationary period extension must be held to the same standard—not a higher or more stringent one—to which other candidates without such an extension are held. This is also true in the case where the candidate's dossier is considered on the original schedule for review. However, in this instance where an approved extension was granted but not utilized, the tenure review is not considered mandatory and can be conducted again in the subsequent year without penalty. A probationary extension also normally extends the time frame for each subsequent review and reappointment during the probationary period. For example, an extension granted prior to the fourth year review and reappointment typically delays that review by one year.

### 3.4.3 Guidelines for the Calculation of Prior Service

At the time of a faculty member's initial appointment, the department head/chair notifies the new faculty member of his or her standing regarding the tenure system. Excepting temporary appointments with limited terms, the faculty appointees are given clear notice of when their appointments will be considered for renewal and when consideration for tenure will be given.

In this latter calculation, appropriate full-time service in another accredited four-year American college or university is credited toward probationary service at Virginia Tech only if the appointed faculty member requests such credit.

In such a request, all prior service is presented if undertaken after the faculty member completes the terminal degree appropriate to the field. A maximum of three years may be credited toward probationary service at Virginia Tech. The request must be made in writing within one year of the initial appointment. The specification of credit for prior service toward the probationary period is subject to the approval of the provost on the recommendation of the department head or chair and dean.

## 3.4.4 Evaluation Procedures for Promotion and Tenure

Promotion to a higher rank and appointment with tenure may be granted to faculty members on a regular faculty appointment who demonstrate outstanding accomplishments in an appropriate combination of learning, discovery, engagement, and other professional activities. Every faculty member should maintain a current *curriculum vitae*, with copies filed in the department and college (or equivalent academic units, as appropriate). The *curriculum vitae* together with annual reports, student evaluations, reprints of publications, reference letters, and other similar documents comprise a dossier that furnishes the principal basis for promotion and tenure decisions.

Faculty members being considered for either promotion or the awarding of tenure have their dossiers reviewed at as many as three levels: by a departmental committee and the head or chair; by a college committee and the dean; and by a university committee and the provost.

Although some participants in the review process may serve at more than one level—for example a departmental committee member may also serve on the college committee—participants may only vote once on a case. A faculty member may not serve on any committee that is evaluating a spouse, family member, or other individual with whom the faculty member has a close personal relationship (see section 2.16.2, "Potential Conflicts Involving Spouses and Immediate Family Members").

Each candidate for tenure and/or promotion to associate professor is evaluated in the light of the triple mission of the university: learning, discovery, and engagement. Although not all candidates are expected to have equal levels of commitment or equal responsibilities in each of these missions, a high level of general competence is expected, in recognition of the need for flexibility in the future establishment of priorities in academic programs. Beyond that basic foundation of competence, decisions related to tenure or promotion to associate professor require evidence of excellence in at least one area.

The award of tenure is based on the achievement of distinction in an area of learning and the prediction of eminence throughout the individual's professional career. The documentation and evaluation should recognize some significant impact of the candidate's contributions beyond the borders of the university. If the primary strength is in instruction, there should be recognition that the candidate's pedagogical contributions have influence beyond the immediate classroom; if in research, that there is significant impression on colleagues nationally; if in outreach that the influence of the contributions reaches beyond the immediate clientele.

The criteria by which faculty with part-time appointments are evaluated for tenure does not differ from the criteria by which full-time faculty are evaluated. Promotion and tenure committees consider years of full-time equivalent service when reaching decisions, excluding any approved probationary period extensions granted under the extending the tenure clock policy.

Each candidate for the rank of professor must demonstrate a high level of competence in an appropriate combination of instruction, outreach, and professional activities relevant to their assignment. Because of the university's mission and commitment as a major research institution,

successful candidates for the rank of professor must demonstrate excellence in research, scholarship, or creative achievement, as appropriate for the candidate's discipline and assignment. Promotion to the rank of professor is contingent upon national or international recognition as an outstanding scholar and educator.

The university recognizes and encourages appropriate international involvement of its faculty as a mission of the university that cuts across the three traditional missions of learning, discovery, and engagement. Occasionally faculty members are placed on international assignments at full salary from the university, with responsibilities that require their residence far from the campus for a considerable period. Under such circumstances, faculty members should be given the usual consideration for tenure, promotion, and salary advancement, with the recognition that international assignments can be an important stimulus to professional growth.

In cases of tenure recommendation—besides evaluation of the candidate's professional abilities—consideration should be given, at all stages of evaluation and review, to future departmental program directions and concern for maintaining currency and flexibility by preserving opportunities to appoint new faculty members in the various sub-fields of the department.

Levels of expectation vary, of course, with the level of the decision. Where probationary reappointments recognize, in part, perceived potential instead of accomplishment, recommendations for tenure should suggest that the potential is being achieved and should imply few, if any, lingering doubts about the value of the candidate to the department's program for a "lifetime." And promotion to professor, which leaves limited opportunity for further university recognition of professional development, should be reserved for those whose achievements are broad and noteworthy.

Besides consideration of specific professional criteria, evaluation for promotion or tenure should consider the candidate's integrity, professional conduct, and ethics. To the extent that such considerations are significant factors in reaching a negative recommendation, they should be documented as part of the formal review process.

### 3.4.4.1 Departmental Evaluation

Each department has one or more committees with appropriate faculty representation to evaluate candidates for promotion and/or tenure and make recommendation to the department head or chair. The department head/chair may chair the committee or may remain separate from the committee's deliberations and subsequently receive its recommendations. (See guidelines at the end of this section.)

The committee reviews the cases of all members who the head or chair of the committee believes deserve consideration for promotion or tenure, including those faculty members in the sixth year of probationary service. The department head or chair furnishes the committee with a dossier for each candidate.

Guidelines for compiling the dossier are recommended by the University Promotion and Tenure Committee and are available on the <u>provost's website</u>.

The committee makes a recommendation on each candidate to the department head or chair, including a written evaluation that assesses the quality of the candidate's performance in each relevant area. The division of the vote is conveyed to the college-level committee and administrator, but must otherwise remain confidential outside the committee. In the absence of a unanimous recommendation, a minority report may be included. Whenever the department head or chair does not concur with the committee's recommendation, the committee is so notified.

In all cases of mandatory (sixth year) tenure decision, the head or chair passes on to the dean the dossier of every candidate, which includes the committee's evaluation and recommendation and the head's or chair's own recommendation, whether concurring or not. If not concurring, the head or chair includes a letter specifying the reasons. If concurring, the head or chair may submit a letter that combines the committee's and the head's or chair's evaluation and recommendation. Should the committee and the head or chair agree on a negative recommendation, the dean may declare that to be the final decision or may choose to have the recommendation reviewed by the college committee. In all other cases (promotion or tenure before the sixth year of probationary service), the head or chair follows the same procedures except that, when the committee's recommendation is negative and the head or chair concurs, the head or chair declares a final decision and no further review is carried out. The head or chair informs the faculty member of a negative decision if no further review is scheduled. In that case the faculty member is notified in writing of appeal options.

In sending dossiers to the college level, the head or chair may hold back supplementary materials not deemed central to the review but indicates their nature and their availability. Accompanying the set of dossiers is a statement from the head or chair describing the formation and procedures of the departmental committee and summarizing the number of candidates considered in each category (mandatory tenure, pre-sixth-year tenure, promotion at each rank).

On recommendation of the Commission on Faculty Affairs, University Council approved guidelines for the careful consideration by colleges and departments in the composition and method of selection of departmental promotion and tenure committees. They are presented as guidelines in the recognition that some flexibility is necessary to accommodate the diversity in size, structure, and composition of departments and in the desire to preserve some degree of department and college autonomy in such matters.

Composition and Size: Individual departments must develop and publish written policies to guide their promotion and tenure review processes, including the rules governing eligibility and selection of committee members. Individual departments determine who is eligible to serve on committees from among tenured faculty members. A balance between adequate representation and effectiveness of operation as a committee suggests that a size between four to seven members is most appropriate.

Method of selection: Some significant elements of faculty choice must be a part of the selection procedure. Some possibilities are: a combination of elected and appointed representatives; an

elected slate significantly larger than the committee size, allowing the department head or chair to appoint the committee from the slate; or a committee elected by the faculty.

Role of the department head or chair: Given their responsibility to make a separate and independent recommendation on each case, department heads or chairs may not vote as members of committees. Department heads or chairs may convene committees and may discuss each candidate with committees as appropriate. However, it is recommended that committees discuss the merits of the candidates and frame their recommendations without heads or chairs in attendance.

### 3.4.4.2 College Evaluation

Each college shall have a committee with appropriate faculty representation to review the recommendations on promotion and tenure sent by the department head or chair. The committee reviews the cases of any candidates recommended by the departmental committee and/or the head or chair and, if requested by the dean, reviews cases of mandatory tenure receiving negative recommendations by both the departmental committee and the head or chair.

The purposes of the review are to verify that the recommendations are consistent with the evidence, reflecting college-wide standards, and that they consider the goals, objectives, and programmatic priorities of the college as components of the university mission.

The committee makes a recommendation on each candidate to the dean. The division of the vote at both the departmental and college level is conveyed to the university-level committee and provost, but must otherwise remain confidential. If the recommendation is at variance with that received from the department head or chair, reasons for that variance should be specified in the recommendation.

Whenever the dean does not concur with the committee's recommendation, the committee is so notified. The dean sends to the provost the full dossier of every candidate for whom the dean makes positive recommendation and also the dossiers of those cases where the dean does not concur with the college committee's positive recommendation. The dean includes a letter specifying the reasons for any reversal of the committee's recommendation and, in cases of concurrence, may include a letter to bring out additional points not raised in earlier evaluations.

In the case of any candidate for promotion or tenure whose dossier is not being sent to the provost, the dean informs the department head or chair of the rejection and the department head or chair so notifies the departmental committee and the faculty member. In that case, the faculty member is notified in writing of appeal options.

The dossiers that the dean sends to the provost are accompanied by a statement describing the formation and procedures of the college committee and by a summary of the number of candidates considered by the committee in each category.

On recommendation of the Commission on Faculty Affairs, University Council approved the following further guidelines on formation and procedures of the college-level evaluation:

Committee Composition: Rules governing eligibility and selection of college committee members and the committee chair, and operating guidelines for the committee's deliberations must be documented in written college policies, formally approved by the faculty.

- 1. Individual colleges determine who is eligible to serve on committees from among tenured faculty members.
- 2. The college committee may include department heads, chairs, or department-level promotion and tenure committee members. However, none of these members may vote on cases from their departments since each has already had an opportunity to vote or make a recommendation on those candidates.
- 3. As far as possible, each department within the college should be represented on the committee.
- 4. A significant element of faculty choice must be part of the committee selection procedure. Some possibilities are: election by the college faculty, appointment by an elected college executive committee, a combination of elected and appointed (by the dean or college executive committee) representatives, or an elected slate significantly larger than the required committee size, thus allowing the dean or college executive committee to appoint the committee from the elected slate.
- 5. The dean may appoint up to three tenured faculty members to serve on the college committee in order to assure appropriate representation of disciplines or very large departments, participation by members of underrepresented groups, or other critical considerations to help assure fairness of the process in both fact and perception. Appointments by the dean may not constitute more than a third of the committee's total membership.
- 6. If department heads or chairs serve on college committees, their total number must be less than that of other faculty members.
- 7. Committee appointments should be staggered to assure continuity from one year's deliberation to the next. If possible, members should not serve more than two successive terms.
- 8. Selection of the committee chair is determined in accordance with college policies, approved by the faculty.
- 9. The dean may be present at college committee deliberations and serve in an advisory capacity to the committee to assure compliance with college and university procedures and fairness and equity of treatment of candidates. The dean does not vote on committee recommendations, but provides a separate recommendation to the provost.

10. Faculty members appointed to serve on the university-level promotion and tenure committee are encouraged to observe college-level deliberations to better prepare for their roles, but should not participate or attempt to influence college-level recommendations.

Committee Procedures and Recommendations: The college committee may ask the department head or chair, the candidate, and/or a representative(s) of the department committee to appear before the college committee to present additional information or clarification of recommendations.

The committee makes a recommendation on each candidate to the dean and prepares a letter summarizing its evaluation to forward with the dossier. A record of the committee's vote is documented and forwarded to the dean.

Review and Recommendations by the Dean: The dean sends forward to the provost the full dossier of every candidate for whom there is a positive recommendation from either the college committee or the dean, or both. The dean prepares a separate letter of recommendation to be forwarded with the dossier. Whenever the dean does not concur with the committee's recommendation, the committee is so notified.

The dossiers that the dean sends to the provost are accompanied by a statement describing the formation and procedures of the college committee and a summary of the number of candidates considered by the committee in each category. The division of the vote at both the departmental and college levels is conveyed to the university-level committee and provost, but must otherwise remain confidential.

If both the college committee and the dean of the college reject a positive department recommendation, the normal process of review is concluded and the dossier is not sent forward to the provost. The dean informs the department head or chair of the rejection and the department head or chair notifies the departmental committee and the faculty member. In that case, the dean informs the faculty member in writing of the specific reasons for the decision and provides notification of appeal options outlined in section 3.4.5, "Appeals of Decisions on Reappointment, Tenure, or Promotion."

### 3.4.4.3 University Evaluation

The University Promotion and Tenure Committee is appointed and chaired by the senior vice president and provost. The committee reviews the qualifications of the candidates recommended for promotion or tenure by each college dean. It also reviews those cases in which the dean does not concur with the college committee's positive recommendation. The purpose of the reviews is to verify that the recommendations are consistent with the evidence, reflecting university standards, and that they are consistent with university objectives, programmatic plans, and budgetary constraints.

Guidelines for submission of candidates' dossiers are available on the provost's website.

The committee makes a recommendation on each candidate to the provost. The provost makes recommendations to the president, informing the committee of those recommendations, including the basis for any non-concurrence with committee recommendations. The provost informs the president of any variation between the provost's recommendations and those of the committee.

The president makes recommendation to the board of visitors from among those candidates reported by the provost with the board of visitors being responsible for the final decision.

The provost notifies the appropriate dean of any negative decision reached by the provost, the president, or the board of visitors. The dean, when notifying the faculty member in writing, notes appeal options.

On recommendation of the Commission on Faculty Affairs, the University Council approved the following further guidelines on formation and procedures of the university committee:

- The university committee consists of the college deans and tenured faculty members of the rank of associate professor or higher, one from each college and one faculty member-at-large. The selection of the faculty members should be based on demonstrated professional excellence.
- All members of the committee hold voting privileges. Regardless of the size of the committee, the faculty must always have at least a majority of the potential votes. Consistent with the principle that participants at all levels of the promotion and tenure review process vote only once on an individual case, deans do not vote on cases from their own college. Similarly, faculty members serving on the university committee do not vote on any case they previously voted on, should this circumstance occur.
- Some significant element of faculty choice should be part of the selection procedure; therefore, each college faculty, by means deemed suitable by them, nominates two faculty members for each vacancy, from which the provost selects one. The Faculty Senate nominates two faculty members for the at-large appointment, from which the provost selects one.
- The faculty members of the committee hold rotating terms of three years.
- The provost chairs the committee, but does not hold voting privileges.
- All voting within the committee should be by written secret ballot; the division of any ballot must remain confidential.

### 3.4.4.4 Promotion and Tenure Guidelines

The promotion and tenure guidelines and a standard cover page are available on the provost's website. All candidate dossiers must be submitted to the University Promotion and Tenure Committee according to the guidelines on the provost's website.

### 3.4.5 Appeals of Decisions on Reappointment, Tenure, or Promotion

A faculty member who is notified of a negative decision following evaluation for a term reappointment during the probationary period, for a tenured appointment, or for promotion, and

who believes that the decision was improperly or unfairly determined may appeal for review of the decision under conditions and procedures specified in this section.

Such an appeal must be filed, in writing, within 14 calendar days of formal notification of the decision, which shall make reference to appeal procedures. The appeal can only be based on grounds that certain relevant information was not provided or considered in the decision, or that the decision was influenced by improper consideration. The written appeal must specify the grounds and the basis for such an allegation.

Although the provisions for appeal described below are designed to give faculty members protection against capricious or arbitrary decisions, the faculty member who believes that these procedures have been improperly followed may, at any point, seek advice from the Faculty Senate Committee on Reconciliation and/or file a grievance in accordance with the grievance procedure in section 3.7, "Faculty Grievance Policy and Procedures."

### 3.4.5.1 Probationary Reappointment

Faculty members on probationary term appointments should make no presumption of reappointment. The department head or chair with the advice of the departmental personnel committee or the faculty development committee determines non-reappointment. Notice of non-reappointment is furnished according to the schedule in section 2.11, "Retirement, Resignation, and Non-Reappointment." The specific reasons for the decision are furnished to the faculty member in writing, if requested.

If the decision is based primarily on evaluation of the faculty member's performance, including perceived lack of potential for further professional development, then the faculty member may request a review of the decision by the dean of the college. If the dean sustains the departmental decision, the faculty member may request, through the dean, the further and independent review of the decision by the properly constituted college committee on promotion and tenure.

The faculty member presents the appeal in writing as specified in section 3.4.5, "Appeals of Decisions on Reappointment, Tenure, or Promotion." The faculty member may elect to present oral arguments to the committee. The college committee makes recommendation to the dean, who informs the faculty member of the committee's recommendation and the dean's subsequent decision. The dean's decision closes the appeal process, unless it is at variance with the college committee's recommendation, in which case the faculty member may appeal to the provost for a final decision.

### 3.4.5.2 Tenure Decision

Occasionally faculty members are evaluated for a tenured appointment during the probationary period but before the final probationary year. In such a case, there is no recourse to appeal or review of a negative decision, at whatever level it is reached, because of the certainty that the evaluation will be undertaken again within a limited time.

Evaluation for a tenured appointment is mandated in the sixth year of probationary service unless the faculty member has given written notice of resignation from the faculty. If both the departmental committee and the department head or chair agree that the faculty member's record does not warrant a tenured appointment, there is an automatic review of the candidate's dossier by the dean. If the dean concurs, the faculty member is notified by the dean, in writing, of the decision and the specific reasons for it.

The faculty member may then request, through the dean, that the college committee on promotion and tenure independently review the decision. The faculty member presents the appeal in writing as specified in 3.4.5, "Appeals of Decisions on Reappointment, Tenure, or Promotion." The faculty member may elect to present oral arguments to the committee as well. If the committee concurs with the decision, the decision is final, the dean so notifies the faculty member in writing, and no further appeal is provided.

During the automatic review of the candidate's dossier, the dean may wish to reserve judgment. In such a case, the dean notifies the faculty member of the departmental decision and tells the faculty member that he or she is requesting the college committee on promotion and tenure to undertake an independent review, as specified in the previous paragraph, and to make a recommendation. Should the college committee and the dean concur with the departmental decision, the decision is declared final, the faculty member is so notified, and no further appeal is provided. The specific reason for the decision is provided to the faculty member in writing.

In any case of college-level review of a negative departmental decision, a positive recommendation by either the college committee or the dean is sent with the dossier to the University Promotion and Tenure Committee in the same way as in the normal review process.

If the college committee and the dean undertake the review based on a positive recommendation of either or both the departmental committee and the department head or chair and if the college committee recommends that tenure not be awarded and the dean concurs, the faculty member is notified of the negative decision with reference to appeal procedures. The specific reasons for the decision are furnished to the faculty member in writing. The faculty member may then appeal through the provost for review of the decision by the university committee, which makes a recommendation to the provost for a final decision. The faculty member presents the appeal in writing as specified in section 3.4.5, "Appeals of Decisions on Reappointment, Tenure, or Promotion." No further appeal is provided. The university committee may choose to hear oral arguments.

Should the provost not concur with a positive recommendation from the University Promotion and Tenure Committee, whether that recommendation culminates a normal review or an appeal, the faculty member is so notified in writing of the specific reason for the decision. The faculty member may appeal to the Faculty Review Committee. That committee investigates the case and, if the differences cannot be reconciled, makes a recommendation to the president on the matter. The president's decision is final.

During review following an appeal, the college committee may find reason to believe that the departmental evaluation was biased or was significantly influenced by improper considerations.

In that case, the reviewing committee may request that the college dean form an *ad hoc* committee to re-initiate the evaluation. The *ad hoc* committee is composed, as feasible, of faculty members in the candidate's department or in closely allied fields and does not contain any members of the original committee.

Should the university committee make such a finding in the review of an appeal relative to the college evaluation, it requests the dean to form a new *ad hoc* committee at the college level. The *ad hoc* committee makes a recommendation to the committee that requested its formation.

### 3.4.5.3 Promotion Decision

There is no specification for minimum or maximum time of service in any rank. Consideration for promotion in rank may be requested of the department head or chair by a faculty member at any time if the department head or chair or committee has not chosen to undertake such an evaluation. However, appeal of a negative promotion decision is provided only if the faculty member has been in rank for at least six years and if the faculty member has formally requested, in writing, consideration for promotion in a previous year. In such a case, for a member of a collegiate faculty, or a member of the administrative and professional faculty seeking promotion in rank through an academic department, an appeal follows the same procedures as in section 3.4.5, "Appeals of Decisions on Reappointment, Tenure, or Promotion."

# 3.5 Annual Evaluation, Post-Tenure Review, and Periodic Review of College and Departmental Administrators

## 3.5.1 Annual Evaluation and Salary Adjustments

All departments are required to have written guidelines outlining the process and criteria used in faculty evaluations. The adoption of such guidelines promotes consistency and transparency in this important aspect of faculty life. Guidelines and procedures for the annual review of university or alumni distinguished professors are established by the president and/or provost, who are responsible for their evaluations.

Every faculty member's professional performance is evaluated annually and written feedback is provided separately from confirmation of any merit adjustments. The process begins with submission of a Faculty Activity Report (FAR). All non-temporary faculty members must submit a FAR annually. These reports form part of the basis for performance evaluations, awarding merit adjustments, and promotion, tenure, and post-tenure reviews.

Department heads/chairs are responsible for conducting annual faculty evaluations, either independently or in consultation with an appropriately charged committee in accordance with departmental procedures. All evaluations must be in writing and include a discussion of contributions and accomplishments in all areas of the faculty member's responsibilities, comments on the faculty member's plans and goals, and any recommendations for improvement or change. Faculty members should receive their written evaluations within 90 days of submission of required materials, and they acknowledge receipt by signing and returning a copy for their departmental file, or the electronic equivalent. Acknowledging receipt of the evaluation

does not imply agreement with it. If a faculty member substantially disagrees with the evaluation, that member may submit a written response to the department head/chair for inclusion in his or her personnel file.

In addition to their annual evaluation letters, all pre-tenure faculty members receive at least two thorough reviews during the normal six-year probationary period and written feedback on their progress toward tenure by their departmental promotion and tenure committee prior to reappointment in accordance with guidance included in section 3.4.2, "Probationary Period."

Faculty with part-time appointments are reviewed on the normal annual review cycle. For purposes of annual review, the fraction of the appointment must be taken into account when considering the appropriate level of achievement in that year.

Salary adjustments are based on merit; they are not automatic. Recommendations for salary adjustments originate with the department head or chair and are reviewed by the dean, the provost, and the president. Because salary adjustments are determined administratively on an annual basis and based significantly on the quality of the faculty member's response to assigned responsibility, they do not necessarily reflect an accurate measure of the full scope of the faculty member's professional development as evaluated by relevant committees in the tenure and promotion process.

The salary adjustments of continuing faculty members are approved by the board of visitors, and each faculty member is informed in writing of the board's action as soon as possible. (See section 2.4, "Faculty Compensation Plan.")

### 3.5.2 Unsatisfactory Performance

Failure to meet the minimal obligations and standards the department has stipulated for its faculty results in an "unsatisfactory" rating. Written notification of an unsatisfactory rating and the considerations upon which it was based is given to the faculty member, with copies to the dean and provost. A single unsatisfactory evaluation indicates a serious problem, which should prompt remedial action. Faculty members may respond in writing with a letter to the head or chair for inclusion in their personnel file, or they may seek redress through either the reconciliation or grievance procedures. Two successive annual ratings of unsatisfactory performance for a tenured faculty member results in a post-tenure review.

### 3.5.3 Departmental Minimal Standards

Each academic department shall develop, maintain, and publish a statement of minimal standards for satisfactory faculty performance using the following process. Standards should be written with the participation of faculty in the department and approved by a vote of the tenured and tenure-track faculty in the department. Standards developed and approved by departments and the head or chair are then reviewed by the college-level promotion and tenure committee and the dean, and reviewed and approved by the provost. Once approved, the department's standards are published and made available to all faculty in the department. Revisions of departmental standards also follow these procedures.

The following guidance is provided for the development of departmental minimal standards:

- Departments should carefully assess and state the overall standards of professional performance and contribution they consider minimally acceptable for tenured faculty. Each department's evaluation mechanism should allow a distinction between performance that is deficient in one or more areas requiring improvement, and performance that is so seriously deficient as to merit the formal designation, "unsatisfactory."
- Departmental standards should embrace the entire scope of faculty contributions. Expectations should recognize differences in faculty assignments within the same department. Departmental standards should typically address the individual's skill, effort, and effectiveness in contributing to all aspects of the instructional mission; the individual's activity in and contributions to the academic discipline; the individual's contributions to the collective life of the department, college, and university; and the individual's activity in and contributions to the university's outreach mission.
- Departmental statements should affirm support for the basic principles of academic freedom and should express tolerance for minority opinions, dissent from professional orthodoxies, and honest and civil disagreement with administrative actions.
- Departmental statements should include the expectation that tenured faculty will adhere to the standards of conduct and ethical behavior as stated in the *Faculty Handbook* and/or promulgated through other official channels.

### 3.5.4 Post-Tenure Review

Nothing in this section should be interpreted as abridging the university's right to proceed directly to dismissal for cause as defined in 3.6.3, "Dismissal for Cause," or the right of individual faculty members to pursue existing mechanisms of reconciliation and redress.

A post-tenure review is mandatory whenever a faculty member with tenure receives two consecutive annual evaluations of unsatisfactory performance. Annual reviews for years spent on leave without pay are disregarded for the purpose of this calculation. The departmental promotion and tenure committee conducts the review, unless the same committee was involved in the original unsatisfactory annual evaluations. In this case, the department elects a committee to carry out the review function.

Upon recommendation of the head or chair and with the approval of the dean, a post-tenure review may be waived or postponed if there are extenuating circumstances (such as health problems).

The purpose of a post-tenure review is to focus the perspective of faculty peers on the full scope of a faculty member's professional competence, performance, and contributions to the department, college, and university mission and priorities.

The faculty member has the both the right and the obligation to provide a dossier with all documents, materials, and statements he or she believes are relevant and necessary for the review. Ordinarily, such a dossier includes at least the following: an up-to-date *curriculum vitae*, the past two or more faculty activity reports, teaching assessments, and a description of activities

and accomplishments since the last faculty activity report. The faculty member is given a period of no less than four weeks to assemble the dossier for the committee. The head or chair supplies the review committee with the last two annual evaluations, all materials that were considered in those evaluations, any further materials deemed relevant, and other materials the committee requests. Copies of all materials supplied to the committee are given to the faculty member. The faculty member has the right to provide a written rebuttal of evidence provided by the head or chair.

The committee weighs the faculty member's contributions to the discipline, the department, and the university through learning, discovery, and engagement. The burden of proving unsatisfactory performance is on the university. The committee prepares a summary of its findings and makes a recommendation to the head or chair, with copies to the dean and provost. Final action and notification of the faculty member is the responsibility of the head or chair and dean, with the concurrence of the provost.

The review may result in one of the following outcomes:

**Certification of satisfactory performance:** The committee may conclude that the faculty member's competence and professional contributions are satisfactory to meet the department's minimal expectations, thus failing to sustain the assessment of the head or chair. The review is then complete. An unsatisfactory rating in any subsequent year is counted as the first in any future sequence.

**Certification of deficiencies:** The committee may concur that the faculty member's competence and/or professional contributions are unsatisfactory to meet the department's minimal expectations. The committee may recommend dismissal for cause, a sanction other than dismissal for cause, or a single period of remediation not to exceed two years.

**Remediation:** If a period of remediation is recommended, the committee specifies in detail the deficiencies it noted, defines specific goals and measurable outcomes the faculty member should achieve, and establishes a timeline for meeting the goals. The head or chair meets with the faculty member at least twice annually to review the individual's progress. The head or chair prepares a summary report for the committee following each meeting and at the end of the specified remediation period, at which time the committee either certifies satisfactory performance or recommends dismissal for cause or a sanction other than dismissal for cause following the procedures described below.

Sanction other than dismissal for cause: A departmental recommendation to impose a severe sanction, as defined in section 3.6.2, "Imposition of a Severe Sanction," shall be referred to the college-level promotion and tenure committee, which reviews the case as presented to the departmental committee, provides an opportunity for the faculty member to be heard, and determines whether the recommendation is consistent with the evidence. The college-level committee may reject, uphold, or modify the specific sanction recommended by the departmental committee. If the college-level committee also recommends imposition of a severe sanction, then the same procedures used for dismissal for cause guides the process. The reviews conducted by the department- and college-level committees satisfy the requirement in step two of section 3.6.3,

"Dismissal for Cause," for an informal inquiry by an *ad hoc* or standing personnel committee. Thus, in the case of a post-tenure review, this step is not repeated.

If a severe sanction is imposed or ultimately rejected, then the post-tenure review cycle is considered complete. An unsatisfactory rating in any subsequent year is counted as the first in any future sequence.

**Dismissal for cause:** If dismissal for cause is recommended, the case shall be referred to the college-level promotion and tenure committee as described in section 3.4.4.2, "College Evaluation," which reviews the case as presented to the departmental committee and determines whether the recommendation is consistent with the evidence. If the college-level committee upholds the recommendation for dismissal, then the procedures specified in section 3.6.3, "Dismissal for Cause," begin immediately. The committee review satisfies the requirement in section 3.6.3 for an informal inquiry by a standing personnel committee.

## 3.5.5 Periodic Review of Academic Deans, Department Heads, Senior Administrators, and Academic Vice Presidents

In addition to an annual performance evaluation, and in accordance with university policy 6105, academic deans and academic vice presidents are subject to periodic evaluations every five years. If the review of a dean cannot be conducted in the fifth year as would normally be the case, the senior vice president and provost will inform the officers of the relevant college faculty association as to the reason for delay. A review may also be initiated at any time by the senior vice president and provost and/or at the request of at least one-third of the tenure-track faculty in the college. If the review of a vice president who reports to the provost cannot be conducted in the fifth year, the senior vice president and provost will notify the deans and the chairs of the college faculty associations as to the reason for the delay. In the semester prior to a periodic evaluation, the faculty association will be notified of the review and the association may schedule a faculty meeting with the Provost to discuss the upcoming review. University policy 6105 outlines the procedures for periodic reviews.

University policy 6100, "Department Head or Chair Appointments," outlines the review process for academic department heads or chairs. In addition, colleges should adopt more detailed procedures in accordance with the broad guidelines below so that reviews may be conducted consistently and appropriately across the college for those serving in academic leadership roles.

The purpose of the periodic review is to support the success of the university's academic units by providing developmental feedback to promote fair and effective academic leadership. Reappointment of an individual to academic department head, chair, academic dean, or academic vice president must be preceded by a periodic review conducted in accordance with guidelines outlined in university policy 6105 or in university policy 6100.

### 3.6 Imposition of a Severe Sanction or Dismissal for Cause

### 3.6.1 Adequate Cause

Adequate cause for imposition of a severe sanction or dismissal is related, directly and substantially, to the fitness of faculty members in their professional capacity as teachers and scholars. Imposition of a severe sanction or dismissal will not be used to restrain faculty members in their exercise of academic freedom or other rights of American citizens.<sup>1</sup>

Adequate cause includes: violation of professional ethics (see section 2.7, "Professional Responsibilities and Conduct"); incompetence as determined through post-tenure review; willful failure to carry out professional obligations or assigned responsibilities; willful violation of university and/or government policies; falsification of information relating to professional qualifications; inability to perform assigned duties satisfactorily because of incarceration; or personal deficiencies that prevent the satisfactory performance of responsibilities (e.g., dependence on drugs or alcohol).

Reason to consider the imposition of a severe sanction or dismissal for cause is usually determined by a thorough and careful investigation by an appropriately-charged faculty committee (as in the case of allegations of ethical or scholarly misconduct, or through a post-tenure review) or by the relevant administrator (for example, the department head/chair, compliance and conflict resolution officer, internal auditor, or Virginia Tech Police). Generally, these investigations result in a report of findings; some reports also include a recommendation for sanctions. The report is directed to the relevant administrator for action; it is also shared with the faculty member. Imposition of a severe sanction or initiation of dismissal for cause proceedings, if warranted, follows the procedures set forth below.

### 3.6.2 Imposition of a Severe Sanction

**Definition and Examples:** A severe sanction generally involves a significant loss or penalty to a faculty member such as, but not limited to, a demotion in rank and/or a reduction in salary or suspension without pay for a period not to exceed one year, imposed for unacceptable conduct and/or a serious breach of university policy.

Routine personnel actions such as a recommendation for a below average or no merit increase, conversion from a calendar year to an academic year appointment, reassignment, or removal of an administrative stipend do not constitute "sanctions" within the meaning of this policy. A personnel action such as these may be a valid issue for grievance under procedures defined in the *Faculty Handbook*.

**Process for Imposing a Severe Sanction:** The conduct of a faculty member, although not constituting adequate cause for dismissal, may be sufficiently grave to justify imposition of a severe sanction. Imposition of a severe sanction follows the same procedures as dismissal for cause beginning with step one. If the matter is not resolved at the first step, a standing or *ad hoc* 

<sup>&</sup>lt;sup>1</sup> The procedures specified follow closely, but differ in occasional detail from, the "1976 Institutional Regulations on Academic Freedom and Tenure" approved by Committee A of the AAUP.

faculty committee conducts an informal inquiry (step two). The requirement for such an informal inquiry is satisfied if the investigation was conducted by an appropriately charged faculty committee (as would be the case with an alleged violation of the ethics or scholarly misconduct policies) and, having determined that in its opinion there is adequate cause for imposing a severe sanction, refers the matter to the administration.

#### 3.6.3 Dismissal for Cause

The following procedures apply to faculty members with tenure or for dismissal of a tenure-track faculty member before the end of their current contract.

Dismissal is preceded by:

- 1. Step one: Discussions between the faculty member, department head or chair, dean, and/or provost, looking toward a mutual settlement.
- 2. Step two: Informal inquiry by a standing (or, if necessary, *ad hoc*) faculty committee having concern for personnel matters. This committee attempts to effect an adjustment and, failing to do so, determines whether in its opinion dismissal proceedings should be undertaken, without its opinion being binding on the president's decision whether to proceed.
- 3. Step three: The furnishing by the president (in what follows, the president may delegate the provost to serve instead) of a statement of particular charges, in consultation with the department head or chair and dean. The statement of charges is included in a letter to the faculty member indicating the intention to dismiss, with notification of the right to a formal hearing. The faculty member is given a specified reasonable time limit to request a hearing, that time limit is no less than 10 days.

Procedures for conducting a formal hearing, if requested: If a hearing committee is to be established, the president asks the Faculty Senate, through its president, to nominate nine faculty members to serve on the hearing committee. These faculty members should be nominated on the basis of their objectivity, competence, and regard in which they are held in the academic community. They must have no bias or untoward interest in the case and be available at the anticipated time of hearing. The faculty member and the president each have a maximum of two challenges from among the nominees without stated cause. The president then names a five-member hearing committee from the remaining names on the nominated slate. The hearing committee elects its chair.

Pending a final decision on the dismissal, the faculty member is suspended only if immediate harm to him or herself or to others is threatened by continuance. If the president believes such suspension is warranted, consultation takes place with the <u>Reconciliation Committee</u> of the Faculty Senate concerning the propriety, the length, and other conditions of the suspension. Ordinarily, salary continues during such a period of suspension.

The hearing committee may hold joint pre-hearing meetings with both parties to simplify the issues, effect stipulations of facts, provide for the exchange of documentary or other information,

and achieve such other appropriate pre-hearing objectives as will make the hearing fair and expeditious.

Notice of hearing of at least 20 days is made in writing. The faculty member may waive appearance at the hearing, instead responding to the charges in writing or otherwise denying the charges or asserting that the charges do not support a finding of adequate cause. In such a case, the hearing committee evaluates all available evidence and rests its recommendation on the evidence in the record.

The committee, in consultation with the president and the faculty member, exercises its judgment as to whether the hearing is public or private. During the proceedings, the faculty member is permitted to have an academic advisor and legal counsel. At the request of either party or on the initiative of the hearing committee, a representative of an appropriate educational association is permitted to attend the hearing as an observer.

A verbatim record of the hearing is taken.

The burden of proof that adequate cause exists rests with the institution.

The hearing committee grants adjournment to enable either party to investigate evidence about which a valid claim of surprise is made. The faculty member is afforded an opportunity to obtain necessary witnesses and documentary or other evidence. The administration cooperates with the hearing committee in securing witnesses and evidence. The faculty member and administration have the right to confront and cross-examine all witnesses. The committee determines the admissibility of statements of unavailable witnesses and, if possible, provides for interrogatories.

The hearing committee is not bound by strict rules of legal evidence, and may admit any evidence that is of probative value in determining the issues involved. Every possible effort is made to obtain the most reliable evidence available.

The findings of fact and the recommendation are based solely on the hearing record. The president and the faculty member are notified of the recommendation in writing and are given a written copy of the record of the hearing.

If the hearing committee concludes that adequate cause for dismissal has not been established, it so reports to the president. In such a case, the committee may recommend sanctions short of outright dismissal or may recommend no sanctions. If the president rejects the recommendation, the hearing committee and the faculty member are so informed in writing, with reasons, and each is given an opportunity for response.

Appeal to the board of visitors: If the president decides to impose dismissal or other severe sanction, whether that is the recommendation of the hearing committee, the faculty member may request that the full record of the case be submitted to the board of visitors (or a duly constituted committee of the board). The board's review is based on the record of the committee hearing, and it provides opportunity for argument, written or oral or both, by the principals at the hearing or their representatives. If the recommendation of the hearing committee is not sustained, the

proceeding returns to the committee with specific objections. The committee then reconsiders, taking into account the stated objections and receiving new evidence if necessary. The board makes a final decision only after study of the committee's reconsideration.

Notice of Termination/Dismissal: In cases where gross misconduct is decided, termination is usually immediate. The standard for gross misconduct is behavior so egregious that it evokes condemnation by the academic community generally and is so utterly blameworthy as to make it inappropriate to offer additional notice or severance pay. The first faculty committee that considers the case determines gross misconduct. In cases not involving gross misconduct: (a) a faculty member with tenure receives up to one year of salary or notice, and (b) a probationary faculty member receives up to three months salary or notice. These terms of dismissal begin on the date of final notification of dismissal.

## **3.7** Faculty Grievance Policy and Procedures

The following procedure is provided as the means for resolution of grievances against a supervisor or member(s) of the university administration brought by tenured or tenure-track faculty members.

## **3.7.1** Faculty Reconciliation and Mediation Services

Informal Dialogue: It should be possible to resolve most faculty concerns or complaints through informal communication among colleagues working together in the academic enterprise. Accordingly, a faculty member who feels he or she has a grievance is encouraged to take it to his or her immediate supervisor in the normal collegial spirit of problem solving rather than as a confrontation between adversaries.

Reconciliation: At the initiation of the grievance procedure, or at any earlier time, the grievant may request the assistance of the Faculty Senate Committee on Reconciliation in fashioning an equitable solution. Contacting the Faculty Senate Committee on Reconciliation is not required in filing a grievance, but it may be useful if the grievant feels that the issue may be amenable to, but will require time for, negotiation; or if the grievant is unsure whether his or her concern is a legitimate issue for a grievance; or if personal relations between the parties involved in the grievance have become strained.

For a potential grievance issue to qualify for consideration by the Faculty Senate Committee on Reconciliation, the grievant must contact the Faculty Senate Committee on Reconciliation within 30 calendar days of the time when the grievant knew or should have known of the event or action that is the basis for the potential grievance, just as if beginning the regular grievance process. If the grievant requests assistance from the Faculty Senate Committee on Reconciliation, that committee requests a postponement of the time limits involved in the grievance procedure while it deals with the case. The request is submitted in writing to the vice provost for faculty affairs by the chair of the Faculty Senate Committee on Reconciliation. Also, the grievant should reach an understanding with the Faculty Senate Committee on Reconciliation of the time frame planned for that committee's work on the case, such time not to exceed 60 calendar days.

Faculty members may also consult the Faculty Senate Committee on Reconciliation about serious disagreements with immediate supervisors or other university administrators concerning issues that may not be eligible for consideration within the grievance process. In such instances, the committee contacts the relevant administrator to determine if there is an interest and willingness to explore informal resolution of the dispute; it is not necessary to notify the Office of the Senior Vice President and Provost.

Additional information regarding the faculty reconciliation process is available on the <u>provost's</u> website.

Mediation: <u>Mediation</u> is a voluntary, confidential process through which trained neutral third persons (mediators) assist people to express their concerns and develop solutions to the dispute in a safe and structured environment. Because mediation is voluntary, both parties must agree to participate in order for mediation to occur. Faculty members and supervisors are encouraged to consider using mediation to resolve disputes between them, or to help address a conflict between a faculty member and another member of the Virginia Tech community.

Role of Mediators: Mediators do not make judgments, determine facts, or decide the outcome; instead they facilitate discussion between the participants, who identify the solutions best suited to their situation. No agreement is made unless and until it is acceptable to the participants.

Requesting Mediation: Mediation is available at any time, without the filing of a grievance. Additionally, mediation may be requested by any party during the grievance process prior to step four. If, after the initiation of a formal grievance, both parties agree to participate in mediation, the grievance is placed on administrative hold until the mediation process is complete. If the parties come to a resolution of the dispute through mediation, the parties are responsible to each other for ensuring that the provisions of the agreement are followed. In the event that the parties are not able to reach a mutual resolution to the dispute through mediation, the grievant may request that the grievance be reactivated and the process continues.

Mediation differs from faculty reconciliation in that mediators do not engage in fact-finding or in evaluation of decisions. Both mediation and reconciliation, however, are voluntary; no party is required to participate in either process.

To learn more about mediation and other forms of informal conflict resolution processes, contact the <u>conflict resolution program</u> manager in human resources at 540-231-9331.

### **3.7.2** The Formal Grievance Procedure

If the assistance of the Faculty Senate <u>Committee on Reconciliation</u> is not desired or is not requested; or if that committee determines that it cannot provide assistance in the matter; or if the grievant finds that the length of time the Committee on Reconciliation plans or takes with the case is excessive; or if the grievant is not satisfied with the recommendations of that committee, the grievant may pursue the issue as a formal grievance through the following procedure. Department heads or chairs, deans, directors, and other administrative faculty will cooperate with

the grievant in the mechanics of processing the grievance, but the grievant alone is responsible for preparation of his or her case.

- 1. Step one: The grievant must meet with his or her immediate supervisor (normally the department head or chair) within 30 calendar days of the date that grievant knew or should have known of the event or action that is basis for the grievance and orally identifies the grievance and the grievant's concerns. The supervisor provides an oral response to the grievant within five weekdays following the meeting. If the supervisor's response is satisfactory to the grievant, that ends the matter.
- 2. Step two: If a satisfactory resolution of the grievance is not achieved by the immediate supervisor's oral response, the grievant may submit a written statement of the grievance and the relief requested to the immediate supervisor. This statement must be on the faculty grievance form, must define the grievance and the relief requested specifically and precisely, and must be submitted to the immediate supervisor within five weekdays of the time when the grievant received the immediate supervisor's oral response to the first step meeting. Faculty grievance forms are available on the provost's website.

Within five weekdays of receiving the written statement of the grievance, the immediate supervisor, in turn, gives the grievant a written response on the faculty grievance form, citing reasons for action taken or not taken. If the written response of the immediate supervisor is satisfactory to the grievant, that ends the matter.

3. Step three: If the resolution of the grievance proposed in the written response by the immediate supervisor is not acceptable, the grievant may advance the grievance to the next higher level of university administration by checking the appropriate place on the faculty grievance form, signing and sending the form to the next higher administrator within five weekdays of receiving the written response from the immediate supervisor. The next higher level of administration for collegiate faculty is normally the college dean. The administrator involved at this next level is hereafter referred to as the second-level administrator.

Following receipt of the faculty grievance form, the second-level administrator or designated representative meets with the grievant within five weekdays. The second-level administrator may request the immediate supervisor of the grievant be present; the grievant may similarly request that a representative of his or her choice from among the university faculty be present. Unless the grievant is represented by a member of the faculty who is also a lawyer, the second-level administrator does not have legal counsel present. The second-level administrator gives the grievant a written decision on the faculty grievance form within five weekdays after the meeting, citing reasons for his or her decision. If the second-level administrator's written response to the grievance is satisfactory to the grievant, that ends the matter.

4. Step four: If the resolution of the grievance proposed in the written response from the second-level administrator is not acceptable, the grievant may advance the grievance within five weekdays to the level of the provost, including consideration by an impartial hearing

panel of the Faculty Review Committee. A description of the charge and membership of the <u>Faculty Review Committee</u> is included in section 1.4.5.2, "Faculty Review Committee."

Upon receiving the faculty grievance form requesting step four review, the provost, or appropriate designated representative, acknowledges receipt of the grievance within five weekdays and forwards a copy of the <u>Procedures of the Faculty Review Committee</u> to parties in the grievance process. The provost immediately forwards a copy of the grievance to the <u>president of the Faculty Senate</u>, who also writes to the grievant to acknowledge receipt of the grievance within five weekdays of receipt of the faculty grievance form from the provost.

The grievant may petition the provost to bypass the Faculty Review Committee and rule on the grievance. If the provost accepts the request, there is no subsequent opportunity for the grievance to be heard by a hearing panel. The provost's decision, however, may be appealed to the president, as described in step five. If the provost does not accept the petition, the Faculty Review Committee hears the grievance as outlined in these procedures.

• **Hearing Panel:** A hearing panel consists of five members appointed by the chair of the Faculty Review Committee from among the members of the Faculty Review Committee. The chair of the Faculty Review Committee polls all appointees to ensure that they have no conflict of interest in the case. Both parties to the grievance may challenge one of the appointments, if they so desire, without need to state cause, and the chair of the Faculty Review Committee appoints the needed replacement or replacements. Other replacements are made only for cause. The chair of the Faculty Review Committee rules on issues of cause.

To ensure uniformity in practice, the chair of the Faculty Review Committee or his or her designee serves as the non-voting chair of each hearing panel. In the event that the chair of the Faculty Review Committee has a conflict of interest concerning a case, the chair appoints a disinterested third party from among the members of the Faculty Review Committee not already appointed to the hearing panel for the case to serve as chair of the hearing panel.

• **Hearings:** After a hearing panel is appointed, the chair of the Faculty Review Committee requests that each party to the grievance provide relevant documentation to be shared among the parties and the hearing panel. The panel holds its initial hearing with both principals present within 15 weekdays of receipt of the grievance by the Faculty Senate president. If the panel feels it needs to investigate the case further, or requires more information, or desires to hear witnesses, the hearing is adjourned until the panel completes the necessary work or scheduling. The hearing is then reconvened as appropriate.

Each party to the grievance may have a representative present during the sessions of the hearing at which testimony is presented. The representative may speak on their behalf if so requested. Representatives may be legal counsel, if both parties are so represented, but if the grievant does not wish to have legal counsel at a hearing, neither party to the grievance may have legal counsel present.

These impartial panel hearings are administrative functions, not adversarial proceedings. Therefore, if legal counsels are present they must understand that the proceedings do not follow courtroom or trial procedures and rules. Participation by legal counsel is at the invitation of the parties they represent and is subject to the rulings of the chair of the hearing panel. Detailed procedures followed in hearings are specified in the "Procedures of the Faculty Review Committee" as approved by the Faculty Senate.

• **Findings and Recommendations:** The hearing panel concludes its work and makes its recommendations within 45 weekdays of receipt of the grievance by the Faculty Senate president. The time limit for consideration may be extended by agreement of both parties.

The hearing panel formulates written findings and recommendations regarding disposition of the grievance and forwards copies to the provost, the grievant, and the chair of the Faculty Review Committee.

- **Provost's Action:** The provost meets with the grievant within 10 weekdays after receiving the findings and recommendations of the hearing panel to discuss the case and advise the grievant about the prospects for disposition of the case. Within 10 weekdays of that meeting, the provost sends to the grievant his or her decision in writing concerning the disposition of the grievance. If the provost's decision is fully consonant with (or exceeds) the recommendations of the hearing panel, or if it is satisfactory to the grievant even if it differs from the recommendations of the hearing panel, that ends the matter.
- 5. Step five: If the provost's decision is not acceptable to the grievant and not consonant with the recommendations of the hearing panel, the grievant may appeal in writing to the president within 20 calendar days. The president acts as he or she sees fit. The president's decision is final.

## 3.7.3 Timeliness of Grievance and Procedural Compliance

A grievance must be brought forward in a timely manner. It is the responsibility of the grievant to initiate the grievance process within 30 calendar days of the time when he or she knew or should have known of the event or action that is the basis for the grievance. The university administration is not required to accept a grievance for processing if the grievant does not meet the 30-day deadline, except in cases of demonstrated good cause.

Scheduled commitments made prior to the time of filing or advancement of a grievance that preclude action by either of the parties to the grievance automatically extend time limits for their duration unless this would be demonstrably harmful to the fair processing of the grievance. In such cases, on written request by the grievant to the appropriate office for that step, the grievance is advanced to the next step in the grievance process.

If the grievant does not follow the time limits specified in the grievance procedure it is assumed that he or she accepted the last proposed resolution as satisfactory. If the grievant desires to advance the grievance after the appropriate specified time limits have lapsed, the administrator

who receives the late submission notifies the chair of the <u>Faculty Review Committee</u> in writing, and the chair of the Faculty Review Committee determines if there was good cause for the delay. If so, the grievance proceeds. If not, the process ends with the most recently proposed resolution in force. The finding on the matter by the chair of the Faculty Review Committee is communicated to both parties in writing.

If either party to a grievance charges the other with procedural violations other than time limit issues, a special committee of the president of the Faculty Senate, the chair of the Faculty Senate Committee on Reconciliation, and the chair of the Faculty Review Committee (or the vice president of the senate if the president is also chair of the Faculty Review Committee) is convened to rule on the question, as in disputes about the validity of issues qualifying for the grievance procedure. The special committee has the following options. It can either find no significant procedural violation occurred, in which case the grievance process continues unaffected, or that a significant procedural violation did occur. If the administrator committed a significant procedural violation, the grievance automatically qualifies for advancement to the next step in the grievance process. If the grievant committed a significant procedural violation, the grievance process ends at that point with the last proposed resolution established as the final disposition of the case.

### 3.7.4 Valid Issues for Grievance

For this process, a grievance is defined as a complaint by a faculty member alleging a violation, misinterpretation, or incorrect application of a policy, procedure, or practice of the university that directly affects the grievant. Some examples of valid issues for filing a grievance are: improperly or unfairly determined personnel decisions that result in an unsatisfactory annual performance evaluation, unreasonable merit adjustment or salary level, or excessive teaching load/work assignments; substantive violations of promotion and tenure procedures (see appeal process in section 3.4.5, "Appeals of Decisions on Reappointment, Tenure, or Promotion"); reprisals; substantive error in the application of policy; and matters relating to academic freedom.

Issues not open to grievance: While most faculty disputes with the university administration may be dealt with by this grievance policy, the following issues may not be made the subject of a grievance: determination of policy appropriately promulgated by the university administration or the university governance system; those items falling within the jurisdiction of other university policies and procedures (for example, complaints of unlawful discrimination or harassment, or an appeal related to the merits of a promotion and/or tenure decision); the contents of personnel policies, procedures, rules, regulations, ordinances, and statutes; the routine assignment of university resources (e.g., space, operating funds, parking, etc.); normal actions taken, or recommendations made, by administrators or committee members acting in an official capacity in the grievance process; termination of appointment by removal for just cause, non-reappointment, or abolition of position; or allegations of misconduct in scholarly activities.

Adjudication of disputes on the validity of issues qualifying for consideration under the faculty grievance procedures: If a university administrator rules that an issue does not qualify for the grievance process, the grievant may write to the <u>chair of the Faculty Review Committee</u> within five weekdays of receiving such notification and request a ruling from a special committee

consisting of the president of the Faculty Senate, the chair of the Faculty Senate Committee on Reconciliation, and the chair of the Faculty Review Committee. The special committee considers the matter (including consultations with both parties if deemed necessary) and rules by majority vote on the admissibility of the matter to the grievance process. This special committee is called together by the chair of the Faculty Review Committee, who also sends a written report of the results of the deliberations of the committee to all parties concerned.

### 3.7.5 Particular Concerns and Definitions

Time limits are subject to extension by written agreement of both parties. The grievant and the administrator involved at that particular step of the discussion are the makers of such an agreement. (An agreement form to extend the grievance response time is available on the <u>provost's website</u>.)

Grievances that advance to step four during or close to the summer and/or teaching breaks during the academic year may require some extension of the stipulated time limits. The principals and the chair of the Faculty Review Committee negotiate such an extension. Every effort is made, however, to stay within the stipulated time limits. In case of disagreement, the Faculty Senate president rules on time extension and procedure questions, which might include a hearing conducted by three rather than five panel members, or other recommendations designed to expedite the proceedings while providing peer review of the grievance.

If a faculty member is away from his or her assigned work location at the time he or she discovers the event or action that is the basis for a grievance, the 30-day period during which the grievant must meet with his or her immediate supervisor to initiate the grievance process begins when the faculty member returns to his or her assigned work location. If the date of return causes a delay of such length that the grievance, or its resolution, is not timely, the grievant may submit the grievance in writing to the immediate supervisor (step two), omitting personal meetings until such time as the faculty member returns to his or her assigned work location.

"Weekdays," as used in this procedure, include Monday through Friday only and only when those days are not national, state, or religious holidays relevant to the principals in the grievance.

To protect a grievant from undue pressure in the pursuit of a grievance, if a grievant becomes ill and takes sick leave the grievance process stops until such time as the grievant is able to resume his or her duties. Exceptions to this provision are made at the request of the grievant, but only if the grievant obtains and produces medical certification that proceeding with the grievance will not be harmful to the health of the grievant, or exacerbate the ailment that required taking sick leave.

All costs of legal counsel employed by a grievant are borne by the grievant.

If a grievant is employed away from Blacksburg, and he or she is required to travel away from their duty station in resolution of their grievance, the university pays all travel costs permitted under state regulations.

In the event that a faculty member discovers he or she has a grievance about actions by an administrator above the level of his or her immediate supervisor that directly involve the faculty member, or with actions by an administrator not in his or her department that directly involve the faculty member, the grievant initiates the grievance process by seeking the mediation of his or her immediate supervisor within 30 calendar days of the discovery of the event or action that is the basis for the grievance. If that effort does not resolve the grievance satisfactorily, the grievant, after consulting his or her immediate supervisor, may file the faculty grievance form at the appropriate level or with the appropriate administrative office to initiate response from the administrator perceived as the source of the action causing the grievance. The grievance process then proceeds from that level onward in normal fashion.

A grievance filed by a faculty member concerning an action of the provost is handled by the chair of the Faculty Review Committee and a regular impartial hearing panel, but the findings and recommendations of the hearing panel are sent to the president of the university for his or her ruling, rather than to the provost. A grievance filed by a faculty member concerning an action of the president of the university is dealt with by a special panel appointed by the provost in consultation with the president of the Faculty Senate.

Any final resolution of a grievance must be consonant with the laws of the Commonwealth of Virginia and university policy.

Once a grievance is resolved, either to the satisfaction of the grievant, or if not to the satisfaction of the grievant, by the action of the provost in consonance with the hearing panel recommendations, or by the ruling of the president, that specific grievance is closed and may not be made the subject of another grievance.

## 3.7.6 Overview of the Formal Grievance Process for Tenured and Tenure-Track Faculty

Below is an *abbreviated* overview of the grievance process and deadlines. Refer to section 3.7.2, "The Formal Grievance Procedure," for specific details and options available in each step of the grievance process.

Time limits are subject to extension by written agreement of both parties. The grievant and the administrator involved at that particular step of the discussion are the makers of such an agreement. (An agreement form to extend the grievance response time is available on the provost's website.)

### Step one:

Within 30 days of event 1a. Grievant meets with immediate supervisor (usually department head).

Within 5 weekdays 1b. Department head provides oral response.

- 1c. If department head's response is satisfactory to grievant, that ends the matter.
- 1d. If department head's response is not satisfactory to grievant, move to step two within 5 weekdays.

Step two: Within 5 weekdays Within 5 weekdays	2a. 2b. 2c. 2d.	Grievant submits written grievance to department head. Department head responds in writing on grievance form. If department head's response is satisfactory to grievant, that ends the matter. If department head's response is not satisfactory to grievant, move to step three within 5 weekdays.
Step three:		
Within 5 weekdays	3a.	Grievant advances grievance form to the second-level administrator (usually dean).
Within 5 weekdays	3b.	Dean meets with grievant; dean may request department head to be present.
W/in 5 weekdays	3c.	Dean responds in writing on grievance form.
	3d.	If dean's written response is satisfactory to grievant, that ends the matter.
	3e.	If dean's written response is not satisfactory to grievant, move to step four within 5 weekdays.
Step four:		
Within 5 weekdays	4a.	Grievant advances grievance form to the provost.
Within 5 weekdays	4b.	Provost acknowledges receipt of grievance and forwards copy to Faculty Senate president to receive recommendation of an impartial hearing panel of the Faculty Review Committee.
Within 5 weekdays	4c.	Faculty Senate president acknowledges in writing to grievant that copy of grievance has been received from provost.
Within 15 weekdays	4d.	Faculty Review Committee chair appoints hearing panel from among Faculty Review Committee members; panel holds its initial meeting with both principals.
Within 45 weekdays	4e.	Hearing panel concludes its work and makes recommendations to provost and grievant.
Within 10 weekdays	4f.	Provost meets with grievant.
Within 10 weekdays	4g.	Provost notifies grievant in writing of his or her decision.
	4f.	If the provost's decision is fully consonant with (or exceeds) the recommendations of the hearing panel, or if it is satisfactory to the grievant even if it differs from the recommendations of the hearing panel, that ends the matter.

## Step five:

Within 20 calendar days 5a. Grievant appeals in writing to university president.

5b. University president's decision is final.

to step five within 20 calendar days.

4h. If the provost's decision is not acceptable to the grievant and not consonant with the recommendations of the hearing panel, move

## 3.8 Study-Research Leave

Study-research leave may be granted to faculty members for research and/or advanced study necessary to enhance the competencies of those faculty members to carry out their obligations to the university.

Full-time tenured faculty members with the rank of associate professor or higher, having accrued a minimum of six years of service, are eligible for study-research leaves. Following a period of study-research leave, an additional six years of full-time service are necessary before a faculty member is eligible for another leave. Requests may be submitted prior to completion of six years of service, but faculty members must have completed the sixth year before the leave period begins. Tenure-track faculty members are not eligible to apply for study-research leave until after tenure has been awarded.

Time spent on study-research leave, educational leave, or leave without pay is not considered in compiling minimum service requirements for further leaves.

As part of the commonwealth's educational leave program, recipients of study-research leaves are provided with partial salary (not to exceed one-half salary). Full employee benefits remain in force while faculty members are on study-research leaves. Calendar year faculty on study-research assignment earn annual leave at a rate of half their normal annual leave earnings.

Instead of a proposal for a leave of a full academic or calendar year, faculty members may propose a sequence of semester leave periods at half-salary over several years, not to exceed totally one academic year (for a faculty member on academic year appointment) or 12 months (for a faculty member on calendar year appointment). If such a sequence of leaves is undertaken, all intervening periods of full-time appointment at Virginia Tech accrue toward the six-year minimum service required before eligibility for another study-research leave or sequence of leaves.

Alternatively, following completion of any study-research leave, faculty members may propose a single semester of study-research leave at half-salary following three years of full-time service to the university.

Recipients of a study-research leave may receive additional compensation from other approved sources up to a total equal to their annual salary from the university. Faculty members may receive the additional half salary from sponsored grants or contracts, resulting in a one-year period at full salary from university sources; the appropriate level of effort must be expended on grant-related activities. They may also obtain additional funds from external sources to cover expenses for travel, research, secretarial assistance, and the purchase of relevant materials. Documentation of all external earnings and expected payments is required and must be reviewed and approved by the department head and provost. Engagement in consulting activities must be consonant with existing university policy.

Requests for study-research leave to be taken in the following academic year should be submitted to the department head or chair by November 1 for consideration by the board of

visitors. Requests are forwarded to the board, subject to recommendation of the department head or chair, dean, and the provost, with the same recognition of the need for effective continuation of the department's program as in the case of educational leaves. Specific leave request due dates are established annually and made available in the Calendar of Important Dates found on the <u>provost's website</u>.

Study-research leave is granted with the expectation that the increased competence of the recipient will redound to the benefit of the university. The faculty member must, therefore, return to full-time service with the university for a minimum of at least one academic year at the end of the approved leave or repay the university the salary received plus interest. If less than this required period of service is met, repayment is required of the *pro rata* portion of the compensation provided by the university during the leave period. The faculty member, before undertaking the leave, must sign a memorandum of agreement to this effect.

Within 60 days of returning to full-time status, the faculty member must send a letter to the provost summarizing his or her accomplishments.

## 3.9 Research Assignment

Research assignment is a special category of study-research leave that is awarded to a tenured academic faculty member for one semester of intensive study or research that increases the quality of the individual's professional stature and future contribution to the university. It may be taken in lieu of an ordinary year-long study-research leave.

Full-time tenured faculty members with the rank of associate professor or higher, having accrued a minimum of six years of service, are eligible for research assignment leaves. Following such a leave, an additional six years of full-time service is necessary before a faculty member may be considered for another research assignment. Requests may be submitted prior to completion of six years of service, but faculty members must have completed the sixth year before the leave period begins. Faculty members on calendar year appointments may take research assignment leave for up to six months. Tenure-track faculty members are not eligible to apply for research assignment leave until after tenure has been awarded.

Approval for research assignment provides the faculty member with full salary and related benefits for the period of the leave; faculty members may not take on additional responsibilities for outside income except as allowed by the university's consulting policy. Modest stipends associated with competitive visiting scholar programs at other institutions, competitive national or international fellowships, the Fulbright Scholar Program, and similar prestigious opportunities to support study and/or scholarly research may be approved where there is clear benefit to the faculty member and the university. Similarly, externally funded reimbursements or allotments for travel, temporary relocation, and other expenses associated with the proposed research assignment may be approved. Documentation of all external earnings and expected payments is required and must be reviewed and approved by the department head and provost. When a faculty member proposes a period of paid employment greater than 50 percent of the annual salary in a corporate or governmental setting, leave without pay or a contract through Intergovernmental Personnel Act may be more appropriate than a research assignment.

The primary privilege of a research assignment is entire relief from teaching and administrative duties for one semester. A secondary privilege is that the assignment may be carried out at any location approved by the dean, although research programs that require facilities, library resources, or collaborations not available at the university are given special consideration.

An application for research assignment should be submitted to the appropriate department head or chair by November 1 of the academic year preceding that in which the assignment will be made. Application forms are available from the <u>provost's website</u>. The application should be in the form of a letter, which includes a detailed description of the proposed research or other scholarly project, the location of that activity, and the relevance of the proposed activity in contributing to the faculty member's own scholarly research program. The department head or chair reviews the application and forwards it with a recommendation to the college dean by mid-November indicating the provisions that will be made to accommodate the faculty member's teaching and advising responsibilities. The dean is expected to weigh fiscal and academic load considerations to assure an equitable distribution of the awards. The dean forwards research assignment requests to the provost by mid-December. The provost reviews the recommendations, communicates with the deans, and announces the results to each candidate, following approval by the board of visitors. Specific leave request due dates are established annually and made available in the Calendar of Important Dates found on the provost's website.

Research assignment is granted with the expectation that the increased competence of the recipient will redound to the benefit of the university. The faculty member must, therefore, return to full-time service with the university for a minimum of at least one academic year at the end of the approved leave. If less than this required period of service is met, repayment is required of the *pro rata* portion of the compensation provided by the university during the leave period. The faculty member, before undertaking the leave, must sign a memorandum of agreement to this effect.

Within 60 days of returning to full-time status, the faculty member must send a letter to the provost summarizing his or her accomplishments.

### 3.10 Modified Duties

The university recognizes the need for all tenured and tenure-track faculty members to balance the commitments of family and work. Special family circumstances, for example, birth or adoption of a child, severe illness of an immediate family member, or even issues of personal health, can cause substantial alterations to one's daily routine, thus creating a need to construct a modified workload and flexible schedule for a period of time.

Since the circumstances may vary widely for faculty members at different stages of their careers and with different family and workload situations, this policy does not prescribe the exact nature of the accommodation. In many cases, it may be a reduction or elimination of a teaching assignment while the faculty member continues to meet ongoing, but more flexible research and graduate student supervision obligations. In general, the commitment is to work with a faculty member to devise a modified workload and schedule that enables the faculty member to remain

an active and productive member of the department. Because there is no reduction in salary, the faculty member is expected to have a set of full-time responsibilities.

An eligible faculty member is encouraged to speak with his or her department head or chair as soon as possible about the need for modified duties to ensure the maximum amount of time for planning. A department chair, in conjunction with the relevant dean, is responsible for working with a faculty member to ensure a fair plan for modified duties is implemented if possible, budgetary constraints are considered, and student or other needs are met. The policy does not create an entitlement if there are legitimate business-related reasons for denying the request. Final decisions about the nature of the modified duties are the responsibility of the department head or chair in consultation with the dean.

Provisions of this policy cannot adequately address all individual circumstances. Sick leave (including disability), leave without pay, or permanent reduction in appointment to part-time status may be options to consider for longer-term or more demanding needs. This policy is not intended to provide release time from teaching for the purpose of allowing additional time for research. Reduction in teaching assignment for research purposes is the prerogative of the department and a function of the university's program of study-research leaves.

Extension of the probationary period (see section 3.4.2.1, "Extending the Tenure Clock") is available to faculty members on tenure-track appointments who are confronted with extenuating personal or family circumstances, or birth or adoption of a child. The extension may be requested as a complement to a request for modified duties. However, the semester of modified duties does not automatically affect the tenure probationary period.

**Eligibility:** Modified duties may be requested by any faculty member in a full-time tenured or tenure-track appointment for the purpose of managing family responsibilities or, in exceptional cases, personal health issues not addressed by sick leave. The policy applies to eligible faculty upon employment.

**Guidelines:** The period of modified duties is one semester, or an equivalent amount of time for those faculty members whose responsibilities are not tied directly to teaching on the academic calendar.

Modification of duties should not result in additional duties during the subsequent semester, e.g., the faculty member should not be asked to make up the released teaching before or after the semester of modified duties. The faculty member cannot be employed by another institution during the period of modified duties, nor can the release time be used for extensive professional travel or other increased professional activities (including consulting) that do not meet the goals of the policy.

Medical documentation is required if the period of modified duties is requested related to a health issue not addressed by sick leave.

A semester of modified duties should be considered in addition to, not as a substitute for, sick leave and family leave available to those giving birth or adopting during the period of the

appointment (i.e., during the academic year for those on academic year appointments, or any time for those on calendar year appointments). There are no work expectations for individuals on approved sick or family leave.

Requests for outside consulting during the period of modified duties are not normally approved.

A faculty member should submit a request for modified duties as early as possible so the department can plan appropriately. The request form is available on the <u>provost's website</u>. The plan of proposed activities is developed in consultation with the department head/chair and the dean. The duties can be department-based, college-based, or a combination thereof.

Subject to available funding, the Office of the Senior Vice President and Provost provides an allotment to the faculty member's unit to replace teaching (or to use in other ways relevant to the duties) that is lost through the granting of a term of modified duties. Additional support from departments and colleges is strongly encouraged, and should be noted in the request.

Approval of the department head or chair, dean, and provost are necessary. If the department head or chair does not support the request, the reasons for denial are provided in writing, and the request is automatically forwarded to the dean for further review.

### **CHAPTER FOUR**

- 4.0 Employment Policies and Procedures for Faculty with Continued Appointment or on the Continued Appointment-Track
- 4.1 University Libraries Faculty with Continued Appointment or on the Continued Appointment-Track

As the primary means through which students and faculty gain access to the storehouse of organized knowledge, the University Libraries perform a unique and indispensable function in the educational process. In this function, faculty members of the University Libraries share many of the professional concerns of their collegiate colleagues. The university recognizes the need to protect the academic freedom of librarians in their responsibility to ensure the availability of information and ideas, no matter how controversial, so that teachers may freely teach and students may freely learn.

Continued appointment is the extra-collegiate equivalent of tenure. Extra-collegiate faculty in the library may hold continued appointment or may be on the continued appointment-track; just as collegiate faculty may be tenured or on the tenure-track (see chapter three). Provisions for term appointments during a probationary period are parallel to those for members of the collegiate faculty. Evaluation for continued appointment (in contrast to term appointment) is made no later than the sixth year of such a probationary period.

A library faculty member with continued appointment will have continued employment until retirement with termination of employment based only on unsatisfactory performance, proof of misconduct, discontinuance or reduction in a segment of the university's research or educational program, or library reorganization because of changing patterns of library service or technological advances.

If a position held by a library faculty member with continued appointment is eliminated or changes to such a degree that the incumbent can no longer fulfill the requirements, every effort will be made to reassign the faculty member to another position. If the position of a library faculty member with continued appointment is terminated, it will not be re-established and refilled within a period of two years unless the appointment has been offered to and declined by the faculty member who was originally displaced.

Tenure awarded to faculty of the University Libraries before July 1, 1983 continues to be recognized.

To instill the highest professional standards in the library faculty, procedures for probationary appointment, continued appointment, and promotion for faculty of the University Libraries, including evaluative criteria, have been developed by the Library Faculty Association and the dean of University Libraries. These procedures are contained in *Procedures on Promotion and Continued Appointment: University Libraries*.

The University Promotion and Continued Appointment Committee reviews recommendations for continued appointment and for promotion in rank above the level of assistant professor, and makes recommendations to the senior vice president and provost. (See section 4.5.4.5, "University Evaluation.")

## 4.2 Virginia Cooperative Extension Faculty with Continued Appointment or on the Continued Appointment-Track

Extension faculty, as defined in section 2.3.3, "Virginia Cooperative Extension Faculty with Continued Appointment or on the Continued Appointment-Track," may or may not hold appointments in the collegiate faculty. They are, nonetheless, subject to high professional standards in the dissemination of knowledge through the Virginia Cooperative Extension programs and in the planning and delivery of educational programs and programs of assistance to industries and local governmental agencies. In these functions, extension faculty share many of the professional concerns of their collegiate colleagues, including the need for the protection of academic freedom in these responsibilities.

Continued appointment is the extra-collegiate equivalent of tenure. Extra-collegiate faculty in extension may hold continued appointment or may be on the continued appointment-track; just as collegiate faculty may be tenured or on the tenure-track (see chapter three). Provisions for term appointments during a probationary period are parallel to those for members of the collegiate faculty. Evaluation for continued appointment (in contrast to term appointment) is made no later than the sixth year of such a probationary period.

An extension faculty member with continued appointment can expect continued employment until retirement with termination of employment based only on unsatisfactory performance, proof of misconduct, discontinuance or reduction in a segment of the university's extension programs, or reorganization because of changing patterns of programming.

If a position held by an extension faculty member with continued appointment is eliminated or changes to such a degree that the incumbent can no longer fulfill the requirements, every effort will be made to reassign the faculty member to another position. If the position of an extension faculty member with continued appointment is terminated, it will not be re-established and refilled within a period of two years unless the appointment has been offered to and declined by the displaced faculty member.

Tenure awarded to any member of the Virginia Cooperative Extension faculty before July 1, 1983 continues to be recognized.

Several relevant programmatic units develop standards and procedures for probationary appointments and continued appointment of extension faculty, including evaluative criteria and procedures for promotion in rank.

The University Promotion and Continued Appointment Committee reviews promotions in rank and/or continued appointments and makes recommendations to the senior vice president and provost.

## 4.3 Library and Extension Faculty Ranks

Specification of faculty rank in the library or extension does not imply a particular rank in any collegiate department. Library and extension faculty may be invited to hold concurrent adjunct status in a collegiate department in order to formally recognize their contributions to the undergraduate or graduate program.

## **4.3.1** Extra-Collegiate Instructor

The rank of extra-collegiate instructor is for either library faculty or for Virginia Cooperative Extension faculty on the continued appointment-track whose positions have been designated for continued appointment-track and who have not completed the terminal degree. Annual appointments may be renewed within the limits of a probationary period. Ordinarily, continued appointment would not be awarded at the instructor rank, although time spent at this rank counts in the probationary period leading to continued appointment. A master's degree or significant professional experience is the minimum expectation for appointment at this rank.

Extra-collegiate extension instructors who complete their terminal degree may be recommended for promotion to assistant professor by the unit chair with the approval of the director, dean, provost, president, and the board of visitors. Extra-collegiate instructors in the library may be recommended for promotion to assistant professor by the dean of University Libraries with the approval of the provost, the president, and the board of visitors.

Promotion of library and extension faculty to the ranks of associate or full professor is conducted in accordance with procedures in section 4.5.4, "Evaluation Procedures for Promotion and Continued Appointment."

#### 4.3.2 Assistant Professor

The rank of assistant professor is the usual rank of initial appointment for faculty on the continued appointment track. Appointment to the rank of assistant professor carries with it professional responsibilities in learning, discovery, and engagement. An assistant professor may be assigned responsibility for teaching graduate courses and for supervising master's theses and doctoral dissertations, as well as serving on graduate student committees. The terminal degree appropriate to the field is expected for appointment to this rank. (Further information regarding appropriate credentials for teaching faculty is found in section 2.5.9, "Faculty Credentialing Guidelines," and on the provost's website.)

#### 4.3.3 Associate Professor

In addition to the requirements for assistant professor, a person appointed as associate professor must have demonstrated substantial professional achievements by evidence of an appropriate combination of outstanding teaching, creative scholarship, and recognized performance in extension, outreach, library, or related academic and professional service.

#### 4.3.4 Professor

In addition to the requirements for associate professor, appointment to the rank of professor is contingent upon national recognition as an outstanding scholar and educator.

## 4.3.5 Emeritus or Emerita Designation

The title of emeritus or emerita is conferred on retired full professors, associate professors, administrative officers, extra-collegiate faculty with continued appointment, and senior extension agents who have given exemplary service to the university, and who are specifically recommended to the board of visitors by the president and approved. Their names are carried in the *University Catalog* until death. University policy 4405, "Emeritus Faculty," provides further guidance for department heads or chairs and retiring faculty members concerning emeriti status and continued involvement with the life of the university.

## 4.4 Procedures for Appointments with Continued Appointment

An offer of faculty appointment with continued appointment may be made with the review and approval of the department head or chair, the divisional promotion and continued appointment committee, the library dean or director of extension, the provost, and the president. Ultimately, final approval rests with the board of visitors.

The dean or director forwards the following information to the provost and president for their consideration and decision: the candidate's application package, including cover letter, *curriculum vitae*, and letters of reference; documentation of the division-level promotion and continued appointment committee's approval of rank and continued appointment, and concurrence of the dean or director with as much supporting evidence as deemed appropriate; and a brief overview of the search itself—how many candidates applied, were interviewed, and what is the compelling case for the candidate.

In general, faculty recruited from a comparable university should be recommended for a position at Virginia Tech at a similar level with continued appointment. If the recommended appointment involves a promotion or the initial awarding of continued appointment, the case must be strongly justified. If an individual is coming from a university with a less extensive research mission, or a university of lesser stature, the case must also be strongly justified.

# 4.4.1 Part-Time Continued Appointment and Continued Appointment-Track Appointments

While continued appointment and continued appointment-track appointments are normally full time, Virginia Tech recognizes the importance of allowing flexibility in the percent employment so that faculty members can better manage the balance between their professional work and family or personal obligations over a defined period of time, or perhaps permanently. This policy is intended to encourage departments to accommodate reasonable requests for part-time appointments; however, part-time appointments are not an entitlement, and requests may be turned down when the faculty member and the department cannot agree upon a workable plan.

Departments continue to advertise for full-time continued appointment or continued appointment-track positions and must have funding for a full-time hire. Advertisements include information about university policies for flexible appointments. If desired, the faculty member requests and negotiates a part-time appointment at or after the point of hire if acceptable to the department.

Continued appointment-track faculty members may request a part-time appointment only for reasons of balancing work and family for the arrival or care of a child, the care of a family member, or for personal circumstances related to the health of the faculty member. In addition, they may request a *term* part-time appointment only, allowing the issue to be revisited on a defined cycle. While such term appointments can be renewed throughout the probationary period, a *permanent* part-time appointment may not be granted until continued appointment is awarded.

Faculty members with continued appointment may request either *term* or *permanent* part-time appointments for reasons stated above, or to balance work at Virginia Tech with professional practice or significant community or public service, for example, a professor who wishes to serve as a consultant in addition to an appointment at Virginia Tech; a professor who wishes to engage in entrepreneurial activity outside of university responsibilities; or a professor who runs for office for a limited term and wishes to reduce the workload at Virginia Tech for that period. Other reasonable justifications may be considered if approved by the department head or chair, library dean or director of extension, and provost.

## **4.4.1.1** Part-Time *Term* Continued Appointment and Continued Appointment-Track Appointments

Part-time continued appointment and continued appointment-track appointments are either *term* or *permanent*. *Term* part-time appointments are in increments from one semester up to two years. During the duration of a part-time term appointment, terms of the appointment are only changed via the agreement of all parties. A term agreement must specify the date on which the faculty member is expected to return to full-time status. Renewal of a term appointment should be negotiated no less than three months before the end of the current term so that the department can plan accordingly. For term part-time appointments, departments are able to use the salary savings to replace the work of the faculty member on part-time appointment.

Only the faculty member may initiate a request for conversion from full-time to part-time appointment. The reasons for the request for a change in the percentage of appointment should be clearly stated. The department chair should make a careful assessment of the needs of the department, and works with the faculty member requesting the part-time appointment to facilitate the request whenever possible. The period for which this part-time appointment is granted shall be clearly stated (renewable terms from one semester up to two years, or permanent).

The written agreement should include a careful and thorough statement of work expectations for the part-time appointment. Generally, faculty members continue to contribute to all areas of responsibility, but with reduced expectations for accomplishment proportional to the fractional appointment. Service responsibilities for faculty members on part-time appointments are generally proportional to their appointments. Faculty members on part-time appointments are not excused from regular departmental, division, or university service because of the part-time appointments.

The written agreement for either an initial appointment or a conversion of a full-time appointment to part-time status and any subsequent renewal requires the approval of the faculty member, department head or chair, library dean or director of extension, and provost.

An initial *term* part-time appointment, either continued appointment or continued appointment track, may be approved to accommodate a dual career hire if funding is not immediately available to support a full-time position, or if the faculty member seeks a part-time appointment consistent with the intent of this policy. The expectation is that the subsequent reappointment, if recommended, is for a full-time position, unless the faculty member requests a renewal of the term part-time appointment in accordance with these guidelines. A part-time appointment created for a dual career hire is approved through the usual approval processes for dual career hires. (See section 2.6.7, "Dual Career Assistance Program.")

## 4.4.1.2 *Permanent* Part-Time Continued Appointment and Continued Appointment-Track Appointments

For *permanent* part-time continued appointment and continued appointment-track appointments with no end date, a return to a full-time appointment is not guaranteed. If holding continued appointment, the faculty member remains entitled to the continued appointment on the part-time basis only. However, an increase in the percent appointment up to full-time may be renegotiated between the faculty member and department head or chair if mutually agreeable and funds are available. The department and the division determine the best way to cover the costs of the work in the case of conversion to a permanent part-time appointment.

Faculty members on part-time appointments, whether term or permanent, retain all rights and responsibilities attendant to their appointment as a continued appointment and continued appointment-track faculty member.

Part-time appointments are made for any fraction 50 percent or greater of a full appointment; faculty members receive proportional salary. Faculty members considering such appointments are strongly encouraged to meet with representatives in the <u>benefits office</u> to gain a clear understanding of the consequences of the change to their benefits. Office and laboratory space may be adjusted for longer term or permanent part-time appointments.

## 4.5 Continued Appointment and Promotion

The university has a tradition of upholding academic freedom. It endorses the "1940 Statement of Principles on Academic Freedom and Tenure" of the American Association of University Professors and the Association of American Colleges (*AAUP Bulletin*, September 1970, pp. 323-326).

Members of the library faculty and extension faculty not holding appointments in a collegiate department may be considered for continued appointment or for promotion in faculty rank in recognition of appropriate professional accomplishments as noted above in sections 4.3.1 - 4.3.4.

### 4.5.1 Continued Appointment Eligibility

Like tenure, continued appointment is granted to protect the academic freedom of the library and extension faculty who are engaged in creating new programs and scholarship. Eligibility for continued appointment consideration is limited to faculty members holding regular faculty appointments of 50 percent to 100 percent in the University Libraries or Virginia Cooperative Extension. Continued appointment is not granted to faculty members with temporary appointments. Individuals holding continued appointment who are subsequently appointed to an administrative position retain the status and privileges of continued appointment.

#### 4.5.2 Probationary Period

The term "probationary period" is applied to the succession of term appointments, which an individual undertakes on a full- or part-time regular faculty appointment, and during which evaluation for reappointment and for an eventual continued appointment takes place. The beginning of the probationary period for faculty members on term appointments is taken as July 1 or August 10 of the calendar year in which their initial full-time appointment begins, depending on whether they are on a calendar year or academic year appointment, regardless of the month in which their services are initiated. (The probationary period for new faculty appointed for spring semester begins the following fall even though the spring contract period officially begins December 25.)

The initial appointment for instructors and assistant professors (or those appointed to higher ranks) without continued appointment is ordinarily for a period of not less than two years. Multiple-year reappointments may be subsequently recommended.

The maximum total period for full-time probationary appointments is six years, unless an approved extension is granted. Decision about continued appointment, if not made earlier, is made in the sixth year of the probationary appointment. If the continued appointment decision made in the sixth year is negative, a one-year terminal appointment is offered.

Pre-continued appointment faculty members may request a term part-time appointment as described in section 4.4.1, "Part-Time Continued Appointment and Continued Appointment-Track Appointments," for reasons of balancing work and family or personal health issues. In such cases, the probationary period is extended proportionately. For example, two years of service at 50 percent counts as one year of full-time service. The term appointment may be renewed. (A permanent part-time appointment may be requested and granted following award of continued appointment.)

In determining the mandatory continued appointment review year for those with partial appointments, general equivalency to full-time appointments is expected, so that approximately five years of full-time equivalent service is expected prior to the mandatory continued

appointment review year if no continued appointment clock extensions have been granted; six years if one year of extension is granted, and seven years if two extensions are granted. (In summing partial years of service, a total resulting in a fraction equal to or less than .5 is rounded down, and a fraction greater than .5 is rounded up.) However, review for continued appointment must occur no later than the tenth year of service, resulting in somewhat less full-time equivalent service (4.5 years) for a faculty member with 50 percent appointment throughout all nine probationary years prior to review. If denied continued appointment following a mandatory review, a one-year terminal appointment is offered.

Faculty members on part-time appointments may request a continued appointment clock extension in accordance with section 4.5.2.1, "Extending the Continued Appointment Clock." (Extensions are granted in one-year increments, not prorated by the part-time appointment percentage.) However, the extension is not approved if it results in a mandatory review date beyond the tenth year.

Up to three years of appropriate service at other accredited American four-year colleges and universities may be credited toward the six-year probationary period, as specified in section 4.5.3, "Guidelines for the Calculation of Prior Service."

A faculty member on probationary appointment who wishes to request a leave of absence consults with the dean or director about the effect of the leave on the probationary period, taking into account the professional development that the leave promises. The request for leave addresses this matter and the provost's approval of the leave request specifies whether the leave is to be included in the probationary period.

Under normal circumstances, division-level promotion and continued appointment committees review pre-continued appointment faculty members twice during the probationary period, usually their second and fourth, or third and fifth, years of service. The timing of the reviews depends upon the nature of the faculty member's discipline and must be clearly indicated in written division-level policies. The terms of offer identifies the initial appointment period. Precontinued appointment reviews may be delayed if there is an approved extension as described below. Changes or variations in the standard review cycle must be are documented in writing.

The initial review for a part-time faculty member should be no later than the third year of service (regardless of percent of employment) to give early feedback on their progress. At least two reviews should be conducted for part-time faculty members during their probationary period; more are recommended. The anticipated schedule for such reviews for reappointment and for the mandatory review for continued appointment should be documented in writing as part of the agreement for the part-time appointment. Changes should be agreed upon by the faculty member and division.

These reviews are substantive and thorough. At a minimum, division-level promotion and continued appointment committees must review the faculty member's relevant annual activity reports, peer evaluations of teaching, and authored materials.

The pre-continued appointment reviews should analyze the faculty member's progress toward promotion and continued appointment and offer guidance regarding future activities and plans. All reviews must be in writing, with the faculty member acknowledging receipt by signing and returning a copy for his or her division-level file. In addition, the promotion and continued appointment committee and the dean or director meet with the faculty member to discuss the review and recommendations. Individual faculty members are also encouraged to seek guidance and mentoring from senior colleagues. Pre-continued appointment faculty members bear responsibility for understanding division-level expectations for promotion and continued appointment and for meeting those expectations.

### 4.5.2.1 Extending the Continued Appointment Clock

A one-year probationary period extension is *automatically* granted to either parent (or both, if both parents are continued appointment-track faculty members) in recognition of the demands of caring for a newborn child or a child under five newly placed for adoption or foster care. The request should be made within a year of the child's arrival in the family.

An extension of the probationary period may also be approved on a discretionary basis for other extenuating non-professional circumstances that have a significant impact on the faculty member's productivity, such as a serious personal illness or major illness of a member of the immediate family. In rare cases, extraordinary professional circumstances not of the faculty member's own making may be acceptable justification for a probationary period extension, for example exceptional delays in providing critical equipment, laboratory renovations, or other elements of the committed start-up package essential to establishing a viable research program.

Faculty members who benefit from this policy are expected to fulfill their normal responsibilities during the probationary period extension unless they are also granted a period of modified duties or unless other arrangements are made.

Probationary period extensions are granted in one-year increments. A cumulative total of two years is normally the maximum probationary period extension for any combination of reasons. Requests should be made within a year of the qualifying event or extenuating circumstance. The provost may approve exceptions to these limitations.

Requests for a probationary period extension are submitted in writing to the department head or chair. (A form is available on the <u>provost's website</u>.) Approval is automatic for new parents. Documentation of medical reasons (other than childbirth or adoption) is required prior to approval; documentation of other extenuating circumstances may also be required. Approvals by the department head or chair, library dean or director of extension, and provost are required for probationary period extensions. The faculty member may appeal denial of the request to the next higher level.

It is very important that all individuals and committees participating in continued appointment reviews understand that any individual who receives a probationary period extension must be held to the same standard—not a higher or more stringent one—to which other candidates without such an extension are held. This is also true in the case where the candidate's dossier is

considered on the original schedule for review. However, in this instance where an approved extension was granted but not utilized, the continued appointment review is not considered mandatory and can be conducted again in the subsequent year without penalty. A probationary extension also normally extends the time frame for each subsequent review and reappointment during the probationary period. For example, an extension granted prior to the fourth year review and reappointment typically delays that review by one year.

#### 4.5.3 Guidelines for the Calculation of Prior Service

At the time of a faculty member's initial appointment to the library or extension, the dean of University Libraries or the director of Virginia Cooperative Extension notifies the new faculty member of his or her standing regarding the continued appointment system. Excepting temporary appointments with limited terms, the faculty appointees are given clear notice of when their appointments will be considered for renewal and, if on the continued appointment-track, when consideration for continued appointment will be given.

In this latter calculation, appropriate full-time service in another accredited four-year American college or university is credited toward probationary service at Virginia Tech only if the appointed faculty member requests such credit.

In such a request, all prior service is presented if undertaken after the faculty member completes the terminal degree appropriate to the field. A maximum of three years may be credited toward probationary service at Virginia Tech. The request must be made in writing within one year of the initial appointment. The specification of credit for prior service toward the probationary period is subject to the approval of the provost on the recommendation of the dean of University Libraries or the director of Virginia Cooperative Extension.

## 4.5.4 Evaluation Procedures for Promotion and Continued Appointment

The evaluation of candidates for continued appointment parallels the process for tenure consideration for collegiate faculty, and incorporates the same, or similar, elements of procedure whenever relevant or reasonable. Given the small number of faculty members on the continued appointment-track, their dossiers are reviewed at two levels (rather than three as required for collegiate faculty): first by the library or extension promotion and continued appointment committee and dean of University Libraries or director of Virginia Cooperative Extension, and second by the University Promotion and Continued Appointment Committee and the provost.

Although some participants in the review process may serve at more than one level—for example a departmental committee member may also serve on the divisional committee—participants may only vote once on a case. A faculty member may not serve on any committee that is evaluating a spouse, family member, or other individual with whom the faculty member has a close personal relationship (see section 2.16.2, "Potential Conflicts Involving Spouses and Immediate Family Members").

Because the job descriptions and responsibilities of the candidates being considered do not conform to a single pattern or norm, it is not possible to set forth a statement of criteria with

reference to which all recommendations for promotion or continued appointment must be made. Nevertheless, members of the general faculty seeking continued appointment or promotion in faculty rank are generally expected to have records of outstanding accomplishment in an appropriate combination of the following categories:

- Professional responsibilities: Carrying out the responsibilities of the position within the organizational unit by effective staff work, display of leadership, and a high degree of initiative.
- Research and scholarly activities: Publishing in journals, presenting papers at professional meetings, organizing or chairing sessions at professional meetings, and carrying out instructional responsibilities or graduate student advising.
- University activities: Participating in the conduct of the activities of the administrative unit and the university. Such service takes innumerable forms, including serving on committees or in faculty governance positions, or participating in seminars or conferences.
- External activities: Participating in local, state, regional, and national professional associations. Such participation includes activities such as holding office, serving on committees, conducting workshops, serving on panels, and attending conferences, conventions, or meetings.
- Awards and honors: Receiving awards, grants, and honorary titles or being selected for membership in honorary societies.

Activities and accomplishments in other appropriate areas, beyond these five, may be included in dossiers and are considered.

Library and extension faculty are expected to develop within this framework the performance criteria that are most relevant to the responsibilities of those units. These criteria serve both as an aid to faculty development and as a set of measures that the University Promotion and Continued Appointment Committee may apply.

Promotion to a higher rank or an award of continued appointment may be granted to faculty members on a regular faculty appointment who have demonstrated outstanding accomplishments in an appropriate combination of learning, discovery, engagement, and other professional activities. Every faculty member should maintain a current *curriculum vitae*, with copies filed in the library department or extension unit, as appropriate. The *curriculum vitae* together with annual reports, student or client evaluations, reprints of publications, reference letters, and other similar documents comprise a dossier, which furnishes the principal basis for promotion and continued appointment decisions.

The criteria by which faculty with part-time appointments are evaluated for continued appointment does not differ from the criteria by which full-time faculty are evaluated. Promotion and continued appointment committees consider years of full-time equivalent service when reaching decisions, excluding any approved probationary period extensions granted under the extending the continued appointment clock policy.

Besides consideration of specific professional criteria, evaluation for promotion or continued appointment should consider the candidate's integrity, professional conduct, and ethics. To the

extent that such considerations are significant factors in reaching a negative recommendation, they should be documented as part of the formal review process.

#### 4.5.4.1 Division-Level Evaluation

The University Libraries and extension division each have committees with appropriate faculty representation to evaluate candidates for promotion and/or continued appointment. They make recommendations to the dean of University Libraries or to the director of Virginia Cooperative Extension. The dean or director may chair their respective committees or remain separate from the committee's deliberations and subsequently receive its recommendations. (See guidelines in section 4.5.4.2, "Composition of Library and Extension Division-Level Committees.")

These promotion and continued appointment committees review the cases of candidates for promotion and/or continued appointment, including those faculty members in the final probationary year. The dean or director furnishes the committee with a dossier for each candidate.

## 4.5.4.2 Composition of Library and Extension Division-Level Committees

Rules governing eligibility and selection of the library or extension review committee members, and operating guidelines for the committees' deliberations must be documented in written division-level policies, formally approved by the faculty.

- 1. The library and extension divisions each determine who is eligible to serve on committees from among faculty members with continued appointment.
- 2. The committee may include department heads or district directors; however, these members may not vote on cases from their departments or districts since each has already had an opportunity to vote or make a recommendation on those candidates.
- 3. If possible, some significant element of faculty choice should be a part of the committee selection procedure. Where small numbers make an election process impractical, the dean or director appoints the representative.
- 4. If library department heads or extension district directors serve on the review committees, their total number is less than that of other faculty members.
- 5. Committee appointments should be staggered to assure continuity from one year's deliberation to the next. If possible, members should not serve more than two successive terms.
- 6. Selection of the committee chair is determined in accordance with policies approved by the faculty.
- 7. The dean may be present at the library promotion and continued appointment committee deliberations, and the director may be present at the extension promotion and continued

appointment committee deliberations. Each serves in an advisory capacity to the committee to assure compliance with university procedures and fairness and equity of treatment of candidates. The dean and director do not vote on committee recommendations, but provide a separate recommendation to the provost.

8. Faculty members appointed to serve on the university-level promotion and continued appointment committee are encouraged to observe their division-level deliberations to better prepare for their roles, but should not participate or attempt to influence the division-level recommendations.

## 4.5.4.3 Procedures and Recommendations of Library and Extension Division-Level Committees

The committee makes a recommendation on each candidate to the dean of University Libraries or director of Virginia Cooperative Extension, including a written evaluation that assesses the quality of the candidate's performance in each relevant area. The division of the vote is conveyed to the University Promotion and Continued Appointment Committee and provost, but must otherwise remain confidential outside the committee. In the absence of a unanimous recommendation, a minority report may be included. Whenever the dean or director does not concur with the committee's recommendation, the committee is so notified.

Evaluation for continued appointment is mandated in the sixth year of probationary service unless the faculty member has given written notice of resignation from the faculty. If the review committee feels that the faculty member's record does not warrant a continued appointment, there is an automatic review of the candidate's dossier by the dean of University Libraries or the director of Virginia Cooperative Extension. If the dean or director concurs, the faculty member is notified by the dean or director, in writing, of the decision and the specific reasons for it.

The review committees may ask the candidate to appear before the committee to present additional information or clarification of recommendations.

# 4.5.4.4 Review and Recommendations by the Dean of University Libraries and Director of Virginia Cooperative Extension

The dean of University Libraries and the director of Virginia Cooperative Extension send forward to the provost the full dossier of every candidate for whom there is a positive recommendation from either the division-level review committees or the dean/director, or both. The dean and director prepare separate letters of recommendation to be forwarded with the dossiers from their division. Whenever the dean or director does not concur with their division-level committee's recommendation, the committee is so notified.

The dossiers that the dean and director send to the provost are accompanied by a statement describing the formation and procedures of the review committee and a summary of the number of candidates considered by the division in each category. The division of the vote is conveyed to the university-level committee and provost, but must otherwise remain confidential.

## 4.5.4.5 University Evaluation

Promotions and continued appointments are considered by the University Promotion and Continued Appointment Committee.

**Composition of the University Committee**: On recommendation of the Commission on Faculty Affairs, the University Council approved the following guidelines on formation and procedures of the university committee:

- This committee includes as members the provost or his or her designee; the dean of the University Libraries; the director of Virginia Cooperative Extension; and four faculty members with continued appointment—two each from the library faculty and the extension faculty.
- The provost asks for nominations to this committee from the library faculty and the director of Virginia Cooperative Extension. Where possible, some significant element of faculty choice should be part of the selection procedure.
- The faculty members of the committee hold staggered terms of three years; the provost makes the committee appointments.
- The provost or his or her designee chairs the committee but does not vote.
- All other members of the committee hold voting privileges. Regardless of the size of the
  committee, the faculty must always have at least a majority of the potential votes. Consistent
  with the principle that participants at all levels of the promotion and continued appointment
  review process vote only once on an individual case, the dean and director do not vote on
  cases from their own division. Similarly, faculty members serving on the university
  committee do not vote on any case they previously voted on, should this circumstance occur.
- All voting within the committee should be by written secret ballot; the division of any ballot must remain confidential.

Committee Procedures and Recommendations: The University Promotion and Continued Appointment Committee reviews the qualifications of the candidates recommended for promotion and/or continued appointment by the dean of University Libraries and the director of Virginia Cooperative Extension. It also reviews those cases in which the dean or director do not concur with their division-level committee's positive recommendation. (University-level review of a case with differing recommendations from the dean or director and the division-level committee is automatic and does not require an appeal.) The purpose of the reviews is to verify that the recommendations are consistent with the evidence, reflecting university standards, and that they are consistent with university objectives, programmatic plans, and budgetary constraints.

The committee makes a recommendation on each candidate to the provost. The provost makes recommendations to the president, informing the university committee of those recommendations, including the basis for any non-concurrence with the university committee recommendations. The provost informs the president of any variation between the provost's recommendations and those of the university committee.

The president makes recommendations to the board of visitors, with the board of visitors being responsible for the final decision.

The provost notifies the dean and director of any negative decision reached by the provost, the president, or the board of visitors. The dean or director, when notifying the faculty member in writing, notes appeal options.

## 4.5.5 Appeals of Decisions on Reappointment, Continued Appointment, or Promotion

A faculty member who is notified of a negative decision following evaluation for a term reappointment during the probationary period, for continued appointment, or for promotion, and who believes that the decision was improperly or unfairly determined may appeal for review of the decision under conditions and procedures specified in this section.

Such an appeal must be filed, in writing, within 14 calendar days of formal notification of the decision, which shall make reference to appeal procedures. The appeal can only be based on grounds that certain relevant information was not provided or considered in the decision, or that the decision was influenced by improper consideration. The written appeal must specify the grounds and the basis for such an allegation.

Although the provisions for appeal described below are designed to give faculty members protection against capricious or arbitrary decisions, the faculty member who believes that these procedures have been improperly followed may, at any point, seek advice from the <u>Faculty Senate Committee on Reconciliation</u> and/or file a grievance in accordance with the grievance procedure in section 4.8, "Faculty Grievance Policy and Procedures."

#### 4.5.5.1 Probationary Reappointment

Faculty members on probationary term appointments should make no presumption of reappointment. Procedures for term reappointment or the granting of continued appointment for members of the library faculty or extension faculty are developed in those units. A decision for non-reappointment to a term appointment, based primarily on performance evaluation, is final if reached by the division-level committee and is sustained by the dean of University Libraries or the director of Virginia Cooperative Extension, as appropriate. Notice of non-reappointment is furnished according to the schedule in section 2.11, "Retirement, Resignation, and Non-Reappointment."

If the non-reappointment decision is reached by the dean or director in contradiction to the recommendation of the division-level committee, the faculty member may request that the non-reappointment decision be reviewed by the provost for a final decision.

The faculty member presents the appeal in writing as specified in section 4.5.5, "Appeals of Decisions on Reappointment, Continued Appointment, or Promotion." The provost may ask the University Promotion and Continued Appointment Committee to review the case and make recommendation as an aid to that decision.

## 4.5.5.2 Continued Appointment Decision

Occasionally faculty members are evaluated for continued appointment during the probationary period, but before the final probationary year. In such a case, there is no recourse to appeal or review of a negative decision, at whatever level it is reached, because of the certainty that the evaluation will be undertaken again within a limited time.

If a faculty member is denied continued appointment in a mandatory review by both the division-level committee and the dean of University Libraries or director of Virginia Cooperative Extension, the faculty member may appeal the negative decision in writing in accordance with provisions of this section. The appeal is submitted to the provost for review by the university committee, which shall make a recommendation to the provost for a final decision. No further appeal is provided. The university committee may choose to hear oral arguments. Substantive *procedural* violations may be addressed through the grievance process described in section 4.8, "Faculty Grievance Policy and Procedures."

Should the university committee find reason to believe that the review committee's evaluation was biased or was significantly influenced by improper considerations, the university committee may request that the dean/director form a new *ad hoc* review committee. The *ad hoc* committee makes recommendation to the university committee that requested its formation. The university committee then makes a recommendation to the provost.

Should the provost not concur with a positive recommendation from the University Promotion and Continued Appointment Committee, whether that recommendation culminates a normal review or an appeal, the faculty member is so notified in writing of the specific reason for the decision. The faculty member may appeal to the <u>Faculty Review Committee</u>. That committee investigates the case and, if the differences cannot be reconciled, makes a recommendation to the president on the matter. The president's decision is final.

#### 4.5.5.3 Promotion Decision

There is no specification for minimum or maximum time of service in any rank. A faculty member may request at any time a consideration for promotion in rank if the review committee has not chosen to undertake such an evaluation. However, appeal of a negative promotion decision is provided only if the faculty member has been in rank for at least six years and if the faculty member has formally requested, in writing, consideration for promotion in a previous year. Candidates for promotion who have been denied by both the division-level review committee and the dean of University Libraries or director of Virginia Cooperative Extension may appeal to the provost, who asks the University Promotion and Continued Appointment Committee to consider the appeal. The faculty member presents the appeal in writing as specified in section 4.5.5, "Appeals of Decisions on Reappointment, Continued Appointment, or Promotion."

The committee makes recommendation to the provost. If the university committee and the provost concur with the negative decision, the decision is final; if not, the president makes a final decision.

## 4.6 Annual Evaluation and Post-Continued Appointment Review

## 4.6.1 Annual Evaluation and Salary Adjustments

All departments are required to have written guidelines outlining the process and criteria used in faculty evaluations. The adoption of such guidelines promotes consistency and transparency in this important aspect of faculty life.

Every faculty member's professional performance is evaluated annually and written feedback is provided separately from confirmation of any merit adjustments. The process begins with submission of a Faculty Activity Report (FAR). All non-temporary faculty members must submit a FAR annually. These reports form part of the basis for performance evaluations, awarding merit adjustments, and promotion, continued appointment, and post-continued appointment reviews.

The dean and director are responsible for conducting annual faculty evaluations, either independently or in consultation with an appropriately charged committee in accordance with divisional procedures. All evaluations must be in writing and include a discussion of contributions and accomplishments in all areas of the faculty member's responsibilities, comments on the faculty member's plans and goals, and any recommendations for improvement or change. Faculty members should receive their written evaluations within 90 days of submission of required materials, and they acknowledge receipt by signing and returning a copy for their divisional file, or the electronic equivalent. Acknowledging receipt of the evaluation does not imply agreement with it. If a faculty member substantially disagrees with the evaluation, that member may submit a written response to the dean or director for inclusion in his or her personnel file.

In addition to their annual evaluation letters, all pre-continued appointment faculty members receive at least two thorough reviews during the normal six-year probationary period and written feedback on their progress toward continued appointment by their divisional promotion and continued appointment committee prior to reappointment in accordance with guidance included in section 4.5.2, "Probationary Period."

Faculty with part-time appointments are reviewed on the normal annual review cycle. For purposes of annual review, the fraction of the appointment must be taken into account when considering the appropriate level of achievement in that year.

Salary adjustments are based on merit; they are not automatic. Recommendations for salary adjustments originate with the dean or director and are reviewed by the provost and the president. Because salary adjustments are determined administratively on an annual basis and based significantly on the quality of the faculty member's response to assigned responsibility, they do not necessarily reflect an accurate measure of the full scope of the faculty member's professional development as evaluated by relevant committees in the continued appointment and promotion process.

The salary adjustments of continuing faculty members are approved by the board of visitors, and each faculty member is informed in writing of the board's action as soon as possible. (See section 2.4, "Faculty Compensation Plan.")

### 4.6.2 Unsatisfactory Performance

Failure to meet the minimal obligations and standards the division has stipulated for its faculty results in an "unsatisfactory" rating. Written notification of an unsatisfactory rating and the considerations upon which it was based is given to the faculty member, with a copy to the provost. A single unsatisfactory evaluation indicates a serious problem, which should prompt remedial action. Faculty members may respond in writing with a letter to the dean or director for inclusion in their personnel file, or they may seek redress through either the reconciliation or grievance procedures. Two successive annual ratings of unsatisfactory performance for a faculty member with continued appointment results in a post-continued appointment review.

#### 4.6.3 Divisional Minimal Standards

The University Libraries and extension divisions shall develop, maintain, and publish a statement of minimal standards for satisfactory faculty performance using the following process. Standards should be written with the participation of faculty in the division and approved by a vote of the continued appointment-track faculty in the division. Standards developed and approved by division-level promotion and continued appointment committee and the dean or director are then reviewed and approved by the provost. Once approved, the standards are published and available to all faculty in the division. Revisions of divisional standards also follow these procedures.

The following guidance is provided for the development of divisional minimal standards:

- Divisions should carefully assess and state the overall standards of professional performance
  and contribution they consider minimally acceptable for continued appointment faculty. Each
  division's evaluation mechanism should allow a distinction between performance that is
  deficient in one or more areas requiring improvement, and performance that is so seriously
  deficient as to merit the formal designation, "unsatisfactory."
- Divisional standards should embrace the entire scope of faculty contributions. Expectations
  recognize differences in faculty assignments within the same department or unit. Divisional
  standards should typically address the individual's skill, effort, and effectiveness in
  contributing to all aspects of the instructional mission; the individual's activity in and
  contributions to the discipline; the individual's contributions to the collective life of the
  division and university; and the individual's activity in and contributions to the university's
  outreach mission.
- Divisional statements should affirm support for the basic principles of academic freedom and should express tolerance for minority opinions, dissent from professional orthodoxies, and honest and civil disagreement with administrative actions.
- Departmental statements should include the expectation that faculty with continued appointment will adhere to the standards of conduct and ethical behavior as stated in the *Faculty Handbook* and/or promulgated through other official channels.

## 4.6.4 Post-Continued Appointment Review

Nothing in this section should be interpreted as abridging the university's right to proceed directly to dismissal for cause as defined in section 4.7.3, "Dismissal for Cause," or the right of individual faculty members to pursue existing mechanisms of reconciliation and redress.

A post-continued appointment review is mandatory whenever a faculty member with continued appointment receives two consecutive annual evaluations of unsatisfactory performance. Annual reviews for years spent on leave without pay are disregarded for the purpose of this calculation. The divisional promotion and continued appointment committee conducts the review, unless the same committee was involved in the original unsatisfactory annual evaluations. In this case, the division elects a committee to carry out the review function.

Upon recommendation of the dean or director, a post-continued appointment review may be waived or postponed if there are extenuating circumstances (such as health problems).

The purpose of a post-continued appointment review is to focus the perspective of faculty peers on the full scope of a faculty member's professional competence, performance, and contributions to the division and university mission and priorities.

The faculty member has the both the right and the obligation to provide a dossier with all documents, materials, and statements he or she believes are relevant and necessary for the review. Ordinarily, such a dossier includes at least the following: an up-to-date *curriculum vitae*, the past two or more faculty activity reports, teaching or client assessments, if any, and a description of activities and accomplishments since the last faculty activity report. The faculty member is given a period of no less than four weeks to assemble the dossier for the committee. The dean or director supplies the review committee with the last two annual evaluations, all materials that were considered in those evaluations, any further materials deemed relevant, and other materials the committee requests. Copies of all materials supplied to the committee are given to the faculty member. The faculty member has the right to provide a written rebuttal of evidence provided by the dean or director.

The committee weighs the faculty member's contributions to the discipline, the division, and the university through learning, discovery, and engagement. The burden of proving unsatisfactory performance is on the university. The committee prepares a summary of its findings and makes a recommendation to the dean or director and provost. Final action and notification of the faculty member is the responsibility of the dean or director, with the concurrence of the provost.

The review may result in one of the following outcomes:

**Certification of satisfactory performance**: The committee may conclude that the faculty member's competence and professional contributions are satisfactory to meet the division's minimal expectations, thus failing to sustain the assessment of the dean or director. The review is then complete. An unsatisfactory rating in any subsequent year is counted as the first in any future sequence.

**Certification of deficiencies**: The committee may concur that the faculty member's competence and/or professional contributions are unsatisfactory to meet the division's minimal expectations. The committee may recommend dismissal for cause, a sanction other than dismissal for cause, or a single period of remediation not to exceed two years.

**Remediation**: If a period of remediation is recommended, the committee specifies in detail the deficiencies it noted, defines specific goals and measurable outcomes the faculty member should achieve, and establishes a timeline for meeting the goals. The dean or director meets with the faculty member at least twice annually to review the individual's progress. The dean or director prepares a summary report for the committee following each meeting and at the end of the specified remediation period, at which time the committee either certifies satisfactory performance or recommends dismissal for cause or a sanction other than dismissal for cause following the procedures described below.

Sanction other than dismissal for cause: A division-level recommendation to impose a severe sanction, as defined in section 4.7.2, "Imposition of a Severe Sanction," is referred to the University Promotion and Continued Appointment Committee, which reviews the case as presented to the division-level committee, provides an opportunity for the faculty member to be heard, and determines whether the recommendation is consistent with the evidence. The university committee may reject, uphold, or modify the specific sanction recommended by the division-level committee. If the division-level committee also recommends imposition of a severe sanction, then the same procedures used for dismissal for cause guide the process. The review conducted by the division-level committee satisfies the requirement in step two for an informal inquiry by an *ad hoc* or standing personnel committee. Thus, in the case of a post-continued appointment review, this step is not repeated.

If a severe sanction is imposed or ultimately rejected, then the post-continued appointment review cycle is considered complete. An unsatisfactory rating in any subsequent year is counted as the first in any future sequence.

**Dismissal for cause**: If dismissal for cause is recommended, the case shall be referred to the division-level promotion and continued appointment committee as described in section 4.5.4.3, "Procedures and Recommendations of Library and Extension Division-Level Committees," which reviews the case as presented to the departmental committee and determines whether the recommendation is consistent with the evidence. If the University Promotion and Continued Appointment Committee upholds the recommendation for dismissal, then the procedures specified in section 4.7.3, "Dismissal for Cause," begin immediately. The division-level committee review satisfies the requirement in section 4.7.3 for an informal inquiry by a standing personnel committee.

#### 4.7 Imposition of a Severe Sanction or Dismissal for Cause

#### 4.7.1 Adequate Cause

Adequate cause for imposition of a severe sanction or dismissal is related, directly and substantially, to the fitness of faculty members in their professional capacity as teachers and

scholars. Imposition of a severe sanction or dismissal will not be used to restrain faculty members in their exercise of academic freedom or other rights of American citizens.<sup>2</sup>

Adequate cause includes: violation of professional ethics (see section 2.7, "Professional Responsibilities and Conduct"); incompetence as determined through post-continued appointment review; willful failure to carry out professional obligations or assigned responsibilities; willful violation of university and/or government policies; falsification of information relating to professional qualifications; inability to perform assigned duties satisfactorily because of incarceration; or personal deficiencies that prevent the satisfactory performance of responsibilities (e.g., dependence on drugs or alcohol).

Reason to consider the imposition of a severe sanction or dismissal for cause is usually determined by a thorough and careful investigation by an appropriately-charged faculty committee (as in the case of allegations of ethical or scholarly misconduct, or through a post-continued appointment review) or by the relevant administrator (for example, the dean or director, compliance and conflict resolution officer, internal auditor, or Virginia Tech Police). Generally, these investigations result in a report of findings; some reports also include a recommendation for sanctions. The report is directed to the relevant administrator for action; it is also shared with the faculty member. Imposition of a severe sanction or initiation of dismissal for cause proceedings, if warranted, follows the procedures set forth below.

## 4.7.2 Imposition of a Severe Sanction

**Definition and Examples**: A severe sanction generally involves a significant loss or penalty to a faculty member such as, but not limited to, a demotion in rank and/or a reduction in salary or suspension without pay for a period not to exceed one year, imposed for unacceptable conduct and/or a serious breach of university policy.

Routine personnel actions such as a recommendation for a below average or no merit increase, conversion from a calendar year to an academic year appointment, reassignment, or removal of an administrative stipend do not constitute "sanctions" within the meaning of this policy. A personnel action such as these may be a valid issue for grievance under procedures defined in the *Faculty Handbook*.

**Process for Imposing a Severe Sanction**: The conduct of a faculty member, although not constituting adequate cause for dismissal, may be sufficiently grave to justify imposition of a severe sanction. Imposition of a severe sanction follows the same procedures as dismissal for cause beginning with step one. If the matter is not resolved at the first step, a standing or *ad hoc* faculty committee conducts an informal inquiry (step two). The requirement for such an informal inquiry is satisfied if the investigation was conducted by an appropriately charged faculty committee (as would be the case with an alleged violation of the ethics or scholarly misconduct policies) and, having determined that in its opinion there is adequate cause for imposing a severe sanction, refers the matter to the administration.

<sup>&</sup>lt;sup>2</sup> The procedures specified follow closely, but differ in occasional detail from, the "1976 Institutional Regulations on Academic Freedom and Tenure" approved by Committee A of the AAUP.

### 4.7.3 Dismissal for Cause

The following procedures apply to faculty members with continued appointment, or for dismissal of a continued appointment-track faculty member before the end of his or her current appointment.

#### Dismissal is preceded by:

- 1. Step one: Discussions between the faculty member, dean or director, and/or provost, looking toward a mutual settlement.
- 2. Step two: Informal inquiry by a standing (or, if necessary, *ad hoc*) faculty committee having concern for personnel matters. This committee attempts to effect an adjustment and, failing to do so, determines whether in its opinion dismissal proceedings should be undertaken, without its opinion being binding on the president's decision whether to proceed.
- 3. Step three: The furnishing by the president (in what follows, the president may delegate the provost to serve instead) of a statement of particular charges, in consultation with the dean or director. The statement of charges is included in a letter to the faculty member indicating the intention to dismiss, with notification of the right to a formal hearing. The faculty member is given a specified reasonable time limit to request a hearing, that time limit is no less than 10 days.

Procedures for conducting a formal hearing, if requested: If a hearing committee is to be established, the president asks the Faculty Senate, through its president, to nominate nine faculty members to serve on the hearing committee. These faculty members should be nominated on the basis of their objectivity, competence, and regard in which they are held in the academic community. They must have no bias or untoward interest in the case and are available at the anticipated time of hearing. The faculty member and the president each have a maximum of two challenges from among the nominees without stated cause. The president then names a five-member hearing committee from the remaining names on the nominated slate. The hearing committee elects its chair.

Pending a final decision on the dismissal, the faculty member is suspended only if immediate harm to him or herself or to others is threatened by continuance. If the president believes such suspension is warranted, consultation takes place with the <u>Reconciliation Committee</u> of the Faculty Senate concerning the propriety, the length, and other conditions of the suspension. Ordinarily, salary continues during such a period of suspension.

The hearing committee may hold joint pre-hearing meetings with both parties to simplify the issues, effect stipulations of facts, provide for the exchange of documentary or other information, and achieve such other appropriate pre-hearing objectives as will make the hearing fair and expeditious.

Notice of hearing of at least 20 days is made in writing. The faculty member may waive appearance at the hearing, instead responding to the charges in writing or otherwise denying the

charges or asserting that the charges do not support a finding of adequate cause. In such a case, the hearing committee evaluates all available evidence and rests its recommendation on the evidence in the record.

The committee, in consultation with the president and the faculty member, exercises its judgment as to whether the hearing is public or private. During the proceedings, the faculty member is permitted to have an academic advisor and legal counsel. At the request of either party or on the initiative of the hearing committee, a representative of an appropriate educational association is permitted to attend the hearing as an observer.

A verbatim record of the hearing is taken.

The burden of proof that adequate cause exists rests with the institution.

The hearing committee grants adjournment to enable either party to investigate evidence about which a valid claim of surprise is made. The faculty member is afforded an opportunity to obtain necessary witnesses and documentary or other evidence. The administration cooperates with the hearing committee in securing witnesses and evidence. The faculty member and administration have the right to confront and cross-examine all witnesses. The committee determines the admissibility of statements of unavailable witnesses and, if possible, provides for interrogatories.

The hearing committee is not bound by strict rules of legal evidence, and may admit any evidence that is of probative value in determining the issues involved. Every possible effort is made to obtain the most reliable evidence available.

The findings of fact and the recommendation are based solely on the hearing record. The president and the faculty member are notified of the recommendation in writing and are given a written copy of the record of the hearing.

If the hearing committee concludes that adequate cause for dismissal has not been established, it so reports to the president. In such a case, the committee may recommend sanctions short of outright dismissal or may recommend no sanctions. If the president rejects the recommendation, the hearing committee and the faculty member are so informed in writing, with reasons, and each is given an opportunity for response.

Appeal to the board of visitors: If the president decides to impose dismissal or other severe sanction, whether that is the recommendation of the hearing committee, the faculty member may request that the full record of the case be submitted to the board of visitors (or a duly constituted committee of the board). The board's review is based on the record of the committee hearing, and it provides opportunity for argument, written or oral or both, by the principals at the hearing or their representatives. If the recommendation of the hearing committee is not sustained, the proceeding returns to the committee with specific objections. The committee then reconsiders, taking into account the stated objections and receiving new evidence if necessary. The board makes a final decision only after study of the committee's reconsideration.

**Notice of Termination/Dismissal**: In cases where gross misconduct is decided, termination is usually immediate. The standard for gross misconduct is behavior so egregious that it evokes condemnation by the academic community generally and is so utterly blameworthy as to make it inappropriate to offer additional notice or severance pay. The first faculty committee that considers the case determines gross misconduct. In cases not involving gross misconduct: (a) a faculty member with continued appointment receives up to one year of salary or notice, and (b) a probationary faculty member receives up to three months salary or notice. These terms of dismissal begin at the date of final notification of dismissal.

#### **4.8** Faculty Grievance Policy and Procedures

The following procedure is provided as the means for resolution of grievances against a supervisor or member(s) of the university administration brought by members of the library and extension faculty with continued appointment or on the continued appointment-track. The <u>Faculty Review Committee</u> of the Faculty Senate conducts the step four hearing if requested.

## **4.8.1** Faculty Reconciliation and Mediation Services

**Informal Dialogue**: It should be possible to resolve most faculty concerns or complaints through informal communication among colleagues working together in the academic enterprise. Accordingly, a faculty member who feels he or she has a grievance is encouraged to take it to his or her immediate supervisor in the normal collegial spirit of problem solving rather than as a confrontation between adversaries.

**Reconciliation**: At the initiation of the grievance procedure, or at any earlier time, the grievant may request the assistance of the <u>Faculty Senate Committee on Reconciliation</u> in fashioning an equitable solution. Contacting the Faculty Senate Committee on Reconciliation is not required in filing a grievance, but it may be useful if the grievant feels that the issue may be amenable to, but will require time for, negotiation; or if the grievant is unsure whether his or her concern is a legitimate issue for a grievance; or if personal relations between the parties involved in the grievance have become strained.

For a potential grievance issue to qualify for consideration by the Faculty Senate Committee on Reconciliation, the grievant must contact the Faculty Senate Committee on Reconciliation within 30 calendar days of the time when the grievant knew or should have known of the event or action that is the basis for the potential grievance, just as if beginning the regular grievance process. If the grievant requests assistance from the Faculty Senate Committee on Reconciliation, that committee must request a postponement of the time limits involved in the grievance procedure while it deals with the case. The request is submitted in writing to the vice provost for faculty affairs by the chair of the Faculty Senate Committee on Reconciliation. Also, the grievant should reach an understanding with the Faculty Senate Committee on Reconciliation of the time frame planned for that committee's work on the case, such time not to exceed 60 calendar days.

Faculty members may also consult the Faculty Senate Committee on Reconciliation about serious disagreements with immediate supervisors or other university administrators concerning

issues that may not be eligible for consideration within the grievance process. In such instances, the committee contacts the relevant administrator to determine if there is an interest and willingness to explore informal resolution of the dispute; it is not necessary to notify the Office of the Senior Vice President and Provost.

Additional information regarding the faculty reconciliation process is available on the <u>provost's</u> website.

**Mediation**: Mediation is a voluntary, confidential process through which trained neutral third persons (mediators) assist people to express their concerns and develop solutions to the dispute in a safe and structured environment. Because mediation is voluntary, both parties must agree to participate in order for mediation to occur. Faculty members and supervisors are encouraged to consider using mediation to resolve disputes between them, or to help address a conflict between a faculty member and another member of the Virginia Tech community.

**Role of Mediators**: Mediators do not make judgments, determine facts, or decide the outcome; instead they facilitate discussion between the participants, who identify the solutions best suited to their situation. No agreement is made unless and until it is acceptable to the participants.

**Requesting Mediation**: Mediation is available at any time, without the filing of a grievance. Additionally, mediation may be requested by any party during the grievance process prior to step four. If, after the initiation of a formal grievance, both parties agree to participate in mediation, the grievance is placed on administrative hold until the mediation process is complete. If the parties come to a resolution of the dispute through mediation, the parties are responsible to each other for ensuring that the provisions of the agreement are followed. In the event that the parties are not able to reach a mutual resolution to the dispute through mediation, the grievant may request that the grievance be reactivated and the process continues.

Mediation differs from faculty reconciliation in that mediators do not engage in fact-finding or in evaluation of decisions. Both mediation and reconciliation, however, are voluntary; no party is required to participate in either process.

To learn more about mediation and other forms of informal conflict resolution processes, contact the <u>conflict resolution program</u> manager in human resources at 540-231-9331.

#### 4.8.2 The Formal Grievance Procedure

If the assistance of the Faculty Senate <u>Committee on Reconciliation</u> is not desired or is not requested; or if that committee determines that it cannot provide assistance in the matter; or if the grievant finds that the length of time the Committee on Reconciliation plans or takes with the case is excessive; or if the grievant is not satisfied with the recommendations of that committee, the grievant may pursue the issue as a formal grievance through the following procedure. Department heads or chairs, deans, directors, and other administrative faculty will cooperate with the grievant in the mechanics of processing the grievance, but the grievant alone is responsible for preparation of his or her case.

- 1. Step one: The grievant must meet with his or her immediate supervisor (normally the department head) within 30 calendar days of the date that grievant knew or should have known of the event or action that is basis for the grievance and orally identifies the grievance and the grievant's concerns. The supervisor provides an oral response to the grievant within five weekdays following the meeting. If the supervisor's response is satisfactory to the grievant, that ends the matter.
- 2. Step two: If a satisfactory resolution of the grievance is not achieved by the immediate supervisor's oral response, the grievant may submit a written statement of the grievance and the relief requested to the immediate supervisor. This statement must be on the faculty grievance form, must define the grievance and the relief requested specifically and precisely, and must be submitted to the immediate supervisor within five weekdays of the time when the grievant received the immediate supervisor's oral response to the first step meeting. Faculty grievance forms are available on the provost's website.

Within five weekdays of receiving the written statement of the grievance, the immediate supervisor, in turn, gives the grievant a written response on the faculty grievance form, citing reasons for action taken or not taken. If the written response of the immediate supervisor is satisfactory to the grievant, that ends the matter.

3. Step three: If the resolution of the grievance proposed in the written response by the immediate supervisor is not acceptable, the grievant may advance the grievance to the next higher level of university administration by checking the appropriate place on the faculty grievance form, signing and sending the form to the next higher administrator within five weekdays of receiving the written response from the immediate supervisor. The next higher level of administration for extra-collegiate faculty in the library and extension is normally the library dean or dean of agriculture and life sciences. The administrator involved at this next level is hereafter referred to as the second-level administrator.

Following receipt of the faculty grievance form, the second-level administrator or designated representative meets with the grievant within five weekdays. The second-level administrator may request the immediate supervisor of the grievant be present; the grievant may similarly request that a representative of his or her choice from among the university faculty be present. Unless the grievant is represented by a member of the faculty who is also a lawyer, the second-level administrator does not have legal counsel present. The second-level administrator gives the grievant a written decision on the faculty grievance form within five weekdays after the meeting, citing reasons for his or her decision. If the second-level administrator's written response to the grievance is satisfactory to the grievant, that ends the matter.

4. Step four: If the resolution of the grievance proposed in the written response from the second-level administrator is not acceptable, the grievant may advance the grievance within five weekdays to the level of the provost, including consideration by an impartial hearing panel of the Faculty Review Committee. A description of the charge and membership of the Faculty Review Committee is included in section 1.4.5.2, "Faculty Review Committee."

Upon receiving the faculty grievance form requesting step four review, the provost, or appropriate designated representative, acknowledges receipt of the grievance within five weekdays and forwards a copy of the <u>Procedures of the Faculty Review Committee</u> to parties in the grievance process. The provost immediately forwards a copy of the grievance to the <u>president of the Faculty Senate</u>, who also writes to the grievant to acknowledge receipt of the grievance within five weekdays of receipt of the faculty grievance form from the provost.

The grievant may petition the provost to bypass the Faculty Review Committee and rule on the grievance. If the provost accepts the request, there is no subsequent opportunity for the grievance to be heard by a hearing panel. The provost's decision, however, may be appealed to the president, as described in step five. If the provost does not accept the petition, the Faculty Review Committee hears the grievance as outlined in these procedures.

• **Hearing Panel**: A hearing panel consists of five members appointed by the chair of the Faculty Review Committee from among the members of the Faculty Review Committee. The chair of the Faculty Review Committee polls all appointees to ensure that they have no conflict of interest in the case. Both parties to the grievance may challenge one of the appointments, if they so desire, without need to state cause, and the chair of the Faculty Review Committee appoints the needed replacement or replacements. Other replacements are made only for cause. The chair of the Faculty Review Committee rules on issues of cause.

To ensure uniformity in practice, the chair of the Faculty Review Committee or his or her designee serves as the non-voting chair of each hearing panel. In the event that the chair of the Faculty Review Committee has a conflict of interest concerning a case, the chair appoints a disinterested third party from among the members of the Faculty Review Committee not already appointed to the hearing panel for the case to serve as chair of the hearing panel.

• **Hearings**: After a hearing panel is appointed, the chair of the Faculty Review Committee requests that each party to the grievance provide relevant documentation to be shared among the parties and the hearing panel. The panel holds its initial hearing with both principals present within 15 weekdays of receipt of the grievance by the Faculty Senate president. If the panel feels it needs to investigate the case further, or requires more information, or desires to hear witnesses, the hearing is adjourned until the panel completes the necessary work or scheduling. The hearing is then reconvened as appropriate.

Each party to the grievance may have a representative present during the sessions of the hearing at which testimony is presented. The representative may speak on their behalf if so requested. Representatives may be legal counsel, if both parties are so represented, but if the grievant does not wish to have legal counsel at a hearing, neither party to the grievance may have legal counsel present.

These impartial panel hearings are administrative functions, not adversarial proceedings. Therefore, if legal counsels are present they must understand that the proceedings do not

follow courtroom or trial procedures and rules. Participation by legal counsel is at the invitation of the parties they represent and is subject to the rulings of the chair of the hearing panel. Detailed procedures followed in hearings are specified in the <u>Procedures of the Faculty Review Committee</u> as approved by the Faculty Senate.

• **Findings and Recommendations**: The hearing panel concludes its work and makes its recommendations within 45 weekdays of receipt of the grievance by the Faculty Senate president. The time limit for consideration may be extended by agreement of both parties.

The hearing panel formulates written findings and recommendations regarding disposition of the grievance and forwards copies to the provost, the grievant, and the chair of the Faculty Review Committee.

- **Provost's Action**: The provost meets with the grievant within 10 weekdays after receiving the findings and recommendations of the hearing panel to discuss the case and advise the grievant about the prospects for disposition of the case. Within 10 weekdays of that meeting, the provost sends to the grievant his or her decision in writing concerning the disposition of the grievance. If the provost's decision is fully consonant with (or exceeds) the recommendations of the hearing panel, or if it is satisfactory to the grievant even if it differs from the recommendations of the hearing panel, that ends the matter.
- 5. Step five: If the provost's decision is not acceptable to the grievant and not consonant with the recommendations of the hearing panel, the grievant may appeal in writing to the president within 20 calendar days. The president acts as he or she sees fit. The president's decision is final.

## 4.8.3 Timeliness of Grievance and Procedural Compliance

A grievance must be brought forward in a timely manner. It is the responsibility of the grievant to initiate the grievance process within 30 calendar days of the time when he or she knew or should have known of the event or action that is the basis for the grievance. The university administration is not required to accept a grievance for processing if the grievant does not meet the 30-day deadline, except in cases of demonstrated good cause.

Scheduled commitments made prior to the time of filing or advancement of a grievance that preclude action by either of the parties to the grievance automatically extend time limits for their duration unless this would be demonstrably harmful to the fair processing of the grievance. In such cases, on written request by the grievant to the appropriate office for that step, the grievance is advanced to the next step in the grievance process.

If the grievant does not follow the time limits specified in the grievance procedure it is assumed that he or she accepted the last proposed resolution as satisfactory. If the grievant desires to advance the grievance after the appropriate specified time limits have lapsed, the administrator who receives the late submission notifies the <u>chair of the Faculty Review Committee</u> in writing, and the chair of the Faculty Review Committee determines if there was good cause for the delay. If so, the grievance proceeds. If not, the process ends with the most recently proposed resolution

in force. The finding on the matter by the chair of the Faculty Review Committee is communicated to both parties in writing.

If either party to a grievance charges the other with procedural violations other than time limit issues, a special committee of the president of the Faculty Senate, the chair of the Faculty Senate Committee on Reconciliation, and the chair of the Faculty Review Committee (or the vice president of the senate if the president is also chair of the Faculty Review Committee) is convened to rule on the question, as in disputes about the validity of issues qualifying for the grievance procedure. The special committee has the following options. It can either find no significant procedural violation occurred, in which case the grievance process continues unaffected, or that a significant procedural violation did occur. If the administrator committed a significant procedural violation, the grievance automatically qualifies for advancement to the next step in the grievance process. If the grievant committed a significant procedural violation, the grievance process ends at that point with the last proposed resolution established as the final disposition of the case.

#### 4.8.4 Valid Issues for Grievance

For this process, a grievance is defined as a complaint by a faculty member alleging a violation, misinterpretation, or incorrect application of a policy, procedure, or practice of the university that directly affects the grievant. Some examples of valid issues for filing a grievance are: improperly or unfairly determined personnel decisions that result in an unsatisfactory annual performance evaluation, unreasonable merit adjustment or salary level, or excessive teaching load/work assignments; substantive violations of promotion and continued appointment procedures (see appeal process in section 4.5.5 , "Appeals of Decisions on Reappointment, Continued Appointment, or Promotion"); reprisals; substantive error in the application of policy; and matters relating to academic freedom.

Issues not open to grievance: While most faculty disputes with the university administration may be dealt with by this grievance policy, the following issues may not be made the subject of a grievance: determination of policy appropriately promulgated by the university administration or the university governance system; those items falling within the jurisdiction of other university policies and procedures (for example, complaints of unlawful discrimination or harassment, or an appeal related to the merits of a promotion and/or continued appointment decision); the contents of personnel policies, procedures, rules, regulations, ordinances, and statutes; the routine assignment of university resources (e.g., space, operating funds, parking, etc.); normal actions taken, or recommendations made, by administrators or committee members acting in an official capacity in the grievance process; termination of appointment by removal for just cause, non-reappointment, or abolition of position; or allegations of misconduct in scholarly activities.

Adjudication of disputes on the validity of issues qualifying for consideration under the faculty grievance procedures: If a university administrator rules that an issue does not qualify for the grievance process, the grievant may write to the chair of the Faculty Review Committee within five weekdays of receiving such notification and request a ruling from a special committee consisting of the president of the Faculty Senate, the chair of the Faculty Senate Committee on Reconciliation, and the chair of the Faculty Review Committee. The special

committee considers the matter (including consultations with both parties if deemed necessary) and rules by majority vote on the admissibility of the matter to the grievance process. This special committee is called together by the chair of the Faculty Review Committee, who also sends a written report of the results of the deliberations of the committee to all parties concerned.

#### 4.8.5 Particular Concerns and Definitions

Time limits are subject to extension by written agreement of both parties. The grievant and the administrator involved at that particular step of the discussion are the makers of such an agreement. (An agreement form to extend the grievance response time is available on the <u>provost's website</u>.)

Grievances that advance to step four during or close to the summer and/or teaching breaks during the academic year may require some extension of the stipulated time limits. The principals and the chair of the Faculty Review Committee negotiate such an extension. Every effort is made, however, to stay within the stipulated time limits. In case of disagreement, the <u>Faculty Senate president</u> rules on time extension and procedure questions, which might include a hearing conducted by three rather than five panel members, or other recommendations designed to expedite the proceedings while providing peer review of the grievance.

If a faculty member is away from his or her assigned work location at the time he or she discovers the event or action that is the basis for a grievance, the 30-day period during which the grievant must meet with his or her immediate supervisor to initiate the grievance process begins when the faculty member returns to his or her assigned work location. If the date of return causes a delay of such length that the grievance, or its resolution, is not timely, the grievant may submit the grievance in writing to the immediate supervisor (step two), omitting personal meetings until such time as the faculty member returns to his or her assigned work location.

"Weekdays," as used in this procedure, include Monday through Friday only and only when those days are not national, state, or religious holidays relevant to the principals in the grievance.

To protect a grievant from undue pressure in the pursuit of a grievance, if a grievant becomes ill and takes sick leave the grievance process stops until such time as the grievant is able to resume his or her duties. Exceptions to this provision are made at the request of the grievant, but only if the grievant obtains and produces medical certification that proceeding with the grievance will not be harmful to the health of the grievant, or exacerbate the ailment that required taking sick leave.

All costs of legal counsel employed by a grievant are borne by the grievant.

If a grievant is employed away from Blacksburg, and he or she is required to travel away from their duty station in resolution of their grievance, the university pays all travel costs permitted under state regulations.

In the event that a faculty member discovers he or she has a grievance about actions by an administrator above the level of his or her immediate supervisor that directly involve the faculty

member, or with actions by an administrator not in his or her department that directly involve the faculty member, the grievant initiates the grievance process by seeking the mediation of his or her immediate supervisor within 30 calendar days of the discovery of the event or action that is the basis for the grievance. If that effort does not resolve the grievance satisfactorily, the grievant, after consulting his or her immediate supervisor, may file the faculty grievance form at the appropriate level or with the appropriate administrative office to initiate response from the administrator perceived as the source of the action causing the grievance. The grievance process then proceeds from that level onward in normal fashion.

A grievance filed by a faculty member concerning an action of the provost is handled by the <u>chair of the Faculty Review Committee</u> and a regular impartial hearing panel, but the findings and recommendations of the hearing panel are sent to the president of the university for his or her ruling, rather than to the provost. A grievance filed by a faculty member concerning an action of the president of the university is dealt with by a special panel appointed by the provost in consultation with the president of the Faculty Senate.

Any final resolution of a grievance must be consonant with the laws of the Commonwealth of Virginia and university policy.

Once a grievance is resolved, either to the satisfaction of the grievant, or if not to the satisfaction of the grievant, by the action of the provost in consonance with the hearing panel recommendations, or by the ruling of the president, that specific grievance is closed and may not be made the subject of another grievance.

# 4.8.6 Overview of the Formal Grievance Process for Faculty with Continued Appointment or on the Continued Appointment-Track

Below is an <u>abbreviated</u> overview of the grievance process and deadlines. Refer to section 4.8.2, "The Formal Grievance Procedure," for specific details and options available in each step of the grievance process.

Time limits are subject to extension by written agreement of both parties. The grievant and the administrator involved at that particular step of the discussion are the makers of such an agreement. (An agreement form to extend the grievance response time is available on the <u>provost's website</u>.)

#### Step one:

Within 30 days of event 1a. Grievant meets with immediate supervisor (usually department or division head).

Within 5 weekdays 1b. Supervisor provides oral response.

- 1c. If supervisor's response is satisfactory to grievant, that ends the matter.
- 1d. If supervisor's response is not satisfactory to grievant, move to step two within 5 weekdays.

Step two:		
Within 5 weekdays	2a.	Grievant submits written grievance to immediate supervisor.
Within 5 weekdays	2b.	Supervisor responds in writing on grievance form.
·	2c.	If supervisor's response is satisfactory to grievant, that ends the
		matter.
	2d.	If supervisor's response is not satisfactory to grievant, move to
		step three within 5 weekdays.
Step three:		
Within 5 weekdays	3a.	Grievant advances grievance form to the second-level
		administrator (usually the library dean or the dean of agriculture
		and life sciences).
Within 5 weekdays	3b.	Dean meets with grievant; dean may request department or
		division head to be present.
Within 5 weekdays	3c.	Dean responds in writing on grievance form.
	3d.	If dean's written response is satisfactory to grievant, that ends the
	_	matter.
	3e.	If dean's written response is not satisfactory to grievant, move to
G. C		step four within 5 weekdays.
Stan toller		
Step four:	4 -	C.:
Within 5 weekdays	4a.	Grievant advances grievance form to the provost.
	4a. 4b.	Provost acknowledges receipt of grievance and forwards copy to
Within 5 weekdays		Provost acknowledges receipt of grievance and forwards copy to Faculty Senate president to receive recommendation of an
Within 5 weekdays Within 5 weekdays	4b.	Provost acknowledges receipt of grievance and forwards copy to Faculty Senate president to receive recommendation of an impartial hearing panel of the Faculty Review Committee.
Within 5 weekdays		Provost acknowledges receipt of grievance and forwards copy to Faculty Senate president to receive recommendation of an impartial hearing panel of the Faculty Review Committee.  Faculty Senate president acknowledges in writing to grievant that
Within 5 weekdays Within 5 weekdays Within 5 weekdays	4b. 4c.	Provost acknowledges receipt of grievance and forwards copy to Faculty Senate president to receive recommendation of an impartial hearing panel of the Faculty Review Committee. Faculty Senate president acknowledges in writing to grievant that copy of grievance has been received from provost.
Within 5 weekdays Within 5 weekdays	4b.	Provost acknowledges receipt of grievance and forwards copy to Faculty Senate president to receive recommendation of an impartial hearing panel of the Faculty Review Committee.  Faculty Senate president acknowledges in writing to grievant that copy of grievance has been received from provost.  Faculty Review Committee chair appoints hearing panel from
Within 5 weekdays Within 5 weekdays Within 5 weekdays	4b. 4c.	Provost acknowledges receipt of grievance and forwards copy to Faculty Senate president to receive recommendation of an impartial hearing panel of the Faculty Review Committee.  Faculty Senate president acknowledges in writing to grievant that copy of grievance has been received from provost.  Faculty Review Committee chair appoints hearing panel from among Faculty Review Committee members; panel holds its
Within 5 weekdays Within 5 weekdays Within 5 weekdays Within 15 weekdays	4b. 4c. 4d.	Provost acknowledges receipt of grievance and forwards copy to Faculty Senate president to receive recommendation of an impartial hearing panel of the Faculty Review Committee. Faculty Senate president acknowledges in writing to grievant that copy of grievance has been received from provost. Faculty Review Committee chair appoints hearing panel from among Faculty Review Committee members; panel holds its initial meeting with both principals.
Within 5 weekdays Within 5 weekdays Within 5 weekdays	4b. 4c.	Provost acknowledges receipt of grievance and forwards copy to Faculty Senate president to receive recommendation of an impartial hearing panel of the Faculty Review Committee. Faculty Senate president acknowledges in writing to grievant that copy of grievance has been received from provost. Faculty Review Committee chair appoints hearing panel from among Faculty Review Committee members; panel holds its initial meeting with both principals. Hearing panel concludes its work and makes recommendations
Within 5 weekdays Within 5 weekdays Within 5 weekdays Within 15 weekdays Within 15 weekdays	4b. 4c. 4d. 4e.	Provost acknowledges receipt of grievance and forwards copy to Faculty Senate president to receive recommendation of an impartial hearing panel of the Faculty Review Committee. Faculty Senate president acknowledges in writing to grievant that copy of grievance has been received from provost. Faculty Review Committee chair appoints hearing panel from among Faculty Review Committee members; panel holds its initial meeting with both principals. Hearing panel concludes its work and makes recommendations to provost and grievant.
Within 5 weekdays Within 5 weekdays Within 5 weekdays Within 15 weekdays Within 45 weekdays Within 10 weekdays	4b. 4c. 4d. 4e. 4f.	Provost acknowledges receipt of grievance and forwards copy to Faculty Senate president to receive recommendation of an impartial hearing panel of the Faculty Review Committee. Faculty Senate president acknowledges in writing to grievant that copy of grievance has been received from provost. Faculty Review Committee chair appoints hearing panel from among Faculty Review Committee members; panel holds its initial meeting with both principals. Hearing panel concludes its work and makes recommendations to provost and grievant.  Provost meets with grievant.
Within 5 weekdays Within 5 weekdays Within 5 weekdays Within 15 weekdays Within 15 weekdays	4b. 4c. 4d. 4e.	Provost acknowledges receipt of grievance and forwards copy to Faculty Senate president to receive recommendation of an impartial hearing panel of the Faculty Review Committee. Faculty Senate president acknowledges in writing to grievant that copy of grievance has been received from provost. Faculty Review Committee chair appoints hearing panel from among Faculty Review Committee members; panel holds its initial meeting with both principals. Hearing panel concludes its work and makes recommendations to provost and grievant.  Provost meets with grievant.  Provost notifies grievant in writing of his or her decision.
Within 5 weekdays Within 5 weekdays Within 5 weekdays Within 15 weekdays Within 45 weekdays Within 10 weekdays	4b. 4c. 4d. 4e. 4f. 4g.	Provost acknowledges receipt of grievance and forwards copy to Faculty Senate president to receive recommendation of an impartial hearing panel of the Faculty Review Committee. Faculty Senate president acknowledges in writing to grievant that copy of grievance has been received from provost. Faculty Review Committee chair appoints hearing panel from among Faculty Review Committee members; panel holds its initial meeting with both principals. Hearing panel concludes its work and makes recommendations to provost and grievant.  Provost meets with grievant.  Provost notifies grievant in writing of his or her decision. If the provost's decision is fully consonant with (or exceeds) the
Within 5 weekdays Within 5 weekdays Within 5 weekdays Within 15 weekdays Within 45 weekdays Within 10 weekdays	4b. 4c. 4d. 4e. 4f. 4g.	Provost acknowledges receipt of grievance and forwards copy to Faculty Senate president to receive recommendation of an impartial hearing panel of the Faculty Review Committee. Faculty Senate president acknowledges in writing to grievant that copy of grievance has been received from provost. Faculty Review Committee chair appoints hearing panel from among Faculty Review Committee members; panel holds its initial meeting with both principals. Hearing panel concludes its work and makes recommendations to provost and grievant.  Provost meets with grievant.  Provost notifies grievant in writing of his or her decision.
Within 5 weekdays Within 5 weekdays Within 5 weekdays Within 15 weekdays Within 45 weekdays Within 10 weekdays	4b. 4c. 4d. 4e. 4f. 4g.	Provost acknowledges receipt of grievance and forwards copy to Faculty Senate president to receive recommendation of an impartial hearing panel of the Faculty Review Committee.  Faculty Senate president acknowledges in writing to grievant that copy of grievance has been received from provost.  Faculty Review Committee chair appoints hearing panel from among Faculty Review Committee members; panel holds its initial meeting with both principals.  Hearing panel concludes its work and makes recommendations to provost and grievant.  Provost meets with grievant.  Provost notifies grievant in writing of his or her decision.  If the provost's decision is fully consonant with (or exceeds) the recommendations of the hearing panel, or if it is satisfactory to
Within 5 weekdays Within 5 weekdays Within 5 weekdays Within 15 weekdays Within 45 weekdays Within 10 weekdays	4b. 4c. 4d. 4e. 4f. 4g.	Provost acknowledges receipt of grievance and forwards copy to Faculty Senate president to receive recommendation of an impartial hearing panel of the Faculty Review Committee.  Faculty Senate president acknowledges in writing to grievant that copy of grievance has been received from provost.  Faculty Review Committee chair appoints hearing panel from among Faculty Review Committee members; panel holds its initial meeting with both principals.  Hearing panel concludes its work and makes recommendations to provost and grievant.  Provost meets with grievant.  Provost notifies grievant in writing of his or her decision.  If the provost's decision is fully consonant with (or exceeds) the recommendations of the hearing panel, or if it is satisfactory to the grievant even if it differs from the recommendations of the

## Step five:

Within 20 calendar days 5a. Grievant appeals in writing to university president.

5b. University president's decision is final.

to step five within 20 calendar days.

consonant with the recommendations of the hearing panel, move

## 4.9 Study-Research Leave

Study-research leave may be granted to faculty members for research and/or advanced study necessary to enhance the competencies of those faculty members to carry out their obligations to the university. Although the purposes of a study-research leave are distinct from those of an educational leave, both are subject to the statute outlined in *Educational Aid for State Employees*.

Full-time faculty members holding continued appointment with significant responsibility for instruction and scholarly productivity, with the rank of assistant professor or higher, having accrued a minimum of six years of service, are eligible for study-research leaves. Following such a leave, an additional six years of full-time service are necessary before a faculty member is eligible for another study-research leave. Requests may be submitted prior to completion of six years of service, but faculty members must have completed the sixth year before the leave period begins. Faculty on the continued appointment-track are not eligible to apply for study-research leave until after continued appointment has been awarded.

Time spent on study-research leave, educational leave, or leave without pay is not considered in compiling minimum service requirements for further leaves.

As part of the commonwealth's educational leave program, recipients of study-research leaves are provided with partial salary (not to exceed one-half salary). Full employee benefits remain in force while faculty members are on study-research leaves. Calendar year faculty on study-research assignment earn annual leave at a rate of half their normal annual leave earnings.

Instead of a proposal for a leave of a full academic or calendar year, faculty members may propose a sequence of semester leave periods at half-salary over several years, not to exceed totally one academic year (for a faculty member on academic year appointment) or 12 months (for a faculty member on calendar year appointment). If such a sequence of leaves is undertaken, all intervening periods of full-time appointment at Virginia Tech accrue toward the six-year minimum service required before eligibility for another study-research leave or sequence of leaves.

Alternatively, following completion of any study-research leave, faculty members may propose a single semester of study-research leave at half-salary following three years of full-time service to the university.

Recipients of a study-research leave may receive additional compensation from other approved sources up to a total equal to their annual salary from the university. Faculty members may receive the additional half salary from sponsored grants or contracts, resulting in a one-year period at full salary from university sources; the appropriate level of effort must be expended on grant-related activities. They may also obtain additional funds from external sources to cover expenses for travel, research, secretarial assistance, and the purchase of relevant materials. Documentation of all external earnings and expected payments is required and must be reviewed and approved by the department head and provost. Engagement in consulting activities must be consonant with existing university policy.

Requests for study-research leave to be taken in the following academic year are submitted to the dean or director by November 1 for consideration by the board of visitors. Requests are forwarded to the board, subject to recommendation of the dean or director and the provost, with the same recognition of the need for effective continuation of the division's program as in the case of educational leaves. Specific leave request due dates are established annually and made available in the Calendar of Important Dates on the <u>provost's website</u>.

Study-research leave is granted with the expectation that the increased competence of the recipient will redound to the benefit of the university. The faculty member must, therefore, return to full-time service with the university for a minimum of at least one academic year at the end of the approved leave or repay the university the salary received plus interest. If less than this required period of service is met, repayment is required of the *pro rata* portion of the compensation provided by the university during the leave period. The faculty member, before undertaking the leave, must sign a memorandum of agreement to this effect.

Within 60 days of returning to full-time status, the faculty member must send a letter to the provost summarizing his or her accomplishments.

### 4.10 Research Assignment

Research assignment is a special category of study-research leave that is awarded to a faculty member with continued appointment for one semester of intensive study or research that increases the quality of the individual's professional stature and future contribution to the university. It may be taken in lieu of an ordinary year-long study-research leave.

Full-time faculty members holding continued appointment with the rank of assistant professor or higher, and having accrued a minimum of six years of service, are eligible for research assignment or study-research leave. Following such a leave, an additional six years of full-time service are necessary before a faculty member may be considered for another research assignment. Requests may be submitted prior to completion of six years of service, but faculty members must have completed the sixth year before the leave period begins. Faculty members on calendar year appointments may take research assignment leave for up to six months. Faculty on the continued appointment-track are not eligible to apply for research assignment leave until after continued appointment has been awarded.

Approval for research assignment provides the faculty member with full salary and related benefits for the period of the leave; faculty members may not take on additional responsibilities for outside income except as allowed by the university's consulting policy. Modest stipends associated with competitive visiting scholar programs at other institutions, competitive national or international fellowships, the Fulbright Scholar Program, and similar prestigious opportunities to support study and/or scholarly research may be approved where there is clear benefit to the faculty member and the university. Similarly, externally funded reimbursements or allotments for travel, temporary relocation, and other expenses associated with the proposed research assignment may be approved. Documentation of all external earnings and expected payments is required and must be reviewed and approved by the department head and provost. When a faculty member proposes a period of paid employment greater than 50 percent of the annual

salary in a corporate or governmental setting, leave without pay or a contract through Intergovernmental Personnel Act may be more appropriate than a research assignment.

The primary privilege of a research assignment is entire relief from teaching, administrative duties, and other faculty duties for one semester. A secondary privilege is that the assignment may be carried out at any location approved by the dean or director, although research programs that require facilities, library resources, or collaborations not available at the university are given special consideration.

An application for research assignment is submitted to the appropriate department head or chair by November 1 of the academic year preceding that in which the assignment will be made. Application forms are available from the <u>provost's website</u>. The application is in the form of a letter, which includes a detailed description of the proposed research or other scholarly project, the location of that activity, and the relevance of the proposed activity in contributing to the faculty member's own scholarly research program. The dean or director reviews the application and forwards it with a recommendation to the provost by mid-December, indicating the provisions that will be made to accommodate the faculty member's responsibilities. The dean or director is expected to weigh fiscal and academic load considerations to assure an equitable distribution of the awards. The provost reviews the recommendations, communicates with the dean or director, and announces the results to each candidate, following approval by the board of visitors. Specific leave request due dates are established annually and made available in the Calendar of Important Dates on the provost's website.

Research assignment is granted with the expectation that the increased competence of the recipient will redound to the benefit of the university. The faculty member must, therefore, return to full-time service with the university for a minimum of at least one academic year at the end of the approved leave. If less than this required period of service is met, repayment is required of the *pro rata* portion of the compensation provided by the university during the leave period. The faculty member, before undertaking the leave, must sign a memorandum of agreement to this effect.

Within 60 days of returning to full-time status, the faculty member must send a letter to the provost summarizing his or her accomplishments.

#### 4.11 Modified Duties

The university recognizes the need for all continued appointment and continued appointment-track faculty members to balance the commitments of family and work. Special family circumstances, for example, birth or adoption of a child, severe illness of an immediate family member, or even issues of personal health, can cause substantial alterations to one's daily routine, thus creating a need to construct a modified workload and flexible schedule for a period of time.

Since the circumstances may vary widely for faculty members at different stages of their careers and with different family and workload situations, this policy does not prescribe the exact nature of the accommodation. In many cases, it may be a reduction or elimination of a teaching

assignment while the faculty member continues to meet ongoing, but more flexible research and graduate student supervision obligations. In general, the commitment is to work with a faculty member to devise a modified workload and schedule that enables the faculty member to remain an active and productive member of the department. Because there is no reduction in salary, the faculty member is expected to have a set of full-time responsibilities.

An eligible faculty member is encouraged to speak with his or her department head or chair as soon as possible about the need for modified duties to ensure the maximum amount of time for planning. A department chair, in conjunction with the library dean or director of extension, is responsible for working with a faculty member to ensure a fair plan for modified duties is implemented if possible, budgetary constraints are considered, and student or other needs are met. The policy does not create an entitlement if there are legitimate business-related reasons for denying the request. Final decisions about the nature of the modified duties are the responsibility of the department head in consultation with the dean/director.

Provisions of this policy cannot adequately address all individual circumstances. Sick leave (including disability), leave without pay, or permanent reduction in appointment to part-time status may be options to consider for longer-term or more demanding needs. This policy is not intended to provide release time from teaching for the purpose of allowing additional time for research. Reduction in teaching assignment for research purposes is the prerogative of the division and a function of the university's program of study-research leaves.

Extension of the probationary period (see section 4.5.2.1, "Extending the Continued Appointment Clock") is available to faculty members on continued appointment-track appointments who are confronted with extenuating personal or family circumstances, or birth or adoption of a child. The extension may be requested as a complement to a request for modified duties. However, the semester of modified duties does not automatically affect the continued appointment probationary period.

**Eligibility**: Modified duties may be requested by any faculty member in a full-time continued appointment or continued appointment-track appointment for the purpose of managing family responsibilities or, in exceptional cases, personal health issues not addressed by sick leave. The policy applies to eligible faculty upon employment.

**Guidelines**: The period of modified duties is one semester, or an equivalent amount of time for those faculty members whose responsibilities are not tied directly to teaching on the academic calendar.

Modification of duties should not result in additional duties during the subsequent semester, e.g., the faculty member should not be asked to make up the released teaching before or after the semester of modified duties. The faculty member cannot be employed by another institution during the period of modified duties, nor can the release time be used for extensive professional travel or other increased professional activities (including consulting) that do not meet the goals of the policy.

Medical documentation is required if the period of modified duties is requested related to a health issue not addressed by sick leave.

A semester of modified duties should be considered *in addition to*, not as a substitute for, sick leave and family leave available to those giving birth or adopting during the period of the appointment (i.e., during the academic year for those on academic year appointments, or any time for those on calendar year appointments). There are no work expectations for individuals on approved sick or family leave.

Requests for outside consulting during the period of modified duties are not normally approved.

A faculty member should submit a request for modified duties as early as possible so the department can plan appropriately. The request form is available on the <u>provost's website</u>. The plan of proposed activities is developed in consultation with the department head/chair and the library dean or director of extension. The duties can be department-based, division-based, or a combination thereof.

Subject to available funding, the Office of the Senior Vice President and Provost provides an allotment to the faculty member's unit to replace teaching (or to use in other ways relevant to the duties) that is lost through the granting of a term of modified duties. Additional support from departments and divisions is strongly encouraged, and should be noted in the request.

Approval of the department head or chair, library dean or director of extension, and provost are necessary. If the department head or chair does not support the request, the reasons for denial are provided in writing, and the request is automatically forwarded to the dean for further review.

## 4.12 Consulting Activities for Virginia Cooperative Extension Faculty

Consistent with the university's policy and procedures on consulting activities, additional restrictions may be imposed on the consulting activity of Virginia Cooperative Extension faculty members. These restrictions are imposed to give further assurance that consulting approval is not granted for assistance that is the normal responsibility of faculty members within extension.

It is recognized that the outreach responsibilities of extension are broad and, thus, program assistance parameters are difficult to define. Consequently, the following procedures are designed to provide judgmental decisions by appropriate supervisory staff for consulting requests in ambiguous areas of program responsibilities.

The Request to Engage in External Activity Form 13010A is sent to the department head, chair, or immediate supervisor along with a letter outlining the nature of the consulting activity and why it falls outside the normal responsibilities of extension. (The form is available on the <u>provost's website</u>.) Typically, consulting activities do not involve university sponsorship.

The department head, chair, or immediate supervisor reviews the Request to Engage in External Activity Form 13010A and either approves or disapproves it. If approval is granted, the request is sent to the college dean for approval. The director of Virginia Cooperative Extension grants

final approval. If disapproval is exercised at any level, the request is sent back through the department head or supervisor, to the faculty member along with an explanation for the action.

Decisions are based upon, but not limited to, the following: consistency with guidelines stipulated in section 2.14, "Consulting and Outside Employment"; whether the area of consulting is found to be within or outside normal extension responsibilities; and whether the time required falls within the number of consulting days allowed.

#### **CHAPTER FIVE**

# 5.0 Employment Policies and Procedures for Non-Tenure-Track Instructional Faculty

Non-tenure-track faculty members fill critical roles in the learning, discovery, and engagement missions at Virginia Tech. They are intended to complement the efforts and qualifications of tenure-track faculty, provide access to specialized faculty resources, and allow flexibility to address programmatic needs. As valuable contributors to departmental and institutional missions, they are entitled to fair treatment and compensation, access to professional development opportunities, recognition for their accomplishments, and participation in the life of the university community. The following policies address specific aspects of non-tenure-track faculty appointments. In a few cases, faculty members with regular academic rank (assistant, associate, or full professor) hold non-tenure-track appointments because of unusual job responsibilities and historical lack of appropriate alternative ranks. Policies in this section also apply to those individuals.

Ordinarily a graduate or professional degree is required for appointment to one of these ranks. Appointments are made using established <u>university search procedures</u>.

Academic departments retain the authority and responsibility to decide whether to employ non-tenure-track faculty members to deliver aspects of their instructional program. Departmental policies and practices related to the use of non-tenure-track ranks must be approved by an appropriate departmental committee.

#### 5.1 Non-Tenure-Track Instructional Faculty Ranks

#### **5.1.2** Visiting Professor

Appointment to the rank of visiting assistant, associate, or full professor is for a restricted period to carry out learning, discovery, and engagement responsibilities within an academic department. Professional credentials required for the standard professorial ranks are required for appointment as a visiting assistant, associate, or full professor. A visiting faculty member may not serve in such a position beyond six years. Tenure cannot be awarded at this rank.

Full-time service at this rank *may* or *may not* be counted as part of the probationary period if the faculty member is subsequently appointed to a tenure-track position. As with prior service credit from another institution, the decision to include all or some of the years of service from a visiting appointment is at the discretion of the faculty member. However, this decision must be made at the time of appointment to the tenure-track position and documented as part of that initial contract.

#### 5.1.3 Adjunct Professor

Appointment to the rank of adjunct assistant, associate, or full professor is reserved for persons whose primary employment is with another agency, organization, or educational institution, or

with a non-instructional unit of the university. Professional credentials required for the standard professorial ranks are required for appointment as an adjunct assistant, associate, or full professor. Appointments may be renewed annually, but tenure cannot be awarded at this rank. The professor of practice series titles may be used for wage adjunct faculty appointments in lieu of the adjunct assistant professor, associate professor, or professor titles, if appropriate for the assignment and credentials of the individuals. (See section 5.1.4, "Professor of Practice Ranks.")

Adjunct faculty must present credentials appropriate to the level of the course they are teaching. It is the responsibility of the department to verify documentation of appropriate credentials for adjunct faculty members prior to the start of the course. (See section 2.5.9, "Faculty Credentialing Guidelines," or the <u>provost's website</u>.)

If deemed qualified and appropriate by the host department, authorization for an adjunct faculty member to serve as principal investigator on a sponsored project may be requested. The department, with the approval of the dean, submits a written request for such authorization to the Office of the Vice President for Research.

Wage adjunct faculty do not typically submit an annual faculty activity report or have an annual evaluation because their employment is temporary. Although wage adjunct faculty may be hired repeatedly to teach a course or courses, they are not considered to be continuing faculty for the purposes of evaluation. Per course stipends paid to wage adjunct faculty are not fixed university-wide, but rather are determined on a departmental basis. Payments typically reflect the experience and credentials of the wage adjunct faculty member, the level of demand (market) for the necessary expertise, and general salary levels in the discipline.

A "How To Guide" for processing P14 actions is available on the human resources website.

#### **5.1.4** Professor of Practice Ranks

Academic departments retain the authority and responsibility to make decisions about whether to employ professors of practice to deliver aspects of their instructional program or to carry out other aspects of the departmental mission. Departmental policies and practices related to the use of non-tenure-track ranks must be approved by an appropriate standing committee in the department, such as a promotion and tenure or executive/personnel committee, the department head or chair, and dean.

For disciplines where professional preparation of students is a major goal, the involvement of experienced practitioners in teaching the skills and values of the profession, overseeing internships and project experiences, and career advising, for example, are a vital aspect of a successful program. Professional programs often have a deep commitment to on-going continuing education of practitioners in the field, resulting in a greater commitment to delivery of outreach programs than is typical of a tenure-track appointment outside of extension. Programs in the arts may wish to attract resident artists or performers for a period of time to contribute to the program. The professor of practice rank series may be appropriate in these and other roles that typically do not reflect the same range of responsibilities required for tenure-track faculty members.

The professor of practice series provides for short- or long-term, full- or part-time, non-tenure-track faculty appointments for individuals who bring specialized expertise to the instructional programs of the university, thereby complementing the qualifications and contributions of tenure-track faculty. (These rank titles may also be used for wage appointments in lieu of adjunct assistant, associate, or professor, if appropriate for the assignment and credentials of the individual.) Individuals appointed to these ranks are expected to be successful and effective professionals in a given field. They must be effective teachers of the profession or discipline and they are expected to be able to understand and evaluate the research that applies to their field and to teach it to students. While professor of practice faculty members may conduct research and present their findings in professional venues, there are no expectations for an extensive research program as is typical of tenure-track faculty appointments.

Professor of practice faculty members are expected to remain active in their professions in ways that contribute to their assignment—teaching, consulting or doing outreach, serving in technical and professional societies and associations, and similar activities. Where appropriate to their assignment, they may interact with graduate students and interns, serve on graduate committees, and chair graduate advisory committees with the approval of the academic unit and the graduate school. They may also be expected to serve on departmental, college, or university committees as contributing members of their departments and the broader university community.

Individuals appointed to a professor of practice rank must have a graduate or professional degree in the discipline (or a related discipline), professional certification(s) if relevant, and/or significant professional experience. Any appointment without the relevant *terminal* degree in the field must be certified by the department as appropriately credentialed for the faculty member's particular instructional assignment in accordance with guidelines for regional accreditation and university policy and procedures. Documentation supporting alternative credentials certification is required. Further information regarding appropriate credentials for the teaching faculty is found in section 2.5.9, "Faculty Credentialing Guidelines," and on the provost's website.

A record of significant professional achievement is expected for appointment at the associate or full level; initial appointments at such ranks require approval of the appropriate departmental committee and head or chair. Appointment to one of these ranks may be from one to five years and is renewable without limit.

Promotion within these ranks may be pursued through procedures outlined in this section. Tenure will not be awarded at any of these ranks and service at these ranks is excluded from the probationary period if the faculty member is subsequently appointed to a tenure-track position.

**Assistant Professor of Practice**: Persons appointed at this rank have a graduate or professional degree in the discipline (or a related discipline), professional certification(s) if relevant, and/or significant professional experience. Experience and a demonstrated competence in practice of the profession are expected. Credentials must be relevant to the field and type of assignment.

Associate Professor of Practice: Persons appointed at the associate professor of practice rank have a graduate or professional degree in the discipline (or a related discipline), professional certification(s) if relevant, and/or significant professional experience. Credentials for

appointment or promotion to this rank must document a record of significant professional experience and accomplishments relevant to the field and type of assignment.

**Professor of Practice**: Professor of practice is the capstone rank in the series. Appointment to this rank denotes distinguished professional achievement, and regional, national, or international prominence in the field. Credentials for appointment or promotion to this rank must document a record of significant professional experience and accomplishments relevant to the field and type of assignment. External validation of such accomplishments and leadership in the field is expected at the time of appointment or promotion.

#### 5.1.5 Clinical Faculty Ranks

General collegiate faculty members with responsibilities primarily in instruction and/or service in a clinical setting, such as veterinary medicine, are considered clinical faculty. The following clinical faculty appointments are intended to promote and retain clinical educators and to complement the clinical activities of the university. The clinical faculty track provides for long-term, full-time or part-time faculty appointments to individuals whose primary responsibilities are in clinical settings and in the instructional programs. While clinical faculty may conduct clinical research and present their findings in professional venues, there are no expectations for an extensive research program as is typical of tenure-track faculty appointments. Tenure cannot be earned in these ranks and time spent in one of these ranks is not applicable toward probationary tenure-track faculty service. The clinical faculty ranks include:

**Clinical Instructor**: Persons appointed to this rank must have the appropriate professional degree. Preference is given to individuals eligible for, or certified by, the most appropriate specialty college or organization recognized by the profession. Appointments at this rank are typically for one year and are renewable.

The clinical professor series is designed for clinical faculty members who have extended appointments and who are expected to interact with graduate students/residents and interns, serving on committees or supervising their training. Appointment to one of these ranks may be from one to five years and is renewable without limit. Normally a national search is conducted for appointment at one of these ranks (or an approved exemption sought for exceptional skills or similar justification).

Clinical Assistant Professor: Persons appointed to this rank must have the appropriate professional degree and eligibility for, or certification by, the most appropriate specialty college recognized by the professional organization. Credentials shall be consistent with those for appointment to assistant professor, with an expectation for primary commitment to the instructional and clinical teaching setting.

**Clinical Associate Professor**: Persons appointed to this rank must have the appropriate professional degree and be a diplomate in the appropriate specialty college recognized by their professional organization. Credentials shall be consistent with those for appointment to associate professor, with an emphasis on clinical accomplishments.

**Clinical Professor**: Persons appointed to this rank must have the appropriate professional degree and be a diplomate in the most appropriate specialty college recognized by their professional organization. Credentials shall be consistent with those for appointment to professor, with an emphasis on clinical accomplishments.

Further detail on the duties and responsibilities of these ranks, criteria and the process for promotion, and the terms and conditions of employment for clinical faculty are established by the respective academic departments and approved by an appropriate college-level committee and the dean.

#### 5.1.6 Instructor Faculty Ranks

The responsibilities of a person appointed to one of the instructor ranks in an academic department are focused on undergraduate education, with minimal or no expectation for development of an independent program of research or scholarship. A master's degree is the usual minimum educational credential for an appointment to the instructor ranks, and generally a minimum of 18 graduate credits in the teaching discipline is required to meet accreditation standards. (Further information regarding appropriate credentials for the teaching faculty is found in section 2.5.9, "Faculty Credentialing Guidelines," and on the provost's website.)

While initial appointment is typically at the entry rank, prior experience may be considered for a recommendation of appointment at a higher rank with the approval of the appropriate departmental committee and head. Up to three years of similar instructional service at another accredited American four-year college or university may be counted toward the designated period required prior to review for promotion in rank.

Tenure is not awarded at any of these ranks and all service at any instructor rank is excluded from the probationary period if the faculty member is subsequently appointed to a tenure-track position. Faculty members within the instructor ranks may not chair a graduate committee.

**Instructor**: The instructor rank is the initial rank for appointment of a full- or part-time faculty member. Primary responsibilities are usually to the instructional program, but assignments vary depending on the faculty member's expertise and experience and departmental needs. Typically they include teaching undergraduate courses, advising students, developing or revising courses and curricula, and fulfilling other instructional, administrative, or service responsibilities. Appointment at this rank consists of a series of one- or two-year renewable appointments with a minimum of five years of completed service before consideration for promotion.

**Advanced Instructor**: Consideration for promotion to the rank of advanced instructor may be requested by the instructor or recommended by the department based on excellence in instructional responsibilities and significant evidence of related professional growth and development. Mentoring more junior colleagues or graduate teaching assistants, student advising, course or curriculum development, or exemplary service or outreach are examples of ways in which instructors can make valuable contributions to the instructional programs in a department. Advanced instructors are expected to demonstrate mastery in teaching with significant impact on student learning and the department's undergraduate programs. Scholarship

and publication are not typically an assigned responsibility of instructor positions, but such accomplishments may be considered as part of the evaluation for promotion. Promotion to the advanced instructor rank is generally accompanied by a renewable three-year contract.

A minimum of five years of completed service at the advanced instructor rank is required before consideration for promotion to senior instructor.

**Senior Instructor**: Senior instructor is the capstone rank in the instructor series and promotion to this rank denotes exemplary instruction, demonstrated continued professional development, and significant contributions to undergraduate education. In addition to teaching courses, senior instructors may have considerable responsibility in mentoring junior colleagues or graduate teaching assistants, overseeing course development or special instructional initiatives, student advising, or other non-teaching responsibilities reflecting their role as instructional leaders. Promotion to the rank of senior instructor is generally accompanied by a renewable five-year contract.

## 5.2 Policies Related to Non-Tenure-Track Instructional Appointments

## 5.2.1 Initial Appointment

All initial non-tenure-track faculty appointments are normally for a period of one year, including appointments at the more senior ranks. Subsequent reappointments may be multi-year, as appropriate for the rank.

Appointments may be regular (renewable) or restricted (with a defined end date), calendar year or academic year, and full- or part-time depending on job responsibilities and available funding. Visiting and adjunct appointments are intended to be temporary in nature and are almost always restricted.

Non-tenure-track faculty positions are term appointments eligible for renewal based on quality of performance, continuing need for services in the unit, and available funding.

## 5.2.2 Reappointment

Non-tenure-track faculty members on restricted contracts whose appointments will be continued are issued a reappointment contract specifying the new ending date for their appointment. Reappointments typically occur on the anniversary of the hire date or are realigned to coincide with the academic year or other relevant appointment cycle. The practice of issuing repeated one-year restricted contracts for an individual faculty member over many years is explicitly discouraged. It results in exclusion of the faculty member from promotion consideration and can be exploitative over an extended time.

Reappointments for faculty members on regular contracts are usually effective July 1 or August 10, reflecting either calendar year or an academic year appointment. Notice of non-reappointment is in accordance with periods identified in section 2.11, "Retirement, Resignation, and Non-Reappointment."

## 5.2.3 Annual Evaluations and Merit Adjustments

Continuing faculty members must submit an annual faculty activity report in accordance with departmental and college procedures and timelines. Timely submission of the annual faculty activity report is required for consideration for a merit adjustment. Annual evaluation of performance by the department head/chair or supervisor (or appropriately charged committee) and feedback to the faculty member are required, and should be consistent with university policies and practices for annual evaluation of tenure-track faculty members.

Non-tenure-track faculty members are entitled to full consideration for merit adjustments as available and warranted by their performance.

#### 5.2.4 Promotion Guidelines for Non-Tenure-Track Faculty Appointments

Non-tenure-track faculty members are eligible for promotion in rank in accordance with guidelines established by academic departments and approved by an appropriate college-level committee and the dean. Such guidelines should outline the process and criteria for promotion in rank; they should be widely available along with other departmental and college documents related to promotion and tenure. Faculty members must be in a *regular* rather than *restricted* appointment to be considered for promotion.

Promotion in rank for any non-tenure-track position is not a requirement of continued employment, or an entitlement for years of service without evidence of exceptional merit, continued professional development, and contribution in the assigned role. An approved promotion in rank is recognized by a change in title, increasing length of appointment contract, and a base salary adjustment as identified in the annual faculty compensation plan approved by the board of visitors. (See section 2.4, "Faculty Compensation Plan.")

Consideration for promotion in rank includes preparation of a dossier using a common university format, which may be based on relevant elements of the promotion and tenure dossier format for tenure-track faculty members. Typically such a dossier includes a statement of professional direction and accomplishment, a full *curriculum vitae*, and documentation of contributions associated with the instructor's appointment. These contributions can be to instructional programs or administrative and/or service responsibilities. Colleges and departments may request supplemental materials. The appropriate college committee and dean approve guidelines for dossier development and departmental policies and procedures for the promotion process. External evaluations of credentials are not necessary for promotion consideration for non-tenure-track faculty, except for promotion to clinical professor.

The promotion dossier is reviewed at three levels: by an appropriately charged departmental committee and the department head or chair, by an appropriately charged college-level committee and the dean, and by the provost. Given the wide variation in representation of non-tenure-track instructional appointments in the various academic colleges, some latitude is provided in the nature and make up of such committees. For those departments with significant numbers of instructors, it is expected that the committee charged with such reviews would consist of majority representation of advanced and senior instructors (or associate or full clinical

professors). In departments with very few such appointments, the existing departmental promotion and tenure committee may review the dossier(s) and make recommendations. Similarly, at the college level, either a special committee may be formed to review promotions of non-tenure-track instructional faculty with majority representation of those in the advanced level of such ranks, or existing promotion and tenure committees may be assigned such review and recommendation responsibilities. While the procedures vary to recognize practical issues and varying numbers, the guidelines for review should be in writing and adhered to for consistency and fair treatment of all candidates. A faculty member may not serve on any committee that is evaluating a spouse, family member, or other individual with whom the faculty member has a close personal relationship (see section 2.16.2, "Potential Conflicts Involving Spouses and Immediate Family Members").

The department head or chair and dean make separate recommendations to the subsequent review levels. The provost reviews college and dean recommendations and makes recommendations to the president. The board of visitors grants final approval.

Given that promotion decisions do not carry the same "up or out" decision associated with tenure, a negative recommendation on a promotion request need not translate into termination of employment. Indeed, a faculty member may remain at the initial rank as long as their performance warrants continued employment and serves departmental needs. If the promotion request is not supported on the first submission, it may not be appealed until at least a second review has taken place in a subsequent or later year.

Following the same pattern as review of tenure-track faculty members, a positive recommendation from either or both the departmental committee and the department head or chair automatically advances the dossier for promotion to the college committee level. Similarly, at the college level, a positive recommendation from either or both the college committee and the dean results in automatic advancement of the recommendation to consideration by the provost. The decision of the provost is final and cannot be appealed.

Faculty members should be provided written feedback in the case of a negative recommendation at either the department or college level so that they might improve their performance or dossiers for a later submission.

#### **5.2.5** Appeal Procedures

Following a second negative review by both the departmental committee and department head or chair, the decision may be appealed to the college committee, but only on grounds that relevant information was not considered or that the decision was influenced by improper consideration. The appeal must be filed within 14 days of official notification. A negative recommendation from both the college and the dean ends the process. There is no appeal available when both the college committee and dean vote "no."

Significant *procedural* violations may be grieved under the faculty grievance process described in section 5.7, "Faculty Grievance Policy and Procedures."

## 5.3 Termination Procedures for Non-Tenure-Track Faculty on Regular Appointments

Members of the non-tenure-track faculty may be removed from their position by one of the following three procedures: removal for just cause, non-reappointment, or termination of position because of insufficient funds or no further need for services. Notice of non-reappointment for non-tenure-track instructional faculty members on regular appointments is addressed in section 2.11, "Retirement, Resignation, and Non-Reappointment."

#### **5.3.1** Dismissal for Cause

Stated causes for removal shall include, but are not limited to, professional incompetence; unacceptable or unsatisfactory performance after due notice; unethical conduct or misconduct that interferes with the capacity of the employee to perform effectively the requirements of the position; violation of university policy; or falsification of credentials, experience, leave reports, or other official university documents. Filing a grievance shall not constitute just cause for termination.

When it becomes necessary to terminate a non-tenure-track faculty member for unsatisfactory performance prior to the end of the appointment period, the following procedures apply:

- The department head or chair writes a letter to the faculty member detailing the areas of
  performance that are deficient and setting clear expectations for acceptable performance and
  continued employment. The college dean receives a copy. The letter states the time period in
  which the deficiencies must be addressed. This time period will be not less than 30 calendar
  days.
- At the end of that period, the department head or chair must again write the faculty member with an evaluation of his/her performance with a copy to the college dean. If performance continues to be unsatisfactory, this second letter may contain a notice of termination. The termination notice will have an effective date of 45 calendar days or more from the date of the second letter. In cases where there is a threat to health or safety, the 45-day period may be waived.

For termination for cause for reasons other than unsatisfactory performance, the faculty member shall receive written notification of the reasons for termination and shall be allowed an opportunity to respond within five workdays. With the approval of the provost, a faculty member may be suspended with or without pay pending an investigation into allegations of wrongdoing.

The faculty member may appeal notification of termination for cause to the college dean and the termination will be held in abeyance until the appeal process is complete. The appeal must be made in writing within five working days of receipt of the notification of termination. The dean must respond in writing within ten working days. If the dean's response is unsatisfactory to the appellant, an appeal may be made to the provost in writing within five working days. The provost appoints a committee of three members of the general faculty who review the case and make recommendations to the provost. The decision of the provost is final. The above time limits

of the appeal process may be altered by extenuating circumstances and the agreement of both parties.

#### 5.3.2 Termination of Appointment During the Contract Period

Occasionally a decline in funding resources makes it necessary to terminate an appointment before the end of a contract. While department heads or chairs are encouraged to make every effort to assure continuity of employment to individuals performing satisfactorily, there are circumstances in which this may not be possible or in the best interest of the university. Nontenure-track faculty appointments may be terminated in the case where there are insufficient funds or no further need for services. Written notice of termination within the contract period shall be at least three months for those who have been in a regular appointment less than two year and at least one semester (if academic year) or six months (if calendar year) for those who have been in regular appointments two years or more. A proposed notice of termination during the contract period because of insufficient funds or lack of need for services requires the approval of the dean and provost.

## **5.4** Participation in Governance

Salaried non-tenure-track instructional faculty members are eligible to participate in departmental, college, and university committees as appropriate for their assignments. Non-tenure-track faculty members should have meaningful engagement in program planning at the department level, especially as it relates to aspects of the curriculum for which they bear teaching responsibility. Although non-tenure-track instructional faculty members cannot be involved in reviewing cases of promotion and tenure for tenure-track or tenured faculty members, they may otherwise be voting members of the departmental faculty in accordance with the policy set by individual departmental governance.

Those faculty members at the rank of instructor, assistant, associate, or professor, or related rank variations, such as clinical assistant professor, professor of practice, or senior instructor are eligible to serve as voting members of the Faculty Senate.

#### 5.5 Participation on Graduate Committees

Non-tenure-track instructional faculty members with appropriate credentials may serve on graduate advisory committees and interact with graduate students and interns where relevant to their assignment and with approval of the departmental graduate program, department head or chair, and graduate school. Faculty members within the instructor ranks may not chair a graduate committee.

## 5.6 Eligibility to Serve as a Principal Investigator

Faculty members in a non-tenure-track rank may serve as a principal investigator for a sponsored project or contract with the approval of the department head or chair, the dean, and the Office of the Vice President for Research. A written request for authorization may be submitted by the department, with the approval of the dean, to the Office of the Vice President for Research.

## **5.7** Faculty Grievance Policy and Procedures

The following procedure is provided as the means for resolution of grievances against a supervisor or member(s) of the university administration brought by members of the non-tenure-track instructional faculty. The grievance process is the same as that for tenured and tenure-track faculty. The <u>Faculty Review Committee</u> of the Faculty Senate conducts the step four hearing if requested.

## **5.7.1** Faculty Reconciliation and Mediation Services

**Informal Dialogue**: It should be possible to resolve most faculty concerns or complaints through informal communication among colleagues working together in the academic enterprise. Accordingly, a faculty member who feels he or she has a grievance is encouraged to take it to his or her immediate supervisor in the normal collegial spirit of problem solving rather than as a confrontation between adversaries.

**Reconciliation**: At the initiation of the grievance procedure, or at any earlier time, the grievant may request the assistance of the Faculty Senate <u>Committee on Reconciliation</u> in fashioning an equitable solution. Contacting the Faculty Senate Committee on Reconciliation is not required in filing a grievance, but it may be useful if the grievant feels that the issue may be amenable to, but will require time for, negotiation; or if the grievant is unsure whether his or her concern is a legitimate issue for a grievance; or if personal relations between the parties involved in the grievance have become strained.

For a potential grievance issue to qualify for consideration by the Faculty Senate Committee on Reconciliation, the grievant must contact the Faculty Senate Committee on Reconciliation within 30 calendar days of the time when the grievant knew or should have known of the event or action that is the basis for the potential grievance, just as if beginning the regular grievance process. If the grievant requests assistance from the Faculty Senate Committee on Reconciliation, that committee must request a postponement of the time limits involved in the grievance procedure while it deals with the case. The request is submitted in writing to the vice provost for faculty affairs by the chair of the Faculty Senate Committee on Reconciliation. Also, the grievant should reach an understanding with the Faculty Senate Committee on Reconciliation of the time frame planned for that committee's work on the case, such time not to exceed 60 calendar days.

Faculty members may also consult the Faculty Senate Committee on Reconciliation about serious disagreements with immediate supervisors or other university administrators concerning issues that may not be eligible for consideration within the grievance process. In such instances, the committee contacts the relevant administrator to determine if there is an interest and willingness to explore informal resolution of the dispute; it is not necessary to notify the Office of the Senior Vice President and Provost.

Additional information regarding the faculty reconciliation process is available on the <u>provost's</u> website.

**Mediation**: Mediation is a voluntary, confidential process through which trained neutral third persons (mediators) assist people to express their concerns and develop solutions to the dispute in a safe and structured environment. Because mediation is voluntary, both parties must agree to participate in order for mediation to occur. Faculty members and supervisors are encouraged to consider using mediation to resolve disputes between them, or to help address a conflict between a faculty member and another member of the Virginia Tech community.

**Role of Mediators**: Mediators do not make judgments, determine facts, or decide the outcome; instead they facilitate discussion between the participants, who identify the solutions best suited to their situation. No agreement is made unless and until it is acceptable to the participants.

**Requesting Mediation**: Mediation is available at any time, without the filing of a grievance. Additionally, mediation may be requested by any party during the grievance process prior to step four. If, after the initiation of a formal grievance, both parties agree to participate in mediation, the grievance is placed on administrative hold until the mediation process is complete. If the parties come to a resolution of the dispute through mediation, the parties are responsible to each other for ensuring that the provisions of the agreement are followed. In the event that the parties are not able to reach a mutual resolution to the dispute through mediation, the grievant may request that the grievance be reactivated and the process continues.

Mediation differs from faculty reconciliation in that mediators do not engage in fact-finding or in evaluation of decisions. Both mediation and reconciliation, however, are voluntary; no party is required to participate in either process.

To learn more about mediation and other forms of informal conflict resolution processes, contact the <u>conflict resolution program</u> manager in human resources at 540-231-9331.

#### **5.7.2** The Formal Grievance Procedure

If the assistance of the Faculty Senate <u>Committee on Reconciliation</u> is not desired or is not requested; or if that committee determines that it cannot provide assistance in the matter; or if the grievant finds that the length of time the Committee on Reconciliation plans or takes with the case is excessive; or if the grievant is not satisfied with the recommendations of that committee, the grievant may pursue the issue as a formal grievance through the following procedure. Department heads or chairs, deans, directors, and other administrative faculty will cooperate with the grievant in the mechanics of processing the grievance, but the grievant alone is responsible for preparation of his or her case.

1. Step one: The grievant must meet with his or her immediate supervisor (normally the department head or chair) within 30 calendar days of the date that grievant knew or should have known of the event or action that is basis for the grievance and orally identifies the grievance and the grievant's concerns. The supervisor provides an oral response to the grievant within five weekdays following the meeting. If the supervisor's response is satisfactory to the grievant, that ends the matter.

2. Step two: If a satisfactory resolution of the grievance is not achieved by the immediate supervisor's oral response, the grievant may submit a written statement of the grievance and the relief requested to the immediate supervisor. This statement must be on the faculty grievance form, must define the grievance and the relief requested specifically and precisely, and must be submitted to the immediate supervisor within five weekdays of the time when the grievant received the immediate supervisor's oral response to the first step meeting. Faculty grievance forms are available on the provost's website.

Within five weekdays of receiving the written statement of the grievance, the immediate supervisor, in turn, gives the grievant a written response on the faculty grievance form, citing reasons for action taken or not taken. If the written response of the immediate supervisor is satisfactory to the grievant, that ends the matter.

3. Step three: If the resolution of the grievance proposed in the written response by the immediate supervisor is not acceptable, the grievant may advance the grievance to the next higher level of university administration by checking the appropriate place on the faculty grievance form, signing and sending the form to the next higher administrator within five weekdays of receiving the written response from the immediate supervisor. The next higher level of administration for collegiate faculty is normally the college dean. The administrator involved at this next level is hereafter referred to as the second-level administrator.

Following receipt of the faculty grievance form, the second-level administrator or designated representative meets with the grievant within five weekdays. The second-level administrator may request the immediate supervisor of the grievant be present; the grievant may similarly request that a representative of his or her choice from among the university faculty be present. Unless the grievant is represented by a member of the faculty who is also a lawyer, the second-level administrator does not have legal counsel present. The second-level administrator gives the grievant a written decision on the faculty grievance form within five weekdays after the meeting, citing reasons for his or her decision. If the second-level administrator's written response to the grievance is satisfactory to the grievant, that ends the matter.

4. Step four: If the resolution of the grievance proposed in the written response from the second-level administrator is not acceptable, the grievant may advance the grievance within five weekdays to the level of the provost, including consideration by an impartial hearing panel of the Faculty Review Committee. A description of the charge and membership of the Faculty Review Committee is included in section 1.4.5.2, "Faculty Review Committee."

Upon receiving the faculty grievance form requesting step four review, the provost, or appropriate designated representative, acknowledges receipt of the grievance within five weekdays and forwards a copy of the <u>Procedures of the Faculty Review Committee</u> to parties in the grievance process. The provost immediately forwards a copy of the grievance to the <u>president of the Faculty Senate</u>, who also writes to the grievant to acknowledge receipt of the grievance within five weekdays of receipt of the faculty grievance form from the provost.

The grievant may petition the provost to bypass the Faculty Review Committee and rule on the grievance. If the provost accepts the request, there is no subsequent opportunity for the grievance to be heard by a hearing panel. The provost's decision, however, may be appealed to the president, as described in step five. If the provost does not accept the petition, the Faculty Review Committee hears the grievance as outlined in these procedures.

• **Hearing Panel**: A hearing panel consists of five members appointed by the chair of the Faculty Review Committee from among the members of the Faculty Review Committee. The chair of the Faculty Review Committee polls all appointees to ensure that they have no conflict of interest in the case. Both parties to the grievance may challenge one of the appointments, if they so desire, without need to state cause, and the chair of the Faculty Review Committee appoints the needed replacement or replacements. Other replacements are made only for cause. The chair of the Faculty Review Committee rules on issues of cause.

To ensure uniformity in practice, the chair of the Faculty Review Committee or his or her designee serves as the non-voting chair of each hearing panel. In the event that the chair of the Faculty Review Committee has a conflict of interest concerning a case, the chair appoints a disinterested third party from among the members of the Faculty Review Committee not already appointed to the hearing panel for the case to serve as chair of the hearing panel.

• **Hearings**: After a hearing panel is appointed, the chair of the Faculty Review Committee requests that each party to the grievance provide relevant documentation to be shared among the parties and the hearing panel. The panel holds its initial hearing with both principals present within 15 weekdays of receipt of the grievance by the Faculty Senate president. If the panel feels it needs to investigate the case further, or requires more information, or desires to hear witnesses, the hearing is adjourned until the panel completes the necessary work or scheduling. The hearing is then reconvened as appropriate.

Each party to the grievance may have a representative present during the sessions of the hearing at which testimony is presented. The representative may speak on their behalf if so requested. Representatives may be legal counsel, if both parties are so represented, but if the grievant does not wish to have legal counsel at a hearing, neither party to the grievance may have legal counsel present.

These impartial panel hearings are administrative functions, not adversarial proceedings. Therefore, if legal counsels are present they must understand that the proceedings do not follow courtroom or trial procedures and rules. Participation by legal counsel is at the invitation of the parties they represent and is subject to the rulings of the chair of the hearing panel. Detailed procedures followed in hearings are specified in the "Procedures of the Faculty Review Committee" as approved by the Faculty Senate.

• **Findings and Recommendations**: The hearing panel concludes its work and makes its recommendations within 45 weekdays of receipt of the grievance by the Faculty Senate president. The time limit for consideration may be extended by agreement of both parties.

The hearing panel formulates written findings and recommendations regarding disposition of the grievance and forwards copies to the provost, the grievant, and the chair of the Faculty Review Committee.

- **Provost's Action**: The provost meets with the grievant within 10 weekdays after receiving the findings and recommendations of the hearing panel to discuss the case and advise the grievant about the prospects for disposition of the case. Within 10 weekdays of that meeting, the provost sends to the grievant his or her decision in writing concerning the disposition of the grievance. If the provost's decision is fully consonant with (or exceeds) the recommendations of the hearing panel, or if it is satisfactory to the grievant even if it differs from the recommendations of the hearing panel, that ends the matter.
- 5. Step five: If the provost's decision is not acceptable to the grievant and not consonant with the recommendations of the hearing panel, the grievant may appeal in writing to the president within 20 calendar days. The president acts as he or she sees fit. The president's decision is final.

## **5.7.3** Timeliness of Grievance and Procedural Compliance

A grievance must be brought forward in a timely manner. It is the responsibility of the grievant to initiate the grievance process within 30 calendar days of the time when he or she knew or should have known of the event or action that is the basis for the grievance. The university administration is not required to accept a grievance for processing if the grievant does not meet the 30-day deadline, except in cases of demonstrated good cause.

Scheduled commitments made prior to the time of filing or advancement of a grievance that preclude action by either of the parties to the grievance automatically extend time limits for their duration unless this would be demonstrably harmful to the fair processing of the grievance. In such cases, on written request by the grievant to the appropriate office for that step, the grievance is advanced to the next step in the grievance process.

If the grievant does not follow the time limits specified in the grievance procedure it is assumed that he or she accepted the last proposed resolution as satisfactory. If the grievant desires to advance the grievance after the appropriate specified time limits have lapsed, the administrator who receives the late submission notifies the chair of the Faculty Review Committee in writing, and the chair of the Faculty Review Committee determines if there was good cause for the delay. If so, the grievance proceeds. If not, the process ends with the most recently proposed resolution in force. The finding on the matter by the chair of the Faculty Review Committee is communicated to both parties in writing.

If either party to a grievance charges the other with procedural violations other than time limit issues, a special committee of the president of the Faculty Senate, the chair of the Faculty Senate

Committee on Reconciliation, and the chair of the Faculty Review Committee (or the vice president of the senate if the president is also chair of the Faculty Review Committee) is convened to rule on the question, as in disputes about the validity of issues qualifying for the grievance procedure. The special committee has the following options. It can either find no significant procedural violation occurred, in which case the grievance process continues unaffected, or that a significant procedural violation did occur. If the administrator committed a significant procedural violation, the grievance automatically qualifies for advancement to the next step in the grievance process. If the grievant committed a significant procedural violation, the grievance process ends at that point with the last proposed resolution established as the final disposition of the case.

#### 5.7.4 Valid Issues for Grievance

For this process, a grievance is defined as a complaint by a faculty member alleging a violation, misinterpretation, or incorrect application of a policy, procedure, or practice of the university that directly affects the grievant. Some examples of valid issues for filing a grievance are: improperly or unfairly determined personnel decisions that result in an unsatisfactory annual performance evaluation, unreasonable merit adjustment or salary level, or excessive teaching load/work assignments; substantive violations of promotion procedures (see appeal process in section 3.4.5, "Appeals of Decisions [on Promotion]"); reprisals; substantive error in the application of policy; and matters relating to academic freedom.

Issues not open to grievance: While most faculty disputes with the university administration may be dealt with by this grievance policy, the following issues may not be made the subject of a grievance: determination of policy appropriately promulgated by the university administration or the university governance system; those items falling within the jurisdiction of other university policies and procedures (for example, complaints of unlawful discrimination or harassment, or an appeal related to the merits of a promotion decision); the contents of personnel policies, procedures, rules, regulations, ordinances, and statutes; the routine assignment of university resources (e.g., space, operating funds, parking, etc.); normal actions taken, or recommendations made, by administrators or committee members acting in an official capacity in the grievance process; termination of appointment by removal for just cause, non-reappointment, or abolition of position; or allegations of misconduct in scholarly activities.

Adjudication of disputes on the validity of issues qualifying for consideration under the faculty grievance procedures: If a university administrator rules that an issue does not qualify for the grievance process, the grievant may write to the chair of the Faculty Review Committee within five weekdays of receiving such notification and request a ruling from a special committee consisting of the president of the Faculty Senate, the chair of the Faculty Senate Committee on Reconciliation, and the chair of the Faculty Review Committee. The special committee considers the matter (including consultations with both parties if deemed necessary) and rules by majority vote on the admissibility of the matter to the grievance process. This special committee is called together by the chair of the Faculty Review Committee, who also sends a written report of the results of the deliberations of the committee to all parties concerned.

#### 5.7.5 Particular Concerns and Definitions

Time limits are subject to extension by written agreement of both parties. The grievant and the administrator involved at that particular step of the discussion are the makers of such an agreement. (An agreement form to extend the grievance response time is available on the provost's website.)

Grievances that advance to step four during or close to the summer and/or teaching breaks during the academic year may require some extension of the stipulated time limits. The principals and the chair of the Faculty Review Committee negotiate such an extension. Every effort is made, however, to stay within the stipulated time limits. In case of disagreement, the Faculty Senate president rules on time extension and procedure questions, which might include a hearing conducted by three rather than five panel members, or other recommendations designed to expedite the proceedings while providing peer review of the grievance.

If a faculty member is away from his or her assigned work location at the time he or she discovers the event or action that is the basis for a grievance, the 30-day period during which the grievant must meet with his or her immediate supervisor to initiate the grievance process begins when the faculty member returns to his or her assigned work location. If the date of return causes a delay of such length that the grievance, or its resolution, is not timely, the grievant may submit the grievance in writing to the immediate supervisor (step two), omitting personal meetings until such time as the faculty member returns to his or her assigned work location.

"Weekdays," as used in this procedure, include Monday through Friday only and only when those days are not national, state, or religious holidays relevant to the principals in the grievance.

To protect a grievant from undue pressure in the pursuit of a grievance, if a grievant becomes ill and takes sick leave the grievance process stops until such time as the grievant is able to resume his or her duties. Exceptions to this provision are made at the request of the grievant, but only if the grievant obtains and produces medical certification that proceeding with the grievance will not be harmful to the health of the grievant, or exacerbate the ailment that required taking sick leave.

All costs of legal counsel employed by a grievant are borne by the grievant.

If a grievant is employed away from Blacksburg, and he or she is required to travel away from their duty station in resolution of their grievance, the university pays all travel costs permitted under state regulations.

In the event that a faculty member discovers he or she has a grievance about actions by an administrator above the level of his or her immediate supervisor that directly involve the faculty member, or with actions by an administrator not in his or her department that directly involve the faculty member, the grievant initiates the grievance process by seeking the mediation of his or her immediate supervisor within 30 calendar days of the discovery of the event or action that is the basis for the grievance. If that effort does not resolve the grievance satisfactorily, the grievant, after consulting his or her immediate supervisor, may file the faculty grievance form at

the appropriate level or with the appropriate administrative office to initiate response from the administrator perceived as the source of the action causing the grievance. The grievance process then proceeds from that level onward in normal fashion.

A grievance filed by a faculty member concerning an action of the provost is handled by the chair of the Faculty Review Committee and a regular impartial hearing panel, but the findings and recommendations of the hearing panel are sent to the president of the university for his or her ruling, rather than to the provost. A grievance filed by a faculty member concerning an action of the president of the university is dealt with by a special panel appointed by the provost in consultation with the president of the Faculty Senate.

Any final resolution of a grievance must be consonant with the laws of the Commonwealth of Virginia and university policy.

Once a grievance is resolved, either to the satisfaction of the grievant, or if not to the satisfaction of the grievant, by the action of the provost in consonance with the hearing panel recommendations, or by the ruling of the president, that specific grievance is closed and may not be made the subject of another grievance.

## 5.7.6 Overview of the Formal Grievance Process for Non-Tenure-Track Instructional Faculty

Below is an <u>abbreviated</u> overview of the grievance process and deadlines. Refer to section 5.7.2, "The Formal Grievance Procedure," for specific details and options available in each step of the grievance process.

Time limits are subject to extension by written agreement of both parties. The grievant and the administrator involved at that particular step of the discussion are the makers of such an agreement. (An agreement form to extend the grievance response time is available on the provost's website.)

#### Step one:

Within 30 days of event 1a. Grievant meets with immediate supervisor (usually the department head).

Within 5 weekdays 1b. Department head provides oral response.

- 1c. If department head's response is satisfactory to grievant, that ends the matter.
- 1d. If department head's response is not satisfactory to grievant, move to step two within 5 weekdays.

#### Step two:

Within 5 weekdays
Within 5 weekdays

- 2a. Grievant submits written grievance to department head.
- 2b. Department head responds in writing on grievance form.
- 2c. If department head's response is satisfactory to grievant, that ends the matter.
- 2d. If department head's response is not satisfactory to grievant, move to step three within 5 weekdays.

Step three:		
Within 5 weekdays	3a.	Grievant advances grievance form to the second-level administrator (usually the dean).
Within 5 weekdays	3b.	Dean meets with grievant; dean may request DH to be present.
Within 5 weekdays	3c.	Dean responds in writing on grievance form.
	3d.	If dean's written response is satisfactory to grievant, that ends the matter.
	3e.	If dean's written response is not satisfactory to grievant, move to step four within 5 weekdays.
Step four:		-
Within 5 weekdays	4a.	Grievant advances grievance form to the provost.
Within 5 weekdays	4b.	Provost acknowledges receipt of grievance and forwards copy to Faculty Senate president to receive recommendation of an impartial hearing panel of the Faculty Review Committee.
Within 5 weekdays	4c.	Faculty Senate president acknowledges in writing to grievant that
within 5 weekdays	ъ.	copy of grievance has been received from provost.
Within 15 weekdays	4d.	Faculty Review Committee chair appoints hearing panel from among Faculty Review Committee members; panel holds its initial meeting with both principals.
Within 45 weekdays	4e.	Hearing panel concludes its work and makes recommendations
vi imin ie vietnaajs		to provost and grievant.
Within 10 weekdays	4f.	Provost meets with grievant.
Within 10 weekdays	4g.	Provost notifies grievant in writing of his or her decision.
·	4h.	If the provost's decision is fully consonant with (or exceeds) the
		recommendations of the hearing panel, or if it is satisfactory to
		the grievant even if it differs from the recommendations of the
		hearing panel, that ends the matter.
	4i.	If the provost's decision is not acceptable to the grievant and not consonant with the recommendations of the hearing panel, the

## Step five:

Within 20 calendar days 5a. Grievant may appeal in writing to university president.

calendar days.

5b. University president's decision is final.

grievant may appeal in writing to the president within 20

#### **CHAPTER SIX**

#### **Employment Policies and Procedures for Research Faculty**

#### 6.1 Office of Research Human Resources

The Office of Research Human Resources provides support and service to the Office of Research senior management area. The Office of Research senior management area includes university research institutes, university research centers, and the offices of research/interdisciplinary programs administration, sponsored programs, research compliance, export and secure research compliance, and Virginia Tech Intellectual Properties. The Office of Research Human Resources supports and serves research faculty regarding the policies, employment, retention, promotions, employee relations, and separations. Support and service include, but are not limited to, assistance in areas such position descriptions, classifications, position postings, screening, salary offers, promotions, salary adjustments for increased responsibilities, equity and retention, employment contracts, visas, and referrals. The Office of Research Human Resources is located at 300 Turner Street NW, suite 4100, mail code 0921, 540-231-9359.

## **6.2** Research Faculty Appointments

Research faculty appointments are designated to promote and expedite university research activities. The rank descriptions create several series common to current sponsored research or outreach projects. For example, employees involved in the conduct of research are generally appointed as research associates or research scientists (or to the "senior" titles for either of these). Those individuals who are appointed to a research traineeship for a period of up to four years following receipt of their doctorate are usually appointed as postdoctoral associates (see Guidelines for the Hiring of Postdoctoral Associates). Typically, postdoctoral associates work closely with a faculty mentor in preparation for a career in academe or research; if they remain involved with research projects at Virginia Tech over a period of time, they are promoted into another appropriate rank.

The "project associate" series was designed for employees involved in sponsored activity other than traditional research, such as delivery of service or technical assistance, consultation with particular clients, preparation of manuals and materials, and so on. The project associate series is also appropriate for personnel involved primarily in the administration of large and complex sponsored programs.

While there is logic to the progression between and among ranks, employees may change ranks as appropriate or necessary to reflect a change in role or project. Promotions from one rank to another in order to recognize a faculty member's increased responsibilities, credentials, and/or contributions to the program over time may be recommended by the supervisor. Recommendations for promotions are done during the annual evaluation and merit adjustment process. They require approval by the department head, dean, and Office of the Vice President for Research. Approval by a departmental committee is also required for certain ranks (see below). A faculty member may not serve on any committee that is evaluating a spouse, family

member, or other individual with whom the faculty member has a close personal relationship (see section 2.16.2, "Potential Conflicts Involving Spouses and Immediate Family Members").

Research faculty members may be assigned a functional title in addition to their official faculty rank in order to facilitate their work or clarify their role to internal or external constituencies. (A functional title may not be an official faculty rank other than that held by the research faculty member.) In some cases, increased responsibilities may lead to a change in functional title and possibly a salary adjustment rather than a promotion in faculty rank.

Appointments to research faculty ranks, except the rank of postdoctoral associate, are indefinitely renewable. However, tenure cannot be earned in any of the research faculty ranks and service is not applicable toward the probationary period if the employee is subsequently appointed to a tenure-track position.

The source of funds is not the determining factor as to whether a position carries a research faculty rank, but rather the nature and purpose of the assignment. Thus, a research faculty member may be funded by sponsored projects, overhead, state dollars, or other sources. Policies related to research faculty apply, regardless of the source of funding.

Research faculty may participate in activities outside of their direct source of funding, such as providing significant contributions to the conceptualization or writing of new proposals, or teaching; however, support for any time or effort spent on activities outside of their sponsored research must come from non-sponsored research funds. Special attention should be given in the development of position descriptions where funding is limited to only sponsored funding. (See sections 6.15, "Effort Certification Compliance Issues for Research Faculty Members," and 10.4, "Effort Certification and Salary Charges to Sponsored Grants and Contracts.")

Original appointments and reappointments, including rank, salary, and other conditions, require the approval of the department head, dean (or next level administrator), and the Office of the Vice President for Research *before* an offer is extended. Requests for principal investigator status may be submitted to the Office of the Vice President for Research. Such requests require the approval of the department head, dean, and the Office of the Vice President for Research.

#### 6.3 Research Associate Ranks

When establishing positions in this series, particularly at the research associate level, departments must ensure that the work anticipated for the employee is sufficiently complex and sophisticated to warrant a faculty position. There are a variety of staff roles that are appropriate for research personnel, depending on the nature of the work proposed and on the credentials being required. For example, laboratory and research specialist I or II is usually the appropriate staff role for personnel overseeing laboratory, animal care, or research support; or, conducting routine tests, compiling data, collecting and preparing samples. The Office of Research Human Resources is available to advise on the options. Careful preparation of the position description by the principal investigator and/or department head is essential in making a determination whether the position is staff or research faculty. Staff positions must be used where appropriate; exceptions are granted only in very rare cases.

#### **6.3.1** Research Associate

The research associate rank is the entry, or most junior, rank for research faculty members involved in sponsored projects. However, the work may vary from that which is appropriate for a relatively new professional to broader or more significant responsibilities expected of more experienced research faculty personnel. Research associates generally conduct research under supervision using standard and non-standard procedures appropriate to the field. They may provide input into the preparation of proposals or supervise staff or student personnel, but typically they have limited responsibility and authority in these areas. Research associates are not eligible to be principal investigators; however, requests for principal investigator status may be submitted to the Office of the Vice President for Research.

In keeping with all faculty appointments at the university, a master's degree in a relevant field is the minimum qualification for appointment as a research associate. The Office of Research Human Resources must approve exception requests for individuals with a bachelor's degree and significant related experience *before* an offer is extended.

## **6.3.2** Senior Research Associate

The rank of senior research associate requires greater qualifications than the research associate either in education, experience, or both. The doctorate or a terminal degree in the field and some experience are required. The Office of Research Human Resources must approve an appointment or promotion to this rank for individuals with a lesser degree and substantial related experience *before* an offer is extended.

Senior research associates conduct research under limited supervision using standard or non-standard techniques appropriate to the field. Typically, they have some significant supervisory responsibility for lower-level personnel and may contribute to the conceptualization and preparation of research proposals, reports, and resource acquisition. Senior research associates are not eligible to serve as principal investigators; however, requests for principal investigator status may be submitted to the Office of Vice President for Research.

#### 6.3.3 Research Scientist

Research scientists fulfill a senior role in the research program. They carry out independent research under limited supervision. A doctorate or terminal degree in the field and significant experience are expected. By virtue of their expertise and experience, research scientists make significant contributions to the conceptualization and conduct of the research. They may be involved in the preparation of proposals, reports, and publications, presentation of research results, and development of patents. Research scientists may serve as principal investigators.

#### 6.3.4 Senior Research Scientist

The senior research scientist is the highest rank in the research faculty series for those who do not also have involvement in the graduate program. The rank of senior research scientist is parallel to that of research associate professor or research professor. A doctorate or terminal

degree in the field and a considerable record of research are expected. As experienced research faculty members, senior research scientists are often responsible for the design and execution of a project and interpretation of research results. Faculty members at this rank usually serve as principal investigators of sponsored projects related to their own expertise or provide leadership to a research team. Typically, they have significant responsibility for supervision of personnel, budget preparation and execution, and organization and management of the research project.

#### 6.3.5 Postdoctoral Associate

Appointment to this rank is normally reserved for persons who have been awarded a doctoral degree no more than four years prior to the effective date of the appointment with a minimum of one year of eligibility remaining and are engaged in research for a restricted period under the direction of a faculty mentor. The position of postdoctoral associate is intended to be a limited-term traineeship lasting two to four years (not to exceed four years), during which the individual works under the supervision of one or more senior faculty mentors in preparation for a career in academe or research. Funding is usually from a grant, contract, or a postdoctoral fellowship.

The cumulative allowable duration for all postdoctoral appointments held by a single individual, even at multiple institutions, may not exceed five years. The maximum allowable time an individual may be employed in the rank of postdoctoral associate at Virginia Tech is four years. Continued appointment beyond four years would require a promotion in rank.

Typically, postdoctoral associates have very limited responsibility for project management, supervision of personnel, or design of the research project on which they are funded. Rather, the position enables the individual to continue studies in a specialty area while gaining practical experience in the field. The postdoctoral traineeship may include opportunities to write and submit grant proposals, and the postdoctoral associate may serve as co-principal investigator with the approval of the department head or chair. A postdoctoral associate may be permitted in certain cases, such as proposals for young career awards, to submit a grant as a principal investigator. Requests for principal investigator status may be submitted to the Office of the Vice President for Research.

#### 6.4 Project Associate Ranks

The project associate rank series is most appropriate for faculty personnel involved in sponsored projects that provide technical assistance, outreach, consultation, project management, preparation of specialized materials, or delivery of educational or other services, rather than conducting traditional research or scholarship. These ranks may also be appropriate for faculty personnel who are involved in the administration of large and complex research centers or programs.

There are staff or administrative and professional (A/P) faculty roles appropriate to many of these responsibility sets. The Office of Research Human Resources provides advice on the options available. Careful preparation of the position description by the principal investigator or department head/chair is essential in making a determination whether the position is staff, A/P faculty, or research faculty.

Like all research faculty members, those in the project associate series may be assigned a functional title in addition to their official faculty rank in order to facilitate their work. Given the range of activities included under sponsored programs, use of the functional title may be the most effective way to clarify the faculty member's role to internal or external constituencies. (A functional title may not be an official faculty rank other than that held by the research faculty member.)

## 6.4.1 Project Associate

The project associate rank is the entry, or most junior rank, for faculty members involved in sponsored projects that may deliver services to clients or involve program responsibilities other than traditional research. The work may vary from that appropriate for a relatively new professional to broader or more significant responsibilities expected of more experienced faculty personnel. Project associates work under supervision and carry out project responsibilities, such as technical assistance or consultation, which require professional preparation and application of accepted principles and practices of the field. They may be involved in preparing reports, documents, or manuals for review by project leaders. They may develop and deliver educational programs, or coordinate activities involving a number of project members. They may provide input into the preparation of proposals or supervise project personnel, but typically they have limited responsibility and authority in these areas. Project associates are not eligible to be principal investigators; however, requests for principal investigator status may be submitted to the Office of the Vice President for Research.

In keeping with all faculty appointments at the university, a master's degree in a relevant field is the minimum qualification for appointment as a project associate. The Office of Research Human Resources must approve exception requests for individuals with a bachelor's degree and significant related experience *before* an offer is extended.

#### 6.4.2 Senior Project Associate

The rank of senior research associate requires greater qualifications than the project associate either in education, experience, or both. The doctorate or a terminal degree in the field and some experience, or a lesser degree and substantial related experience, are required. The Office of Research Human Resources must approve exemption requests *before* an offer is extended. Senior project associates carry out project activities under limited supervision, providing a high level of professional service and expertise. They may create original materials or methods, requiring a high level of knowledge of and/or experience in the subject matter. Typically, they have some significant supervisory responsibility for project personnel. They may contribute to the conceptualization and preparation of project proposals, reports, resource acquisition, and interaction with stakeholders. Senior project associates are not eligible to serve as principal investigators; however, requests for principal investigator status may be submitted to the Office of the Vice President for Research.

#### 6.4.3 Project Director

Project director is the most senior rank in the project associate series. Appointment to this rank requires a doctorate or terminal degree and two or more years' experience, or a lesser degree and related experience of substantial scope and duration. By virtue of their expertise and experience, project directors or leaders make significant contributions to the conceptualization and implementation of the project. They typically have significant independence, responsibility, and authority for all aspects of the project. Project directors may be involved in the preparation of proposals, reports, or publications; and presentation of results to sponsors or other stakeholder groups. They have significant responsibility for hiring and managing project personnel, assigning resources, and evaluating project effectiveness. Those functioning in an administrative capacity may serve in a role equivalent to an assistant center director, managing complex and varied business and other administrative responsibilities for a large research center. Project directors may serve as principal investigators with the approval of the department head.

#### 6.5 Research Professor Ranks

The research professor ranks are designed for research faculty members whose appointments are expected to last more than one year and whose credentials are comparable to those of the tenure-track faculty of similar rank. This series is parallel to research scientist and senior research scientist, not necessarily above it. Appointment to research professor ranks is not appropriate for those with short-term or limited appointments since this would disadvantage the graduate students with whom they might work. Research faculty members whose primary appointment is in a research unit not affiliated with a degree-granting academic department may also be appointed to these ranks with appropriate credentials and approvals. Tenure cannot be earned in any of the research faculty ranks and service is not applicable toward the probationary period if the employee is subsequently appointed to a tenure-track position.

With approval by the degree-granting program, those appointed to any rank in the research professor series may direct graduate theses and dissertations and serve on graduate committees consistent with program and graduate school policy. Faculty members in this series may teach occasionally in their areas of expertise in accordance with guidelines below and by providing the appropriate credentials required of instructional faculty. (See section 2.5.9, "Faculty Credentialing Guidelines," or the provost's website.) Faculty in the research professor series may teach one course per year or two courses in a two-year period. They may teach more if funding is appropriately charged to the instructional budget and approved by the principal investigator/supervisor, department head/chair and dean.

At the discretion of the academic department, departmental faculty membership with or without voting privileges may be extended to an assistant, associate, or research professor. However, a research faculty member is not eligible to vote on matters relating to faculty appointment, retention, promotion, or tenure.

Promotion to a higher rank may be granted to research professorial faculty who have sources of continued funding and demonstrate outstanding accomplishments in appropriate activities. The *curriculum vita* together with annual reports, reprints of publications, reference letters, and other

similar documents comprise a dossier, which furnishes the principal basis for promotion decisions. Besides consideration of specific professional criteria, evaluation for promotion should consider the candidate's integrity, professional conduct, and ethics. To the extent that such considerations are significant factors in reaching a negative recommendation, they should be documented as part of the formal review process.

Research professorial faculty members being considered for promotion have their dossiers reviewed at as many as three levels: (1) by a departmental committee and the head or chair; (2) by a college committee and the dean; and (3) by the Office of the Vice President for Research. A parallel process for review, approved in advance by the senior vice president and provost and the vice president for research, is required for promotion of a member of the research professor series whose primary appointment is not in an academic department.

#### 6.5.1 Research Assistant Professor

Persons appointed as research assistant professors are required to have a doctoral degree or terminal degree appropriate to the field and credentials consistent with those for appointment to the rank of assistant professor. Research assistant professors are equivalent to research scientists in terms of their credentials; however, appointment to this rank indicates actual or anticipated involvement with the academic program.

Research assistant professors are expected to contribute significantly to the design and execution of research projects. They may serve as principal investigators with the approval of the department head. They carry out independent research in their field of specialization under general supervision. They may have supervisory responsibility for project personnel and contribute to project management.

Original appointments including rank, salary, and other conditions require the approval of the department head, dean (or next level administrator), and the Office of the Vice President for Research *before* an offer is extended.

#### **6.5.2** Research Associate Professor

Persons appointed to this rank are required to have a doctoral degree or terminal degree appropriate to the field and credentials consistent with those for appointment to the rank of associate professor. Research associate professors are typically responsible for design and execution of research projects and interpretation of research results. They are expected to serve as principal investigators and conduct independent research in their area of specialization. They may have significant supervisory responsibility for project personnel and contribute to project management.

Original appointments follow standard departmental procedures for tenured and tenure-track faculty appointments. Typically, this involves approval by the appropriate departmental/unit committee, followed by approval of the department/unit head, dean (or next level administrator), and Office of the Vice President for Research. Promotion to this rank requires evidence of continuous professional development, documentation of excellence in their disciplinary field,

contribution to research or creative activity supported through grants and contracts, and at least regional recognition. (See section 3.4.4, "Evaluation Procedures for Promotion and Tenure.") The promotion review and approval process is described in section 6.5, "Research Professor Ranks."

#### 6.5.3 Research Professor

Persons appointed to this rank are required to have a doctoral degree or terminal degree appropriate to the field and credentials consistent with those for appointment to the rank of professor. Research professors are typically responsible for design and execution of research projects and interpretation of research results. They are expected to serve as principal investigators and conduct independent research in their area of specialization. They may have significant supervisory responsibility for project personnel and contribute to project management.

Original appointments follow standard departmental procedures for tenured and tenure-track faculty appointments. Typically, this involves approval by the appropriate departmental/unit committee, followed by approval of the department/unit head, dean (or next level administrator), and the Office of the Vice President for Research. Promotion to this rank requires evidence of continuous professional development, documentation of excellence in their disciplinary field, outstanding research or creative activity supported by grants and contracts, and national and/or international recognition. (See section 3.4.4, "Evaluation Procedures for Promotion and Tenure.") The promotion review and approval process is described in section 6.5, "Research Professor Ranks."

#### 6.6 Matrix of Research Ranks

A matrix of qualifications, approval requirements, general expectations, salary guidelines, and typical position responsibilities for employees in the research faculty ranks is found on the Office of Research Human Resources website.

## 6.7 Affiliated Research Faculty

Occasionally, individuals outside of university employment are identified to team with university faculty to enhance research opportunities through departments. To support these associations, the university has developed the affiliated research faculty program. Affiliated research appointments may be made for individuals connected to specific academic departments, or may be made through research institutes or centers. Affiliated research appointments may also be established to facilitate research partnerships. This program may address occasions where a university faculty member has a spouse or partner who also has professional academic credentials, but who has not found appropriate employment opportunities. The vice president for research reviews nominations from departments, colleges, or the provost, and approves appropriate applications. Applicants must have academic credentials equivalent to those of university faculty, including the terminal degree normally required of faculty in the discipline. Applications for appointment must have the endorsement of the head or chair of the Virginia Tech department relevant to the applicant's discipline.

Typically, an affiliated research faculty member has unpaid adjunct status in the academic department of his/her discipline. The appointment is initiated by the host department submitting to the vice president for research a letter of support, the applicant's *curriculum vitae*, and the request for unpaid faculty affiliation with Virginia Tech academic department, approved by the department head or chair and dean or director. The appointments are renewable. The rank is the research faculty designation commensurate with the credentials of the candidate.

## **6.8** Searches for Research Faculty

Virginia Tech is committed to announcing the availability of positions so that a diverse pool of qualified candidates is developed for faculty positions of all types. In the case of research faculty, many of whom are hired on short-term grants and contracts sometimes requiring specialized skills and abilities, there is a need to balance the demands of the sponsored grant or contract with the institution's commitment to open and aggressive recruitment practices to attract a diverse workforce.

Competitive searches are required for salaried appointments to the research faculty, except in a limited number of cases warranting an exemption. Appointments of less than one year do not require a search. Searches are not required to fill a postdoctoral associate position or in the case where the person to be appointed is the author of the grant or is listed as the principal investigator or co-principal investigator, or appointment of a dual career hire. Consult the Office of Research Human Resources regarding other very limited exemptions. Occasionally the identified candidate may have previously worked on the project in a significant role and continuation of personnel is critical to the success of the project and a search may not be required.

## 6.9 Terms of Faculty Offer and Documentation of Credentials

Employment is contingent upon presentation of satisfactory documentation of credentials prior to employment. In accordance with federal law, new employees must also provide on the first day of their employment documentation of U.S. citizenship or lawful authorization to work in the United States.

The terms of faculty offer, including salary, effective date, rank, and other critical information concerning faculty appointments to the university is prepared by the department and approved by the department head for each new research faculty member. (See the Office of Research Human Resources website for terms of faculty offer templates for research faculty.) The contract specifies whether the appointment is *restricted* (usual appointment type for research faculty) or *regular*. See below for conditions under which research faculty appointments may be "regular."

All letters of appointment make reference to further terms and conditions of employment contained in the *Faculty Handbook*.

The <u>terms of faculty offer</u> for a restricted appointment clearly defines the length of the appointment. In cases where there is no expected opportunity for continuation, the document also serves as a notice of termination. The terms of faculty offer also makes clear that continuation of

appointment, even during the initial year, is subject to the availability of funds, the need for services, and satisfactory performance. Related offer or appointment letters should not contain promises that the hiring unit may not be able to keep; the university looks to the department to make good on defaults. Approval of the department head, dean, and the Office of the Vice President for Research are required *before* an offer is extended.

It is the responsibility of the hiring department to obtain and verify documentation of credentials on all faculty prior to employment. Information regarding appropriate credentials for instructional faculty is found in section 2.5.9, "Faculty Credentialing Guidelines," and on the provost's website.

## **6.9.1** Restricted Appointments

Salary support for research faculty typically comes from one or more sponsored projects. While some research faculty may be employed for years on successive grants, particularly in large, ongoing research programs, many are employed only for the duration of a specific funded project—in some cases six months and in other cases perhaps several years. Sponsored funding is seldom certain and never permanent. As a result, research faculty are usually appointed as "restricted" faculty members whose employment depends on availability of funding, the need for services, and satisfactory performance.

Procedures for terminating employees are addressed in section 6.14, "Termination Procedures for Research Faculty."

### **6.9.2** Multi-Year Restricted Appointments

An initial appointment, or a reappointment, for a research faculty member may be for up to three years provided that documented funding for the salary and fringe benefits is available from a multi-year grant, multiple grants, or other appropriate source(s) and that a search has been conducted or an approved exemption obtained. In such cases, the <u>terms of faculty offer</u> specifies the length of the restricted appointment.

If a person on restricted appointment will be continued, a reappointment is required. (See section 6.13, "Reappointment.") The reappointment contract again defines the conditions of the appointment. Any changes should be made explicit. If a salary increase is approved, it should be a part of the reappointment contract. The department is expected to execute the reappointment contract prior to the current contract end date.

Annual evaluation and merit adjustments occur on the same cycle as for all other faculty members. A performance review must be done annually, shared with the appointee, and documented in writing.

#### 6.9.3 Regular Appointments

A research center or program, including research entities established by state or board of visitors' action, may seek approval from the relevant department head, dean (or appropriate

administrators as defined in university policy 3020, "Centers and University Institutes: Financial and Administrative Policy and Procedures," and university policy 13005, "Centers and University Institutes: Establishment, Governance, and Programmatic Oversight") and the vice president for research (or designee) to advertise and fill certain research faculty positions as regular rather than restricted appointments under certain conditions. The criteria and expectations for such approval are as follows:

- The research program or center must have a documented record of substantial past funding, usually from diversified sources, generally over more than five years. In the case of a new center with multi-year funding, documentation of the new funding supported by the history of funding for the principal researchers may be considered. Research programs supported in full or in part by state funds are eligible for consideration for regular positions.
- For the purposes of this policy, "research programs or centers" are defined very broadly to include those entities established by state or board of visitors' action for the purposes of conducting research, as well as those that meet the definition of departmental, college, or university centers/institutes as defined in university policy 3020, "Centers and University Institutes: Financial and Administrative Policy and Procedures," and university policy 13005, "Centers and University Institutes: Establishment, Governance, and Programmatic Oversight." Departments intending to support specific research faculty members on state funds, whether or not those individuals participate in a "program" or "center," may also seek approval to appoint a research faculty member to a regular position.
- The research program or center must have documented prospects for continued funding at a level equal to or greater than its current funding.
- The unit must be able to guarantee payment of salary and fringe benefits from sponsored grants or contracts (or other appropriate sources) for a minimum of three years in order to advertise a research faculty position as a regular appointment.
- The unit must be able to guarantee funding of annual leave, sick leave, and salary following non-reappointment in the case of insufficient grant funds. The source of such payouts or salary support must be non-sponsored funds, such as indirect or state funds.
- The unit will advertise and conduct a national search for regular positions. Search exemptions may be approved only under certain very limited conditions, such as unique qualifications or unit restructuring. However, a search *must* be conducted if there is an intention of supporting an international candidate for permanent residency. In such a case, the hiring unit should work closely with the International Support Services Office to ensure compliance with current Immigration and Naturalization Service (INS) regulations. Appropriate documentation of the search process and selection is a critical element in supporting an application for permanent residency.
- In supporting the request for a regular faculty appointment, the unit and/or department (or approving unit) is thereby committing itself to covering shortfalls in funding between grants, or whenever there is insufficient funding for the salary, from other sources. Should this occur, duties may be reassigned in order to match the available source of funds.
- The Office of the Vice President for Research may grant approval to the unit to conduct searches for regular positions for a period of three to five years at which time the financial capabilities and commitments of the unit are reviewed and authorization granted for another three to five year period, if appropriate.

- Approval for the unit to advertise and fill some research faculty positions as regular appointments does not in any way suggest that all positions in the unit should be so designated. Indeed, careful thought should go into the shaping of such positions, the identification of talents and skills needed in the research group, and the availability of qualified individuals that may necessitate this more generous commitment of resources.
- Postdoctoral associates are considered temporary or short-term appointments by definition and are not appropriate for regular appointments.

Research faculty members on regular appointments are entitled to notice of non-reappointment, as specified in section 2.11, "Retirement, Resignation, and Non-Reappointment."

## 6.9.4 Calendar Year versus Academic Year Appointments

The nature of the research enterprise generally dictates that research faculty are appointed to a calendar year (12-month) position. However, there may be circumstances in which an academic year (nine-month) appointment is justified and appropriate. The justification for an academic year appointment should accompany the faculty search authorization or <u>search exemption request</u>.

Academic year research appointments do not earn or accrue annual leave. Faculty members on *academic year restricted* appointments earn and accrue sick leave at the rate of five hours per pay period during the academic year; those on *regular* appointments are entitled to 1040 hours of paid sick leave upon employment. Sick leave is addressed in more detail in section 2.9.6, "Sick Leave."

Research faculty members on academic year appointments may accept summer research wage (P14) or summer school teaching employment during the summer months in the same department or program, or elsewhere in the university. Research faculty on H1-B visas qualify for summer wage employment only in very limited cases. Contact the International Support Services Office to verify eligibility. Those who have documentation of additional months of funding from sponsored grants or contracts may be eligible for consideration of a calendar year research conversion under the terms of university policy 6200, "Policy on Research Extended Appointments." (See sections 2.6.3, "Research Extended Appointments," 6.15, "Effort Certification Compliance Issues for Research Faculty Members," and 10.4, "Effort Certification and Salary Charges to Sponsored Grants and Contracts.")

#### 6.10 Position Descriptions

Every research faculty member must have an up-to-date position description that forms the basis for initial hiring and assignment of duties and, through subsequent updates, for annual evaluation. Position descriptions should include a relatively detailed list or narrative description of assigned responsibilities and expectations for performance. A position description created and entered using the on-line hiring process. Further information is available on the Office of Research Human Resources website.

#### **6.11 Annual Evaluations**

Supervisors, usually principal investigators, are responsible for conducting an annual evaluation for any and all research faculty. A performance review must be shared with the appointee, and documented in writing. This documentation supports the request for annual merit and/or special adjustments. Giving regular and constructive feedback is essential to the development of employees, and it is the responsibility of the department head to be certain that research faculty are appropriately and consistently evaluated. The cycle for evaluation is the same as that for all other faculty members.

## 6.12 Merit and Special Adjustments

Research faculty members, both regular and restricted, are eligible for annual merit adjustments (and special adjustments when necessary and appropriate) on the same cycle as all other faculty members. The result is then communicated to the research faculty member. Promotions will require generating new terms of faculty offers.

Following the annual evaluation, supervisors make recommendations to department heads that, in turn, make their recommendations to college and university officials. Review and approval by the Office of the Vice President for Research is required for exceptions and promotions. Final recommendations are approved by the board of visitors. The result is then communicated to the research faculty member. Merit recommendations for research faculty members are generally expected to track the merit adjustments for teaching and research faculty members. Guidance on this point is issued annually by the Office of the Vice President for Research. In some cases, available funds may limit, delay, or even preclude a merit adjustment. However, performance evaluation and feedback to the employee are still required even if a merit adjustment is not possible.

Special adjustments, outside of the annual merit process, may be recommended in accordance with the guidelines for faculty salary adjustments available from the Office of Research Human Resources. Examples of such adjustments might be for promotion in rank, increased responsibility, retention, equity, or other reasons critical to retaining productive research faculty members and compensating them fairly. Justification and appropriate approvals are required as outlined in the guidelines.

#### 6.13 Reappointment

Research faculty members on restricted appointment whose employment will be continued are issued a reappointment specifying the new ending date for their appointment. Reappointments typically occur on the anniversary of the hire date or in relation to the funding cycle of the grant or appointment. Reappointments may be for less than one year in such situations where additional funding is anticipated but not confirmed. Multiple reappointments may occur during the span of the research faculty member's employment. The reappointment date may be adjusted based as a result of other significant actions (e.g., promotion), or by issuing appointments that move the employee to the same effective date as their merit adjustment. Multi-year reappointments are possible in cases where funding is available for the proposed reappointment period.

Research faculty members on regular appointment are typically reappointed on a cycle coinciding with the merit adjustment. Reappointments may be for one or more years. Since faculty members on regular appointments do not have fixed ending dates to their contracts, they must receive written notice of non-reappointment as described section 2.11, "Retirement, Resignation, and Non-Reappointment," should it become necessary to end the appointment.

#### 6.14 Termination Procedures for Research Faculty

Members of the research faculty may be removed from their position by one of the following three procedures: removal for just cause, non-reappointment, or termination of position because of insufficient funds or no further need for services. The period of notice for non-reappointment of research faculty members on regular appointments is addressed in section 2.11, "Retirement, Resignation, and Non-Reappointment."

#### 6.14.1 Dismissal for Cause

Research faculty members may be removed for just cause. Stated causes for removal shall include, but are not limited to: professional incompetence; unacceptable or unsatisfactory performance after due notice; unethical conduct or misconduct that interferes with the capacity of the employee to perform effectively the requirements of the position; violation of university policy; falsification of credentials, experience, leave reports, or other official employment documents. Filing a grievance does not constitute just cause for termination.

When it becomes necessary to terminate a member of the research faculty for unsatisfactory performance prior to the end of the appointment period, the following procedures apply:

- 1. When the faculty supervisor determines that performance is unsatisfactory, the supervisor writes a letter to the individual detailing the areas of performance that are deficient. This letter should indicate specific expectations of improvement by the employee during a specified time period of not less than 30 calendar days. The department head and college dean receive copies. In cases where there is some likelihood of threat to health or safety, the 30-day period may be waived.
- 2. At the end of the above period, the faculty supervisor must again write the research faculty member with an evaluation of his/her performance during the interim since the first letter, with copies to the department head and college dean. If performance continues to be unsatisfactory, this second letter may contain a termination notice. The termination notice has an effective date 45 calendar days from the date of the second letter.
- 3. In the event of termination, the research faculty member may appeal to the department head. Should the appeal process be initiated, the termination is held in abeyance until the appeal process is complete.

The appeal must be made in writing within five working days of receipt of the letter. (If the department head has a conflict of interest, the head refers the matter to the college dean.)

The department head (or dean) must respond in writing within 10 working days. If the recommendations of the department head (or dean) are unsatisfactory to either party, an appeal may be made to the vice president for research in writing within five working days.

The vice president for research appoints a committee of three members of the general faculty who make recommendations to the senior vice president and provost within 10 working days.

The decision of the provost is final and is rendered within 10 working days of receiving the report.

The above time limits of the appeal process may be altered by extenuating circumstances and the agreement of both parties.

4. If the research faculty member is a member of an interdisciplinary research center, the center director as well as the department head and college dean of the faculty supervisor are copied on all correspondence.

## 6.14.2 Termination of Appointment for Faculty on Restricted Contracts

Occasionally a sponsor terminates funding before the end of a contract, or directs a change in the research program resulting in the need to terminate the services of an employee. While principal investigators and research centers are encouraged to make every effort to assure continuity of employment to individuals performing satisfactorily, there are circumstances in which this may not be possible or in the best interest of the research program or university. Research faculty appointments may be terminated in the case where there are insufficient funds or no further need for services. The date of termination for a restricted faculty member is at least 30 calendar days from the date of notification. A faculty member on a regular appointment is entitled to notice of non-reappointment, as stated above. A proposed notice of termination because of insufficient funds or lack of need for services requires the approval of the department head, and dean (or appropriate administrator), and the Office of the Vice President for Research Human Resources.

## **Effort Certification Compliance Issues for Research Faculty Members**

As described in section 10.4, "Effort Certification and Salary Charges to Sponsored Grants and Contracts," a research faculty member with regular, well-defined responsibilities for new proposal preparation, teaching, or administrative duties is prohibited from charging 100 percent of salary to sponsored projects during an effort reporting period in which such activity occurred, unless those activities are specifically allowed on the sponsored project.

Research faculty members are typically on standard 12-month appointments, which earn and accrue annual leave by university policy. Use of annual leave is recognized as an acceptable charge to a sponsored project when such leave is part of the standard university appointment.

#### 6.16 Instructional Responsibilities for Research Faculty Members

The primary responsibility of a research faculty appointment is the conduct of research and contribution to the institution's research mission through the acquisition of and successful implementation of sponsored grants and contracts. Federal guidelines require truthful and auditable documentation of the faculty member's efforts on a semester basis. If the faculty member's salary is paid for by sponsored grants and contracts then there is a concomitant expectation that the faculty member's time is allocated to those projects.

While keeping the primacy of the research role in mind, there are circumstances in which the university and its instructional programs benefit from the occasional participation of research faculty member with the appropriate credentials, expertise, and interest. The normal limitation on teaching by research faculty members is one (three-credit) course per academic year, or no more than two courses in a two-year period. The principal investigator/supervisor, department head/chair, and dean must approve exceptions. The academic department provides instructional funding for the teaching appointment and research duties are adjusted accordingly. A three-credit course equates to .25 FTE during an academic semester; this is the usual basis for salary charges to the instructional budget.

Research faculty members with appropriate credentials and experience may serve on graduate student committees in accordance with policies of the graduate school and the department. Those with appointment to the research professor ranks may chair a committee, if approved by the degree-granting department. Involvement in supervision of graduate student research may be directly related to fulfillment of sponsored grants and contracts and thus may have a synergistic effect.

Contributions to the instructional program are monitored and evaluated by the academic department and by the supervisor.

#### **6.17** Faculty Grievance Policy and Procedures

The following procedure is provided as the means for resolution of grievances against a supervisor or member(s) of the university administration brought by members of the research faculty.

#### **6.17.1** Faculty Reconciliation and Mediation Services

**Informal Dialogue**: It should be possible to resolve most faculty concerns or complaints through informal communication among colleagues working together in the academic enterprise. Accordingly, a faculty member who feels he or she has a grievance is encouraged to take it to his or her immediate supervisor in the normal collegial spirit of problem solving rather than as a confrontation between adversaries.

**Reconciliation**: At the initiation of the grievance procedure, or at any earlier time, the grievant may request the assistance of the Faculty Senate Committee on Reconciliation in fashioning an

equitable solution. Contacting the Faculty Senate Committee on Reconciliation is not required in filing a grievance, but it may be useful if the grievant feels that the issue may be amenable to, but will require time for, negotiation; or if the grievant is unsure whether his or her concern is a legitimate issue for a grievance; or if personal relations between the parties involved in the grievance have become strained.

For a potential grievance issue to qualify for consideration by the Faculty Senate Committee on Reconciliation, the grievant must contact the Faculty Senate Committee on Reconciliation within 30 calendar days of the time when the grievant knew or should have known of the event or action that is the basis for the potential grievance, just as if beginning the regular grievance process. If the grievant requests assistance from the Faculty Senate Committee on Reconciliation, that committee requests a postponement of the time limits involved in the grievance procedure while it deals with the case. The request is submitted in writing to the vice provost for faculty affairs by the chair of the Faculty Senate Committee on Reconciliation. Also, the grievant should reach an understanding with the Faculty Senate Committee on Reconciliation of the time frame planned for that committee's work on the case, such time not to exceed 60 calendar days.

Faculty members may also consult the Faculty Senate Committee on Reconciliation about serious disagreements with immediate supervisors or other university administrators concerning issues that may not be eligible for consideration within the grievance process. In such instances, the committee contacts the relevant administrator to determine if there is an interest and willingness to explore informal resolution of the dispute; it is not necessary to notify the Office of the Senior Vice President and Provost.

Additional information regarding the faculty reconciliation process is available on the <u>provost's</u> <u>website</u>.

**Mediation**: Mediation is a voluntary, confidential process through which trained neutral third persons (mediators) assist people to express their concerns and develop solutions to the dispute in a safe and structured environment. Because mediation is voluntary, both parties must agree to participate in order for mediation to occur. Faculty members and supervisors are encouraged to consider using mediation to resolve disputes between them, or to help address a conflict between a faculty member and another member of the Virginia Tech community.

**Role of Mediators**: Mediators do not make judgments, determine facts, or decide the outcome; instead they facilitate discussion between the participants, who identify the solutions best suited to their situation. No agreement is made unless and until it is acceptable to the participants.

Requesting Mediation: Mediation is available at any time, without the filing of a grievance. Additionally, mediation may be requested by any party during the grievance process prior to step four. If, after the initiation of a formal grievance, both parties agree to participate in mediation, the grievance is placed on administrative hold until the mediation process is complete. If the parties come to a resolution of the dispute through mediation, the parties are responsible to each other for ensuring that the provisions of the agreement are followed. In the event that the parties are not able to reach a mutual resolution to the dispute through mediation, the grievant may request that the grievance be reactivated and the process continues.

Mediation differs from faculty reconciliation in that mediators do not engage in fact-finding or in evaluation of decisions. Both mediation and reconciliation, however, are voluntary; no party is required to participate in either process.

To learn more about mediation and other forms of informal conflict resolution processes, contact the conflict resolution program manager in human resources at (540) 231-9331.

#### 6.17.2 The Formal Grievance Procedure

If the assistance of the Faculty Senate <u>Committee on Reconciliation</u> is not desired or is not requested; or if that committee determines that it cannot provide assistance in the matter; or if the grievant finds that the length of time the Committee on Reconciliation plans or takes with the case is excessive; or if the grievant is not satisfied with the recommendations of that committee, the grievant may pursue the issue as a formal grievance through the following procedure. Department heads or chairs, deans, directors, and other administrative faculty will cooperate with the grievant in the mechanics of processing the grievance, but the grievant alone is responsible for preparation of his or her case.

1. Step one: The grievant shall provide a written description of the event or action on the faculty grievance form and relevant supporting documentation of the grievance within 30 calendar days of the date when he or she should have known the event or action that is the basis of the grievance to his or her immediate supervisor. Research faculty grievance forms are available on the <u>provost's website</u>.

The supervisor meets with the grievant and provides a written response within five weekdays to the grievant citing reasons for action taken or not taken and the final decision. If the supervisor's response is satisfactory to the grievant, that ends the matter.

If the response is not satisfactory to the grievant or the supervisor does not respond within five weekdays, the grievant will indicate "no resolution" on the faculty grievance form, return a copy of the form to the immediate supervisor and proceed to step two.

2. Step two: The grievant advances the written description of the event or action, relevant supporting documentation, research faculty grievance form and the written response of the immediate supervisor (or statement of non-response if the supervisor did not respond within five the weekdays at step one) to the next level administrator within five weekdays of receipt of the decision. The next level administrator is the department head. If the department head has a conflict of interest, he or she refers the matter to the college dean. The administrator involved at this level is hereafter referred to as the second-level administrator.

Within five weekdays of receipt of the research faculty grievance form, the second-level administrator meets with the grievant and may request the presence of the immediate supervisor. The grievant may similarly request that a representative of his or her choice from among the university faculty be present. Unless the grievant is represented by a member of

the faculty who is also a lawyer, the second-level administrator does not have legal counsel present.

The second-level administrator returns the research faculty grievance form and provides a written response and final decision to the grievant with copies to the immediate supervisor within five weekdays after the meeting. The decision of the second-level administrator takes precedence over the decision of the immediate supervisor. If the response is satisfactory to the grievant, that ends the matter. If the response is not satisfactory or if there is no response within five weekdays by the second level administrator, the grievant may so indicate on the grievance form and return a copy to the second-level administrator and proceed to step three.

3. Step three: The grievant may advance the written description of the event or action, relevant supporting documentation grievance form and written responses of the immediate supervisor and second-level administrator to the dean within five weekdays of receipt of the decision in step two.

The dean will inform the immediate supervisor within five weekdays that the procedure has advanced to step three.

Within five weekdays of receipt of the grievance form, the dean meets with the grievant and may request the presence of the immediate supervisor. The grievant may similarly request that a representative of his or her choice from among the university faculty may be present. Unless the grievant is represented by a member of the faculty who is also a lawyer, the dean does not have legal counsel present.

The dean shall return the grievance form and provide a written response and final decision to the grievant with copies to the immediate supervisor and second-level administrator within five weekdays after the meeting. The decision of the dean takes precedence over the decision of the second-level administrator. If the response is satisfactory to the grievant, the procedure is terminated. If the response is not satisfactory or if there is no response within five weekdays by the dean, the grievant may so indicate on the research faculty grievance form and return a copy to the dean and proceed to step four.

4. Step four: The grievant will advance the written description of the event or action, relevant supporting documentation, research faculty grievance form, and written responses of the immediate supervisor, second-level administrator, and dean to the senior vice president and provost within five weekdays of the decision of step four. The provost will make a decision and may wish to consult faculty members unfamiliar with the grievance for an opinion.

The decision of the senior vice president and provost will be final and will be rendered to the grievant and immediate supervisor within five weekdays of receipt of the grievance.

The above time limits of the appeal process may be altered by extenuating circumstances and the agreement of both parties.

If the research faculty member is a member of an interdisciplinary research center, the center director as well as the department head and dean are copied on all correspondence.

## 6.17.3 Timeliness of Grievance and Procedural Compliance

A grievance must be brought forward in a timely manner. It is the responsibility of the grievant to initiate the grievance process within 30 calendar days of the time when he or she knew or should have known of the event or action that is the basis for the grievance. The university administration is not required to accept a grievance for processing if the grievant does not meet the 30-day deadline, except in cases of demonstrated good cause.

Scheduled commitments made prior to the time of filing or advancement of a grievance that preclude action by either of the parties to the grievance automatically extend time limits for their duration unless this would be demonstrably harmful to the fair processing of the grievance. In such cases, on written request by the grievant to the appropriate office for that step, the grievance is advanced to the next step in the grievance process.

If the grievant does not follow the time limits specified in the grievance procedure it is assumed that he or she accepted the last proposed resolution as satisfactory. If the grievant desires to advance the grievance after the appropriate specified time limits have lapsed, the administrator who receives the late submission notifies the chair of the Faculty Review Committee in writing, and the chair of the Faculty Review Committee determines if there was good cause for the delay. If so, the grievance proceeds. If not, the process ends with the most recently proposed resolution in force. The finding on the matter by the chair of the Faculty Review Committee is communicated to both parties in writing.

If either party to a grievance charges the other with procedural violations other than time limit issues, a special committee of two research faculty appointed by the vice president for research and the chair of the Faculty Review Committee is convened to rule on the question, as in disputes about the validity of issues qualifying for the grievance procedure. The special committee has the following options. It can either find no significant procedural violation occurred, in which case the grievance process continues unaffected, or that a significant procedural violation did occur. If the administrator committed a significant procedural violation, the grievance automatically qualifies for advancement to the next step in the grievance process. If the grievant committed a significant procedural violation, the grievance process ends at that point with the last proposed resolution established as the final disposition of the case.

#### 6.17.4 Valid Issues for Grievance

For this process, a grievance is defined as a complaint by a faculty member alleging a violation, misinterpretation, or incorrect application of a policy, procedure, or practice of the university that directly affects the grievant. Some examples of valid issues for filing a grievance are: improperly or unfairly determined personnel decisions that result in an unsatisfactory annual performance evaluation, unreasonable merit adjustment or salary level, or excessive teaching load/work assignments; substantive violations of promotion procedures (see section 6.5, "Research

Professor Ranks"); reprisals; substantive error in the application of policy; and matters relating to academic freedom.

Issues not open to grievance: While most faculty disputes with the university administration may be dealt with by this grievance policy, the following issues may not be made the subject of a grievance: determination of policy appropriately promulgated by the university administration or the university governance system; those items falling within the jurisdiction of other university policies and procedures (for example, complaints of unlawful discrimination or harassment, or an appeal related to the merits of a promotion decision); the contents of personnel policies, procedures, rules, regulations, ordinances, and statutes; the routine assignment of university resources (e.g., space, operating funds, parking, etc.); normal actions taken, or recommendations made, by administrators or committee members acting in an official capacity in the grievance process; termination of appointment by removal for just cause, non-reappointment, or abolition of position; or allegations of misconduct in scholarly activities.

Adjudication of disputes on the validity of issues qualifying for consideration under the faculty grievance procedures: If a university administrator rules that an issue does not qualify for the grievance process, the grievant may write to the chair of the Faculty Review Committee within five weekdays of receiving such notification and request a ruling from a special committee consisting of the president of the Faculty Senate, the chair of the Faculty Senate Committee on Reconciliation, and the chair of the Faculty Review Committee. The special committee considers the matter (including consultations with both parties if deemed necessary) and rules by majority vote on the admissibility of the matter to the grievance process. This special committee is called together by the chair of the Faculty Review Committee, who also sends a written report of the results of the deliberations of the committee to all parties concerned.

## 6.17.5 Particular Concerns and Definitions

Time limits are subject to extension by written agreement of both parties. The grievant and the administrator involved at that particular step of the discussion are the makers of such an agreement. (An agreement form to extend the grievance response time is available on the <u>provost's website</u>.)

Grievances that advance to step four during or close to the summer and/or teaching breaks during the academic year may require some extension of the stipulated time limits. The principals and the chair of the Faculty Review Committee negotiate such an extension. Every effort is made, however, to stay within the stipulated time limits.

If a faculty member is away from his or her assigned work location at the time he or she discovers the event or action that is the basis for a grievance, the 30-day period during which the grievant must meet with his or her immediate supervisor to initiate the grievance process begins when the faculty member returns to his or her assigned work location. If the date of return causes a delay of such length that the grievance, or its resolution, is not timely, the grievant may submit the grievance in writing to the immediate supervisor (step two), omitting personal meetings until such time as the faculty member returns to his or her assigned work location.

"Weekdays," as used in this procedure, include Monday through Friday only and only when those days are not national, state, or religious holidays relevant to the principals in the grievance.

To protect a grievant from undue pressure in the pursuit of a grievance, if a grievant becomes ill and takes sick leave the grievance process stops until such time as the grievant is able to resume his or her duties. Exceptions to this provision are made at the request of the grievant, but only if the grievant obtains and produces medical certification that proceeding with the grievance will not be harmful to the health of the grievant, or exacerbate the ailment that required taking sick leave.

All costs of legal counsel employed by a grievant are borne by the grievant.

If a grievant is employed away from Blacksburg, and he or she is required to travel away from their duty station in resolution of their grievance, the university pays all travel costs permitted under state regulations.

In the event that a faculty member discovers he or she has a grievance about actions by an administrator above the level of his or her immediate supervisor that directly involve the faculty member, or with actions by an administrator not in his or her department that directly involve the faculty member, the grievant initiates the grievance process by seeking the mediation of his or her immediate supervisor within 30 calendar days of the discovery of the event or action that is the basis for the grievance. If that effort does not resolve the grievance satisfactorily, the grievant, after consulting his or her immediate supervisor, may file the faculty grievance form at the appropriate level or with the appropriate administrative office to initiate response from the administrator perceived as the source of the action causing the grievance. The grievance process then proceeds from that level onward in normal fashion.

A grievance filed by a faculty member concerning an action of the provost is handled by the chair of the Faculty Review Committee and a regular impartial hearing panel, but the findings and recommendations of the hearing panel are sent to the president of the university for his or her ruling, rather than to the provost. A grievance filed by a faculty member concerning an action of the president of the university is dealt with by a special panel appointed by the provost in consultation with the president of the Faculty Senate.

Any final resolution of a grievance must be consonant with the laws of the Commonwealth of Virginia and university policy.

Once a grievance is resolved, either to the satisfaction of the grievant, or if not to the satisfaction of the grievant, by the action of the provost in consonance with the hearing panel recommendations, or by the ruling of the president, that specific grievance is closed and may not be made the subject of another grievance.

## 6.17.6 Overview of the Formal Grievance Process for Research Faculty

Below is an <u>abbreviated</u> overview of the grievance process and deadlines. Refer to section 6.17.2, "The Formal Grievance Procedure," for specific details and options available in each step of the grievance process.

Time limits are subject to extension by written agreement of both parties. The grievant and the administrator involved at that particular step of the discussion are the makers of such an agreement. (An agreement form to extend the grievance response time is available on the provost's website.)

tten the
the
e to
evel
nce
1100
to
ιο
, to
ιο
iate
Tate
tha
the
e to
the
ing.
j

The provost's decision is final.

#### CHAPTER SEVEN

# 7.0 Employment Policies and Procedures for Administrative and Professional Faculty

The categories of administrative and professional (A/P) faculty were introduced by the state in 1989. Under the Higher Education Restructuring Act, level-three institutions were given flexibility to redefine and realign university personnel in the A/P categories. In August 2008, the board of visitors adopted slightly modified titles, definitions, and qualification criteria for these positions to address changes that better meet the university's need to recruit and retain a highly trained workforce by converting higher level staff positions to A/P faculty appointments. The revised titles, definitions, and criteria recognize the professional training and experience required of a wide variety of positions at the university.

**Senior Administrators**: Administrative faculty members are referred to as senior administrators to more accurately reflect the nature of these appointments.

Senior administrators perform work related to the management of the educational and general activities of the institution at least 50 percent or more of their contractual time. Senior administrators typically serve in executive level leadership roles such as vice president, dean, and assistant or associate vice president or dean. The organizational reporting relationship is normally not lower than three levels below the president or the next most senior position to the president.

#### Qualifications criteria:

- Must have an advanced degree or training and work experience at a level that equates to an advanced degree; a master's degree is the typical minimum entry qualification. Many of these positions, particularly academic leadership roles, may require a terminal degree.
- Must regularly exercise discretionary actions.

**Managers and Professionals**: Professional faculty are referred to as managers and professionals to reflect the variety of roles and appointments included in this category.

Professional faculty positions designated as continued appointment-track are not governed by policies outlined in this chapter. Policies applicable to library faculty on the continued appointment-track and extension faculty on the continued appointment-track are described in chapter four. Policies in this chapter *do* apply to extension agents, who are A/P faculty members.

Managers (and directors) typically have responsibility for supervision and evaluation of a significant number of staff and/or professional faculty, and budgetary responsibility for their unit or a substantive program. Incumbents exercise discretion and independent judgment and perform managerial or directorial functions at least 50 percent of their contractual appointment. Managers and directors typically report to a senior administrator and provide leadership and oversight for their unit or a significant program.

Professionals provide direct service to students, other university constituencies, or clients external to the university as part of the university's missions of learning, discovery, and engagement. They may direct or provide support for academic, administrative, extension, outreach, athletic, or other programs. Professional faculty may also provide vital university functions such as information technology, budget or finance, human resources, public relations, development, and architectural or engineering functions. Incumbents must regularly exercise professional discretion and judgment and are expected to take professional initiative in carrying out their primary roles and assignments. Professionals include, but are not limited to, extension agents, librarians, coaches, physicians, lawyers, engineers, architects, student or academic affairs professionals, development officers, specialists in public relations, human resources, or information technology, and financial specialists.

### Qualifications criteria:

- Must have an advanced degree or training and work experience at a level that equates to an
  advanced degree; although a master's degree is the typical entry qualification, this category
  also includes individuals with a bachelor's degree and professional training or certifications
  critical to their fields. In some cases, individuals with substantial professional level
  experience or expertise that equates to the minimum educational qualifications may be
  considered for appointment.
- Must regularly exercise discretionary actions.
- The work must be intellectual and varied in character, in contrast to positions that carry out more standardized or routine tasks and activities.

## 7.1 Faculty Rank and Title

Members of the administrative and professional faculty have the nominal faculty rank of lecturer and a functional title appropriate to the position (e.g., lecturer and assistant dean of students). Professional development is recognized by salary adjustment and/or functional title advancement rather than promotion in faculty rank. Administrative and professional faculty to whom initial ranks other than lecturer were assigned prior to July 1, 1983 retain such ranks.

## 7.1.1 Standard Faculty Rank

The rank of lecturer is generally reserved for administrative and professional faculty appointments. A master's degree or significant professional experience is the minimum expectation for appointment to the administrative and professional faculty. Tenure cannot be awarded at this rank. Promotion for the administrative and professional faculty is usually recognized by changes in functional title rather than promotion in rank. Appointments are renewable annually.

Members of the administrative and professional faculty whose credentials and professional development activities are similar to those of instructional faculty and who are involved in the instructional mission of an academic department may be assigned a standard faculty rank. Initial assignment of a standard faculty rank (assistant, associate, or full professor) for non-tenure-track administrative and professional faculty is recommended by using standard personnel appointment or review procedures for that department, such as the departmental promotion and

tenure committee or personnel committee. A faculty member may not serve on any committee that is evaluating a spouse, family member, or other individual with whom the faculty member has a close personal relationship (see section 2.16.2, "Potential Conflicts Involving Spouses and Immediate Family Members"). Departmental recommendations for rank are forwarded to the dean and subsequently to the provost for administrative approval.

Administrative and professional faculty who hold a standard faculty rank with an academic department are considered for promotion in rank by submitting their credentials through the usual department promotion process for collegiate faculty, including consideration by the college- and university-level committees. The department head or chair works closely with the committee to develop reasonable guidelines for consideration of rank promotions for A/P faculty affiliated with the department and with the individual A/P faculty member so that the appropriate materials are submitted for committee consideration. Appeal of a negative promotion decision is handled in accordance with appeal procedures for collegiate faculty. (See section 3.4.5, "Appeals of Decisions on [Reappointment or Promotion].")

The assignment of, or change in, a standard faculty rank carries no aspect of tenure.

## 7.2 Policies Related to Administrative and Professional Faculty Appointments

#### 7.2.1 Protection of Academic Freedom

The university recognizes the need to protect the academic freedom of administrative and professional faculty members when their responsibilities include instruction, direct support of the academic programs of the university, or an activity in which academic freedom is respected.

## 7.2.2 Initial Appointment and Reappointment

Search procedures for administrative and professional faculty positions are the same as those for instructional faculty positions. Faculty search resources are available on the <u>human resources</u> website. Please refer to that website for detailed information on the search process. Further guidance for searches involving certain administrative faculty positions (e.g., dean, vice president, or president) is included in other sections of this handbook.

Some administrative and professional faculty may be appointed on a "restricted" rather than "regular" appointment. The special conditions of temporary, restricted appointments are described in section 2.6.4, "Restricted Appointments."

Appointments to administrative and professional faculty positions are term appointments. No aspects of tenure or continued appointment are involved. Initial appointments to an administrative or professional faculty position are normally for a minimum of one calendar year or academic year, as appropriate. If the evaluation is positive, the faculty member can normally expect to continue employment for another year. Important determinants in any reappointment decision are job performance at a high level of productivity and effectiveness and the continued need of the university for the scope and level of services being provided.

As a means to address budget reductions and with a six-month written notice to the employee, the university is authorized to convert administrative and professional faculty on calendar year appointments to the traditional academic year appointment period, or to an alternate 9-, 10-, or 11-month appointment.

Appointment to an administrative or professional faculty position does not carry any aspect of tenure. Faculty who achieve tenure or continued appointment in an academic department retain their tenure or continued appointment. On occasion, requests are made by outside candidates, particularly for administrative appointments in academic areas, for an appointment with tenure and rank in an academic department comparable to that held at the previous institution. Review and approval by the department head or chair, the departmental promotion and tenure or continued appointment committee, the dean, the provost, and the president is required before a decision is made to extend a firm offer that includes the granting of tenure or continued appointment. Review and approval by the department promotion and tenure or continued appointment committee shall be sought before a decision is made to extend a firm offer at a rank other than lecturer to an administrative or professional faculty.

## 7.2.3 Degree Verification

Prior to employment, human resources verifies the highest degree earned for salaried administrative and professional faculty members. Verification is conducted through the National Student Clearinghouse or other certified vendor. In cases where human resources is unable to complete the verification for any reason, the candidate is responsible for providing an original transcript to human resources within 30 days of notification by the university. Employment is contingent upon verification of appropriate credentials.

Administrative or professional faculty who teach credit courses are responsible for providing an original transcript to the teaching department for verification of appropriate credentials in accordance with the faculty credentialing guidelines found in section 2.5.9, "Faculty Credentialing Guidelines," and on the provost's website.

## 7.2.4 Academic Year Appointments for Administrative and Professional Faculty

Most administrative and professional faculty positions are 12-month appointments with the appointment period extending from July 1 to June 30. Some administrative and professional faculty positions are established as 9-, 10-, or 11-month appointments based on programmatic need.

Regular administrative and professional (A/P) faculty members on academic year appointment earn annual leave only during the period of their appointment at the same rate as regular A/P faculty members on calendar year appointment. That is, two days (16 hours) of annual leave credit are earned per month in accordance with leave regulations; after 20 years of continuous employment by the commonwealth, 18 hours of annual leave are earned per month.

#### 7.3 Annual Evaluations

The supervisor is responsible for maintaining an up-to-date job description for each administrative and professional faculty member in the unit and for determining acceptable standards of performance. Goals and objectives are developed annually in consultation with the faculty member. These should relate closely to the functional title and job description of the position and should become criteria for judging professional performance at the end of the performance cycle. All administrative and professional faculty members should complete an annual faculty activity report at a time determined by the appropriate administrator, but usually near the end of the academic or fiscal year, referencing their goals and objectives and citing their successes, shortfalls, and future directions. Additional items to report are service to the university, creative scholarship, and other professional activities and recognitions during the year. The performance of each administrative and professional faculty member is evaluated annually in a discussion with the supervisor and by written response. The annual faculty activity report and evaluation are part of the basis for salary adjustments and other personnel matters.

## 7.3.1 Periodic Evaluation of Deans, Vice Presidents, and Directors of Major Organizational Units

In addition to annual reviews by the supervisor, periodic reviews (approximately every five years) are required for vice presidents and directors of major organizational units. Administrators serving in other senior leadership roles should also be considered for periodic review where appropriate and identified by the supervisor and president. The review is intended to be formative and to assist both the administrator and the department to improve.

#### 7.3.1.1 Reviews for Senior Academic Administrators

University policy 6105, "Periodic Evaluation of Academic Deans," outlines the procedures used in the conduct of a periodic review of academic deans. The same general process has been adapted for the periodic reviews of other senior academic leaders who report to the senior vice president and provost, such as the vice presidents for research, graduate education, undergraduate education, outreach and international affairs, and student affairs, and the vice provosts or associate vice presidents. The periodic reviews occur every five years, are conducted by an appointed committee with diversified membership, and include an extensive survey and interview process. For periodic reviews of senior administrators reporting directly to the provost, the provost oversees the appointment of members and provides the charge to the committee, and the committee submits a confidential report to the provost with findings and recommendations. Reviews conducted for directors of major units are managed by the relevant vice president to whom they report. The periodic review is used to inform the decision for reappointment.

The determination of participants, the schedule, and specific procedures for periodic review of academic administrators are the responsibility of the senior vice president and provost.

## 7.3.1.2 Reviews for the Provost, Vice Presidents, and Senior Staff Reporting to the President, and Other Non-Academic Administrators

The <u>Administrative Evaluation</u> and <u>Development Program</u> documents the process to be used for periodic evaluations and executive development for the vice presidents reporting directly to the president, the president's direct reports, and direct reports to the vice presidents. The vice presidents participate in an external leadership assessment and development program appropriate to the stage of their career. The selected program must include an opportunity for 360-degree feedback, with responses shared with the president. Participation in the evaluation and development program is required within two years of initial appointment and every five years thereafter. The president may also solicit feedback from other individuals and/or constituencies as may be appropriate to the administrator being reviewed. Preparation of an individual development plan and executive coaching for a limited time period are also standard elements of the process. The president provides oral and written feedback to the administrator.

The review process for those senior staff who report directly to the president and direct reports to the various vice presidents will follow a similar pattern, generally using internal assessment instruments and professional development resources.

The determination of participants for the Administrative Evaluation and Development Program, the schedule, and specific procedures for periodic evaluations of administrators reporting through the president are the responsibility of the president's office.

## 7.4 Salary Adjustments

Salary increases are based on merit; they are not automatic. Recommendations for salary adjustments are approved by the appropriate supervisor, dean (where relevant), vice president, and president prior to approval by the board of visitors.

Merit encompasses more than adequate performance of assigned duties. Although no faculty member can simultaneously engage successfully in activities in all areas below, administrative and professional faculty should work with their supervisor to develop a long-range plan to become involved and demonstrate a high level of competence in the areas below.

- **Performance**: Administrative and professional faculty members have an obligation to maintain a high level of performance in carrying out their job-related duties and responsibilities. A high level of competence in the performance of one's duties is the major factor in any evaluation. Evaluations are based upon standards set by the supervisor with the participation of the faculty member and relate closely to the duties inherent in the functional title and job description of the position. Annually set expectations are important criteria for judging professional job performance at the end of the performance cycle.
- **Service to the university**: Historically, Virginia Tech depends on the administrative and professional faculty for service on a wide variety of committees and as leaders and staff for important university projects and initiatives. Demonstrated participation in and leadership of departmental or university committees, special university-wide assignments, or similar

activity on behalf of important university priorities is expected of those who seek higher-level administrative positions.

- **Professional and scholarly activities**: Administrative and professional faculty have an obligation to maintain a high level of professional competence and to stay abreast of developments in their field. Effective administrators also benefit from active involvement in the intellectual and scholarly development of one's field, which often leads to contributions to the profession.
- Teaching in appropriate credit or non-credit programs: Many administrative and professional faculty at Virginia Tech contributed directly to academic programs by teaching undergraduate or graduate courses or becoming involved in continuing and professional education activities. (See below for information regarding the teaching of credit classes and overload compensation for administrative and professional faculty.)

Salary adjustments may also be recommended to address such issues as equity, increased responsibility, retention for especially meritorious employee, or completion of the doctorate.

# 7.5 Teaching Credit Classes and Overload Compensation for Administrative and Professional Faculty

University policy 4072, "Teaching Credit Classes and Overload Compensation for Administrative and Professional Faculty Members," allows appropriately credentialed administrative and professional (A/P) faculty to teach graduate or undergraduate courses at the university if requested to do so by an academic program. The A/P faculty member must have full responsibility for teaching a class to be eligible for overload compensation. Occasional lectures, supervision of interns or practicum students, or other minor instructional support activities typically are not compensated.

Academic departments may employ full-time administrative and professional faculty to teach one course per semester. The A/P faculty member must hold at least the minimum credentials required for teaching courses at various levels in accordance with the Southern Association of Colleges and Schools expectations. It is the responsibility of the teaching department to verify and document appropriate credentials for all teaching faculty. (See section 2.5.9, "Faculty Credentialing Guidelines," or the provost's website.)

The A/P faculty member may receive overload payment for teaching credit courses only when such teaching is not part of the normal expectation for the administrative and professional position. Teaching for supplemental compensation is limited by overall time and income restrictions defined in the consulting policy (section 2.14, "Consulting and Outside Employment").

The specific requested teaching assignment should be approved in advance by the faculty member's own department head/chair or supervisor. The department head/chair determines whether the teaching assignment is within the A/P faculty member's normal job responsibilities, and therefore not eligible for additional compensation. In approving or disapproving the teaching assignment, the department head/chair or supervisor considers the A/P faculty member's ability to manage additional work outside of usual job expectations, whether the course occurs during

normal hours of work, and whether scheduled absences and additional responsibilities will create undue disruption. The benefit to the A/P faculty member for his or her professional development and contribution to the academic program is also considered.

Engagement of administrative and professional faculty in the instructional mission of the university is encouraged; however, teaching on an overload basis is not a right. Continued satisfactory performance in the primary position is essential and is the basis of the annual performance evaluation and merit adjustment.

Further information regarding compensation to administrative and professional faculty for approved supplemental teaching activities is found in university policy 4072, "Teaching Credit Classes and Overload Compensation for Administrative and Professional Faculty Members."

## 7.6 Non-Reappointment, Reassignment, Removal, and Imposition of Sanctions Other Than Dismissal

Members of the administrative and professional faculty may be removed from their position by one of the following four procedures: non-reappointment, reassignment, removal for just cause, or abolition of position.

# 7.6.1 Non-Reappointment of Administrative and Professional Faculty on *Regular* Appointments

Monitoring the progress of newly appointed administrative and professional faculty members is the responsibility of the supervisor. An evaluation is made prior to the end of the first term appointment to ascertain that the faculty member is performing the assigned duties in a highly satisfactory manner. If the evaluation is positive, the faculty member can normally expect to be reappointed for another year. Notice of non-reappointment for administrative and professional faculty on regular appointment is given in writing in accordance with the standards of notice in section 2.11, "Retirement, Resignation, and Non-Reappointment."

# 7.6.2 Non-Reappointment of Administrative and Professional Faculty on *Restricted* Appointments

Restricted appointments may be terminated for a number of reasons including discontinuation of funding, or a change in research or other program priorities, resulting in the need to terminate the services of an employee. Administrative and professional faculty appointments may be terminated in the case where there are insufficient funds or no further need for services. The date of termination for a restricted A/P faculty member is at least 30 calendar days from the date of notification. A proposed notice of termination because of insufficient funds or lack of need for services requires the approval of the department head, dean (or appropriate administrator), and the provost or president (or their designees).

A faculty member on a regular appointment is entitled to notice of non-reappointment as stated in section 2.11, "Retirement, Resignation, and Non-Reappointment."

## 7.6.3 Reassignment

The university may reassign administrative and professional faculty members at any time. Reassignment may involve change in administrative title or supervisory responsibilities, reassignment to another position or department, transfer to a staff position, and/or reduction in salary commensurate with reduced responsibilities. Neither notice of non-reappointment nor removal for cause is required to effect a reassignment. The university's responsibility under reassignment is to make available a substitute position or duties reasonably commensurate with the person's education, experience, and performance. Reassignment that involves a geographic transfer of more than 50 miles is conducted in accordance with the geographical transfer policy. (See section 2.22, "Geographical Transfer Policy.")

In cases of reduction in salary and/or transfer to a staff position, the proposed salary reduction or reassignment to a staff position must be reviewed and approved by the senior administrator. The effective date of the reassignment shall be no sooner than 90 days following senior administrator approval.

## 7.6.4 Removal for Just Cause

Members of the administrative and professional faculty may be removed for just cause. Stated causes for removal shall be documented and shall include, but are not limited to, unacceptable or unsatisfactory performance; unethical conduct; misconduct that interferes with the capacity of the employee to perform effectively the requirements of the position; unsatisfactory attendance; falsifying credentials or any records—including but not limited to vouchers, reports, insurance claims, time records, leave records, or other official state or federal documents; unauthorized removal or damage of records or property belonging to others; acts of physical violence; criminal convictions for acts of conduct occurring on or off the job that are plainly related to job performance or are of such a nature that to continue the employee in the assigned position would constitute negligence in regard to the agency's duties to the public, students, or to other state employees; or violation of university policies. With approval by the provost or the associate vice president for human resources, as appropriate, a faculty member may be suspended with or without pay during an internal or external investigation of any act(s) that may lead to removal.

Removal for just cause is preceded by a meeting of the supervisor and a next-level administrator with the faculty member to review the reasons for termination, which are presented in writing to the employee. The meeting requirement may be satisfied in ways other than a face-to-face on-campus session, if there is a likelihood of threat to the health or safety of students, other employees, or property. With approval of the provost or associate vice president for human resources, as appropriate, the supervisor may suspend the faculty member with or without pay until the effective date of termination or until the employee is authorized to return to work.

The faculty member is given a minimum of three working days to respond to the reasons for termination. The response is made to the supervisor, who then makes a final decision and communicates it to the faculty member. The faculty member may invoke the applicable grievance procedure.

## 7.6.5 Imposition of Sanctions Other Than Dismissal

Minor sanctions include, but are not limited to, verbal or written reprimand. As compared to severe sanctions, minor sanctions usually do not involve a financial loss or penalty.

A severe sanction generally involves a significant loss or penalty to a faculty member such as, but not limited to, a reduction in title, responsibilities, and salary; or suspension without pay for a period not to exceed one year imposed for unacceptable conduct and/or a serious breach of university policy.

Routine personnel actions such as a recommendation for a below average or no merit increase, reassignment, or removal of an administrative stipend do not constitute "sanctions" within the meaning of this policy. A personnel action such as these may be a valid issue for grievance under procedures defined in section 7.7.4, "Valid Issues for Grievance."

Process for imposing a minor sanction: If a supervisor believes the conduct of a faculty member justifies imposition of a minor sanction, the faculty member is notified in writing of the proposed sanction and provided an opportunity to respond. A faculty member who believes that a severe sanction has been incorrectly imposed under this section, or that a minor sanction has been unjustly imposed, may file a grievance following procedures outlined in section 7.7, "Grievance Policy and Procedures."

Process for imposing a severe sanction: The conduct of a faculty member, although not constituting adequate cause for dismissal, may be sufficiently grave to justify imposition of a severe sanction. Imposition of a severe sanction follows the same procedures as dismissal for cause.

#### 7.6.6 Abolition of Position

Members of the administrative and professional faculty on regular appointments may be removed in the event of financial hardship within a department that cannot be alleviated by ordinary budgeting practices, or upon reduction of the specific services for which they were employed. A minimum of 90 calendar days notice is given in such circumstances. If an A/P faculty member on a regular appointment is separated involuntarily due to budget reduction, reorganization, or workforce downsizing, the faculty member may be eligible for severance in accordance with university policy 4245, "Severance Benefits Policy for University Employees."

Administrative and professional faculty with tenure or continued appointment and whose A/P position is abolished return to their academic department.

# 7.7 Grievance Policy and Procedures for Administrative and Professional Faculty

The following procedure is provided as the means for resolution of grievances against a supervisor or member(s) of the university administration brought by members of the administrative and professional faculty. The steps in the grievance process will, in part, be

guided by the reporting relationships of the employees involved in the grievance. Step one and two administrators involved in responding to a grievance should consult with the associate vice president for human resources or the vice provost for faculty affairs who may involve additional parties as appropriate.

## 7.7.1 Faculty Reconciliation and Mediation Services

**Informal Dialogue**: It should be possible to resolve most faculty concerns or complaints through informal communication among colleagues working together in the academic enterprise. Accordingly, a faculty member who feels he or she has a grievance is encouraged to take it to his or her immediate supervisor in the normal collegial spirit of problem solving rather than as a confrontation between adversaries.

**Reconciliation**: Reconciliation is useful if the individual feels the issue may be amenable to, but will require time for, negotiation or if the individual is unsure whether his or her concern is a legitimate issue for a grievance, or if personal relations between the parties involved in the matter have become strained. Information regarding faculty reconciliation processes is available on the website for the Senior Vice President and Provost.

The Commission on Administrative and Professional Faculty Affairs (CAPFA) designates the method by which reconciliation services are offered to administrative and professional faculty members. In consultation with the associate vice president for human resources or the vice provost for faculty affairs, CAPFA may designate the <u>Faculty Senate Committee on Reconciliation</u>, which typically includes participation by one or more administrative and professional faculty members as members of the reconciliation team, to conduct reconciliation between a faculty member and his or her supervisor. Reconciliation may include fact-finding and engaging the appropriate parties in negotiating a resolution. The designated reconciliation service is referred to hereafter as the "reconciliation team." Engaging the reconciliation team is not required prior to filing a grievance.

For a potential grievance issue to qualify for consideration by the reconciliation team, the faculty member must contact the chair of the reconciliation team within 30 calendar days of the date the grievant knew, or should have known, of the event or action that is the basis for the potential grievance. (The chair of CAPFA, the associate vice president for human resources or the vice provost for faculty affairs can provide the name of the chair of the reconciliation team.)

If the chair of the reconciliation team is unable to resolve the matter within 30 calendar days, the chair sends a letter to the faculty member stating such, providing the appropriate form for initiation of a formal grievance if the faculty member should choose to pursue the matter, and documenting that the matter was brought forward within the prescribed 30-day period. A copy of this letter is provided to the associate vice president for human relations with a copy to the vice provost for faculty affairs when appropriate. The faculty member has five weekdays after receiving the letter to initiate a formal grievance by following the procedures below and providing a copy of the letter from the chair of the reconciliation team to the supervisor, validating the timeliness of the grievance.

Administrative and professional faculty members may also consult the reconciliation team about serious disagreements with immediate supervisors or other university administrators concerning issues that may not be eligible for consideration within the grievance process. In such instances, in consultation with the associate vice president for human resources or the vice provost for faculty affairs, the CAPFA chair contacts the relevant administrator to determine if there is an interest and willingness to explore informal resolution of the dispute.

**Mediation**: <u>Mediation</u> is a voluntary, confidential process through which trained neutral persons (mediators) assist people to express their concerns and develop solutions to the dispute in a safe and structured environment. Because mediation is voluntary, both parties must agree to participate in order for mediation to occur. Faculty members and supervisors are encouraged to consider using mediation to resolve disputes between them, or to help address a conflict between a faculty member and another member of the Virginia Tech community.

**Role of Mediators**: Mediators do not make judgments, determine facts, or decide the outcome; instead they facilitate discussion between the participants, who identify the solutions best suited to their situation. No agreement is made unless and until it is acceptable to the participants.

**Requesting Mediation**: Mediation is available at any time, without the filing of a grievance. Additionally, mediation may be requested by any party during the grievance process prior to step three. If, after the initiation of a formal grievance, both parties agree to participate in mediation, the grievance is placed on administrative hold until the mediation process is complete. If the parties come to a resolution of the dispute through mediation, the parties are responsible to each other for ensuring that the provisions of the agreement are followed. In the event that the parties are not able to reach a mutual resolution to the dispute through mediation, the grievant may request that the grievance be reactivated and the process continues.

Mediation differs from faculty reconciliation in that mediators do not engage in fact-finding or in evaluation of decisions. Both mediation and reconciliation, however, are voluntary; no party is required to participate in either process.

To learn more about mediation and other forms of informal conflict resolution processes, contact the <u>conflict resolution program</u> manager in human resources.

#### 7.7.2 The Formal Grievance Procedure

The grievant may pursue the issue as a formal grievance through the following procedure. Supervisors and administrators will cooperate with the grievant in the mechanics of processing the grievance, but the grievant alone is responsible for preparation of his or her case (See section 7.7.4: Valid issues for Grievance).

The number of steps in the process is determined by the reporting line of the grievant. Thus, if three steps do not exist between the grievant and the president, then the available number of steps is used. Whether a complaint can be grieved under university policy can be determined at any point in the grievance process.

1. Step one: The grievant must submit a written statement of the grievance to his or her director or department head/chair (for A/P faculty in extension, the district director) within 30 calendar days of the date that he or she knew, or should have known, of the event or action that is the basis for the grievance. For purposes of the grievance process, the department head/chair (or district director) is the step one administrator. The written complaint should use the appropriate form and outline the relief requested. Grievance forms and procedures for A/P faculty are available on the Human Resources and Office of the Senior Vice President and Provost websites.

The grievance must be well described and the relief requested must be specified on the grievance form. The grievance form is signed and delivered to the step one administrator. If appropriate, the step one administrator will provide a copy of the grievance to the supervisor of the grievant.

Within five weekdays of receiving the written grievance, the step one administrator gives the grievant a written response, citing reasons for action taken or not taken. If the written response of the director or department head/chair is satisfactory to the grievant, that ends the matter.

2. Step two: If the resolution of the grievance proposed in the written response by the step one administrator is not acceptable, the grievant may advance the grievance to the step two administrator (usually a dean or vice president) by checking the appropriate place on the grievance form and sending it within five weekdays of receiving the written response. The step two administrator for extension A/P faculty (such as extension agents) is the dean of the College of Agriculture and Life Sciences.

Following receipt of the grievance form, the step two administrator or designated representative meets with the grievant within five weekdays. The grievant may request that a representative of his or her choice from among the university general faculty be present. Unless the grievant is represented by a member of the faculty who is also a lawyer, the step two administrator does not have legal counsel present.

If the grievance involves a programmatic issue for an extension A/P faculty member where responsibility for that program lies with a different dean, the designated step two administrator consults with the programmatic dean before rendering a decision.

The step two administrator gives the grievant a written decision within five weekdays after the meeting, citing reasons for his or her decision. If the step two administrator's written response to the grievance is satisfactory to the grievant, that ends the matter.

3. Step three: If the resolution of the grievance proposed in the written response by the step two administrator is not acceptable, the grievant may advance the grievance to the step three administrator. The step three administrator is the associate vice president for human resources. The grievant must advance his or her complaint to the step three administrator within five weekdays of receiving the written response from the step two administrator. Depending on the reporting structure of the grievant, the step three administrator will provide

a copy of the grievance to the senior vice president and provost. Advancement of a grievance to step three includes consideration by an impartial CAPFA hearing panel, unless the grievant petitions the step three administrator to bypass the hearing panel and rule on the grievance. If the step three administrator accepts the request to rule on the grievance, there is no subsequent opportunity for the grievance to be heard by a CAPFA hearing panel. If the step three administrator does not accept the petition, a CAPFA hearing panel is formed to review the grievance as outlined in these procedures. A CAPFA hearing panel may also be convened to determine whether a complaint may be grieved under university policy.

Within five weekdays, the step three administrator, or appropriate designated representative, acknowledges receipt of the grievance and forwards a copy of the "Hearing Procedures of the Committee on Administrative and Professional Faculty Grievances" to parties in the grievance process. The step three administrator also forwards a copy of the grievance immediately to the chair of CAPFA.

• **Hearing Panel**: A grievance hearing for A/P faculty is conducted by an *ad hoc* panel selected by the CAPFA chair from the A/P faculty at large. The chair of CAPFA solicits volunteers on a regular basis so that a pool of willing participants is available. A hearing panel consists of three members, an alternate, and a non-voting chair. Panel members are selected from the volunteers by the CAPFA chair and one or more of the *ex officio* members of CAPFA. The chair polls all appointees to ensure that they have no conflict of interest in the case. Either party may challenge one of the appointments, including the alternate. Other replacements are made only for cause. The alternate serves as a replacement panel member if the need arises.

To ensure uniformity in practice, the chair of CAPFA serves as the non-voting chair of each hearing panel. In the event that the chair of CAPFA has a conflict of interest concerning a case, the chair appoints a replacement from among the administrative and professional faculty at large to serve as chair of the hearing panel.

• **Hearings**: After the members of the hearing panel are appointed, the chair of CAPFA requests that each party to the grievance provide relevant documentation to be shared among the parties and the hearing panel. The panel holds its initial hearing with both principals present within 15 weekdays of receipt of the grievance by the chair of CAPFA. If the panel feels it needs to investigate the case further, or requires more information, or desires to hear witnesses, the hearing is adjourned until the panel completes the necessary work or scheduling can occur. The hearing is then reconvened as appropriate (and within the 45 day time frame required by this policy, unless agreed upon by both parties)

Each party to the grievance may have a representative present during the sessions of the hearing at which testimony is presented. The representative may speak if so requested. Representatives may be legal counsel, if both parties are so represented, but if the grievant does not wish to have legal counsel at a hearing, neither party to the grievance may have legal counsel present.

These impartial panel hearings are administrative functions, not adversarial proceedings. Therefore, if legal counsels are present, they must understand that the proceedings do not follow courtroom or trial procedures and rules. Participation by legal counsel is at the invitation of the parties they represent and is subject to the rulings of the chair of the hearing panel.

• **Findings and Recommendations**: The hearing panel concludes its work and makes its recommendations within 45 weekdays of receipt of the grievance by the chair of CAPFA. The time limit for consideration may be extended by agreement of both parties.

The hearing panel formulates written findings and recommendations regarding disposition of the grievance and forwards copies to the step three administrator and parties to the grievance.

- Action of the Step Three Administrator: The step three administrator meets with the grievant within 10 weekdays after receiving the findings and recommendations of the hearing panel to discuss the case and advise the grievant about the prospects for disposition of the case. Within 10 weekdays of that meeting the step three administrator sends to the grievant his or her decision in writing concerning the disposition of the grievance. If the step three administrator's decision is fully consonant with (or exceeds) the recommendations of the hearing panel, or if it is satisfactory to the grievant even if it differs from the recommendations of the hearing panel, that ends the matter.
- 1. **Step four**: If the step three administrator's decision is not acceptable to the grievant <u>and</u> not consonant with the recommendations of the hearing panel, the grievant may appeal in writing to the president within 20 calendar days. The president acts as he or she sees fit. The president's decision is final.

## 7.7.3 Timeliness of Grievance and Procedural Compliance

A grievance must be brought forward in a timely manner. It is the responsibility of the grievant to initiate the grievance process within 30 calendar days of the date when he or she knew, or should have known, of the event or action that is the basis for the grievance. The university administration is not required to accept a grievance for processing if the grievant does not meet the 30-day deadline, except in cases of demonstrated good cause.

Scheduled commitments made prior to the time of filing or advancement of a grievance that preclude action by either of the parties to the grievance automatically extend time limits for their duration unless this would be demonstrably harmful to the fair processing of the grievance. In such cases, on written request by the grievant to the appropriate office for that step, the grievance is advanced to the next step in the grievance process.

If the grievant does not follow the time limits specified in the grievance procedure it is assumed that he or she accepted the last proposed resolution as satisfactory. If the grievant desires to advance the grievance after the appropriate specified time limits have lapsed, the administrator who receives the late submission notifies the chair of CAPFA in writing, who determines if there

was good cause for the delay. If so, the grievance proceeds. If not, the process ends with the most recently proposed resolution in force. The finding on the matter by the chair of CAPFA is communicated to both parties in writing.

If either party to a grievance charges the other with procedural violations other than time limit issues, the chair of CAPFA rules on the question, as in disputes about the validity of procedural issues qualifying for the grievance procedure. The CAPFA chair has the following options: The chair can either find no significant procedural violation occurred, in which case the grievance process continues unaffected, or that a significant procedural violation did occur. If the step one or step two administrator committed a significant procedural violation, the grievance automatically qualifies for advancement to the next step in the grievance process. If the grievant committed a significant procedural violation, the grievance process ends at that point for that grievance with the last proposed resolution established as the final disposition of the case.

#### 7.7.4 Valid Issues for Grievance

For this process, a grievance is defined as a complaint by a faculty member alleging a violation, misinterpretation, or incorrect application of a policy, procedure, or practice of the university that directly affects the grievant. Some examples of valid issues for filing a grievance are: termination for cause, improperly or unfairly determined personnel decisions that result in an unsatisfactory annual performance evaluation, unreasonable merit adjustment or salary level, or excessive teaching load/work assignments; reprisals; substantive error in the application of policy; and matters relating to academic freedom.

Issues not open to grievance: While most faculty disputes with the university administration may be dealt with using this grievance policy, the following issues may not be made the subject of a grievance: determination of policy appropriately promulgated by the university administration or the university governance system; those items falling within the jurisdiction of other university policies and procedures (for example, complaints of unlawful discrimination or harassment, or an appeal related to the merits of a promotion decision); the contents of personnel policies, procedures, rules, regulations, ordinances, and statutes; the routine assignment of university resources (e.g., space, operating funds, parking, etc.); normal actions taken, or recommendations made, by administrators or committee members acting in an official capacity in the grievance process; non-reappointment or abolition of position; or allegations of misconduct in scholarly activities.

The subject of a grievance is normally not considered by the Commission on Administrative and Professional Faculty Affairs while it is simultaneously under review by another committee or panel of the university.

Adjudication of disputes on the validity of issues qualifying for consideration under the faculty grievance procedures: If either the step one or step two administrator rules that an issue does not qualify for the grievance process, the grievant may write to the chair of the Commission on Administrative and Professional Faculty Affairs (CAPFA) within five weekdays of receiving such notification and request a ruling. The chair of CAPFA, two A/P faculty members from CAPFA selected by the CAPFA chair in consultation with the associate vice president for human

resources and, depending on the reporting structure of the grievant, the vice provost for faculty affairs, and the chair of the reconciliation team deliberate and determine the admissibility of the matter to the grievance process. A written report of the results of the deliberations is sent to all parties concerned.

#### 7.7.5 Particular Concerns and Definitions

Timelines stated in the policy indicate the number of days within which the other party should receive notification. Submission by fax from a departmental office within the specified time frame is acceptable. This is immediately followed by submission by mail of the original form and any related materials.

Time limits are subject to extension by written agreement of both parties. The grievant and the administrator involved at that particular step of the discussion are the makers of such an agreement. (An agreement form to extend the grievance response time is available on the human resources and provost's websites.)

The principals and the chair of CAPFA, if necessary, negotiate extensions of time limits at step three. In case of disagreement, the chair of CAPFA rules on time extension and procedural questions or recommendations designed to expedite the proceedings while providing peer review of the grievance.

If a faculty member is away from his or her assigned work location at the time he or she discovers the event or action that is the basis for a grievance, the 30-day period during which the grievant must meet with his or her supervisor or step one administrator to initiate the grievance process begins when the faculty member returns to his or her assigned work location. If the date of return causes a delay of such length that the grievance, or its resolution, is not timely, the grievant may submit the written grievance as prescribed in step one by mail or email attachment in his or her absence from the primary work location.

"Weekdays," as used in this procedure, include Monday through Friday only and only when those days are not national, state, or religious holidays relevant to the principals in the grievance.

To protect a grievant from undue pressure in the pursuit of a grievance, if a grievant becomes ill and takes sick leave, the grievance process stops until such time as the grievant is able to resume his or her duties. Exceptions to this provision are made at the request of the grievant, but only if the grievant obtains and produces medical certification that proceeding with the grievance will not be harmful to the health of the grievant, or exacerbate the ailment that required taking sick leave.

All costs of legal counsel employed by a grievant are borne by the grievant.

If a grievant is employed away from Blacksburg, and he or she is required to travel away from their duty station in resolution of their grievance, the university pays all travel costs permitted under state regulations.

In the event that a faculty member discovers he or she has a grievance about actions by an administrator above the level of his or her supervisor that directly involve the faculty member, or with actions by an administrator not in his or her department/unit that directly involve the faculty member, the grievant initiates the grievance process by seeking the intervention of his or her supervisor within 30 calendar days of the date when the grievant knew or should have known of the event or action that is the basis for the grievance. If that effort does not resolve the grievance satisfactorily, the grievant, after consulting his or her supervisor, may file the grievance form for A/P faculty at the appropriate level or with the appropriate administrative office to initiate response from the administrator perceived as the source of the action causing the grievance. The grievance process then proceeds from that level onward in normal fashion.

A grievance filed by an A/P faculty member concerning an action of either the provost or the associate vice president for human resources is handled by the chair of CAPFA and an impartial hearing panel, but the findings and recommendations of the hearing panel are sent to the president of the university for his or her ruling. A grievance filed by a faculty member concerning an action of the president of the university is dealt with by a special panel appointed by the provost in consultation with the chair of the Commission on Administrative and Professional Faculty Affairs.

Any final resolution of a grievance must be consonant with the laws of the Commonwealth of Virginia and university policy.

Once a grievance is resolved, either to the satisfaction of the grievant, or if not to the satisfaction of the grievant, by the action of the senior administrator in consonance with the hearing panel recommendations, or by the ruling of the president, that specific grievance is closed and may not be made the subject of another grievance.

## 7.7.6 Overview of the Formal Grievance Process for Administrative and Professional Faculty

Below is an <u>abbreviated</u> overview of the grievance process and deadlines. Refer to section 7.7.2, "The Formal Grievance Procedure," for specific details and options available in each step of the grievance process.

Time limits are subject to extension by written agreement of both parties. The grievant and the administrator involved at that particular step of the discussion are the makers of such an agreement. (An agreement form to extend the grievance response time is available on the human resources and <u>provost's websites</u>.)

The validity of a grievance under university policy can be determined by CAPFA at any point in the process.

### Step one:

Within 30 days of event 1a. Grievant submits written grievance to step one administrator; for extension A/P faculty, this is the district director.

#### Within 5 weekdays 1b. Step one administrator determines whether the complaint can be grieved, and responds in writing on grievance form. If step one administrator's response is satisfactory to grievant, that ends the matter. 1d. If step one administrator's response is not satisfactory to grievant, move to step two within 5 weekdays. Step two: Within 5 weekdays Grievant submits written grievance to the step two administrator, 2a. usually the dean or vice president; for extension A/P faculty, this is the dean of agriculture and life sciences. Step two administrator meets with the grievant. Within 5 weekdays 2b. Step two administrator determines whether the complaint can be grieved and responds in writing on the grievance form. 2d. If step two administrator's response is satisfactory to grievant, Within 5 weekdays that ends the matter. If step two administrator's response is not satisfactory to 2e. grievant, move to step three within 5 weekdays. Step three: Within 5 weekdays 3a. Grievant advances grievance form to the step three administrator who then, depending on reporting structure, shares a copy of the grievance with the senior vice president and provost. 3b. Step three administrator acknowledges receipt of grievance and Within 5 weekdays forwards copy to chair of CAPFA. CAPFA chair acknowledges in writing to grievant that copy of Within 5 weekdays 3c. grievance has been received from the step three administrator. 3d. Questions of grievability are resolved by the CAPFA chair in accordance with Section 7.7.4. Within 15 weekdays 3e. CAPFA chair appoints hearing panel from among A/P faculty members; panel holds its initial meeting with both principals. Within 45 weekdays Hearing panel concludes its work and makes recommendation to 3f. step three administrator and grievant. Step three administrator meets with grievant. Within 10 weekdays 3g. Within 10 weekdays Step three administrator notifies grievant in writing of his or her 3h.

## Step four:

Within 20 calendar days 4a. Grievant may appeal in writing to university president.

within 20 calendar days.

decision.

3i.

4b. University president's decision is final.

If the step three administrator's decision is fully consonant with (or exceeds) the recommendations of the hearing panel, or if it is satisfactory to the grievant even if it differs from the

If the step three administrator's decision is not acceptable to the grievant and not consonant with the recommendations of the hearing panel, the grievant may appeal in writing to the president

recommendations of the hearing panel that ends the matter.

## 7.8 Leave

Administrative and professional faculty are eligible for the following types of leaves defined in section 2.9, "Leaves": administrative leave, annual leave, disaster relief leave, educational leave, family leave, leave without pay, military leave, sick leave, and special leave. Members of the administrative and professional faculty who have tenure or continued appointment may, under certain special conditions, request study-research leave or research assignment, particularly when they are returning to instructional faculty status. All study-research leaves and research assignments, require approval by the board of visitors.

## 7.9 Consulting Activities for Virginia Cooperative Extension Faculty

Consistent with the university's policy and procedures on consulting activities, additional restrictions may be imposed on the consulting activity of Virginia Cooperative Extension faculty members. These restrictions are imposed to give further assurance that consulting approval is not granted for assistance that is the normal responsibility of faculty members within extension.

It is recognized that the outreach responsibilities of extension are broad and, thus, program assistance parameters are difficult to define. Consequently, the following procedures are designed to provide judgmental decisions by appropriate supervisory staff for consulting requests in ambiguous areas of program responsibilities.

The Request to Engage in External Activity Form 13010A should be sent to the department head, chair, or immediate supervisor along with a letter outlining the nature of the consulting activity and why it falls outside the normal responsibilities of extension. (The form is available on the provost's website.) Typically, consulting activities do not involve university sponsorship.

The department head, chair, or immediate supervisor reviews the Request to Engage in External Activity Form 13010A and either approves or disapproves it. If approval is granted, the request is sent to the college dean for approval. The director of Virginia Cooperative Extension must grant final approval. If disapproval is exercised at any level, the request is sent back through the department head or supervisor, to the faculty member along with an explanation for the action.

Decisions are based upon, but not limited to, the following: consistency with guidelines stipulated in section 2.14, "Consulting and Outside Employment"; whether the area of consulting is found to be within or outside normal extension responsibilities; and whether the time required falls within the number of consulting days allowed.

#### **CHAPTER EIGHT**

## 8.0 Employment Policies and Procedures for Graduate Assistants

## 8.1 Graduate Student Appointments

Fully enrolled graduate students may be appointed to one of several assistantship categories. The responsibilities of such assistants are to serve the learning, discovery, and engagement functions of the university. The normal workload of graduate students may be defined in hours: a full assistantship is 20 hours' effort.

Since the responsibilities or requirements of graduate students may vary by academic discipline, each department is required to define clearly the expectation of its students. Although these positions are described here, the appointments do not carry faculty status or other faculty rights or responsibilities.

The <u>graduate school</u> reviews and approves each assistantship appointment recommended by the department. Only full-time graduate students may receive assistantships under criteria established by the graduate school.

All graduate assistantships carry stipends according to a schedule of steps as approved by the board of visitors. This schedule is the same for all three kinds of assistantships.

**Graduate Assistant**: Graduate assistants (GAs) are graduate students who provide academic and program support. GA responsibilities may be administrative in nature and consist of duties unrelated directly to teaching or research (such as academic advising, program planning, advising student groups, and assisting with the administration of student services offices). GA responsibilities may also be academic in nature and include grading examinations, problem sets, and/or lab assignments, setting up displays for lectures or laboratory sections, and preparing or maintaining equipment used in laboratory sections.

**Graduate Research Assistant**: Graduate research assistants (GRAs) are graduate students conducting academically significant research under the direction of a faculty member, who is generally a principal investigator on an external grant or contract.

Graduate Teaching Assistant: Graduate teaching assistants (GTAs) may provide academic program support under the supervision of a faculty member. GTAs may assist faculty members in the department in teaching undergraduate courses, including laboratory teaching assignments, or in providing other appropriate professional assistance, including grading examinations, problem sets, and/or lab assignments, setting up displays for lectures and laboratory sections, and preparing or maintaining equipment used in laboratory sections. GTAs must have 18 hours of graduate-level course work in their teaching discipline to be assigned full responsibility for teaching an undergraduate course. GTAs lacking this training are assigned to work under the supervision of a faculty member who is the instructor of record for the course. Further information regarding appropriate credential requirements for graduate assistants who are assigned as instructors of record is found below and on the provost's website.

## **8.2** Required Teaching Credentials for Graduate Teaching Assistants

Graduate assistants assigned as the instructor of record must also have documented teaching credentials. A master's degree in the teaching discipline or 18 graduate semester hours in the teaching discipline, direct supervision by a faculty member experienced in the teaching discipline, regular in-service training, and planned and periodic evaluations are required. For further information, see the teaching faculty credentialing guidelines available in section 2.5.9, "Faculty Credentialing Guidelines," and on the provost's website.

## 8.3 Additional Employment by Graduate Students with a Full-Time Assistantship Contract

Given individual circumstances, graduate students on full-time graduate assistantship may, at times, wish to pursue additional employment. Unless specified otherwise in the assistantship agreement, graduate students receiving full-time assistantships are not prohibited from seeking additional employment. In the interest of their professional development and maintenance of satisfactory academic progress, students seeking additional employment should consult with their academic advisors, and when appropriate their assistantship providers, regarding the fulfillment of their assistantship and graduate study responsibilities. The graduate school must be notified of additional employment agreements. The graduate school should be consulted to assist with the resolution of any conflicts that may arise.

In evaluating the merits of outside employment, graduate students and their advisors should consider the following:

- Employment by a company owned in whole or part by the faculty chair of the student's
  dissertation or thesis committee presents the potential for serious conflicts of interest. In such
  cases, another faculty member of equal or greater rank must serve as chair or co-chair of the
  advisory committee.
- It is inappropriate for any student to receive remuneration directly from the external funding organization while also being employed as a graduate assistant or wage earner on a contract from that same organization.
- It is inappropriate for any student to work for an employer who is in direct competition with a current funding source.
- International graduate students on assistantships may be prohibited from any additional employment by their specific visa status.

#### **CHAPTER NINE**

#### 9.0 Instruction-Related Policies

## 9.1 Assignment of Academic Responsibilities

Assignments of teaching load and academic advising are the responsibility of the department head or chair and may vary from one term to the next depending on the departmental requirements. Assignments should involve consultation with the faculty member, and in cases involving non-routine assignments—such as those requiring extensive travel—consultation is required. Ultimately, authority rests with the department head or chair to make the final assignment. Although the normal load for those engaged only in teaching is 12 didactic hours, the loads vary widely and are usually adjusted to permit time for other scholarly activities—for outreach which is related to the mission of the university and the faculty member's disciplinary expertise, and for faculty development related to the quality of instruction. A didactic hour is defined here as one contact hour in a lecture course or 0.60 hour for each contact hour in a course designated as a laboratory course.

Faculty members are expected to be available two weeks prior to the first day of classes and two weeks following commencement. The discretion of the department head or chair is recognized in assigning duties during periods when the university is not in session. Faculty members are expected to inform their department heads or chairs of their whereabouts during such periods.

#### 9.1.1 Summer and Winter Sessions

Teaching loads during the summer and winter terms are more tightly controlled. Summer teaching appointments are the responsibility of the department head or chair. (See section 2.6.5, "Summer and Winter Appointments.")

## 9.1.2 Independent Study and Undergraduate Research

The courses designated as Undergraduate Independent Study and Undergraduate Research are generally unique educational experiences between an instructor and a student. Such studies require prior approval by the instructor's department head or chair and by the dean. Courses designated as Graduate Independent Study/Special Study require approval of the instructor's department head or chair only. Approval forms are available in the colleges. Normally, these courses do not count in the teaching load of a faculty member.

## 9.1.3 Graduate Program Standards and Policies

Each graduate degree-granting program in the university is responsible for the conduct of the program and designates a faculty member to serve as liaison with the appropriate college dean(s) and the vice president and dean for graduate education. Further, each graduate degree-granting program formulates and retains a current policy statement that spells out criteria governing its program. Copies are filed with the appropriate college dean(s) and the vice president and dean for graduate education. Policy statements address faculty participation on graduate student

advisory committees (thesis and dissertation advisors; advisory committee membership); admissions procedures and requirements; and management of graduate students (orientation/advising; manuals, guides, handbooks; assistantships—selection procedures, obligations; evaluation of satisfactory progress towards the degree).

## 9.2 Scheduling of Classes

The Office of the University Registrar coordinates the preparation of the timetable of classes and disseminates this information. The department head or chair or a designated scheduler prepares proposed schedules in response to a call from the university registrar. The university registrar reconciles the material provided with the approved catalog of university courses, established scheduling patterns and allocations, and requests of other departments. Individual professors address scheduling issues through the department head or chair or the designated scheduler. The timetable of classes is available via *Hokie SPA* and *Faculty Access*.

The university registrar assigns classrooms. Moving the location of courses is possible only with approval of the department scheduler and the university registrar. Enrollments may not exceed the posted room capacity. Commonwealth of Virginia fire code specifications do not allow students to be seated in the aisles or on the floor.

## 9.3 Registration Procedures

The registration period for each term occurs during the regular preceding term. New and transfer students register for the fall semester during summer orientation. Students who enter the university for the first time in the spring semester or a summer term register in the normal manner.

The university registrar works with each department to amend course offerings by increasing the capacity of the section within limits for the assigned classroom; creating new sections with times subject to availability of suitable classrooms; and canceling sections for which the demand is too small to justify retention.

Undergraduate classes with fewer than 15 students and graduate classes with fewer than six are reviewed by the department head or chair and the academic dean and canceled unless there are compelling reasons for retention. In the summer and winter terms, the department head or chair and the director of summer and winter sessions review undergraduate classes with an enrollment of fewer than 18 and graduate classes with fewer than eight. Unless there are compelling reasons to offer the course, it is canceled.

Students register via <u>Hokie SPA</u> during the published pre-registration period. After departments make adjustments based on course requests, completed schedules are available to students via <u>Hokie SPA</u>. Students may adjust their schedules before the end of the preceding term and in the first five days of the term of the registration.

## 9.3.1 Drop-Add Period

At the beginning of the term, students may add courses through the end of the fifth day of classes and drop courses through the 30th day of classes. During the summer, students may add courses through the end of the third day of classes and drop courses through the end of the fifth. During the winter term, students may add courses through the first day of the part of term and drop classes through the second day of the part of term.

Dropping or adding courses becomes necessary if a student has an incomplete schedule, changes curriculum, fails a course in the previous term, or fails to pay fees on time, which removes the student from all classes.

A professor may require a student who processes changes during the drop/add period to show a printed copy of his or her class schedule from <u>Hokie SPA</u> before being admitted to the class. The names of students who have properly added the course immediately appear on the instructor's class listing available via *Faculty Access*.

## 9.3.2 Force-Add Requests

Force-add transactions are final solutions for critical scheduling problems in required courses. The force-add request requires approval by the course instructor or designated departmental representative. (Students may check with the department for departmental policy.) If a force-add request is approved, the student must retain his or her copy of the approval to verify enrollment with the instructor. An add processed through the force-add process overrides all other courses on a student's schedule and may create schedule conflicts. Do not process force-adds above the physical capacity of the scheduled room; doing so will not guarantee relocation of the course. Enrollments may not exceed the posted room capacity. Commonwealth of Virginia fire code specifications do not allow students to be seated in the aisles or on the floor.

#### 9.3.3 Class Rolls

Faculty may check with their departments to determine whether printed class rolls are provided. Up-to-date class rolls are available to instructors via <u>Faculty Access</u>. To obtain view access to the class roll and listserv capability, a faculty member must be the instructor of record in the student record system. The instructor of record is expected to inform students whose names do not appear on the displayed class roll. A student should contact his or her academic dean for assistance to correct inconsistencies. Graded work should not be returned to these students until their names are officially added to the class roll.

#### 9.4 Textbooks and Other Instructional Materials

The <u>University Bookstore</u> is responsible for providing textbooks and related teaching supplies to the university community. Each department has a person designated as the departmental bookstore representative. This person serves as the main contact between the bookstore and the instructor. Pursuant to *Higher Education Authorization Act, Public Law 110-315 (HEOA 2008)*, administrators, faculty, university bookstores, and publishers are "to ensure that students have access to affordable course materials by decreasing costs to students and enhancing transparency

and disclosure with respect to the selection, purchase, sale, and use of course materials." Provisions of the law require publication of textbook requirements prior to registration. Requests to the bookstore for textbooks and other instructional materials are routed through this representative. The University Bookstore must be notified of the selection of textbooks and other materials for any fall semester class no later than April 15. The deadline for spring semester is October 15 of the year immediately preceding the spring semester. The faculty member must confirm whether he or she intends to use all items ordered—particularly individual items sold as a part of a bundled package. If the faculty member does not intend to use each item in the bundled package, he or she must notify the bookstore. Before adoption of a particular textbook, the respective department determines that a copy of the textbook is on reserve in the University Library during any period that the textbook is to be used. Under provisions of the law, publishers must offer bundled materials for sale individually. Additionally, publishers are to document the substantive differences in any revised publication of texts. Faculty members are encouraged to limit their use of new edition textbooks when previous editions do not differ in a substantive way. Additional information on the textbook requirements federal guidelines may be found at the University Registrar's website. http://www.registrar.vt.edu/faculty\_staff/Textbook\_facts.php.

Faculty members should not engage in direct sale of instructional materials to students. *The Code of Virginia* §23-4.3:1 states that, "No employee at a Virginia public college or university shall demand or receive any payment, load, subscription, advance, deposit of money, services or anything, present or promised, as an inducement for requiring students to purchase a specific textbook required for course work or instruction; with the exception that the employee may receive (i) sample copies, instructor's copies, or instructional material, not to be sold; and (ii) royalties or other compensation from sales of textbooks that include such instructor's own writing or work." See also Policy 13010 "Individual Conflicts of Interest and Commitment."

## **9.4.1** Faculty-Authored Course Materials

A faculty member teaching a course may not receive a royalty and/or other fees beyond direct cost of production and sales for any material used as part of class activity, except for material that has received an independent external review, that has been copyrighted, and a portion of the copyright is owned by a publisher other than the author. Faculty accused of abusing the distribution of classroom material for personal financial gain are subject to review by the Committee on Faculty Ethics.

## 9.5 Grading Systems

"A" to "F" system (undergraduate students): The majority of course enrollments by undergraduate students at the university are graded on the traditional A-F basis, with a 12-point plus/minus grading scale. The grades "A" through "D-" represent passing grades and "F" is a failing grade. The grade of "A" should be assigned to students who meet the learning objectives outlined for the course at a level of comprehension and performance deemed excellent. The grade of "F" should be used for those students who have not demonstrated acceptable achievement with regard to the learning objectives of the course of study. An instructor may choose not to use the plus/minus system in the assignment of grades.

"A" to "F" system (graduate students): The grading system for graduate students is similar to the A-F system with "D" as the lowest passing grade.

**Pass/Fail system (undergraduate students)**: A pass/fail grading system is available to encourage students to enrich their academic programs and explore more challenging courses outside their major without the pressures and demands of the regular grading system. The pass/fail grading option is available to all undergraduates who have completed a minimum of 30 credit hours at Virginia Tech and have a cumulative Virginia Tech grade point average (GPA) of 2.0 or above.

The following restrictions apply: (1) for students in a four-year program, the total maximum allowable credits for pass/fail shall not exceed 12 semester hours, exclusive of courses offered only pass/fail; (2) for students in a five-year program, the total maximum allowable credits for pass/fail shall not exceed 12 semester hours, exclusive of courses offered only pass/fail, *or* 10 percent of the required hours for graduation completed at Virginia Tech—whichever is greater; (3) a student may not enroll for more than two courses per semester on a pass/fail basis—excluding physical education activity courses and required courses offered on a P/F basis only; (4) courses may not be changed from A-F to the P/F basis beyond the last day to drop classes without penalty; and (5) courses may not be changed from P/F to A-F beyond the last day to resign without penalty.

For courses offered only on a pass/fail basis, the 30-hour and 2.0 GPA requirement does not apply. Any courses taken beyond the number of hours required for graduation also may be taken pass/fail, except that no more than two courses may be taken on the P/F option per semester.

Under the pass/fail grading system, a "P" is granted for earning a "D" or better in the course; otherwise an "F" is given. The "P" or "F" is recorded on the student's transcript and credit given if the course is passed; if the course is failed, the "F" is considered as equivalent to an "F" received under the A-F grading system and is included in calculation of the GPA. The GPA is unaffected by a "P." Once credit is received for a course taken on pass/fail, the course cannot be repeated under the A-F grading system.

**Pass/Fail system (graduate students)**: A limited pass/fail grading system is available to encourage graduate students to explore courses outside their major. Subject to approval of the major professor, graduate students may take an unlimited number of hours of graduate course work (5000- and 6000-level) on a pass/fail basis, if outside the department and not on the plan of study. These courses may not be used to satisfy minimum degree requirements. All courses on the plan of study, including supporting courses, must be taken on a letter grade (A-F) basis except for those courses offered on a pass/fail basis only.

Under the pass/fail grading system, a "P" is granted for earning a "C-" or better in the course; otherwise an "F" is given. The GPA is unaffected by a "P." Grades of "F" are counted in the calculation of the GPA.

Audit grade (undergraduate students): A student may choose to audit a course, without the necessary prerequisites, to enhance his or her educational experience. Permission of the course instructor is required, in accordance with university policy 6360, "Auditing Courses," and policy memorandum 250, "Assignment of an Audit Grade for Undergraduate Courses," and policy memorandum 21, "Modification of Auditing Policy." An audit is a mechanism for a student to reserve a seat in a course, with no performance evaluation required. If the student or the instructor expects evaluation of course work, then the student must enroll either for the P/F option or for a letter grade. If the instructor of record wishes to restrict the participation of auditing students in selected activities, then that is stated in the syllabus. Students are assessed the same rate of tuition and fees for audited courses as for courses taken for credit. Audited courses do not count toward full-time enrollment.

Students are assessed the same rate of tuition and fees for audited courses as for courses taken for credit. Audited courses do not count toward full-time enrollment. Unsatisfactory audit should be left blank in grade column. An e-mail should be forwarded to the Graduate School requesting deletion of the course from the student's record.

The "I" grade: The "I" grade signifies incomplete work, but does not affect a student's GPA. It is assigned at the discretion of the instructor only. The "I" may be used when a student is unable to take the final examination during examination week, but the instructor may wish to confirm the legitimacy of the request with the <u>Schiffert Health Center</u> or the student's academic dean. Except for certain laboratory courses, "I" grades must be removed by the end of the student's first subsequent semester of enrollment or one calendar year from the date of the original "I" grade. Official change-of-grade cards must be used to remove an "I" grade and submitted to the department of the course. Incompletes not removed during the designated time are changed to "F" and calculated in the student's GPA.

The "NG" grade: The "NG" grade is given when a student's name appears on the class roll, but he or she has never attended class or submitted work for grading.

The "X" grade: The "X" mark shows that pursuit of the project begun in the course will be continued. The "X" does not compute in the student's GPA. The "X" may be assigned only for courses pre-established as eligible for this treatment. Changes from the "X" to the final grade must be submitted on change-of-grade cards; the regular grade marked on a grade sheet for an "X"-eligible course will process to that term's enrollment only.

The "EQ" grade: The "EQ" grade is reserved for graduate students enrolled in research and thesis (5994), or research and dissertation (7994). The awarding of this grade shows that the enrollment has been reviewed and the credits are to be sent to the grade report system. Failure to assign an "EQ" grade will result in the computation of the credits as failing.

The "NR" grade: The "NR" (not reported) grade is automatically entered when an instructor fails to award a grade to a student. The "NR" grade computes as an "F."

The "W" grade: The "W" (withdrawn) grade is given to an undergraduate student who has applied the course withdrawal policy to a course. The "W" grade is automatically awarded based

on the course option of "W." A regular grade cannot be awarded if a student has applied the withdrawal policy or "W" option to the course.

**Mid-term grade reports**: Mid-term grade reports are issued for first-term undergraduates and first semester transfer students for the purpose of informing them about their progress early in their first academic year. Courses that are oriented toward freshmen should be designed to include at least one substantial graded assignment in time for the mid-term grade report.

**Projected grades for the graduating students, spring term**: Projected grades for graduating students—all levels—must be submitted by the published tentative grade entry deadline in the spring semester. All students are completed for spring term based on the projected (tentative) grades received. Failure to submit tentative grades results in the student's non-completion and non-receipt of diploma at the college or department ceremony. Entry of tentative grades follows the same process as the end of term entry via *Faculty Access*.

## 9.6 Course Grading

The instructor of record has sole responsibility for assigning final course grades and may not delegate the task to other colleagues or teaching assistants. Department heads or chairs may ask instructors in their department to explain unusual profiles of grades or schemes of evaluation.

Faculty are expected to adhere to principles of professionalism, fairness, and clear communication when assigning grades. This includes consistent treatment of all students in the class; clear criteria—communicated directly to the class—about the basis on which course work is evaluated and grades are assigned; timely return of graded work to the student; sufficient feedback through the grading process for the student to improve performance on future assignments; and attention to fair and reasonable measures of course content and student performance.

During the term (i.e., before final grades are assigned), the grading process is not only a record of evaluation for work completed, it is also an important device for providing information to the student about how his or her work could improve in the future. Grading is a teaching tool that provides specific feedback to students. Faculty should keep this in mind when designing assignments and course work.

Students have the right to see their grades for a course and to lodge a grade appeal if they believe a grade was assigned unfairly. (See sections 9.6.4, "Undergraduate Student Appeals," and 9.6.5, "Graduate Student Appeals.")

The U.S. Department of Education stipulates that posting of grades using *even a portion* of the student identification number, is considered a violation of the Family Educational Rights and Privacy Act (FERPA). FERPA protects the confidentiality of educational records and prohibits distribution of that record unless with the student's written consent. Faculty may *not* post any grades as a class listing using any portion of the student identification number, either via paper or electronically. (This policy applies whether the student identification is the social security number or a generated identification number.)

# 9.6.1 Syllabus and Performance Expectation

Each semester on the first day of classes, faculty are expected to provide students with a course syllabus that includes course objectives, topical outlines, expected performance for which grades will be assigned, and the instructor's attendance policy, if any. The syllabus should also include a statement on the honor system and its application to the course, reference to accommodations for students with disabilities, and information regarding office hours and how the instructor can be reached directly or through the departmental office during normal working hours. (Further information regarding the undergraduate honor system is found in section 9.8. Further information regarding accommodations for students with disabilities is available on the Services for Students with Disabilities website.)

An explicit statement concerning prerequisites for the course must be included on the course syllabus or assignment sheet. The instructor should call attention to the prerequisites during the first week of classes. Before the official drop-add deadline, the instructor may require specific students without prerequisites to drop the course. The student granted permission to enroll without prerequisites should be informed that course expectations and grading practices are the same for all students regardless of whether prerequisites were satisfied or waived.

The syllabus is a very important document because it provides explicit information to the student about course content, schedule, grading scale, and expectations of the instructor. The instructor should design the syllabus as a useful means for setting the tone of the course. Substantial changes in the syllabus constitute modifications in the structure or content of the course, which should be communicated clearly and in writing to students in a revised syllabus. These might include changes in the grading scale, significant departures from the schedule, or modifications of assignments.

All written work, with the exceptions noted below, should be given at such time that it may be graded and then returned during a regularly scheduled class meeting. To the extent feasible, instructors should **not** schedule major assignments or tests for the last three calendar days of scheduled classes or reading day. Students should be allowed time to prepare for their final exams and benefit from feedback on material relevant to exams.

# Common exceptions include:

- Due dates for term papers and project reports may be set at the instructor's discretion, if the student will not be held responsible for the subject matter therein on the final examination.
- If a lab course or other course does not warrant a final examination during the exam period, but if the department and/or instructor requires that there be a final examination, the exam should be given during the last regularly scheduled laboratory or class period.
- Final examinations for master's and doctoral candidates, if approved by the vice president and dean for graduate education.

#### 9.6.2 Class Attendance

Class meetings are an integral part of most courses and the central component of many. Therefore, both faculty and students are expected to meet at all regularly scheduled times, except for cancellations announced on a university-wide basis by appropriate authority.

If a faculty member cannot meet a class, it is his or her responsibility to follow departmental procedures so that appropriate measures are taken to provide for the missed class.

When students cannot attend a class, it is their responsibility to make arrangements for any work missed as soon as possible. In cases of prolonged absences, students may ask their academic deans or the <u>Dean of Students</u> to notify their instructors of the reason for their absence. Staff members from the Office of the Dean of Students may verify absences from class for students who have documentation to support an absence beyond their control. The staff members send an absence verification notice to the college dean, who then forwards the verification to faculty members. A staff member from the Office of the Dean of Students reinforces with the student that he or she is responsible for contacting the faculty member to arrange make-up of any missed work.

# 9.6.2.1 Religious Holidays

Consistent with the university's tradition of religious tolerance, faculty are encouraged to be sensitive to students who wish to observe religious holidays. The student is responsible for requesting and providing justification for a religious accommodation, preferably during the first two weeks of classes or as soon as the student becomes aware of the need for an accommodation. Faculty should inform students of their willingness to make accommodations for reasonable requests. Faculty are not required to compromise the academic integrity of the course to accommodate religious practices. Thus, the faculty member determines accommodations for religious practices that are consistent with his or her class attendance policy.

#### 9.6.3 Final Examinations

Instructors must adopt an appropriate means for evaluating and measuring student performance relative to the course objectives. A final examination schedule is displayed on <u>Hokie SPA</u> and <u>Faculty Access</u> in the timetable of classes for each academic term and final exams, if used, must follow this schedule unless the dean of the college has granted special permission. The method of evaluation must be made known to students in the course syllabus at the beginning of the term (see section 9.6.1, "Syllabus and Performance Expectation"). Faculty members will make available to students any final graded material at least through the following academic term.

A student with conflicting examinations or with more than two examinations within 24 hours may reschedule an examination with permission of the student's college dean at least 10 days before the beginning of the examination period and by arrangement with the appropriate instructor.

A re-examination in one course, in which the final grade is C- or below, may be authorized when the student was enrolled in the course during the final term of his or her senior year and a satisfactory re-examination in the course would qualify the student for graduation. A re-examination request must be made and the exam must be completed by the student as soon as possible, but no later than one academic term after the original examination in the course. Re-examination approval by the instructor, the student's department head or chair, and the student's college dean is required, with consideration given to class performance and completion of assigned work.

# 9.6.4 Undergraduate Student Appeals

The university provides a process for student appeal of a grade. If a student feels that a grade was calculated incorrectly or was assigned in a prejudiced or capricious manner, the student must first discuss the matter with the instructor. If discussion between the instructor and the student does not resolve the issue, the student then has the option of requesting a formal appeal of the grade to the department head or chair who examines the student's allegation, discusses the matter with the instructor, and makes every effort to resolve the matter at the department level. In the unusual circumstance that resolution does not occur at the departmental or divisional level, the student may appeal to the instructor's college dean. The dean reconciles the matter by whatever mechanism is most appropriate for that college and that case. The decision of the college dean is final in undergraduate appeals.

A grade appeal must be made by the student as soon as possible, but no later than the end of the next academic term of the regular academic year (i.e., fall or spring).

# 9.6.5 Graduate Student Appeals

Graduate education is a complex activity involving a high order of student-faculty relationship.

It follows that the evaluation of the graduate student's progress is, and must be, dependent in large part on the judgment of their major professor, augmented by the collective judgment of the members of their assigned committees. The university, through the agency of the <u>graduate school</u>, defines minimal entrance standards and prescribes general rules governing eligibility for continuation. But the crucial agency in student evaluation is the department in which the student's work is centered, and the crucial evaluator is the faculty advisor.

It is important, therefore, that each graduate student be fully informed—not only of the university's expectations, but of the department's expectations as well. Each department shall prepare, in outline form, a statement for each of its graduate degrees. The statement should cover course requirements, the nature and timing of oral and written examinations, and the evaluation that is given to the thesis. A copy of each departmental statement should be on file in the graduate school and made available to each student at the time of matriculation.

Most disputes over evidence of unsatisfactory progress are informally discussed and reconciled at the departmental level. Discussions of this kind occur among the student, the major professor, and the other members of the advisory committee. Nonetheless, from time to time serious

questions arise regarding both the status of a graduate student (whether in a given course or as a candidate for the degree) and the basis of the evaluation that placed the student's status in jeopardy. On these occasions it is important that the university provides full opportunity for the student's grievance to be reviewed in a judicious manner.

The procedures for a formal graduate student appeal are described in the <u>Graduate Catalog</u> or may be obtained from the graduate school.

# 9.6.6 Grade Adjustments for Undergraduate Honor Suspects

The undergraduate honor system is described in section 9.8. Grades may not be unilaterally adjusted in a course to compensate for suspected dishonesty. When an alleged violation of the honor system occurs, the incident should be reported to the honor system by submission of a violation report form. The Office of the Senior Vice President and Provost recommends a grade change for undergraduate students found guilty of class I, II, and III offenses that are instituted by the instructor of the class. The university registrar adjusts grades for class V and class VI offenses.

# 9.6.7 Change of Grade

A change in grade is authorized only under unusual circumstances. Faculty submit the change of grade via the department change of grade card. The change of grade process requires the instructor's signature and the electronic approval of the department head or chair and dean for all grade changes—including removal of "I" grades. Grade change requests should carry a statement regarding the circumstances necessitating the change, which includes a description of the circumstances for an original award of an "I" grade. It is improper to permit a student to improve a grade by doing extra work unless all students in the class are given the same opportunity.

# 9.6.8 Final Grade Reports

Final grades are reported via the Web, using <u>Faculty Access</u>. Two methods of entry are available—direct entry or upload from an external data file. Grades must be submitted within 48 hours of the last final examination on the published schedule. Student grade reports are generated from these submissions and displayed via <u>Hokie SPA</u>.

Faculty may not post grades, either via paper or electronically, using even a portion of the student identification number. Pursuant to the Family Educational Rights and Privacy Act (FERPA), using a portion of the student identification number in conjunction with the course grade is not allowed without the written permission of the student. Faculty may wish to remind students that grades are available via <u>Hokie SPA</u> within 48 hours of the end of the term.

# 9.7 Instruction-Related Responsibilities

#### 9.7.1 Office Hours

As a part of their teaching responsibilities, faculty members are expected to provide several regularly scheduled office hours each week for consultation with students. These hours should be reasonably spaced over the week at times mutually convenient to the instructor and students. Although a specific number of office hours is not stated in university policy, faculty members should ensure that they are readily available, both through office hours and by message at other times during the normal workweek. Information about office hours and how to contact the faculty member through the department office should be included on the course syllabus.

The instructor should encourage students in need of counsel to seek clarification about their work. Those in need of non-academic or personal counseling outside the purview of the faculty members' professional capabilities may be referred to the <u>Cook Counseling Center</u>.

# 9.7.2 Tutoring

Faculty members and graduate teaching assistants do not accept fees for tutoring students enrolled in their classes, either on a group or single-student basis. They are free to tutor for payment otherwise under university consulting policies.

#### 9.7.3 Students with Disabilities

The university, as a federal aid recipient and state agency, is required to provide opportunities and reasonable accommodation to all identified students with disabilities. Section 504 of the Rehabilitation Act, the Americans with Disabilities Act, the Virginians with Disabilities Act, and policy memorandum 178, "Accommodation of Disabilities," provide guidelines and requirements for colleges and universities in providing academic assistance. Accommodation means more than the removal of architectural barriers and the provision of auxiliary services such as notetakers, readers, and interpreters for the deaf. It means reasonable accommodation must be made in the instructional process to ensure full educational opportunity. For faculty, this means that teaching strategies and methods, including Web page design and distance learning, as well as instructional policies, must be sensitive to the laws and the needs of students with disabilities and responsive to the university's legal obligations.

Students with disabilities may self-identify and may qualify for accommodations through <u>Services for Students with Disabilities (SSD)</u>. Students must present medical or extensive psychoeducational documentation of physical, medical, psychological, or learning disabilities to SSD. Accommodations for students with disabilities are established by the SSD, in accordance with medical and professional information in the student's record, legal precedent, and national standards for services for students with disabilities. Faculty are urged to include a syllabus statement that encourages the student with a disability to disclose their need for accommodation to the professor as early in the semester as possible. Examples of inclusive disability syllabi statements are found on the <u>SSD website</u>. Contact SSD for more information regarding accommodation and services.

# 9.8 The Undergraduate Honor System

The functions of the undergraduate honor system are to communicate the meaning and importance of intellectual honesty to all undergraduate students of the university; to articulate and support the interest of the community in maintaining the highest standards of conduct in academic affairs; and to identify, sanction, and educate those students who fail to live up to the stated expectations with regard to those standards. The honor code defines the standards of student conduct in academic affairs. Details are found in the <u>constitution</u> of the Virginia Tech undergraduate honor system and on the <u>honor system website</u>.

# 9.8.1 Undergraduate Offenses and Sanctions

The undergraduate honor system classifies offenses into one of six categories depending upon the severity of the offense and the circumstances of its occurrence. Each classification carries several sanctions, one or more of which may be imposed on those convicted of violating the honor code. These include probation, grade adjustments on assignments in question, permanent grade of "F," university service, suspension, and dismissal.

Among common offenses are: cheating (i.e., giving or receiving of any unauthorized aid, assistance, or unfair advantage on any form of any academic work); plagiarism (i.e., copying the language, structure, ideas, and/or thoughts of another and passing it off as one's own); falsification and tampering with records; purchased work and work for hire.

Any attempt to commit one of these acts is a violation of the honor code as well.

# 9.8.2 Faculty Participation

Although the Virginia Tech undergraduate honor system is a student program, the support of the faculty is essential in making the system an effective and efficient means of handling academic violations. Faculty members are encouraged to support the undergraduate honor system and are expected to abide by procedures designed for the effective implementation of the honor code.

Any suspected violations should be reported promptly, in writing, to the chief justice. Forms for this purpose are available from all department offices, the honor system office, and the honor system website. The chief justice assigns the case to an associate justice for investigation. If warranted, a judicial panel hears the case to determine innocence or guilt. The undergraduate honor system review board reviews the decision; it may return a case for rehearing or reinvestigation, overturn verdicts, or forward recommendations to the provost. Once the provost imposes a sentence, the faculty complainant is informed of the verdict, the classification assigned to the offense, the sentence imposed, and any recommended grade change, if applicable.

In order to protect the accuser and the accused, a faculty member bringing charges of an honor code violation has the right to decline discussion of the case with the accused, or, with the written permission of the accused, to have witnesses present at such a discussion. The faculty member receives a copy of the case coordinator's report summarizing the evidence in the case. A

faculty member involved in a case is expected to cooperate with undergraduate honor system personnel, appear before the judicial panel if requested, and maintain confidentiality.

In addition, the undergraduate honor system offers the following guidelines to faculty:

- When an alleged violation is detected, the suspected student(s)' paper should not be collected until the test is completed, as this would be contrary to the student(s)' right of presumed innocence. However, any evidence that would be necessary in an investigation should be collected immediately. The test should be graded without prejudice and the alleged violation should be reported to the honor system. Please provide the original of the instrument in question in the submission of evidence. Grades should not be adjusted in a course to compensate for suspected dishonesty.
- If a professor suspects that a student or students are cheating, it is permissible to speak with the suspected student(s)—after the test or other work has been completed—and indicate these suspicions. However, it is not permissible to penalize or berate the student(s) or to take any other action that might affect the student(s) or violate the student(s)' rights to due process.
- A statement is to be included on each course syllabus about the undergraduate honor system and its application to the particular course.
- Faculty members are not required to proctor quizzes, tests, and examinations. Faculty are expected to personally administer the examination and to remain within reasonable proximity of the examination room to answer questions that may be raised by the students. However, it is not a compromise of the honor system to stay in the room or visit frequently, when a test is being given. In fact, precautionary measures in the spirit of reducing the opportunity for cheating are advisable, especially in large classes. Seats should be spaced in examination rooms whenever possible. Alternate test forms may be used. In some rare cases such extreme measures as requiring ID when a test is handed in may be necessary to prevent organized "paid substitutes" from taking tests for other students. Under no circumstances can measures be taken that would compromise student(s)' rights.

The faculty, along with the students and other university personnel, share the responsibility for providing, explaining, and disseminating information regarding honor practices and the undergraduate honor system.

#### 9.8.3 The Honor Pledge

The Virginia Tech honor pledge is as follows: "I have neither given nor received unauthorized assistance on this assignment."

The pledge is to be written out on all graded assignments at the university and signed by the student. The honor pledge represents both an expression of the student's support of the honor code and an unambiguous acknowledgment that the student has, on the assignment in question, abided by the obligation that the honor code entails. In the absence of a written honor pledge, the honor code still applies to an assignment.

# 9.9 Honor System for Graduate Students and Students in the College of Veterinary Medicine

Detailed information concerning the graduate honor system applicable to all graduate students is found in the graduate honor system constitution, published in the <u>Graduate Catalog</u>. The graduate honor system constitution describes the rights and responsibilities of students as well as faculty with regard to the honor code.

The honor system for students in the College of Veterinary Medicine is described in the Virginia-Maryland Regional College of Veterinary Medicine Student Honor Code

#### 9.10 Classroom Conduct

Maintaining a good learning environment in the classroom is an important part of a faculty member's responsibility as a teacher. The teacher should endeavor to create a classroom atmosphere that is comfortable and welcoming of all students, including women and members of minority groups. Disruptive classroom conduct on the part of some students may be distracting, annoying, or intimidating to other students and should not be tolerated by the teacher.

As much as possible, the teacher should endeavor to create a classroom environment in which there is an active participation on the part of most of the students, rather than the domination of the class by a few individual students. This may require different teaching strategies such as the use of small groups or teams, as well as different approaches to the structure of classroom presentations. Assistance for faculty who are trying to improve the learning environment of the classroom is available through the <u>Center for Instructional Development and Educational Research</u>.

Faculty have the prerogative of deciding the classroom conduct and the appropriate dress of their students as long as these actions do not infringe upon the students' rights as guaranteed in principles underlying the section in the *Hokie Handbook*. It is the faculty member's obligation to ensure that the classes following find a clean and orderly space.

#### 9.11 Teaching Evaluation

Good teaching evaluation processes are essential for maintaining the quality of academic programs, for reviewing the job performance of faculty members with respect to the instructional mission of the university, and for designing effective faculty development initiatives. All of the colleges have processes of teaching evaluation, which are used in promotion and tenure decisions and in yearly performance evaluations.

#### 9.11.1 Student Evaluation of Courses and Instructors

Student evaluation of courses and instructors is an integral component of a good teaching program. While specific procedures vary across the colleges, in general, committees in each college are responsible for designing appropriate evaluation procedures and for receiving such evaluations. Faculty members should ensure that their college's procedures for conducting

student teaching evaluations are followed in a way that is absolutely free of intimidation or influence by the teacher's presence. Most of the colleges require that evaluation forms be filled out after the faculty member has given instructions and left the room, and that they be collected and delivered to the department by a responsible student in an envelope provided by the teacher for this purpose. Otherwise, students may believe that their grades will be affected by how they answer the questions. The university expectation is that all faculty members will be evaluated in all courses taught each year. More information about this matter is available from departmental offices and from the academic deans.

#### 9.11.2 Other Evaluation of Courses and Instructors

Good teaching evaluation includes more than the student perception of instruction. The university expectation is that in-depth peer evaluation of teaching will be conducted periodically for all faculty members and at least twice during the probationary period for tenure-track faculty.

Colleges, departments, and individuals wishing assistance in devising evaluation forms may consult the <u>Center for Instructional Development and Educational Research</u>, where a variety of such forms are available. Faculty members may find such evaluations helpful in revealing information that leads to improvement of classroom presentation, evaluation of students, and student response to their classes.

# 9.12 Student Record Policy

In response to the Family Educational Rights and Privacy Act (FERPA), a statement of policy on the maintenance and disclosure of student records was adopted by the university. This policy protects the privacy of student records; the specific policy document is available from the <u>university registrar's website</u>.

#### 9.12.1 Academic Records

Names and addresses of current and former students may be selected and released to non-university entities only on the basis of class level (freshman . . . senior), major, or place of residence. It is important to note that email addresses are not directory information and may not be released without the student's prior consent.

The protection of academic records, which exist in enrollment services and in the college and departmental files, is covered by this policy. This includes the student's right to review these records. Responses to telephone inquiries are limited to the following information: whether the student is currently enrolled; dates of enrollment; degree(s) earned if any, date, major, and honors received; address and telephone number. Special note: no information, including directory information, may be released if a student has marked their record confidential.

Grade reports may not be released to parents, guardians, or any other person without prior written approval from the student. Students may not have access to financial aid information about their parents or guardians without written approval from the parent or guardian.

The university may withhold transcripts, certificates, registration materials, or any other information about a student's record if financial obligations are unmet. The university also reserves the privilege of withholding materials if violations of university regulations have not been cleared.

# 9.13 Undergraduate Student Advising

<u>Undergraduate advising</u> at Virginia Tech is a collaborative process between student and advisor, leading to the exchange of information that encourages the individual student to make responsible academic and career decisions. The university is committed to effective advising by recognizing and supporting the needs of both students and advisors. Each undergraduate student is provided information and assistance to aid the student in making academic and career decisions. Each advisor is provided the necessary tools to respond to student needs and the opportunity to be recognized for exemplary advising.

Statement of university responsibility: The university shares responsibility for successful advising. Senior leaders will: review the advising process to assess the impact of recommendations implemented; provide information for students, advisors, parents, and other constituents that clearly explain responsibilities and expectations related to advising; make information available about advising for all new faculty and appropriate staff; collect and disseminate information that contributes to effective advising; assist students in clarification of academic and long term goals; support initiatives that enhance the use of technology in advising; support Web-based interactive advising support systems for students, advisors, parents, and other constituents; and support a Virginia Tech plan that effectively assesses, recognizes, and rewards advising in the annual professional evaluation.

**Statement of student responsibility**: The student shares responsibility for developing an advising partnership with the advisor. Over time, this partnership results in increased responsibility for the student. The student will: communicate goals, needs, wants, and concerns to the advisor in a respectful and sincere manner; keep abreast of their own academic progress and requirements related to their academic programs; make, keep, and be prepared for appointments with the advisor; inform the advisor of changes in plans and/or circumstances that might impact academic performance; know departmental procedures regarding changing advisors; and bring concerns regarding quality of advising to the attention of the advisor.

**Statement of advisor responsibility**: The advisor shares responsibility for developing an advising partnership with undergraduate students. The advisor will: communicate with students and delivering individualized and accurate information in a professional and sincere manner; be informed of, and provide accurate information about current academic policies and procedures; keep appointments and be available for assistance; provide appropriate referrals, contacts, and information; do appropriate follow-up with students; and seek out and take advantage of opportunities for professional development.

# 9.14 Identifying and Referring the Distressed Student

The college years can be very stressful for students. In the contemporary climate of competition and pressure, some students adequately cope with these stresses, but others find that stress becomes unmanageable and interferes with learning. In some cases, these students may even disrupt the learning of others.

Identifying the distressed student: Many students initially seek assistance from faculty. A student in distress may display: excessive procrastination and very poorly prepared work, especially if inconsistent with previous work; infrequent class attendance with little or no work completed; dependency (e.g., the student who hangs around or makes excessive appointments during office hours); listlessness, lack of energy, or frequently falling asleep in class; marked changes in personal hygiene; impaired speech and disjointed thoughts; repeated requests for special consideration; threats to others; expressed suicidal thoughts; excessive weight gain or loss; behavior that regularly interferes with effective class management; frequent or high levels of irritable, unruly, abrasive, or aggressive behavior; inability to make decisions despite repeated efforts to clarify or encourage; bizarre behavior that is obviously inappropriate for the situation; or may appear overly nervous, tense, or tearful.

Guidelines for interacting with the distressed student: Talk to the student in private. Express concern and be as specific as possible in stating your observations and reasons for concern. Listen carefully and repeat the essence of what the student has told you so that your attempts to understand are communicated. Avoid criticizing or sounding judgmental. Consider the Cook Counseling Center as a resource and discuss referral with the student. If the student resists referral and you remain uncomfortable with the situation, contact the Cook Counseling Center or the Dean of Students Office to discuss your concern.

Referring the distressed student to Cook Counseling Center: Suggest that the distressed student call or come in to make an appointment. Give him or her the Cook Counseling Center's phone number (540-231-6557) and location (240 McComas Hall). It is usually more effective to assist the student by calling for an appointment with the student present. When you reach the center's receptionist, identify yourself as a faculty member and ask for an appointment for the student. The student's name and Tech ID number are required for the appointment. Write down the appointment time, date, and name of the counselor for the student. If you feel the situation is an emergency or urgent enough to req9.uire immediate attention, tell the receptionist that the student needs to see a counselor immediately. It may be necessary for you to walk the student to the center. If you are concerned about the student, but unsure about the appropriateness of a referral, call the center for a consultation.

**Receiving the assistance of the Office of the Dean of Students**: The Dean of Students offers several guides and videos that may be helpful in identifying and interacting with the distressed student:

• <u>Responding to Disruptive or Threatening Student Behavior: A Guide for Faculty</u>, which is available online or a print copy may be requested by calling the Office of the Dean of Students at 540-231-3787 or by sending an email request to dean.students@vt.edu.

- A listing of *Resources for Student Referrals* may be obtained by calling the Office of the Dean of Students at 540-231-3787, or by sending an email request to <a href="mailto:dean.students@vt.edu">dean.students@vt.edu</a>. The listing is available in two formats—a re-positional sticker, or a 4 x 6 inch card that includes contact information for student affairs departments able to assist with distressed students.
- Online video resources include:
  - Identifying and Referring the Distressed Student
  - Responding to Student Concerns
  - Assessing Disturbing Writing
  - Taking Action
  - Resources Overview

The Office of Dean of Students partners with faculty and staff members to support students for whom there may be concern. Concerns may be shared by phone at 540-231-3787, email (<u>dean.students@vt.edu</u>), or face-to-face contact with staff in the <u>Dean of Students Office</u>. After regular business hours, contact <u>Virginia Tech Police</u> at 540-231-6411 for connection to the dean of students' on-call staff member.

The Office of Dean of Students also offers an additional tool for faculty members to use in sharing concerns about a student. *This online reporting system should not be used for emergencies*. The Dean of Students Reporting System is available through the *Faculty Access/Hokie SPA* menu. This system closely parallels the academic advising system already used by faculty. As always, matters needing immediate attention should be directed to the <u>Virginia Tech Police</u> at 540-231-6411.

# 9.15 Faculty Awards for Teaching and Advising

In recognition of the university's conviction that excellence in teaching and advising should be a major concern of every faculty member, several awards are presented annually to honor outstanding teaching and advising performance.

#### 9.15.1 The Wine Award for Excellence in Teaching

Three Wine Awards for Excellence in Teaching are made possible by a gift from the <u>Alumni Association</u>, honoring William E. Wine, a former rector of the board of visitors and Alumni Association president.

Students, alumni, and faculty nominate possible recipients of the Wine Award. In each of the academic colleges, a Wine Award Committee composed of previous winners selects one or two candidates in their college from those nominated. Each college Wine Award Committee establishes its own procedures for selecting a nominee from candidates within the college. The university Wine Award Committee, in turn, reviews the nominations submitted by the college committees. The university committee selects and recommends the three award recipients and transmits their names to the vice president for alumni relations. Each Wine Award winner

receives a \$2,000 cash prize and automatic induction into the Academy of Teaching Excellence. Further information regarding the William E. Wine award process, guidelines, dossier requirements, and deadlines is available from the <u>Academy of Teaching Excellence</u>.

# 9.15.2 The University Sporn Award for Excellence in Teaching Introductory Subjects

The University Sporn Award for Excellence in Teaching Introductory Subjects is made possible by gifts from Dr. and Mrs. Philip J. Sporn and alumni of the university. Dr. Sporn was the former president and chief executive officer of American Electric Power Company.

Eligible candidates for the University Sporn Award are Virginia Tech faculty members who are full-time employees; hold the rank of professor, associate professor, assistant professor, or instructor; and teach introductory courses at the 1000/2000 level. Nominations for the award are accepted only from students until February 1<sup>st</sup> each year.

A committee comprised of student representatives from Omicron Delta Kappa and the Golden Key National Honor Society select the award recipient. The Sporn Award winner receives a \$2,000 cash prize and automatic induction into the Academy of Teaching Excellence. Further information regarding the award process, guidelines, dossier requirements, and deadlines is available from the Academy of Teaching Excellence.

# 9.15.3 Certificates of Teaching Excellence

The university annually awards certificates of teaching excellence to 21 faculty members who are selected by the individual colleges. The number of faculty receiving awards is proportionately distributed over the university according to the number of instructional faculty positions allotted to each college, with each college awarding at least one certificate.

# 9.15.4 Alumni Teaching Excellence Awards

Two Alumni Teaching Excellence Awards are presented each year to outstanding faculty, elected by the Academy of Teaching Excellence from among the recipients of the certificates of teaching excellence for the past three years. To be considered for the Alumni Teaching Excellence Award, certificate of teaching excellence winners submit a teaching dossier to the selection committee.

A committee of former recipients reviews those eligible and selects two candidates each year. The Alumni Teaching Excellence Award recipients each receive a \$2,000 cash prize and automatic induction into the Academy of Teaching Excellence. Further information regarding the award process, guidelines, dossier requirements, and deadlines is available from the <u>Academy of Teaching Excellence</u>.

# 9.15.5 Diggs Teaching Scholar Awards

The Diggs Teaching Scholar Award was established in 1992 under the auspices of the university's Faculty Rewards Project. Sponsors of the program are the Diggs Endowed Professorship Fund and the <u>Center for Instructional Development and Educational Research</u>.

Three Diggs Teaching Scholars are selected annually from among all instructional faculty at Virginia Tech to recognize exceptional contributions to the teaching program and learning environment. The selected Diggs Teaching Scholars agree to help lead the Diggs Teaching Roundtable, which is a public discussion held in the fall semester that is devoted to pedagogical and curricular issues at Virginia Tech. (See section 9.17.2, "Diggs Roundtable.") The award recipients also lead departmental efforts to implement their proposed project for enhancing teaching, and they actively participate in the Diggs Teaching Scholar Association by working on its projects to strengthen teaching at the university and by sharing in the leadership of the association. The award recipients each receive a \$500 cash prize. In addition, the departments or programs that nominate the recipients receive a matching award of \$500. The Diggs Teaching Scholar Award is designed to honor departments and programs as well as individuals who are dedicated to teaching excellence and to student learning.

Further information regarding the award process, guidelines, dossier requirements, and deadlines is available from the <u>Center for Instructional Development and Educational Research</u>.

# 9.15.6 Academy of Teaching Excellence

Founded in 1974 to honor and reward excellence in instruction, the <u>Academy of Teaching Excellence</u> membership voted in 1987 to expand its role significantly beyond honorary status. The membership defined the academy as an organization of faculty committed to the enhancement of the educational environment, the highest standards of instruction, the vigilant pursuit of new forms and technologies of teaching, the continual evaluation of practices and standards of teaching effectiveness, and personal and organizational commitment to the importance of teaching in the research university.

The six major award winners each year (three Wine Awards, one Sporn Award for the teaching of introductory subjects, two Alumni Teaching Awards) are inducted into the Academy of Teaching Excellence for a term of three years. Although "active service" in the academy is for three years, membership is for life. An individual who has received one of the six major teaching awards is ineligible for any other for a period of seven years; an individual may receive any particular award only once.

# 9.15.7 Alumni Award for Excellence in Undergraduate Academic Advising

The Alumni Award for Excellence in Undergraduate Academic Advising is given annually to recognize a full-time faculty member holding the rank of professor, associate professor, assistant professor, or instructor, who serves undergraduate advisees in exemplary ways. Faculty, staff, administrators, or students may nominate individuals for the award. The award recipient receives a \$2,000 cash prize and membership in the Academy of Advising Excellence. Further

information regarding the award process, guidelines, dossier requirements, and deadlines is available from Undergraduate Advising.

# 9.15.8 Alumni Award for Excellence in Graduate Academic Advising

The Alumni Award for Excellence in Graduate Academic Advising is given annually to recognize a full-time faculty member holding the rank of professor, associate professor, assistant professor, or instructor, who serves graduate advisees in exemplary ways. Faculty, staff, administrators, or students may nominate individuals for the award. The award recipient receives a \$2,000 cash prize and membership in the Academy of Advising Excellence. Further information regarding the award process, guidelines, dossier requirements, and deadlines is available from <u>Undergraduate Advising</u>.

# 9.15.9 Provost's Award for Excellence in Undergraduate Academic Advising

The Provost's Award for Excellence in Undergraduate Academic Advising is given annually to recognize an individual (faculty or staff) who serves undergraduate advisees in exemplary ways. Faculty, staff, administrators, or students may nominate individuals for the award. The award recipient receives a \$2,000 cash prize, a plaque, and membership in the Academy of Advising Excellence. Further information regarding the award process, guidelines, dossier requirements, and deadlines is available from Undergraduate Advising.

# 9.15.10 Academy of Advising Excellence

The provost established the Academy of Advising Excellence to recognize those advisors who have demonstrated exemplary advising practices. The academy consists of all advisors who have received university-level awards for excellence in advising.

# 9.16 Faculty Awards for Research, Extension, Outreach, and Service

#### 9.16.1 Alumni Award for Research Excellence

In 1975, the <u>Alumni Association</u> established the Alumni Award for Research Excellence to recognize the contributions of members of the university faculty who have made outstanding contributions in research. The Alumni Association awards two Alumni Awards for Research Excellence to individuals (or research teams) who have reached a level of research achievement judged to be the most significant within the university. To be eligible, the recipient must have been a member of the Virginia Tech faculty for a period of at least three years, and his or her research must have been conducted since joining the Virginia Tech faculty, although it may be a continuation of research started prior to coming to Virginia Tech. Alumni, students, faculty, and staff members may nominate faculty members.

The selection committee is composed of a chair appointed by the vice president for research, a non-voting representative from the Alumni Association Office, and award winners of the past five years. Each award recipient receives a \$2,000 cash prize and a plaque. Further information

regarding the award process, guidelines, dossier requirements, and deadlines is available from the Office of the Vice President for Research.

#### 9.16.2 Alumni Award for Extension Excellence

The <u>Alumni Association</u> has established the Alumni Award for Extension Excellence to recognize the contributions of members of the university's <u>Virginia Cooperative Extension</u>, honoring those members of the university and field faculty who have made outstanding contributions outside the classroom. Two awards are conferred each year, with one going to an extension specialist and one going to an extension agent. Alumni and extension faculty members nominate faculty members. The selection committee is composed of a chair appointed by the director of Virginia Cooperative Extension, past award recipients, and alumni representatives.

#### 9.16.3 Alumni Award for Outreach Excellence

The <u>Alumni Association</u> has established the Alumni Award for Outreach Excellence to recognize the contributions of members of the university faculty who have made outstanding contributions extending the university's outreach mission throughout the commonwealth, nation, and world. One award is for individual excellence and another is presented to a nominated "team." Faculty members are nominated by their peers.

The selection committee is composed of a chair appointed by the vice president for outreach and international affairs, faculty, and alumni representatives. The recipient of the Alumni Award for Outreach Excellence (individual achievement) receives a \$2,000 cash prize and a plaque. Those recognized for team achievement share a \$2,000 cash prize and a plaque. Further information regarding the award process, guidelines, dossier requirements, and deadlines is available from the Office of the Vice President for Outreach and International Affairs.

#### 9.16.4 Alumni Award for Excellence in International Education

The <u>Alumni Association</u> has established the Alumni Award for Excellence in International Education to recognize contributions by faculty and staff members who have demonstrated an impact on international education at Virginia Tech. The honor is explicitly aimed at according recognition to individuals whose efforts have resulted in thoughtful programming, curricula, or approaches to international education including service to the community; education abroad; services to international students and scholars; curriculum development; program development and external partnerships, awards, and recognitions. Selection is based upon contributions to the internationalization of Virginia Tech; the impact on students; the impact on the campus and community; the significance of the initiative; and the sustainability of the initiative.

The selection committee is drawn from the membership of the university <u>Commission on Outreach and International Affairs</u>. The selection committee may include a representative of the Alumni Association and a previous award winner. The award recipient receives a \$2,000 cash prize and a suitable memento. Further information regarding the award process, guidelines,

dossier requirements, and deadlines is available from the <u>Office of the Vice President for Outreach and International Affairs</u>.

# 9.16.5 Alumni Award for Excellence in International Outreach

The <u>Alumni Association</u> has established the Alumni Award for Excellence in International Outreach to recognize contributions by faculty and staff members who have demonstrated an impact on international outreach at Virginia Tech. Selection is based on contributions to the internationalization of Virginia Tech; global impact; significance of the outreach; and sustainability of the outreach. The honor is explicitly aimed at according recognition to individuals whose efforts are in the area of global outreach to regions, countries, communities, public, private, and non-government organizations; sponsored projects; and innovation in international outreach.

The selection committee is drawn from the membership of the university <u>Commission on Outreach and International Affairs</u>. The selection committee may include a representative of the Alumni Association and a previous award winner. The award recipient receives a \$2,000 cash prize and a suitable memento. Further information regarding the award process, guidelines, dossier requirements, and deadlines is available from the <u>Office of the Vice President for</u> Outreach and International Affairs.

#### 9.16.6 Alumni Award for Excellence in International Research

The Alumni Association has established the Alumni Award for Excellence in International Research to recognize contributions by faculty and staff members who have demonstrated an impact on international research at Virginia Tech. Selection is based upon contributions to the internationalization of Virginia Tech; global impact; significance of the research; and sustainability of the research. The honor is explicitly aimed at according recognition to individuals whose efforts are in the area of international scholarship; global research involving regions, countries, communities, public, private, and non-governmental organizations; sponsored projects; and innovation in international research.

The university <u>Commission on Outreach and International Affairs</u> selects the award winner. The award recipient receives a \$2,000 cash prize and a suitable memento. Further information regarding the award process, guidelines, dossier requirements, and deadlines is available from the Office of the Vice President for Outreach and International Affairs.

# 9.16.7 Academy of Outreach Excellence

The Academy of Outreach Excellence recognizes members of the faculty who have contributed to excellence in the outreach mission of the university. It also facilitates cooperation with the university in encouraging, promoting, recognizing, and rewarding excellence in outreach.

Academy members elect annually a maximum of four new members including the recipients of the Alumni Extension Award and the Alumni Award for Outreach Excellence and two at-large members who have made major contributions in outreach. At-large members, who are selected based on their outstanding outreach record, may be nominated by any university employee. The annual election of at-large members from the nominees is conducted by those selected to academy membership during the previous three years. All academy members retain their membership while employed by the university. Newly elected members have an "active service" status for three years.

Selection criteria for academy members include innovation, creativity, program results, subject-matter expertise, communication skills, and professional activities in extension and outreach.

# 9.16.8 Academy of Faculty Service

In recognition of exceptional service to the university by faculty members (instructional, administrative, professional, or research faculty) beyond the usual responsibilities of their faculty appointments, the university has established the Academy of Faculty Service. A nomination process is initiated each fall, with appointments to the academy made by the senior vice president and provost in the spring of each academic year. Nominations are made in recognition of individuals who have provided exemplary service in roles such as (but not limited to): committee or commission chair persons, faculty senate officers, chairperson of the Academy of Teaching Excellence, and chairperson of the university Honor System Review Board. Faculty members who have made commendable faculty service contributions outside of ongoing formal governance roles may also be nominated. The senior vice president and provost will appoint a committee to review nominations and make recommendations for appointments to the Academy of Faculty Service.

# 9.17 Awards for Creating Good Teaching and Learning Environments

# 9.17.1 Exemplary Department Award

Most of the university's awards for excellence—whether in teaching, research, or academic advising—are based on individual achievement and merit. Yet the health of the undergraduate program surely depends as much on the collective commitment of faculty members in a given department as it does on the exemplary work of key individuals. Virginia Tech has established several major annual awards that go to departments and academic units that maintain an exemplary teaching and learning environment both for students and faculty.

The university exemplary awards program seeks to clarify the expectations of faculty, and define appropriate rewards for accomplishments. The theme of the awards changes annually and is decided and announced in advance by the committee. Selected departments receive a cash award, are honored at a university reception, and may use the exemplary department designation for a period of five years.

#### 9.17.2 Diggs Roundtable

The Diggs Roundtable is a forum devoted to public discussion of major teaching, pedagogical, or curricular issues facing Virginia Tech or American higher education. Recipients of the Diggs Teaching Scholar Awards agree to help lead the Diggs Teaching Roundtable. Further

information regarding the Diggs Roundtable is available from the <u>Center for Instructional Development and Educational Research</u>.

#### 9.17.3 XCaliber Award

Established in 1996 by the Office of the Provost, the XCaliber Award (shorthand for exceptional, high-caliber contributions to technology-enriched learning activities) is presented annually by the Center for Innovation in Learning. The XCaliber recognizes faculty, graduate teaching assistants, and teams (faculty/staff/students) who have made significant contributions to integrating technology in teaching and learning in a course or in a significant technology-enriched project.

The XCaliber celebrates innovative approaches to teaching with technology that are exemplars of student-centered pedagogy. The nominees must be teaching or interacting in technology-enriched learning environments to be eligible for this award. The award recipients receive a cash prize of as much as \$2,500 (depending on the number of awards in a given year) and a commemorative plaque. Honorees are called upon to demonstrate their award-winning activities with their colleagues.

Further information regarding the award process, guidelines, dossier requirements, and deadlines is available from the <u>Center for Innovation in Learning</u>.

#### **CHAPTER TEN**

# 10.0 Research, Creative, and Scholarly Activities

Research and creative scholarship provide a vigorous learning environment for students and enhance the professional development of faculty—qualities by which the university is judged. They are vital components of the academic life of the university. Accordingly, all research and teaching faculty are expected to become involved in such activities. Departments frequently support research among the faculty by reducing teaching loads for persons who are engaged in research and by designating some departmental operating funds for research. The university provides services in support of researchers, such as computing facilities, laboratory support services, various items of high technology equipment, and library collections.

# 10.1 Procedures for Research and Scholarship

Research at the university is classified as departmental research, core research, and/or sponsored research. Individual research projects may receive funds under one or more of these categories, as described below.

# 10.1.1 Departmental Research

Research supported by departmental operating funds and/or through adjustment of teaching responsibilities is called departmental research. In this category, faculty are free to pursue research to enrich their teaching, scholarship, and greater understanding of their discipline.

#### 10.1.2 Core Research

Core research focuses primarily on the needs of Virginia and is funded by state and federal appropriations through the instructional division and Virginia Cooperative Extension/Agricultural Experiment Station. There are six core research sub-programs: agriculture and forestry research, coal and energy research, environmental and water resources research, industrial and economic development research, veterinary medical research, and supporting research. Faculty, who believe their research relates directly to one or more of the sub-programs and is applicable to problems or concerns of the commonwealth, should contact their department head or chair about procedures for securing core research support.

# 10.1.3 Sponsored Research

Sponsored research is supported through awards resulting from proposals submitted, on a project-by-project basis, by university faculty. Such proposals are submitted to state and federal agencies, corporations, and private foundations. Through sponsored research, faculty obtain the resources needed to conduct expanded research programs and may receive additional months of salary support. Research time is charged when the work activity and work reported are during the same period.

Competitive awards add stature to the recipient and the university; thus, faculty members are encouraged to seek such support. The only restrictions for sponsored research are that it must not constitute undue competition with commercial testing and research laboratories or with private consultants, and that it is compatible with the primary mission of the university. Questions about the appropriateness of a specific research project should be directed to the department head or chair or to the Office of the Vice President for Research.

# 10.1.4 Preparation of Proposals for Sponsored Projects

The Office of the Vice President for Research assists faculty in obtaining research sponsorship. Policies, procedures, and pre-award contacts are available on the Office of Sponsored Programs (OSP) website. Faculty are encouraged to explore research sponsorships by viewing funding opportunities on the Office of the Vice President for Research website. Office of the Vice President for Research support and help faculty locate programs and individuals at government agencies, industry, and private foundations.

Most funding agencies have their own scientific and technical interests and funding restrictions. Therefore prior to writing a formal proposal, faculty are encouraged to review their proposal concept with the appropriate person at the agency to which the proposal will be submitted. Personnel in the Office of the Vice President for Research help the principal investigator find the appropriate person(s) for such discussions. Faculty may wish to discuss proposal preparation with the appropriate member of their college dean's staff. Before final budget preparation, an OSP official must review the proposed budget. The OSP official provides information and guidance about university policies for cost sharing, budgetary matters, confidentiality, publication, and intellectual property undertakings.

Faculty should be aware that some agencies limit the number of proposals or the total dollar amount of proposals that can be submitted by an institution in response to a research sponsor's solicitations. The deadlines and procedures for submitting internal notices of intent for these solicitations or program notices are available at the Limited Submissions website.

The required forms for proposal submission are available on the <u>OSP website</u>. Each proposal requires considerable processing. Deadlines for submitting proposals to OSP in advance of agency deadlines are also available on the OSP website.

# 10.1.5 Laboratory Services and Facilities

Several colleges and departments maintain shops and facilities for design, fabrication, maintenance, and repair of specialized equipment. The <u>Office of the Vice President for Research</u> can assist faculty in locating an appropriate facility.

<u>Environmental Health and Safety</u> (EHS) must be consulted *before* any laboratory is established in a university facility. The EHS staff determines if the proposed laboratory meets all necessary facility and laboratory requirements. The staff ensures that all personnel who will be working in the laboratory are familiar with the various university policies, procedures, and publications that cover laboratory operations. These may include chemical hygiene plans, laboratory safety

manuals for biohazard safety level-2 or biohazard safety level-3 labs, and lab licensing and radiation safety manuals approved by the <u>Radiation Safety Committee</u>.

Research compliance procedures related to Select Agents and Toxins (SAT) are overseen and enforced by the SAT Security and Incident Response Committee. This committee ensures that the university safeguards SATs from unauthorized access, theft, loss or release, and works to meet or exceed the requirements of federal, state, and local regulations, university policies, and standards of practice, as applicable.

New or proposed animal facilities are inspected by the attending veterinarian and approved by the <u>Institutional Animal Care and Use Committee</u> (IACUC) before animals are ordered and housed in the facility. The use of animals in teaching and research is covered in IACUC-approved protocols.

# 10.1.6 Ownership and Control of Research Results

The university asserts its right to the results of research funded wholly, or in part, with university resources. University ownership of intellectual properties is covered in university policy 13000, "Policy on Intellectual Property." These ownership rights extend to all permanent, visiting, or research faculty, staff, wage employees, and students.

The faculty principal investigator or project leader is expected to manage the university's ownership of research results and material (including all data) that best advance the standard routes of publication, presentations, and other usual means of dissemination of research results for that particular field. Creation of intellectual properties must be disclosed to the university by submitting an IP disclosure form, which is available on the <u>Virginia Tech Intellectual Properties</u> website.

It is the responsibility of the faculty principal investigator as project leader to preserve the research material and results in the manner that is customary to the field. This includes all notebooks, computer files, samples, specimens, prototypes, etc. germane to the veracity and validity of the research claims. Sponsored research projects may require additional document retention based on sponsor requirements or fulfillment of the project's data management plan, included in the original proposal and as required by state law.

In the event that the faculty principal investigator or project leader permanently departs the university, it is the responsibility of that person's department head or chair to determine the disposition of the research materials and results.

It is the responsibility of principal investigators, student thesis chairs, and dissertation committee chairs to preserve the confidentiality of research material and results and to establish who may have access to them. Normally, access is limited to university faculty, staff, and students involved in generating these materials and/or needing access to them for *bona fide* research or administrative purposes and as required by state law.

#### 10.1.7 Classified Research

The U.S. government occasionally seeks the expertise of Virginia Tech faculty to engage in classified research. Faculty must realize that working in classified or federally controlled unclassified research requires that they relinquish opportunities to disseminate the knowledge gained in this effort. However, the university does recognize that individual investigators may wish to work in areas that have classified or controlled unclassified aspects, and cannot conduct the research in compliance with applicable federal statues and executive orders without access to classified or controlled unclassified information. To accommodate this need, the university has a continuing compliance and security program administered by the Office of Export and Secure Research Compliance in accordance with government regulations.

# **10.1.8** Securing Theses and Dissertations

The <u>graduate school</u> may secure theses and dissertations for up to one year for purposes of obtaining a patent or for other proprietary reasons. To secure a thesis or dissertation, an electronic thesis/dissertation (<u>ETD</u>) <u>approval form</u> must be completed and signed by the thesis or dissertation author and by his or her advisor, with a request that the thesis or dissertation be secured. Securing involves the withholding of the thesis or dissertation from any publishing process.

#### 10.1.9 Publication of Research

Because of its importance to the faculty, department, and the university, assistance is offered faculty in publishing the results of creative scholarship and research. Department heads or chairs can help determine how best to complete and publish the results of research projects.

# 10.1.10 International Travel Supplemental Grants Program

The supplemental grants program provides funds to support faculty travel to international conferences for presenting invited papers and for establishing international collaborations. Designed to promote professional development, to encourage the involvement of faculty in international activities, and to enhance the international reputation of the university, the program supports requests from tenured, tenure-track faculty and selected ranks of research faculty. The program is administered by the Office of the Vice President for Research. Requests are submitted no later than 30 days before the traveler's departure date. Additional information, application forms, and guidelines are available on the Office of the Vice President for Research website.

# 10.2 Policy on Misconduct in Research

The university endorses high ethical standards in conducting research to ensure public trust in the integrity of research results. The university recognizes that deception in research erodes the credibility of an institution and the confidence of those who might benefit from the research. The university takes all reasonable and practical steps to foster a research environment that promotes the responsible conduct of research and research training (and activities related to that research

or research training), discourages research misconduct, and deals promptly with allegations or evidence of possible research misconduct.

#### 10.2.1 Definition

Misconduct in research (or research misconduct) means fabrication, falsification, plagiarism in proposing, performing, or reviewing research; or in reporting research results. *Fabrication* is making up data or results and recording or reporting them. *Falsification* is manipulating research materials, equipment, or processes, or changing or omitting data or results such that the research is not accurately represented in the research record. *Plagiarism* is the appropriation of another person's ideas, processes, results, or words, including those of a student, colleague, or mentor, without giving appropriate credit.

Research misconduct does not include honest error, differences in opinion, or disputes over authorship except those involving plagiarism. It does not include issues relating to sexual harassment, personnel management, fiscal errors, or the reporting of poor effort. It also does not include abuse or improper procedures with laboratory animals or human subjects.

#### 10.2.2 Activities Covered

The university's policy on misconduct in research applies to allegations of research misconduct (fabrication, falsification, or plagiarism in proposing, performing, or reviewing research, or in reporting research results) involving a person who was employed by, was an agent of, or was affiliated by contract or agreement with the university and was engaged in research under the auspices of the university at the time of the occurrence of the alleged research misconduct. In addition, any student engaged in sponsored research is covered by this policy.

Misconduct in non-research activities and other ethical violations are covered by separate policies. Ethical misconduct of faculty is covered in section 2.7, "Professional Responsibilities and Conduct," which describes the principles of ethical behavior. Violations of ethical conduct by graduate students are guided by the constitution of the graduate honor system, which is available in the graduate policies and procedures section of the *Graduate Catalog*. Violations of ethical conduct by undergraduate students are guided by the university's honor system, as outlined in the *Hokie Handbook*. Standards of conduct and performance, as well as procedures for dealing with alleged violations of unacceptable conduct and grievance procedures for staff, are available on the human resources website.

# 10.2.3 Procedures for Reporting, Investigating, and Resolving Misconduct in Research

The university has established detailed procedures for reporting, investigating, and resolving misconduct in research. Those procedures are available in university policy 13020, "Policy on Misconduct in Research." The research integrity officer is responsible for overseeing the procedural process. Any questions regarding the policy or procedures should be addressed to the <u>research integrity officer</u> in the Office of the Vice President for Research.

**Reporting**: Any instance of observed, suspected, or apparent research misconduct should be reported so that it may be thoroughly investigated and promptly resolved by the university.

**Cooperation**: Faculty will cooperate with the university and provide evidence relevant to reviewing and resolving research misconduct allegations.

**Confidentiality**: University procedures limit disclosure of identities, information regarding misconduct allegations, and proceedings to those with a need to know, consistent with a thorough, competent, objective, and fair research misconduct proceeding, and in accordance with applicable law.

**Protection of Persons Involved**: There will be no retaliation against persons involved in the process. Any potential or actual retaliation shall be immediately reported to an appropriate university official. University officials make all reasonable and practical efforts to protect or restore the reputation of persons alleged to have engaged in research misconduct, but against whom no finding of research misconduct is made.

**Assessment**: Allegations of research misconduct are assessed to determine if they fall within the definition of research misconduct and if they are sufficiently credible and specific so that potential evidence of research misconduct may be identified. An inquiry must be conducted if these criteria are met. Allegations that do not proceed to inquiry, but which indicate that other university policies may have been violated, shall be reported to the appropriate university officials or units for further review and/or action.

**Inquiry**: Inquiry means the information gathering and initial fact-finding to determine whether an allegation or apparent instance of misconduct warrants an investigation. An inquiry committee evaluates the evidence, including the testimony obtained during the inquiry, to determine if an investigation is warranted.

**Investigation**: Investigation is the process of developing a factual record by exploring the allegations in detail and examining the evidence in depth. An investigation committee conducts this process and prepares a final investigation report, which includes recommended findings on whether research misconduct was committed, by whom, and to what extent.

**Final Decision**: The investigation report is provided to the provost for a final decision. The provost consults with the investigation committee regarding the committee's findings. The provost may return the report to the committee with a request for further fact-finding or analysis. The provost determines—in writing—the final disposition of the case, including the recommended institutional actions.

**Appeal**: A person found to have engaged in research misconduct may initiate an appeal of the provost's decision. An appeal shall be in writing to the president and shall specifically identify the subject matter of the appeal and provides basis or evidence to support the appeal. The president consults with the provost, the research integrity officer, the investigation committee, and others as necessary when reviewing the basis for appeal. The president provides a written

decision on the appeal and the actions to be taken. The decision of the president is the final resolution of the appeal.

**Reporting**: The university provides notices and reports to sponsors of the research as required.

**Questions**: Any questions regarding the policy or procedures should be addressed to the <u>research</u> integrity officer in the Office of the Vice President for Research.

# 10.3 Removal of a Principal, Co-Principal, or Lead Investigator

Funding agreements are legal contracts between the sponsor and the university rather than an individual, thereby obligating the university to ensure compliance with any and all applicable policies, regulations, or specific conditions attached to the funding. The removal of a principal, co-Principal, or lead investigator is governed by <u>University policy 13025</u>. Funding agencies and sponsors vary in their requirements; the terms of the specific contract with a sponsor guide the university's actions whenever this policy is invoked. This policy applies to investigators who hold identified responsibilities as principal, co-principal, or lead investigator (hereafter referred to collectively as the investigator).

Removal of an investigator from a sponsored project may be necessary or warranted under unusual circumstances such as incapacity (unable to carry out the responsibilities as an investigator), misuse of funds, failure to comply with university and sponsored programs' policies or state or federal regulations, failure to disclose or appropriately manage a significant conflict of interest, or in response to a request by the sponsor of the project.

Significant issues that are uncovered by the university or brought forward by the sponsor should be addressed with the investigator as soon as possible so that he or she has an opportunity to rectify the problem(s) if possible. The Office of the Vice President for Research must consult with the relevant department head or chair and dean (or other senior manager if the investigator is not in an academic college) in addressing the deficiencies prior to removal, evaluating the validity of the concerns, or determining other appropriate resolutions to the identified issues. Referral to and formal investigation by a university entity charged with such responsibility—such as internal audit in the case of alleged fraud or misuse of funds—follows standard university policies and procedures. Reassignment of responsibility for authorization of project expenditures may be necessary pending the outcome of the investigation.

Should circumstances warrant removal, the vice president for research informs the investigator in writing, including a statement of the justification and supporting evidence for the removal. If the sponsor initiates a request for removal, the vice president requests such a statement in writing from the sponsor. The investigator has at least 10 working days from receipt of the notification to respond. If this exchange does not result in a satisfactory resolution of the issue(s) and the removal becomes involuntary, the investigator may appeal within five working days to the provost.

Notification of the sponsor is in accordance with the terms of the specific sponsored contract and/or requirements of the funding agency whenever the university initiates temporary

suspension or permanent removal of an investigator. Reassignment of project leadership, when necessary, occurs in consultation with the sponsoring program manager.

# 10.3.1 Appeal of Removal as a Principal, Co-Principal, or Lead Investigator

Within 10 working days (or as expeditiously as possible), the provost appoints an appeals committee composed of three non-administrative faculty members chosen by the provost: one member chosen from the Faculty Senate Faculty Review Committee, one member chosen from the Committee on Faculty Ethics, and one member from the general faculty. The committee elects its chair.

The vice president for research presents the appeals committee with the rationale and evidence that led to the recommendation for removal. The investigator, if he or she so elects, has an opportunity to provide a statement to the panel. Documents are submitted in accordance with deadlines and guidelines set by the committee and made available to the other party. The appeals committee may choose to review submitted documents and/or hear from the parties.

The committee's responsibility is to determine if there is reasonable cause for removal, that the investigator has a fair opportunity to rebut the evidence as provided for in this policy, and that university policies and procedures are followed appropriately. The committee provides its recommendation in writing to the provost within 30 calendar days. The provost renders a decision on removal of the investigator within five working days after receiving the committee's recommendation. If the decision of the provost does not agree with the recommendation of the appeals committee, the investigator may appeal within five working days to the president of the university. The president's decision is final.

Time limits above may be extended by necessity and by mutual agreement of the parties. Given that this appeal process provides an opportunity for referral to a faculty panel and review by university executives, the same matter may not also be the subject of a grievance.

# 10.4 Effort Certification and Salary Charges to Sponsored Grants and Contracts

#### 10.4.1 Effort Certification

The purpose of effort certification is to confirm that the salaries and wages charged to each sponsored agreement reflect a reasonable estimate of the work performed. University policy 3105, "Effort Certification," describes the procedures for required effort certification in accordance with federal regulations. Individual investigators, departments, and other university administrators have specific responsibilities under the policy for certifying effort, monitoring compliance, and assuring that only allocable charges are made to grants and contracts. Federal audits have made clear that only effort directly related to a project can be charged to that project and salary expenditures on behalf of the project must occur during the effort reporting period. The university takes its obligations to comply with federal regulations very seriously; failure to comply may mean severe financial penalties and/or loss of opportunity for future grants from the federal sponsor. To be consistent and fair to all sponsors, the same kind of accountability applies to non-federal grants and contracts.

Effort certification is particularly complex for instructional faculty members who manage multiple responsibilities simultaneously, seamlessly moving from class to supervising graduate students, to conducting research and developing the next proposal in the same day or week. Indeed, most instructional faculty members are engaged in teaching, administrative tasks, or other duties in addition to their work on sponsored projects, even during the summer. Yet only activities directly related to a sponsored grant or contract may be charged to that grant or contract; institutional activity must be supported by other, non-sponsored funding (or may be uncompensated during the summer).

If the faculty member (regardless of type of appointment) has responsibilities for competitive proposal writing or participation in well-defined, regular teaching or administrative duties (e.g., committee work, hiring, advising, tenure review), a 100 percent allocation of the salary to sponsored projects would be prohibited during the effort reporting period in which such activity occurred. Incidental, inconsequential non-project activity performed rarely may be considered *de minimus* and need not be part of full load for purposes of effort reporting.

Proposal writing for new competitive awards and competitive renewal awards may not be charged to sponsored projects, nor would such proposal writing be considered *de minimis* activity. Preparation of non-competitive, continuation award proposals (progress reports) may be charged to the applicable sponsored project.

Faculty members who receive summer salary from sponsored projects must certify to the effort expended on those projects during the summer period. Work done on the sponsored project during the academic year cannot be counted toward summer effort on the project.

Failure to follow the provisions of university policy 3105, "Effort Certification," may subject the individuals and departments responsible for the violation(s) to administrative and/or disciplinary actions in accordance with university disciplinary procedures. Sanctions for non-compliance may include, but are not limited to:

- If effort reports are not completed and returned in a timely manner, salary costs associated with uncertified grant activity may be removed and charged to a departmental account.
- Following appropriate notice, faculty members with delinquent or improperly completed
  effort reports may be placed on a suspension list by the <u>Office for Sponsored Programs</u> and
  denied eligibility for OSP services, including but not limited to proposal preparation, account
  set-up, and budget transfers, until effort reports are up to date and properly completed and
  certified.
- Certification of effort reports that are known to be materially inaccurate may expose the individual who completed the reports to personal disciplinary actions.

# 10.4.2 Compliance Issues Related to Summer Research Appointments for Nine-Month Faculty Members

Faculty members on academic year (nine-month) appointments are permitted to earn up to three months of additional salary for effort related to sponsored projects, subject to sponsor policies

and appropriate internal approvals. Summer funding may be accomplished by research extended appointments or as summer wages.

University policy 6200, "Policy on Research Extended Appointments," outlines the requirements and procedures for faculty members to extend their nine-month appointments to 10-, 11-, or 12month appointments depending on the availability of sponsored funding for additional months of salary and full fringe benefits. Although the sponsored funding supports the extended employment contract, salary must be charged to reflect a reasonable estimate of effort throughout the entire appointment period, not just the summer. Given the continuation of some typical university responsibilities during the summer, such as meeting with graduate students, attending professional conferences, or preparing future grant proposals or coursework, faculty members should have a mixture of sponsored and institutional funding to support their summer activities. This can be accomplished by making appropriate charges to the project during the academic year, and deferring some institutional funding to the summer period. Faculty members on research extended appointments earn annual leave proportional to the length of their appointment, and they must record the use of annual leave whenever used during the appointment period (all 10, 11, or 12 months). There is no payout for accrued annual leave at the time of reconversion to the base academic year appointment or at the time of separation from the university.

Instead of research extended appointments, academic-year faculty members may receive support from sponsored grants and contracts as summer research wage payments, without full fringe benefits. This would typically be the case for faculty members with one or two months of "summer salary" included in the funded grant project. For those with three full months of funding, project effort during the academic year may be charged to the grant (with attendant changes in the fringe benefit rate), thereby allowing departmental salary savings to support non-project related responsibilities during the summer. Faculty members certify their effort across the entire summer period, and some flexibility is allowed as long as the overall effort and salary charges during the period are consistent.

# 10.4.3 Compliance Issues for Research Faculty Members

As described above, a research faculty member with regular, well-defined responsibilities for new proposal preparation, teaching, or administrative duties is prohibited from charging 100 percent of salary to sponsored projects during an effort reporting period in which such activity occurred, unless those activities are specifically allowed on the sponsored project.

Research faculty members are typically on standard 12-month appointments, which earn and accrue annual leave by university policy. Use of annual leave is recognized as an acceptable charge to a sponsored project when such leave is part of the standard university appointment.

#### 10.5 Policy on Intellectual Property

University policy 13000, "Policy on Intellectual Property," outlines intellectual property (IP) ownership criteria, resolution of ownership questions, and responsibilities of university employees concerning intellectual properties.

Publicly (state) supported universities have the multiple missions of teaching, research, support of the public interest and fostering of economic development of the area/state in which they are located. Scholarly activities in a university setting create Intellectual Properties (IPs). IPs include research papers, books, software programs, new inventions, journal articles, etc.

The university's mission includes dissemination of IPs in the most efficient and effective manner possible. The identification and optimization of opportunities for the industrial/commercial utilization of some IPs is also part of this mission, as is the protection of the ownership rights of both the individuals and the university.

While many IPs are best disseminated by publication and placing in the public domain, there are a significant number that are most effectively handled by protection under the IP laws (i.e., patenting and copyright) and licensing (or other transfer) to private sector entities, with attendant financial considerations.

This Policy is designed to establish the rationale and the mechanisms to:

- A. Establish ownership criteria and resolve ownership questions if such arise.
- B. Define the responsibilities, rights and privileges of those involved.
- C. Develop basic guidelines for the administration of the IP Policy.

# 10.5.1 Authority and Responsibility of the Intellectual Properties Committee

The Intellectual Properties Committee membership shall be as stated in the By-Laws of University Council. The IPC shall have the following authority and responsibility with respect to Intellectual Property:

- A. To develop and recommend university policy and policy changes dealing with IP to the Commission on Research.
- B. To review all disputed invention disclosures submitted by Virginia Tech faculty, staff and/or students for:
  - 1. Complete and appropriate disclosure of individuals involved in the invention and/or creation of the IP.
  - 2. Confirm the determination of IP ownership by university, originating individuals, research sponsors and/or governmental agencies.
  - 3. Examination and recommendation to the Provost for disposition of (1) and/or (2) above in those cases where a dispute remains.
  - 4. Coordination of evaluation and recommendation to Virginia Tech Intellectual Properties, Inc. (VTIP) of technical merit, economic potential and protection/marketing priority as needed.
- C. To make recommendations to the Senior Vice President and Provost (Provost) for the sharing of royalties between the university and the authors or inventor(s) of the IPs owned by the university.

D. To promulgate such guidelines and procedures as may be necessary for the implementation of this Policy.

Much of the work of the IPC as defined above will be addressed through the normal business of the full committee. However, it may be prudent in the review of certain disputed invention disclosures to have a subgroup of the entire IPC to more fully consider all necessary aspects of the dispute. This shall be the role of the Ownership Review Group, which will be composed of three at-large members of the IPC (selected by the IPC Chair) and be chaired by the IPC Chair. This group shall meet as needed with the following agenda:

- A. Review all disclosures submitted that have ownership in dispute.
- B. Confirm university ownership as necessary for those disclosures in which originator(s) have indicated Virginia Tech ownership.
- C. Review disputes involving sponsor ownership/rights.
- D. Review, discuss and reach preliminary conclusions on ownership disputes and forward recommendations to the full IPC.

# 10.5.2 Policy Guidelines

This section outlines the criteria to be used by the IPC and its working groups in their deliberations, findings and recommendations. To the extent that individual questions are not specifically addressed, these guidelines will, at the least, give a general indication of intent and philosophy and allow proper interpretation.

# A. Ownership of IP

For purposes of this policy creations are divided into two groups:

- 1. The traditional results of academic scholarship, i.e. textbooks, literary works, artistic creations and artifacts.
- 2. The novel results of research such as products, processes, machines, software, biological technology, etc.

Intellectual properties in the first (traditional) group are considered to make their full contribution to the university's benefit by their creation and by continued use by the university in teaching, further development, and enhancement of the university's academic stature; the presumption of ownership is to the author(s). Thus, unless there is explicit evidence that the work was specifically commissioned by the university, the IP rights remain with the author(s) and the university rights are limited to free (no cost) use in teaching, research, extension, etc. in perpetuity.

In the second group, as a condition of employment or other involvement in research and/or related activities using university resources, the ownership is to the university (with the originator having a right to share in the benefits derived therefrom in accord with university sharing guidelines). Thus unless there is convincing and explicit evidence that the IP was developed without the use of university resources and/or facilities (which may include but is not limited to any of the following: use of equipment, lab or office

space, university time of originator and/or personnel under his/her control, funds supplied by the university and/or funds originating from sponsored research projects and/or donations to university/affiliated companies, etc.), ownership of the IP rests with the university and the originator(s) do hereby assign ownership, right, title, and interest in any IP, discovery, or invention to the university.

Within the above general guidelines, the following situations are more specifically defined:

- 1. Sponsor Rights: In the case in which an IP is generated as a result of research funded by a private sector company under a sponsored research project, the IP rights of the sponsor as defined in the applicable clauses ("Patents & Copyrights," "Intellectual Properties," "Inventions," etc.) of the Sponsored Research Agreement (as approved by the Vice President for Research or their designee and signed by an authorized officer of the university) shall take precedence over the rights of the university/inventor(s). Any residual rights not accruing to the sponsor shall be as defined in the general guidelines above.
- 2. Federal Agency Rights: Research projects sponsored by an agency of the federal government have statutory IP rights that are limited (in almost all cases) to a non-exclusive non-transferrable royalty-free license to any patent generated by the research, provided the inventor(s)/university advise the agency in a timely manner of their intent to retain their rights and provide for legal protection (i.e. patenting). It is the responsibility of the researcher to advise the agency of the creation of the IP and (with the assistance of the university IP manager, VTIP) advise of the protection steps being undertaken. The residual rights not belonging to the sponsoring agency shall be as defined in the general guidelines above.
- 3. In the event the following condition(s) apply, students, visiting scholars, and volunteers do hereby assign any IP rights to the University when:
  - a) working on a research project funded by Virginia Tech or an entity outside of Virginia Tech sponsoring the research through Virginia Tech from which the IP was created; or,
  - b) employed or receiving payment from Virginia Tech related to a project from which the IP was created; or,
  - c) university resources not typically available to the public are used in the creation of the IP.
- 4. Joint Inventorship: For IPs generated by a team of inventors in which one or more are not members of the faculty/staff/supported students, each inventor is usually entitled (by law) to shared ownership of the entire right. The university's claim to the shares of university-associated inventors will be as outlined in these guidelines. Ownership of outside inventors will vest in them or their assignees.

- 5. Special Situations: In the event that an IP ownership situation arises which is not addressed in either the general or specific guidelines outlined above, the IPC shall make a recommendation based on the spirit of the guidelines. A record of the rationale used to arrive at their recommendation shall be kept and used as a precedent for the handling of future special situations if applicable.
- 6. Commissioned Works: There are times when the university will choose to enter into a contractual arrangement to commission a specific work or undertaking. The university, as the commissioning party, may be expected to maintain certain rights of third parties. These rights are negotiable on a case by case basis, but generally the university expects, at a minimum:
  - a) Exclusive right to give premiere performances of the commissioned work; and,
  - b) Exclusive performance rights for a limited period of time; and,
  - c) Exclusive right to give premieres in other venues; and,
  - d) Right to make the first commercial recording of the work; and,
  - e) The right to be credited as the commissioner of the work in published editions, recordings, and programs for all future performances; and,
  - f) The nonexclusive right to use the commissioned work, without cost, in teaching, research, outreach, etc., in perpetuity.

### B. Obligation to Disclose

While it is recognized that faculty and staff mission and expertise is concentrated and directed in areas other than commercial utilization, originators of new technology shall submit a disclosure when any IP is developed. Timely (i.e., before publication or other enabling non-confidential disclosure) submission of a disclosure to VTIP may also be critical to the value of the IP.

To the extent (and as soon as) the researcher/inventor/creator obtains research results that may be considered an IP and recognizes that they may have potential for commercial utilization there exists an obligation to bring these results to the attention of VTIP in the form of a disclosure.

If, in the absence of a timely disclosure, commercial utilization of a technology takes place with the direct or indirect involvement of the originator(s) but without involvement by the university it will be deemed that the originator(s) have not fulfilled their obligation to disclose and the university may:

- 1. Take whatever legal and/or business action is necessary to protect its rights and rightful share of financial benefits and ownership.
- 2. Deny to originator(s) any share of revenues which would otherwise accrue to them under this policy.

# C. Revenue Sharing

Revenues generated by the successful commercialization of IPs owned by the university (whether or not protected by patent and/or copyright) shall be shared equally between the university and the originator(s) of the IP, subject to the conditions and exceptions outlined below.

- 1. Revenues subject to sharing include royalties, licensing fees, incentives, etc. received by the assignee licensor organization, less the costs/expenses described below. Specifically excepted from sharing are payments received and designated for specific purposes such as sponsored or unrestricted research grants, services to the university, research equipment and/or materials, consulting fees to researchers, etc. These payments will go directly to the designated entity and purpose.
- 2. Also excepted from sharing are revenues resulting from:
  - a) Tasks and/or activities specifically and explicitly assigned to employees by an administrative unit of the university, or
  - b) Activities and/or tasks clearly defined in the written, university approved, policy of an administrative unit of the university.

Such revenues, flowing through the university assignee organization, will accrue to the originating administrative unit of the university net of development costs.

- 3. Expenses to be subtracted from gross revenue before sharing shall be limited to documented direct and indirect costs for protection (patenting), marketing and development of the IP. Specifically excluded are costs incurred in the generation of the IP (i.e. research costs). Development costs shall include (but not be limited to) payments made to (or retained by) non-affiliated organizations (e.g. Research Corp. Technologies, CIT, etc.) involved in the process of commercializing the IPs owned by the university.
- 4. Non-cash compensation for rights to an IP may be accepted but only with the informed consent of the originator(s) of the IP. The share of net revenue not paid to the originator(s) (50 percent) shall be applied as follows:
  - a) A portion equivalent to at least 10 percent of total net revenue may be distributed to the originator(s)' primary unit(s) (e.g. departments, centers, etc.).
  - b) The remainder to the university assignee organization (VTIP).

#### D. Management Responsibility

Virginia Tech Intellectual Properties, Inc. (VTIP), a non-profit corporation affiliated with the university, has been established and charged with the mission of protecting and utilizing IPs for the benefit of the university.

All IPs assigned to the university shall flow to VTIP by assignment for operational management. The IPC should make appropriate inputs and recommendations as to

disposition and priority of individual IPs. Originator(s)' inputs/suggestions to VTIP are also appreciated.

# E. Right Of Appeal

The originator(s) of an IP covered by this Policy shall have the right to appeal application of the policy to the IPC.

The IPC will formulate recommendations relative to each such appeal, and will forward both the appeal and its recommendations to the Office of the Senior Vice President and Provost in a timely manner. The Provost will determine the university's response to each appeal, and will so notify the originator(s) and the IPC.

If the originator(s) disagree with the IPC recommendation regarding ownership, a written appeal to the Provost must be filed within (30) thirty days of receipt of notification of the IPC recommendation. This appeal should contain an exposition of the facts as seen by the originator(s), any information they deem pertinent to the case, as well as any applicable citations of policy guidelines. A copy of the appeal document should be sent to the IPC via its Chair.

Upon receipt of the appeal, the Provost may elect to consult with any and all concerned prior to reaching a decision in the case.

In the event that any member of the university (faculty, staff or student) perceives and/or becomes aware of any irregularity in the inventorship/authorship of an IP disclosed (or about to be disclosed) to VTIP or the IPC he/she should bring it to the attention of the other inventors/authors involved and/or the Department Chair(s) (or the director or unit leader in situations outside the traditional academic departments) concerned in an attempt to resolve the conflict equitably and amicably. Failing such resolution, the facts of the cases should be submitted in writing within (30) thirty days to the Vice President for Research (with copy to the Chair of the IPC) with a request for review by the Ownership Review Group of the IPC.

Upon receipt of such a request, the Ownership Review Group shall review the facts of the case, convene a hearing for all concerned parties, reach a conclusion and present a synopsis of the case and a recommendation to the full IPC who will, in turn, make a recommendation to the Provost.

#### CHAPTER ELEVEN

# 11.0 Faculty Benefits Program

This section provides an overview of benefits that are available to the faculty. The Human Resources Service Center in human resources should be contacted to obtain detailed information pertaining to the benefits programs or to make changes to one's current benefits.

Faculty benefits are either required benefits that are mandated by federal, state, or university regulations or they are optional benefits that are available to be used at the discretion of faculty members.

#### 11.1 Required Benefits

# 11.1.1 Group Life Insurance

Participation in the group life insurance program is required of all full-time and part-time salaried faculty. The university pays the monthly premium for the group life insurance.

Coverage is effective on the first day of employment. The amount of the insurance is determined using the annual salary rounded up to the nearest thousand dollars; then doubled. Thus, if the salary is \$49,400, the amount of insurance is \$100,000. In cases of accidental death, the insurance is four times the annual salary. In the example above, the insurance payout would be \$200,000.

A faculty member who leaves the university may convert the term insurance policy to a private policy if the request is made within 31 days after termination. Eligibility to make this conversion will depend upon individual circumstances. Please contact the Human Resources Service Center for detailed information. Faculty who retire from the university either on disability or after age 50 with at least 10 years of state service will have continued life insurance coverage.

The life insurance program is administered through the Virginia Retirement System and is underwritten by Minnesota Life.

# 11.1.2 Long-Term Disability Insurance

Long-term disability insurance provides coverage for 60 percent of salary after a six-month waiting period if the employee is deemed disabled. These benefits will be offset by Social Security, federal retirement (if applicable), and Virginia Retirement System benefits. The maximum monthly benefit is \$15,000; and the minimum monthly benefit is \$100. In addition to paying a monthly benefit, a contribution of 10.4 percent (8.5 percent for faculty members hired on or after July 1, 2010) of base salary is paid into the Optional Retirement Plan for-those faculty members receiving disability benefits. The Standard Life Insurance Company underwrites the program. This benefit is not optional, and the faculty member pays the monthly premium, which is .26 percent of the employee's salary.

# 11.1.3 Faculty Retirement

All eligible faculty are required to participate in either the defined-benefit Virginia Retirement System (VRS) plan or a defined-contribution Optional Retirement Plan (ORP), which offers a choice of retirement programs. Faculty have 60 days from the date of appointment to choose either the VRS defined-benefit plan or the ORP. If no choice is made, the retirement default is the defined-benefit plan\*.

\*Effective January 1 2014, all eligible faculty members with no prior state service are required to participate in either the VRS Hybrid Retirement Plan or the defined-contribution Optional Retirement Plan (ORP).

Part-time salaried faculty members working half time or more for at least six months on a calendar year appointment or one semester if on an academic year appointment are eligible to participate in either the university's defined contribution ORP or the Virginia Retirement System to the extent permitted by VRS.

# 11.1.3.1 Virginia Retirement System

The Virginia Retirement System (VRS) plan is qualified under IRS code section 401(a). Vesting as it relates to using VRS as a pension plan occurs after five years of credited service. All employees enrolled in VRS are required to pay 5 percent of their salary into the VRS. These funds may be withdrawn or rolled into other accounts at the time of termination in lieu of leaving them with the state retirement system. If vested, the employee can draw a retirement income at a later date. Investment risks are borne by the employer. The retirement income benefits are based on the following a formula that utilizes:

- the highest consecutive 36 months of salary (for employees hired on or after July 1, 2010; or
- 60 months of the highest consecutive salary for employees not vested on January 1, 2013, and;
- years of state service, and age at retirement. There is post-retirement cost-of-living allowance tied to the Consumer Price Index-Urban not to exceed 5 percent per year under current law. For more information, contact www.varetire.org.

# 11.1.3.1.1 The Hybrid Retirement Plan

The Hybrid Retirement Plan is a qualified plan under IRS code section 401(a) and contains both a defined benefit portion and a defined contribution portion. Both the employee and the employer make contributions to fund the defined benefit portion of the Hybrid Retirement Plan. The employee contributes 4 percent of their creditable compensation. The employer contribution is actuarially determined. The employee is required to contribute 1 percent to the defined contribution portion of the Hybrid Retirement Plan. The employee also has the option to voluntarily contribute up to an additional 4 percent in .5 percent increments. The employee's voluntary contributions will be made to the state deferred compensation plan which is qualified under IRS code 457(b). The employer must match the first 1 percent of voluntary contribution with a corresponding contribution of 1 percent of the employee's creditable compensation. Each

of the employee's additional .5 percent increases will be matched by the employer with a .25 percent contribution. Employees can increase or decrease their contributions on a quarterly basis.

Defined Benefit vesting is the minimum length of service members need to qualify for a future retirement benefit. Vesting occurs when a member has at least 60 months of service credit. If vested, members are eligible to receive all member contributions upon retirement or leaving employment. If members are not vested, employer-paid contributions are forfeited upon retirement or leaving employment.

Defined Contribution vesting is the minimum length of service members need to be eligible to withdraw contributions. Vesting is based upon the length of participation in the plan. Upon retirement or leaving employment, members are eligible to withdraw a percentage of employer contributions based upon the following schedule:

- After 2 years, members are 50 percent vested and may withdraw 50 percent of the employer defined contribution plan contributions.
- After 3 years, members are 75 percent vested and may withdraw 75 percent of the employer defined contribution plan contributions.
- After 4 or more years, members are 100 percent vested and may withdraw 100 percent of employer defined contribution plan contributions.

# 11.1.3.2 Optional Retirement Plan

Within 60 days of the date of appointment, eligible faculty may select the <u>Optional Retirement Plan</u> (ORP) in lieu of the defined benefit Virginia Retirement System (or The Hybrid Plan after January 1, 2014). ORP contribution rates are as follows:

- For faculty members hired before July 1, 2010, the university contributes 10.4 percent of base salary to the 401(a) ORP account.
- For faculty members hired on or after July 1, 2010, the university contributes 8.5 percent of base salary and the employee contributes 5 percent of their base salary to the ORP.

The employee contribution is waived if the employee was previously enrolled in VRS or a Commonwealth of Virginia Optional Retirement Plan and still has an active account. Benefits at retirement are based on contributions, net earnings, and age. The faculty member can choose among a number of investment options. The ORPs are qualified under IRS code section 401(a) and vesting is immediate. Investment risks are borne by the employee with risks varying based on types of funds selected.

# 11.1.4 Short-Term Disability Income Protection

Regular faculty are provided with 1040 hours of sick leave at the time of hire. This equates to six months of income protection, i.e., short-term disability coverage. After short-term disability, is exhausted, the faculty long-term disability program outlined above (see 11.1.2) takes effect and provides disability income to age 65 or for five years if over age 60 at the onset of disability.

This benefit is not available to faculty in restricted positions. Faculty in restricted positions accrue five hours of sick leave per pay period. <u>Beginning July 1, 2012</u> all of Virginia Tech's restricted faculty members who are not enrolled in the Virginia Sickness and Disability Program

(VSDP) will be covered under a short term disability plan administered by the Standard Insurance Company.

Restricted faculty who accrue sick leave each pay period will automatically be enrolled in the Standard Insurance short term disability plan. These employees will continue to accrue sick leave and will not be required to pay any additional premium for the coverage.

The Standard Insurance Short Term Disability plan allows employees who are on approved short term disability leave to collect up to 60 percent of their regular salary for up to six months, after the initial seven day waiting period. Employees who are still considered disabled after six months will transition into the Faculty Long Term Disability plan (see section 11.1.2).

Alternatively, faculty members on restricted appointments who enroll in the Virginia Retirement System may consider the Virginia Sickness and Disability Plan (VSDP), which provides short-term disability for six months and long-term disability to age 65 or later depending on age at the time of disability. There is a one-year waiting period from the date of appointment under VSDP for the short- and long-term disability benefits.

# 11.2 Optional Programs

There are several insurance plans and benefits programs available for faculty members. Some have a portion of the premiums paid by the university; others are the employee's responsibility.

# 11.2.1 Long-Term Care

An optional long-term care plan is available through Genworth. Individuals may also apply for long-term coverage for spouse, parents, and parents-in-law. There is a guarantee issue for new hires who apply within sixty days of appointment.

#### 11.2.2 Accidental Death and Dismemberment Insurance

<u>Accidental death and dismemberment insurance</u> is an optional program available to faculty and staff members on at least half-time appointment. The policy is with Zurich.

Accidental death and dismemberment insurance is available in multiples of \$5,000, in a range of benefits from \$10,000 to \$250,000. The premium is paid entirely by the employee. Coverage is effective the first day of the month following the month in which the application is received by human resources. An employee may enroll at any time in this program.

This coverage has full 24-hour, 365-days-a-year protection against accidents occurring in the course of business or pleasure. The insurance includes accidents whether on or off the job, occurring in or away from the home, or traveling by public and private transportation. The benefits provided under this plan are payable in addition to other insurance that may be in effect at the time of accident. There are no geographical limits on this coverage.

#### 11.2.3 Health Insurance

Health insurance is an optional program available to all full-time and eligible part-time faculty and staff members. Coverage is offered for the member's spouse and for eligible dependents through the end of the calendar year that they reach age 26. Employees may participate in the state employees' health insurance plans by contacting the Human Resources Service Center.

For employees who work at least 80 percent time the employer pays a major portion of the employee's health care premium. Employee plus one and family coverage is also available under this plan. For part-time employees whose FTE is between .50 and .80 the employer contribution varies.

Newly eligible employees (newly hired, rehired, or status change to eligible position) have up to 30 calendar days to enroll in a health plan and/or flexible spending accounts (FSA) offered by the state. The 30-day countdown period begins on the first day of employment and ends 30 days later. If the enrollment action is received within the 30 calendar day time frame, coverage will be effective the first of the month coinciding with or following the date of employment. For employees whose employment starts on the first day of the month, and who have completed an application within 30 calendar days, the coverage will begin on their date of hire.

Enrollment or enrollment changes cannot be made outside of the open enrollment period unless there is a qualifying mid-year event such as marriage, divorce, adding or dropping dependents, which must be made within 60 calendar 31 days of the event.

Open enrollment is normally held during the month of May each year for those employees desiring to enroll or to make changes in their health care program. Any enrollment or changes made during open enrollment are effective on July 1.

# 11.2.4. Medical Flexible Spending Account

Salaried faculty who work at least 20 hours per week (.50 FTE) are eligible to enroll in the commonwealth's medical flexible spending account (FSA), which allows them to pay certain medical care expenses (eligible out-of-pocket medical, dental, and vision care expenses not covered by their selected health benefits plan) with pre-tax dollars, thereby reducing taxes and increasing available income. The minimum that may be set aside is \$10 per pay period; the maximum is \$2,500 per year. A monthly pre-tax administrative fee does apply.

Faculty have 30 days from their date of hire/appointment to enroll in the medical FSA. Each year during open enrollment in the spring, faculty may renew FSA accounts or elect to enroll for the first time. Changes made during open enrollment are effective the following July 1. A mid-year qualifying event (resulting from marriage, divorce, birth, or adoption) may be made outside the open enrollment period if the request for change is made within 60 days of the qualifying event. Contact <a href="https://doi.org/10.1001/july-10.1001/jul

# 11.2.5 Dependent Care Flexible Spending Account

Salaried faculty who work at least 20 hours per week (.50 FTE) are eligible to enroll in the commonwealth's dependent care flexible spending account (FSA). The dependent care FSA allows them to pay expenses with pre-tax dollars (thereby reducing taxes/increasing available income) for the care of a child, disabled spouse, elderly parent, or other dependents who are physically or mentally incapable of self-care so that the faculty member (and spouse) can work or actively seek work. The minimum that may be set aside is \$10 per pay period; the maximum is \$5,000 per year. A monthly pre-tax administrative fee does apply.

Faculty employees have 30 days from their date of hire/appoinment to enroll in the dependent care (FSA). Each year during open enrollment in the spring, faculty must may renew dependent care FSA to continue deductions or enroll for the first time. Changes made during open enrollment are effective the following July 1. A mid-year qualifying event (resulting from marriage, divorce, birth, or adoption) may be made outside the open enrollment period if the request for change is made within 60 days of the qualifying event. Contact <a href="https://doi.org/10.1001/july-10.1001

# 11.2.6 Employee Assistance Program

The Employee Assistance Program (EAP) is a counseling and referral service available to faculty and staff to help deal with the range of problems that might have an impact on their work lives as well as personal lives. The EAP provides confidential short-term intervention, assessment, and referral services. Employees may self-refer to the EAP, or a supervisor or manager may make a referral as either an informal recommendation or as a mandatory requirement.

With appropriate approvals by senior administrators, an employee may be referred for a mandatory fitness-for-duty examination in cases where the employee poses a hazard or risk to self or others, or if a determination of the employee's medical or psychological fitness to perform his/her essential job functions is needed. Protections for employees to assure the privacy of their personal health information and to prevent abuse of mandated referrals by supervisors are included in university policy 4345, "Employee Assistance Program."

# 11.2.7 Tax-Deferred Investments/Deferred Compensation/Cash Match

<u>These programs</u> offer opportunities for employees to invest a portion of their salaries and/or wages and delay tax liabilities until a later date. Full-time or part-time salaried faculty and staff may be eligible for the employer-paid cash match program. Please contact the Human Resources Service Center for detailed information.

Informational packets are available in human resources. These packets contain a brief description of the tax sheltered annuity or deferred compensation plans, and the agent's name and address. Information is also available on the 401(a) cash match plan.

#### 11.2.8 Credit Unions

Employees may use the services of two credit unions. Payroll deductions may be made for the Freedom First Credit Union. Its main office is located at 1204 South Main Street, Blacksburg, with ATMs and/or branch services available in several on-campus areas.

Any full time and part-time faculty member employed by the Commonwealth of Virginia is eligible for membership in the Virginia Credit Union, Inc., P.O. Box 90010, Richmond, VA 23225-9010, 800-285-6609. Payroll deductions are not available for the Virginia Credit Union.

Employees desiring information or membership in either credit union should contact the credit union directly.

#### 11.2.9 Charitable Deductions

Payroll deduction may be made for participation in the Commonwealth of Virginia Campaign (CVC), which provides contributions to many national, state, and local social and health charitable organizations, including several local United Way organizations.

# 11.2.10 Optional Life Insurance

Optional Life insurance for the faculty member, the spouse, and children is available through the Optional Life Insurance Program. Coverage is guaranteed if the faculty member enrolls within 31 days of the date of hire. Coverage may be applied for outside of the 31 days, but underwriting will be required. The coverage is underwritten by Minnesota Life Insurance.

# 11.3 Special Programs

There are two special insurance protection programs covering university faculty and staff. All employees are covered against job-related illness or injury by the Commonwealth of Virginia Worker's Compensation. Employees who lose their job through no fault of their own are covered by unemployment insurance through the Virginia Employment Commission.

# 11.3.1 Unemployment Insurance

All employees of the university are covered by unemployment insurance. If for some reason employees become unemployed by no fault of their own, they may qualify for this insurance. If employees lose their jobs, they should contact the Virginia Employment Commission immediately to file an unemployment claim.

#### 11.3.2 Severance Benefits

The university provides severance benefits for eligible faculty who are involuntarily separated due to budget reduction, agency reorganizations, or workforce downsizings for reasons unrelated to performance or conduct. Faculty hired on restricted appointments funded from sponsored contracts or grants, or term appointments with a specified ending date, regardless of funding

source, are not eligible to receive severance benefits. Non-reappointments and voluntary resignations for any reason are not deemed "involuntary separation" for purposes of the severance policy.

### 11.3.3 Workers' Compensation

All employers are required to provide protection to their employees for job-related injuries and illnesses. The aim of the program is to ensure that all university faculty and staff members injured in the course of their employment and arising out of their employment with the university are offered fixed, certain, and speedy relief. The Commonwealth of Virginia Workers' Compensation services are provided by Managed Care Innovations, LLC.

# 11.3.3.1 Reporting Work-Related Injuries

An injured employee is required to immediately report an accident or illness to his or her direct supervisor. Upon the employee's notification of injury to the departmental supervisor, the department is required to offer the employee a worker's compensation panel physician selection form. The employee must select a physician for treatment by completing the panel physician selection form. The department submits the signed form to human resources' benefits office.

Employees in a life-threatening emergency situation must seek medical treatment from the nearest medical service provider by the quickest means possible. In a non-life threatening emergency situation, employees may seek treatment from the nearest emergency medical service provider. After the emergency situation has ended, the employee must report the injury. The employee will be provided with a panel of physicians. All follow-up treatment must be provided by a <u>worker's compensation panel physician</u>.

Once an employee reports a job-related injury, the supervisor must immediately file the employer's accident report. The employer's accident report must be filed within 24 hours of the date/time of the injury. A claims adjuster from Managed Care Innovations, the university's worker's compensation carrier, will be assigned to handle the claim.

#### **CHAPTER TWELVE**

# 12.0 Additional Information

If you have read through the *Faculty Handbook*, explored all avenues, and still have a question or seek policy clarification, contact Dr. Jack Finney, vice provost for faculty affairs, at 540-231-2350.

For questions regarding benefits, contact the benefits office in human resources at 540-231-9331.

University policies are available at www.policies.vt.edu.

# 12.1 Emergency Information

Please see the emergency information sheet that follows this page. For additional information, visit the website of the Office of Emergency Management <a href="https://www.emergency.vt.edu">www.emergency.vt.edu</a>

# REMAIN & FOLLOW THESE INSTRUCTIONS





# HOW TO REPORT AN EMERGENCY

Report all emergencies by dialing 911 from your cell phone or campus phone. Do not hang up until the dispatcher instructs you to. Inform them of the nature of the emergency, number of injured people, and the exact address or building, floor, area, or department.



#### SHELTER-IN-PLACE

Shelter-in-place during a weather or other outdoor emergency:

- Find a sturdy building and go inside.
   Stay away from windows, glass, and objects that could fall on you.
- Find additional information at www.vt.edu or 540-231-6668.



Secure-in-place during violence or active assailant:

- Secure the room or building by locking or barricading the door.
- Call 911 and indicate what is happening.
- Close blinds, block windows, silence cell phones, and stay quiet.
- · Stay behind solid objects, such as desks.



#### FIRE / EVACUATION

Evacuation routes are posted in hallways usually near stairwells or exits.

- Activate the nearest fire alarm on your way out. If there are no fire alarms, knock on doors and yell "fire" as you exit. Call 911.
- If you hear a fire alarm, you MUST evacuate.
- · Close doors as you leave.
- Know two evacuation routes for your room - look for illuminated EXIT signs.
- · Do not use elevators.
- Assist individuals with disabilities if possible. If a person cannot evacuate, report their location to emergency personnel.



# MEDICAL EMERGENCY

Call 911. Provide adequate space for the victim and emergency personnel.

- Unless they are in imminent danger, do not move any victim(s) until emergency personnel arrive.
- Have someone escort emergency responders to the scene.



#### SUSPICIOUS PERSON / OBJECT

Suspicious person:

- · Do not confront the person.
- Do not let anyone into a locked building or office.
- Do not block the person's access to an exit.
- Call 911. Provide as much information about the individual as possible, including direction of travel.

#### Suspicious package or object:

- · Do not investigate the object.
- Call 911. Notify your instructor or supervisor.
- · Be prepared to evacuate if necessary.



# WARNING SIRENS & VT ALERTS

- If you hear the warning sirens, seek shelter and get inside. Shut all doors and stay away from exterior windows.
- If you receive a VT Alert, do what it says. If you are not currently signed up for VT Alerts, register at www.alerts.vt.edu.
- Find additional information at www.vt.edu or 540-231-6668.

